WOMEN AND CHILDREN FIRST: INTIMATE PARTNER VIOLENCE, CHILDREN'S WELL-BEING & CHILD LABOUR IN ECUADOR

by

Paola Beneras Peñaherrera

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy

 at

Dalhousie University Halifax, Nova Scotia May 2021

© Copyright by Paola Beneras Peñaherrera, 2021

To Beatriz Zapata S. & María Veloz.

Contents

List of	Tables	v
List of	Figures	ii
Abstra	ict	x
List of	Abbreviations Used	x
Ackno	wledgements	ci
Chapt	er 1 Introduction	1
Chapt	er 2 Rule of Thumb: Intimate Partner Violence in Ecuador	9
9.1	Tertus de chiere	0
2.1	Introduction	9
2.2	Empirical Framowork	1 9
2.0 2 A	Empirical Results 3	2 1
2.4 2.5	Discussion	- -7
2.6	Conclusion 5	4
Chapt	er 3 Seen and Not Heard: Household Chores and the Well-	
	Being of Children in Ecuador 6	7
3.1	Introduction	7
3.2	Conceptual Framework and Literature Review	0
3.3	Children and Household Work in Ecuador	1
3.4	Data and Empirical Framework 8	2
3.5	Results	0
3.6	Discussion and Conclusion	4

Chapte	er 4	The Road	Not	Taken	or	The	On	e \mathbf{N}	lost	T	av	elle	ed	By	/:
		The Interg	enera	tional	Tra	nsmi	ssio	ı of	\mathbf{Ch}	ild	La	bo	ur	in	
		Ecuador .				•••				• •	•				106
4.1	Introd	uction													106
4.2	Releva	nt Literature													110
4.3	Legal a	and Contextu	al Fra	meworl	ĸ										121
4.4	Data .														122
4.5	Empiri	cal Strategy													129
4.6	Results	8													132
4.7	Discus	sion and Con	clusio	n		• •				• •	•				143
Chapte	er 5	Conclusion	•••			•••					•				164
Refere	nces .	••••••				•••							•		167
Appen	dix A	Intimate P	artne	er Viol	ence	• •									185
Appen	dix B	Children's	Well-	Being							•				214
Appen	dix C	Child Labo	our .												219

List of Tables

2.1	Descriptive Statistics: Means Values by Marital Status	57
2.2	Percentage of Women Who Have Experienced Violence in Current or Past Relationship by Age Group	57
2.3a.	Results, Married and Common-Law Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Inti- mate Partner Violence in Current Partnership	62
2.4a.	Results, Separated and Divorced Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Intimate Partner Violence in Past Partnership	64
2.5	Summary of Results: Statistically Significant Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence in Current or Past Relationship	66
3.1	Descriptive Statistics: Means Values for Children Aged 8-17 Years Old, Living in Two-Parent Households by Gender	99
3.2	Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Total Hours of Household Work (Self-Reported)	100
3.3a.	Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Total Hours of Household Work by Different Activity Performed (Self-Reported)	101
3.4a.	Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Share of Household Work Performed by Children (Self- Reported)	103

3.5	Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Thresholds of Total Hours of Household Work (Self- Reported)	105
4.1	Descriptive Statistics: Types of Work by Gender of Child $\ .\ .$.	148
4.2	Descriptive Statistics: Means of Values by Whether Child is Child Labourer According to the Ecuadorian Legislation	150
4.3a.	Summary of Main Results: Average Marginal Effects from Bivari- ate Probit Regressions for Work & School Attendance by Type of Work, All Children	151
4.4a.	Summary of Main Results: Average Marginal Effects from Bivari- ate Probit Regressions for Work & School Attendance by Type of Work, Children of Dual Earner Parents	153

List of Figures

24	2.1 Examples of Women Surveyed in Relation to Head of Household in Different Household Compositions	2.
58	2.2 Ever Experienced Intimate Partner Violence by Marital Status	2.
58	2.3 Ever Experienced Intimate Partner Violence Divided by Years in Relationship (Exposure to Violence) by Marital Status	2.
59	2.4 Ownership of Economic Resources by Marital Status (Aggregate Measures)	2.
59	2.5 Strength of Acceptability of Wife Beating (Index) by Marital Status	2.
60	2.6 Strength of Patriarchal/Traditional Beliefs (Index) by Marital Status	2.
60	2.7 Domestic Violence in Family Background (Aggregate Measures) by Marital Status	2.
61	2.8 Ownership of Assets in Marital Estate by Age of Women	2.
98	3.1 Proportion of Children Working in Chores by Gender	3,
98	3.2 Child's Responses of Feeling 'Mostly Happy' by Ranges of Hours Worked	3.
148	4.1 Proportion of Children Working According to Official Definition of Child Labour by Parental Work Histories & Poverty	4.
149	4.2 Proportion of Children Working According to Official Definition of Child Labour by Parental Education	4.
149	4.3 Proportion of Children Only Going to School by Parental Work Histories & Poverty	4.

4.4	Average Marginal Effects at Different Ages for All Children Child Labour: Official Definition	155
4.5	Average Marginal Effects at Different Ages for All Children Child Labour: Work in Market Activities	156
4.6	Average Marginal Effects at Different Ages for All Children Child Labour: Work in Domestic Activities	157
4.7	Average Marginal Effects at Different Ages for All Children Child Labour: Paid Work	158
4.8	Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work	159
4.9	Average Marginal Effects at Different Ages for All Children Child Labour: Light Work	160
4.10	Average Marginal Effects at Different Ages for All Children Child Labour: Heavy Work	161
4.11	Average Marginal Effects at Different Ages for All Children Child Labour: Safe Work	162
4.12	Average Marginal Effects at Different Ages for All Children Child Labour: Unsafe Work	163

Abstract

This dissertation examines the relationship between women's ownership of economic resources, beliefs, values and childhood experiences of violence and intimate partner violence, the relationship between household chores and children's self-assessed well-being and the intergenerational transmission of child labour. The individuals studied are Ecuadorian women and children. Chapter 2 explores whether having economic assets available to women helps protect them from intimate partner violence (psychological, physical, sexual and economic violence), taking into account the importance of beliefs, gender norms and childhood experiences of violence. While asset ownership is somewhat protective of intimate partner violence, access to money for personal expenses is strongly associated with lower likelihoods of all types of violence. 'Traditional' beliefs are, perhaps counterintuitively, protective of violence. Results provide compelling evidence supporting an intergenerational aspect to violence where violent family backgrounds are strongly associated with higher incidences of all types of intimate partner violence. Chapter 3 examines the effects of housework on children's self-assessed well-being. The domestic activities that Ecuadorian children (ages 8-17) perform are negatively associated with their self-assessed well-being, both at the extensive and intensive margin of household work. Although the decreases in happiness associated with household chores may appear small in magnitude given the mean level of happiness, the effect is similar to the decrease in happiness associated with the work children perform in the labour market. Chapter 4 studies the intergenerational transmission of child labour in Ecuador. Findings show that children of parents who were child labourers themselves are more likely to combine school and work and less likely to only attend school when defining 'child work' according to Ecuadorian legislation (i.e., children who are illegally working violating minimum age requirements, exceeding working hour limits, whose work interferes with their schooling or who work in dangerous conditions). The intergenerational effect of child labour for Ecuadorian children remains beyond the effects of parental child labour on current family income, it depends on the type of work the children perform and it is not gender neutral.

List of Abbreviations Used

BDH	Bono de Desarrollo Humano
	(Human Development Transfer)
ENTI	Encuesta Nacional de Trabajo Infantil
	(National Child Labour Survey)
ENVIGMU	Encuesta Nacional Sobre Relaciones Familiares y Violencia de Género
	Contra las Mujeres (National Survey on Family Relations and Gen-
	dered Violence Against Women)
ILO	International Labour Organization
INEC	Instituto Nacional de Estadísticas y Censos
	(National Institute of Statistics and Censuses)
IPV	Intimate Partner Violence
OECD	Organisation for Economic Co-operation and Development
OLS	Ordinary Least Squares
РАНО	Pan American Health Organization
PMK	Person Most Knowledgeable
UNDOC	United Nations Office on Drugs and Crime
UNFPA	United Nations Population Fund
UNICEF	United Nations Children's Fund
UNSNA	United Nations System of National Accounts
WHO	World Health Organization

Acknowledgements

I would like to wholeheartedly thank my supervisor, Dr. Shelley Phipps, for guiding me through this process and for doing so in such an encouraging, patient, kind and helpful way - your support has been invaluable. I wish to sincerely thank Dr. Casey Warman and Dr. Lars Osberg for their advice, insightful comments, thoughtful questions and encouragement. I want to thank Dr. Courtney Ward and Dr. Andrea Giusto for their guidance and inspiration. I am very grateful to Dr. Talan İşcan for the honest, supportive advice and for being an incredible mentor.

I wish to thank Monique Comeau and Jodi Lawrence for their unconditional help. I would like to thank Dr. Barry Watson, Dr. Angela Daley, Dr. Zongming Ma, Dr. Min Hu, Dr. Sarah Zhang, Radwa Abdelraouf and Kathryn Wyatt for all the words of wisdom, help, friendship and laughs. I am very grateful to all the professors who have helped me learn and grow and to the students I have had the privilege to teach to.

A heartfelt thank you to my family, I have been refuelled and recharged constantly with your love and laugher. A big thank you to my parents for the Economics gene, and to my siblings, Daniela and Roberto, for always making life brighter. Thank you to Martina, Julia, Manuela, Simón, and Adela for filling my life with joy. Thank you to my cousins, to Alina, Matías, Sebastián, Sandra, Julia, Sara, Danielle, Stephanie, Hilary, Abbie, Brittany, Nader, Perros Inc., Carolina, Felipe, JS, JB, Rafael, Estefanía, Fausto and Mark. Thank you Erin, Karen, Kyoko and Shell. Words are not enough to express my gratitude for Dr. Lisa White - I would not have been able to accomplish this without you, thank you for everything, especially all the sunshine.

Chapter 1

Introduction

This dissertation examines the determinants of negative outcomes experienced by vulnerable populations in Ecuador. Particularly, the focus is understanding the relationship between women's ownership of economic resources, beliefs, values, childhood experiences of violence and intimate partner violence, the relationship between household chores and children's well-being and understanding how child labour can be transmitted across generations. Economic research has focused on the determinants of intimate partner violence and the work children perform mostly as unidimensional issues. This dissertation addresses the gap in the literature by approaching intimate partner violence and child work with a multidimensional perspective taking into account the values and social norms, as well as the familial histories that shape the outcomes of Ecuadorian women and children. Additionally, the analysis focuses on individuals who are susceptible to further harm. Without having a comprehensive understanding of intimate partner violence and the work children do, the prevalence and depth of these issues may grow and outcomes may worsen.

Globally, approximately one in three women will experience physical or sexual violence from a partner during their lifetime (WHO, 2014). In Ecuador, over six in ten women will experience some form of intimate partner violence in their lives and three in ten would have experienced it in the last twelve months (MJDHC, 2018). While experiencing violence is an undesirable outcome in and of itself, it is also important

given that it can become a learned behaviour that is passed down and affect various generations (Aguero, 2013; Cools and Kotsadam, 2017; Orpinas, 1999; Safranoff and Tiravassi, 2018; Tarabah et al., 2016; WHO and PAHO, 2012). Exposure to violence tends to increase the likelihood of desensitization, habituation, normalization and imitation of violence (Tarabah et al., 2016). The further intimate partner violence is normalized, the easier it can spread and become justified. Even without considering the deep emotional personal costs, the societal costs of intimate partner violence are high. The National Institute of Census and Statistics in Ecuador (INEC for its name in Spanish) tried to quantify the costs related to violence at a national level.¹ In these calculations, for Ecuador, the cost of violence against women was estimated to be between 7-14.2% of gross domestic product (GDP) (Carrion, 2007; MJDHC, 2018).²

Economic theory suggests that, if experiencing intimate partner violence, the more resources available to women (e.g., income, wealth, employment, services, and institutions), the better positioned they are to leave an abusive partnership. While evidence shows that this is indeed the case (Cools and Kotsadam, 2017; Garcia-Moreno et al., 2005; Farmer and Tiefenthaler, 1997; Hidrobo and Fernald, 2013; Friedemann-Sanchez and Svec, n.d.; Jewkes, 2002; Tankard and Ivendard, 2018; Munyo and Rossi, 2015), I extend the analysis to incorporate norms and beliefs as well as the women's childhood experience of violence and their (former) spouses'. In Chapter 2, I explore whether having assets available to women helps protect them from intimate partner violence, while taking into account the importance of gender norms and intergenerational transmissions violence. Specifically, I ask whether having assets in the woman's name is associated with a lower probability of experiencing psychological, physical, and sexual intimate partner violence. Results show that while asset ownership is somewhat protective of intimate partner violence, access to money for personal expenses is strongly associated with a lower likelihood of all types of violence. 'Traditional' beliefs are, perhaps paradoxically, protective of violence. I also find compelling evidence supporting an intergenerational effect of violence with violent family backgrounds being strongly associated with higher incidences of all

¹To do this, direct (costs associated with health, such as medical expenditures, institutional costs, and rehabilitation expenditures as well as costs associated with security including expenditures on prevention and legal fees) and indirect costs (such as loss of productivity and loss of economic activity due to death as well as emotional costs derived from compensation data and surveys on willingness to pay to not be a victim of a crime) were estimated (Carrion, 2007; MJDHC, 2018).

 $^{^2\}mathrm{To}$ put this in perspective, education amounts to 12.6% of GDP.

types of violence. While resources can be helpful for women, social norms and violent experiences in childhood loom large and have the largest associations with intimate partner violence in adulthood.

Children engaged in some form of labour are the second vulnerable group studied. Child labour has been widely discussed in the policy arena over the last few decades. According to the last global estimates on child labour from the International Labour Organization (ILO), as of 2017, there were 152 million children working, 73 million of which engaged in hazardous labour. While a lot of attention has been given to the effects of child labour in the context of market activities, the literature remains thin when studying household work as part of child labour, especially, in terms of how it may affect children's well-being. Household chores are usually excluded from definitions of child labour given the "non-economic" nature of these activities. Yet, the number of children performing domestic chores is large and their contributions to their households are quite meaningful. About 800 million children between the ages of 5 and 17 years are involved in some form of housework weekly (ILO, 2017) and contribute between 4% to 12% of all household work in their homes. Blair (1992a); Goldscheider and Waite (1991); Gershuny and Sullivan (2014).

Chapter 3 explores the effects of housework on children's self-assessed well-being. Specifically, I seek to answer how happy children (between 8 and 17 years) self-report themselves to be, conditional on time spent on housework, i.e., do children perceive the unpaid work they do at home as being detrimental to their happiness? I find that the domestic activities that Ecuadorian children perform are negatively associated with their self-assessed well-being, both in the extensive and intensive margin of household work. From a baseline of 86% of children saying that they are happy most of the time, performing chores is associated with an average 4.0 percentage point (p.p.) lower probability in children's self-assessed happiness. An increase in domestic work of 10 hours per week is associated with a 2.1 p.p. lower probability of children self-reporting as happy. Although the decrease in happiness associated with household work may appear small in magnitude given the mean level of happiness, it is very close to decreases in happiness associated with the work children perform in the labour market. While a lot of attention is placed on the work children perform in economic activities, from the children's perspective, work in chores at home is almost as detrimental to their well-being.

While housework is important, it is not the only form of work children are engaged in. There is considerable heterogeneity in the type of work (e.g., market vs. domestic work; paid vs. unpaid work) and the working conditions (light vs. excessive work; favourable vs. unfavourable - dangerous, unhealthy and abusive - work) children may perform. Most of the concern surrounding child labour stems from human capital implications of preventing children from going to school and affecting their future earnings. When children shift from school to the labour market it may create a vicious cycle where poor households send their children to work and these children grow to become poor adults due to the diminished investment in their education (Emerson and Souza, 2003; Udry, 2004). However, the mechanisms of this intergenerational transmission are not fully understood. It could be that the mechanism is exclusively an economic one, a budget constraint that families experience due to poverty. It could also be that the mechanism for the transmission is partly an attitudinal or behavioural one and child labour may be perpetuated due to beliefs, values, expectations and potentially rites of passage. The type of work that children engage in may depend on the mechanism at play. Presumably, if children work out of sheer necessity, the quality of work and potentially dangerous characteristics of work would not be as important as the need to sustain the household. However, if parents send children to work to develop attitudes and abilities, then, it is likely that children would not be engaged in work that would harm them. Target 8.7 of the United Nations Sustainable Development Goals calls to eradicate all forms of child labour by 2025 (ILO, 2017). If the goal is banning child labour due to the human capital argument and if the behavioural mechanism is important, even if poverty is eliminated, child labour would not be eradicated.

Chapter 4 examines the intergenerational transmission of child labour in Ecuador. I ask whether children are more likely to engage in child labour and less likely to go to school if a parent started working before the minimum age requirement (15 years old), after controlling for current family poverty. I explore whether there are differences in the transmission of child work depending on the type of work performed and on working conditions. I assume that the household's decision to send the children to work or school is a simultaneous one and look at outcomes that combine both alternatives. I find that the children of parents who were child labourers themselves are more likely to combine school and work and less likely to only attend school when defining 'child work' according to Ecuadorian legislation. I also find that the transmission of child labour depends on the type of work the children perform (if both parents were child labourers, children are more likely to combine school and be engaged in work that is potentially harmful, i.e., heavy and unsafe work), and that the transmission of work is not gender neutral. Thus, policies solely aimed at reducing or eliminating child work through poverty alleviation may reduce the worst cases of child labour but will not eradicate all forms of child work given the intergenerational links that remain beyond poverty.

Given that all chapters focus on Ecuadorian individuals, what follows is a brief overview of Ecuador to provide context for the analysis.

Socio-Economic Context of Ecuador

Income Ecuador is an upper-middle income country with a population of 15,243,883 as of 2011 (WDI, 2019; INEC, 2015).³ In 2011, average household income was \$892.9 USD per month and average household expenditures were \$809.6 USD per month (ENIGHUR, 2012). There are several social safety net programs for low-income individuals though the most well-known and used is the "Bono de Desarrollo Humano" (Human Development Transfer, BDH for its name in Spanish), a (now) conditional cash transfer. In 2012 it provided households in poverty with \$35 per month (currently, it provides \$50 to households in extreme poverty). It was instituted in 1998 (due to the country undergoing a financial crisis with hyperinflation, extreme devaluation of the local currency which culminated in dollarization, a bank run and high rates of unemployment). The transfer is given to mothers (Martinez et al., 2017).

Education The educational system of Ecuador divides universal education into "early education" for children between the ages of three to four years old, ten years of "general basic education" for children aged five to fourteen and three years of "baccalaureate education" for children aged fifteen to seventeen (S.R.O. No. 754,

 $^{^{3}}$ Given that the data used for this dissertation are from the years 2011-2012, most of the contextualizing information will be set as closely as possible to these dates.

2012). Once completed, children receive a "Baccalaureate Certificate."⁴ Education is compulsory for general basic education and baccalaureate education, lasting a total of 13 years. Pre-primary education was not deemed mandatory until a reform in 2010.⁵

In 2012, net attendance rates⁶ in general basic education, i.e., for children between the ages of five to fourteen, was 95.6% (95.4% for boys and 95.9% for girls). Net attendance rates for baccalaureate education, i.e., for children between the ages of fifteen to seventeen, was 63.9% (61.6% for boys and 66.3% for girls) (Antamba, 2015). Looking at the children who did not attend school, 35.58% said the main reason for not attending was 'lacking the economic resources to do so,' followed by 'due to work' (16.64%) and 'uninterested in school (11.60%) (Antamba, 2015).

Even though education is compulsory for children, enforcement is low. There are some instances in which enrolment and attendance are monitored. In 2008 a pilot program took place in the three main cities of the country, Quito, Guayaquil and Cuenca to check compliance of requirements of the BDH. Almost a third (32%) of households were suspended from receiving the transfer for two months for not meeting the health and/or education requirements. The same year, households in rural areas in twelve provinces were selected to verify school enrolment and the government found that only 47% of households that received the BDH complied with the school requirement (Martinez et al., 2017). Another factor that potentially is related to completion rates in Ecuador is fertility. In 2012, the government reported that the adolescent (women between fifteen to nineteen years) fertility rate was 172 births per 1,000 teenagers, making 17% of adolescent females mothers, the second highest incidence of teenage pregnancy in Latin America (El Universo, 2012; MSP, 2012). As of the last census in 2010, 20% of all deliveries were from females between the ages of fifteen to nineteen who gave birth to 60,600 children. Additionally, 1,100 children were born to mothers between the ages of ten to fourteen years (El Universo, 2012). Abortion is criminalized in Ecuador, both the doctor and the patient could serve

⁴Prior to 2010, the educational system was divided into pre-primary education, primary education and secondary education. Pre-primary education was for children aged 4-5; primary education for children aged 6-11; and secondary education was for children aged 12-17 (UNESCO, 2014; S.R.O. No. 754, 2012). In 2012, the completion rate for primary education was 97.3%, 85.8% for lower secondary and 65.5% in upper secondary.

 $^{{}^{5}}$ Following the reform, the last year of what used to be pre-primary education, when the child is five years old, became the first year of general basic education. Early education (when a child is three or four years old) is universal (by law, the State is mandated to offer it to everyone) but not required for a child to be allowed to enrol in general basic education (S.R.O. No. 754, 2012).

⁶"Total number of students of the official age group for a given level of education who are attending school at any level of education, expressed as a percentage of the corresponding population" (UNESCO, 2014, np).

between six months to two years in jail following an abortion. The only exception, when abortion is not deemed a crime, is when the woman's life is in danger or in the case of rape to an individual with cognitive disabilities (COIP, 2014).

The Institution of Marriage In 2011, the minimum age to marry in Ecuador was 12 years old and 14 years old for women and men, respectively.⁷ Legally, a marriage can end due to the death of one of its members, if it is deemed null or void, if a person is legally declared missing, or due to divorce. No fault divorce is legal (Civil Code, 2005). Common-law unions become a legally binding marital status after two years of cohabitation.⁸

Following the end of a relationship (either a marriage or common-law), the separation of conjugal property can be filed through a dissolution of union (not necessarily a divorce), so that the union's estate is divided and any other assets purchased following this division of joint property remains solely under the ownership of the purchaser, not the matrimonial estate. During the marriage or if the couple has separated but are still legally married, a spouse can file for espousal support for themselves as well as for their children (Civil Code, 2005; Metro Ecuador, 2016).⁹ Child support can be requested for all children under the age of 18 (or under the age of 21 if the child is a student) by the parent (or legal representative) who has custody or by children themselves if they are over the age of 15 (Children and Adolescents Code, 2002; El Telegrafo, 2017, 2019).¹⁰

Violence In 2014, the Penal Code incorporated and criminalized physical, psychological and sexual violence against women or family members, including femicide (STPTV, 2018). Since violence against women was criminalized (and reports of violence were first recorded) in 2014, reports filed to the State Prosecutor's Office have shown an upward trend in gendered violence. In 2014, between August and December there were 2,016 reports filed for physical violence, 13,919 reports of psychological violence, 66

⁷Following reforms to the Civil Code in 2015, the minimum age to marry became 18 years old (El Comercio, 2015). ⁸In the 2015 reform, the cohabitation requirement was eliminated (El Comercio, 2015).

 $^{^{9}}$ Usually, espousal support is granted based on whether a spouse (commonly, the females) stopped working during the union, worked temporarily or earned significantly less than the other spouse (Metro Ecuador, 2016).

 $^{^{10}}$ Child support is calculated based on the earnings of the supporter and support is paid on a monthly basis (El Telegrafo, 2019). If the parent who provides child support is unemployed, a judge will decide a minimum payment for the children. If the parent misses alimony for two or more months, an arrest warrant can be issued against them (El Telegrafo, 2019).

reports of sexual violence against women, and 27 reports of femicide (MJDHC, 2018). In 2019, there were 9,090 sexual abuse cases reported, an additional 4,811 reports of physical domestic violence, 30,893 reports of psychological domestic violence and 250 cases of sexual domestic violence, and 65 femicides (FGE, 2019). In 6 out of 10 cases of femicide, perpetrators were the spouses, former spouses or live-in partners (STPTV, 2018). Since data was first collected in August of 2014 until November of 2020, 833 women have died by femicide, which approximately amounts to one woman being killed every 72 hours in Ecuador (El Comercio, 2020).¹¹

 $^{^{11}}$ So far, 101 of these women died in 2020 and 82 femicides have occurred during the COVID-19 pandemic (El Comercio, 2020).

Chapter 2

Rule of Thumb: Intimate Partner Violence in Ecuador

"His Honor was of opinion that the defendant had a right to whip his wife with a switch no larger than his thumb, and that upon the facts found in the special verdict he was not guilty in law."

State v. A. B. Rhodes, 61 N. C. 453 (1868)

2.1 Introduction

The home is the most dangerous place for women (UNODC, 2018). On an average day, 137 women are killed by their partners or a member of their family (UNODC, 2018).¹ Although these estimates on gender-related killings reflect extreme violence inflicted on women, throughout their lives, a third of women will have survived physical or sexual violence from an intimate partner (WHO, 2014). In Latin America and the Caribbean, at least one in five women (20%), and as many as three in four women (75%), experience at least one emotionally abusive act by a partner throughout their lives (Garcia-Moreno et al., 2005). In Ecuador, throughout their lifetime, 35% and

 $^{^1}$ Globally, in 2018, 58% of all reported female homicides were perpetrated by intimate partners or family members (UNODC, 2018).

14.5% of all women will have experienced physical and sexual violence, respectively, in their partnerships (INEC, 2012a; MJDHC, 2018).² Intimate partner violence is not slowing down. Even though ending gender-based violence is one of the Sustainable Development Goals, progress has been halted due to the Covid-19 pandemic (UNFPA et al., 2020).³

The goal of this chapter is to understand intimate partner violence in the context of Ecuador. Despite richness in the interdisciplinary literature analysing violence, there is no unified theoretical or empirical consensus providing a holistic understanding of intimate partner violence. In an attempt to provide such an understanding, I merge three major themes from the literature to study intimate partner violence in Ecuador. The first theme explores how economic resources, mainly, assets available to women, are associated with intimate partner violence. The resources available to women in a relationship are hypothesized to affect the well-being she could have outside marriage in the event of divorce (i.e., her 'bargaining position'), which in turn may have an association with the probability of violence. The second theme included is the importance of beliefs and social norms surrounding women. Identifying the beliefs women hold aids in uncovering the extent of the normalization of violence, the problems and backlash that women may face when going against those norms, or the protectiveness of abiding by them.

As Phipps and Woolley (2008) point out, the gains from marriage may not be shared in an equal way, "while women and men gain from cooperation within the family, there is inevitably conflict over how those gains are shared: men may 'bring home the bacon,' women may cook it, but who gets to eat it?" (pg. 592). Yet, in light of conflict, does cooking the bacon implies having the pan as a shield? Or is it the owner of the pan the person who can use it as a weapon? Or in turn, if roles are reversed, and if women bring home the bacon, will they be hit with the pan?

The third theme examines intergenerational transmissions of violence, whether the woman or her partner experienced domestic violence as children. Individuals who were abused as children or who witnessed abuse during their childhoods may be more

²Physical violence is defined here as all acts of force that cause harm, pain or physical suffering to the person on whom it is inflicted upon, independent of the means or consequence. Sexual violence is defined as the imposition of any sexual act to a person, forcing them to have sex or sexual acts with the aggressor or with third parties through the use of physical force, intimidation, threats or any other means of coercion (INEC, 2012a).

 $^{^{3}}$ It is estimated that for every three additional months of lockdown, there will be an additional 15 million cases of gender-based violence worldwide (UNFPA et al., 2020).

prone to inflicting or suffering from more violence later in life.

Specifically, I ask three main questions: (i) does ownership of resources affect intimate partner violence? (ii) are women's traditional or patriarchal beliefs or a stronger acceptance of wife beating beliefs associated with higher incidences of violence? And, (iii) is having a history of violence in childhood associated with higher violence?

Using a novel dataset from Ecuador, I find that assets in women's names are only somewhat protective of intimate partner violence. Alternatively, access to money for personal expenses is more shielding of intimate partner violence than ownership of (longer-term) wealth. 'Traditional' beliefs regarding obedience to spouses are similarly protective of all types of intimate partner violence. Lastly, without exception, for all women, there is a strong intergenerational aspect to violence. Having a violent environment in childhood either in their own family background and/or in their spouse's family background is strongly associated with higher incidences of all types of violence.

An important limitation of this study is unobserved heterogeneity in violence, i.e., women who experience(d) violence could be more likely to divorce or separate from their spouses due to the violence experienced. Even though there is no easy solution for this issue, I aim to address it by analysing the married or common-law women separately from their separated or divorced counterparts. Widowed women are also studied given that rather than there being a choice in separating from a partner, in this case, men 'leave' the marriage by death. Similarly, I include a measure of incidence of violence that accounts for exposure to violence through length of the relationship, 'exposure to violence.' Results remain robust to these sensitivity checks.

The chapter is organized in the following way. Section II provides a background and a literature review. Section III presents contextual information on Ecuador. Section IV outlines the empirical framework. Section V discusses the results and Section VI concludes.

2.2 Background and Review of the Literature

The World Health Organization (WHO) defines intimate partner violence (IPV) as "any behaviour within an intimate relationship that causes physical, psychological

or sexual harm to those in the relationship" (WHO and PAHO, 2012, pg. 1). It is distinguished from domestic violence in that domestic violence includes child abuse or any abuse to other members of the household (WHO and PAHO, 2012). Although women can be perpetrators of violence, when they inflict violence, it is generally agreed that they are mostly acting in self-defence and that in most cases, men are the perpetrators (WHO and PAHO, 2012; Heise et al., 1999).⁴

Theoretical Models of Relationship Between Intimate Partner Violence and Resources

Disciplines such as Economics, Sociology, Psychology, and Medicine, have dedicated significant efforts to understanding intimate partner violence, yet there is no consensus regarding the relationship between resources and intimate partner violence.⁵ Note that when referring to resources, the ones that are commonly used in the literature are income, wages, transfers, wealth, savings, employment and education (Bobonis et al., 2013; Cools and Kotsadam, 2017).

In the 1960s, Gary Becker and Jacob Mincer began exploring 'household' models of how men and women allocate their resources (time and money) to maximize the wellbeing of the family unit (assuming everyone in the household has the same preferences and that income is pooled by all members of the family). The simplest models led to a predicted division of labour between men and women where their comparative advantage determines who would be responsible for home-making and who would be responsible for bread-winning. Specializing, i.e., when one members takes on the activity for which they have the lowest opportunity cost, usually predicted increased gains from the union (Blau & Winkler, 2014; Lundberg and Pollak, 1996).

Treating the family as a unit ignored the internal decision-making process within the household, differences in preferences of husband and wife and how outcomes can

⁴It is understood that all genders can experience intimate partner violence. However, given the larger prevalence of violence inflicted on women, especially in the context of Ecuador, throughout the chapter, when referring to intimate partner violence, it is implied it is in the context of a heterosexual couple where the male is violent towards the female. Statistically, when it comes to intimate partner violence, the most common perpetrators are men (WHO and PAHO, 2012). For example, in Canada, in 2015, in 79% of the 92,000 reports of intimate partner violence, women were the victims. In 2015 it was estimated that in the United States, 36.4% of women and 33.6% of men experience intimate partner violence (Smith et al., 2018). In Ecuador, looking at the complaints to the Commissioner for Women and the Family, in 97% of intimate partner violence cases, men were the perpetrators of violence in 2000 (Carrion, 2003; Ernst, 2002). In 2010, in 86% of all cases of intimate partner violence, women were the victims (La Hora, 2011).

⁵For an excellent review of the literature, see Cools and Kotsadam (2017).

be determined through bargaining (Blau & Winkler, 2014). Cooperative bargaining models introduced (typically) a Nash bargaining game in which agents (husbands and wives) have their own utility functions, representing unique (different) preferences (Blau & Winkler, 2014; Lundberg and Pollak, 1996; Mas-Colell et. al, 1995).⁶ Agents also have their own threat points, the levels of well-being that each spouse would obtain in the absence of cooperation.⁷ Threat points are assumed to reflect each individuals' 'outside options,' and are hypothesized to reflect the income that each partner would have if the marriage ends, given the cultural, political (social safety net) and legal environment (Blau & Winkler, 2014; Lundberg and Pollak, 1996). The bargaining power that each individual has is determined by their threat point. The better-off a spouse would be by walking away from the marriage, the higher the threat point and the more bargaining power that spouse would have in the marriage. In these divorce-threat bargaining models, if there is a cooperative solution (one in which the couple stays together), the outcome would more closely resemble the preferences of the spouse with better 'outside options,' the one with stronger threat points, as they had stronger bargaining positions. A non-cooperative solution would be the end of the relationship (Blau & Winkler, 2014; Lundberg and Pollak, 1996; Mas-Colell et. al, 1995). Threats to leave a marriage can be perceived as real depending on a woman's economic independence: her employment, skills, education, savings, wealth or non-labour income indicate whether or not she has the financial footing to leave a marriage. (Lundberg and Pollak, 1996).

One important aspect to consider is that the decision to stay or leave an abusive relationship may be a repeated decision as suffering abuse may not be a one-time incident in marriages (Tiefenthaler, 2012) and that it may not be such a clean-cut decision as economic theory might suggest. Abusive relationships usually start by tension in the relationship increasing until someone inflicts abuse. In most cases (and usually before escalating to physical aggression) conflict is followed by repentance from the spouse who behaved aggressively so that the party who sustained the abuse remains in the relationship. The result is a "continuously repeating cycle of violence," followed

 $^{^{6}}$ For a detailed explanation of the model, see Lundberg and Pollak (1996).

⁷Given the nature of this chapter, it is worth making a clear distinction regarding the difference between the threat of violence and a threat point. A threat of violence is usually a way to inflict emotional abuse by threatening with but not physically inflicting violence. A threat point, on the other hand, is a theoretical concept that outlines the utility that would be obtained from abandoning cooperation, it does not refer to the menace of violence.

by the victim's inability or reluctance to leave the relationship, usually referred to as "learned helplessness" (Bowlus and Seitz, 2006).

There is also a hierarchy to violence. Less aggressive behaviours tend to be more frequent, but these behaviours could also escalate in their level of violence as time goes on (Orpinas, 1999). For instance, emotional or psychological abuse like insults are usually more frequent than physical aggressions like slapping or hitting, but can escalate to more intense aggressions (Orpinas, 1999). In an intimate partner violence context, husband-inflicted violence could be attributed to stress⁸ and being less "in control" of one's temper, or to inherent pleasure derived from inflicting it, or violence could be used as an 'instrument' in the bargaining process that occurs between a couple, as means of coercion, twisting someone's arm both literally and figuratively (Bloch and Rao, 2002; Friedemann-Sanchez and Svec, n.d.; Jewkes, 2002; Weitzman, 2014).

Resources, Attitudes and Intergenerational Transmissions of Violence: A Review of Empirical Studies

Within the bargaining framework described above, resources available to women affect bargaining power. Resources can be studied in absolute terms, relative terms (i.e., relative to the spouse) and in societal terms (i.e., institutions or social norms).

Absolute Resources Absolute resources refer to the level of means available to women, including wealth, income, education, employment or earning potential. Some research has found that access to resources protects women from intimate partner violence. Generally, women with fewer resources are more vulnerable to abuse (Cools and Kotsadam, 2017) and men who have fewer resources tend to be abusive (Cools and Kotsadam, 2017; Garcia-Moreno et al., 2005; Jewkes, 2002).

The type of resources available may have different implications in terms of protectiveness. In Colombia, for instance, income, employment and asset ownership, are found to be protective factors against intimate partner violence (Friedemann-Sanchez and Svec, n.d.). Ownership of wealth can potentially affect women's well-being for

⁸For instance, when individuals are in a state of recurrent economic insecurity their mental state has been shown to be severely affected (Watson and Osberg, 2017).

much longer than income might (Phipps and Woolley, 2008), especially given the fact that wealth is accumulated, whereas income is a flow in a given period of time.⁹ In Canada, women's employment serves as a significant deterrent of abuse (Bowlus and Seitz, 2006). Women's level of education similarly appears to mitigate the risks of violence (Cools and Kotsadam, 2017; Farmer and Tiefenthaler, 1997; Friedemann-Sanchez and Svec, n.d.; Jewkes, 2002; Tankard and Iyendard, 2018). In Ecuador, unconditional cash transfers that increase women's incomes are shown to significantly decrease emotional violence and controlling behaviours from the husband but only for women who have more than elementary education (Hidrobo and Fernald, 2013). Earning potential is also important as some argue that potential, rather than actual opportunities are what defines the threat points (Munyo and Rossi, 2015).

Increased property rights for women have been shown to benefit women's intrahousehold position. For instance, in Vietnam, land reforms were beneficial to women who held land-use rights exclusively i.e., without their partners. Through agricultural security, access to credit markets and higher control of their earning capacity, land titling increased the well-being of women. Women's control of assets benefited selfemployment, enhanced their economic security, reduced their vulnerability to poverty and overall strengthened women's bargaining position within their homes (Menon et al., 2017). Changes to inheritance laws in India suggest that there were decreases in violence. Women who were eligible to receive an inheritance were 17% less likely to be victims of intimate partner violence. However, although women's security is likely to have increased due to the institutional change, their autonomy did not (Amaral, 2017). Another important aspect is under whose name the couple's wealth is, "owning assets improves a spouse's future bargaining position ... each spouse has an interest in having as many of 'the family's' assets in his or her name" (Phipps and Woolley, 2008, p. 596). Both social norms and reliability of law enforcement could influence the intra-household resource allocation determining under whose name the resources are under (Phipps and Woolley, 2008).

Counter-intuitively, from the bargaining perspective, increasing women's absolute resources has sometimes been found to increase violence via 'male backlash' if men

 $^{^{9}}$ When it comes to a couple, there are various determinants of savings decision. Given that the analysis in this chapter is conducted through evaluating the protectiveness of having crops, vehicles, savings or a house under the wives' names, these will be treated as assets or means of saving. Phipps and Woolley (2008) frame the determinants of saving decisions under the life-cycle model, theories of intra-household bargaining and government policy.

feel threatened and resort to violence to re-establish control or power in their marriage (Cools and Kotsadam, 2017). The relationship between intimate partner violence and absolute resources is not a monotonic one. Resources available to a woman can become protective but only up to a threshold, after which, access to more resources is associated with higher incidences of violence (Cools and Kotsadam, 2017; Jewkes, 2002). This is especially evident when the increase in resources available to women are not enough to leave the relationship (Cools and Kotsadam, 2017). In this case, the relationship between resources and abuse resembles an inverted U-shape (Angelucci, 2008; Cools and Kotsadam, 2017). In Mexico, when looking at the effects of women receiving conditional transfers from Oportunidades (a poverty alleviation program), receiving the smallest transfer, decreased aggressive behaviour by almost 40% (Angelucci, 2008). However, when women received larger transfers and their husbands had lower education levels, there was an increase in intimate partner violence (Angelucci, 2008).¹⁰

An alternative to the bargaining perspective is that household level poverty can increase stress in a couple which may result in violent behaviours. In this case, the pathway between violence and economic resources is stress. In Canada, for instance, job insecurity is associated with increased psychological distress in both men and women (Watson and Osberg, 2018). For males, specifically, a one-time, single, occurrence of economic insecurity is associated with higher levels of psychological distress whereas repeated insecurity has larger associations with mental distress, compared to those employed or job secure (Watson and Osberg, 2017). If the family (or men in the family) are having an economically difficult time, mental health suffers and heightened distress may result in individuals losing their tempers and resorting to violent behaviours. Mental health suffering would conflict with the rational behaviour assumption assumed in the economic models discussed above. However, it could be a mechanism by which the added stress from poverty or job insecurity could lead to intimate partner violence.

Relative Resources Being economically dependent on a partner is likely to increase the risk of intimate partner violence (Cools and Kotsadam, 2017). When women increase their relative economic position and become less dependent on their spouses,

 $^{^{10}}$ In a similar study, Bobonis et al. (2013) show that women who are beneficiaries of Oportunidades are less likely to be victims of physical abuse when compared to non-beneficiary women but more likely to face emotional abuse.

abuse is less likely as they are in a better position to leave the relationship. For instance, in California, between 1990 and 2003, reductions in the gender wage gap explained 9% of the decline in domestic violence (Aizer, 2010). Similarly, in Uruguay, exploiting exogenous changes in the exchange rate Munyo and Rossi (2015) found that a reduction in women's relative potential wages increased intimate partner violence, and when their relative wages increased, violence decreased. In Ecuador, the same unconditional transfers that in absolute terms increased women's incomes and decreased emotional violence were shown to increase emotional abuse when the relative education of women was higher than their spouses' (Hidrobo and Fernald, 2013). 'Male backlash' can also be present if the relative bargaining power is shifted towards women. Men can use violence to re-establish their dominant position in the relationship when resources cannot enable them to maintain it (Weitzman, 2014). The perceived power that an individual may have in a relationship could create conflict when "women's superior resources, relative to their spouse's, undermine patriarchal arrangements that include conservative gender norms" (Weitzman, 2014, p. 55).¹¹

Beliefs, Attitudes and Norms An important aspect of both absolute and relative resources is how relevant gender norms are in the context of resources and violence. It could be that individually-held (micro-level) beliefs or societally-held (macro-level) beliefs affect the incidence of intimate partner abuse. The social resources available to women, such as institutions, economic security or social norms, beliefs and attitudes can also affect incidence of violence (Aguero, 2013; Cools and Kotsadam, 2017).¹² Absolute and relative resource theories are incomplete when social norms (especially those referring to the male identity) or social resources are not included in the analysis. Rather than solely looking at individual or intra-couple resources, contingencies on what can affect a household due to the social environment in which they live are

¹¹In a similar vein, it has been shown that the relative amount of resources is important when it comes to many aspects of a marriage including its survival. In Bertrand et al. (2015) if a wife's potential income is likely to exceed that of her husband's, the wife faces a slew of negative outcomes: the wife is less likely to be in the labour force, and if she does work, there is a strong likelihood that she will earn less than her potential. For the women who do work, the couples report being less happy and having more problems in their relationship becoming more likely to divorce. Lastly, for couples where the wife earns more than the husband, there is a gender gap in household production. Wives have more non-market work (chores) suggesting that a "threatening wife takes on greater share of household work to assuage the husband's unease with the situation." Even though in this case there may not be intimate partner violence, clashing the societal norms tends to be detrimental for females and may result in the end of their marriage.

 $^{^{12}}$ Strong, equitable institutions can be protective of violence. For example, when the United States allowed unilateral divorce, the country experienced increases in the likelihood of ending an abusive relationship, as the bargaining power shifted towards the abused spouse (Stevenson and Wolfers, 2006).

important (Atkinson et al., 2005; Cools and Kotsadam, 2017).

Violence against women is widespread in Latin American countries, it is deepseated culturally and it is among the most pervasive types of violence (Aguero, 2013; Orpinas, 1999). A study conducted in the capital cities of Brazil (two cities), Chile, Colombia, Costa Rica, El Salvador, Venezuela and Spain, explored cultural norms and attitudes surrounding violence and aggression. The study focused on general aggressive behaviours, including perceptions on violence towards children, spouses, and members outside the family (Orpinas, 1999). Results suggest that tolerance and acceptance of domestic violence in Latin American countries are high. For instance, the proportion of respondents who state that slapping their partners is justified ranged from 67% in Costa Rica to 98% in Spain (the highest in South America being 77% in Chile). There are co-morbidities with aggression, for instance, believing in corporal punishment¹³ as a necessary way to rear children is linked to further violence against partners and others. Findings suggest strong associations between the likelihood of aggression to a partner and aggression to a child.

The cultural context in which a woman lives may be detrimental to modifying beliefs that accept women being battered (Cools and Kotsadam, 2017; Jewkes, 2002). It has been shown that when women are surrounded by strong patriarchal norms, there is a greater tolerance for intimate partner violence (Jesmin, 2017). Intimate partner violence may become not only tolerated but normalized, "girls are socialized from very early childhood that men are their protectors, and good wives are submissive to their husbands ... [increasing the] odds of justifying intimate partner violence (Jesmin, 2017, p. 3238).¹⁴ When it comes to male gender roles, in Latin America, masculinity is often associated with a man's ability to provide for their family (Angelucci, 2008). As such,

 $^{^{13}}$ The action of a caretaker inflicting physical pain on a child or adolescent, commonly through spanking, slapping or hitting the child with an object (Orpinas, 1999).

¹⁴According to the World Values Survey, 62.6% of Ecuador is Roman Catholic (97% believe in God), which requires couples to be married by the Church in addition to having a legally binding union (Inglehart et al., 2014). Although this is anecdotal evidence only, a popular passage that is commonly read at weddings is Ephesians 5:26-32, which highlights a wife's submissive role in a marriage, "Wives, submit yourselves to your own husbands as you do to the Lord. For the husband is the head of the wife as Christ is the head of the church, his body, of which he is the Saviour Now as the church submits to Christ, so also wives should submit to their husbands in everything ... [Husbands] each one of you also must love his wife as he loves himself, and the wife must respect her husband." Further, while civil divorce is accepted by the Catholic Church, the only way to dissolve a marriage under the eyes of the church is through an annulment of the marriage. If a divorced individual remarries (through civil procedures) without having had their marriage annulled by the church, they are excommunicated - though divorce itself does not lead excommunication. Grounds for annulment could be lack of consent, not believing in the vows, not wanting children, deceit, betrayal, amongst others. The process can be an arduous one and tends to lasts years in South American countries (Kirchgaessner, 2016).

the economic resources that men bring into their households could be highly reflective of their own identity (Cools and Kotsadam, 2017; Fortin, 2005; Jewkes, 2002). If, or when, women start contributing to their households with resources of their own, the male identity may be threatened which could result in retaliatory violence.

Similarly, if women's identity is strongly associated with housework or childrearing, taking on those roles has the potential of deepening that identity. For instance, for housewives who mostly stay at home, the bulk of their time would potentially be spent in their own household, interacting with those close to them, rather than with co-workers, who may not share and may even challenge their values.¹⁵ This 'domestic isolation,' where women may be isolated due to the time they spend at home and, as such, are not exposed to competing beliefs or values, could result in a confirmation bias in terms of their identity and the beliefs surrounding their identity. "Women who do not work have their traditional gender roles reinforced and domestic isolation hinders change" (Cools and Kotsadam, 2017, p. 214). If domestic isolation prevents change, and if stronger patriarchal norms are associated with a higher tolerance of violence, then the acceptance and normalcy of violence can be reinforced at home. The more households accept violence in their homes, in aggregate, the easier it is for violence to become normalized (and potentially widespread) in society at large. If violence is normalized rather than penalized or rejected in society, violent spouses (or parents) are shielded from negative consequences, allowing impunity for violent behaviour both privately in individuals' homes and socially in their community (Cools and Kotsadam, 2017).

When there are 'atypical' roles taken by women, the male identity could be threatened resulting in male backlash, as previously noted. Even if women are empowered and have enough individual resources but society at large accepts intimate partner violence, actions that bring their independence to light can result in backlash (usually referred to as the 'pioneering hypothesis' which can explain having highly educated women being battered) (Cools and Kotsadam, 2017).¹⁶ It may not only be

¹⁵For instance, a housewife with three children will not work daily next to a woman who works in an office, has never been married and has no desire for children because their work environments will not cross, and they will not be exposed as much to each other's life experiences. If women's schedules and interactions resemble and are restricted to women with similar lifestyles, their exposure to different household dynamics may be restricted.

¹⁶In Bangladesh, a study found that socio-economic status tends to have a negative correlation with views accepting intimate partner violence. Women in poorer communities were more likely to condemn domestic violence and, as illiteracy increased, the justification of intimate partner violence decreased (Jesmin, 2017). Results tend to indicate that, in poorer households, men are not able to exert control over their wives as they are less able to leverage financial

that women face violence from the ownership of resources, but that taking on different gender roles (including jobs outside of the home), away from those deemed as "socially appropriate" may also result in discrimination, harassment and violence. If women embrace different stereotypically "male" attitudes, they may face violent reactions not only at home, but also from society, like co-workers (Akerlof & Kranton, 2010).¹⁷ When identities are threatened, violence may be used to reaffirm power and 'restore' "traditional" power dynamics where men are more powerful than women, forcibly so. As people challenge societal norms, violence 'enables' those norms to be reinstated. If a male's identity is challenged and the female does not respond with submissiveness to fall back into what norms dictate, violence can be used as a tool of 'restoration.' Social norms shape the behaviour that allow or condemn inflicting violence on women.

Intergenerational Transmission of Violence Beliefs and customs can be passed down and tend to be deeply rooted, especially when it comes to gender norms (Akerlof & Kranton, 2010; Fortin, 2005). The more accepted intimate partner violence is, the more likely it is observed intergenerationally (Aguero, 2013; Cools and Kotsadam, 2017; Orpinas, 1999; Safranoff and Tiravassi, 2018; Tarabah et al., 2016; WHO and PAHO, 2012). Pollak (2004) elaborates a theoretical model analysing the intergenerational transmission of domestic violence. The model hinges on three assumptions: first, that violent behaviour from a husband depends on him growing up in a violent home; second, that staying in a violent relationship depends on the wife growing up in a violent home; and, third, that individuals who come from violent backgrounds have a tendency to marry each other.¹⁸ The model highlights the importance of the marriage market conditions and incidence of divorce. Pollak (2004) argues that there is an amplification effect to implementing policies in the short-run. Policies aimed at reducing the incidence of intimate partner violence could further reduce the incidence

contributions as means of control, which results in women being less tolerant of abuse (Jesmin, 2017).

 $^{^{17}}$ For instance, in a workplace environment, characteristics that are identified in men as 'assertive' are usually seen as 'bossy' in women (Fleming, 2018).

¹⁸The transmission pathway in the model is assumed to be witnessing domestic violence in the individual's family of origin. The model relies on the heterogeneity of men's violence and women's divorce decision, some men are violent, while others are not and some women divorce their violent husbands but not all women. The model begins by assuming that realized violence does not affect the marriage or fertility choices and if divorce occurs, it happens before violence materializes (it is triggered by a signal that violence will happen but it has not happened yet) so that divorce becomes protective of violence. This assumption is later relaxed. The model states that violence in the home of origin will determine violence in a partnership (if a man chooses to be violent and the woman chooses not to divorce him, violence will occur), implicitly, men are the perpetrators and women are the victims of violence, in all homes (in the current partnership and in the family of origin).

of domestic abuse in the long run for future generations given the intergenerational aspect of the model and violence.

When it comes to violence in childhood, the literature is vast in portraying how exposure to child abuse (e.g., corporal punishment) tends to have intergenerational effects. Studies point to physical violence in childhood being a risk factor in physically abusing own children. Children who were abused tend to become abusers themselves, to their partners, to their own children and to strangers (Bowlus and Seitz, 2006; Orpinas, 1999; Tarabah et al., 2016; Tankard and Iyendard, 2018; UNODC, 2018; WHO and PAHO, 2012). Evidence shows that, on average, when children are exposed to violence, they become more likely to: be desensitized and habituated to violence, see aggressive behaviours as normal and, potentially imitate such behaviours (Tarabah et al., 2016). Violence (or violence tolerance) is not the only behaviour (or norm) that can also be transmitted intergenerationally. Dynamics within a relationship can also be passed on depending on what an individual witnessed their parents do when the individual was a child. For instance, parental divorce has a positive association with offspring divorce, especially when both spouses have divorced parents (Amato, 1996; WHO and PAHO, 2012). Orpinas (1999) finds that there is a strong association between intimate partner violence and aggression to a child in various Latin American capital cities. These cities had high incidences of supporting attitudes favouring corporal punishment, "violence against children perpetuates the cycle of violence and may increase antisocial behaviour" (Orpinas, 1999, p. 242).

The present chapter seeks to contribute to the literature by integrating the themes of how access to resources, social norms and beliefs and childhood experiences of violence affect intimate partner violence. Ecuadorian women in the National Survey on Family Relations and Gendered Violence Against Women (ENVIGMU for its name in Spanish) answer questions about their experience of intimate partner violence (physical, psychological, sexual and economic violence). Using these reports of violence as dependent variables, I am able to study links with assets, attitudes and childhood experience of violence. Rather uniquely, the survey asks about assets owned by the household and in whose name these assets are registered in addition to more standard economic resources such as income, education and labour force participation. Women are also asked if they believe a husband is justified in hitting his wife and whether they accept patriarchal values. I use these measures to test the association between the experience of intimate partner violence and attitudes and beliefs held by Ecuadorian women. Perhaps most uniquely, women are asked about whether violence was experienced in their own and/or their partner's childhood. Thus, I can test for the presence of an intergenerational transmission of intimate partner violence.

2.3 Empirical Framework

Data In 2011, the Ecuadorian government conducted the National Survey on Family Relations and Gendered Violence Against Women (ENVIGMU), a nation-wide survey covering 18,000 households with the goal of understanding family relations and violence against women. The survey covered all 24 provinces in Ecuador and included households in both urban and rural areas.¹⁹

The survey had a general section for all members of the household, in which basic demographic questions (age, gender and relationship to the head of household) were asked to all members if present or to an informant, the person more knowledgeable. This section included 70,446 people. All surveys were conducted in person. Following the general section, one woman from the household was randomly selected and the domestic violence portion of the survey was conducted on this selected woman. Of the 18,000 household covered, 16,415 women completed the survey fully. All women in these subsequent parts of the survey were women aged 15 years or above. There were three types of questionnaires, where slightly different questions were asked depending on whether the woman was married or in a common-law relationship; separated, divorced or widowed; had never been legally attached.

The first section of the violence portion of the survey addressed general experiences with violence throughout the woman's lifetime. The second section focused on childhood experiences, before the age of twelve. The third section covered sexual abuse, initially as a minor, and then throughout woman's lifetime. The subsequent components of the survey differed depending on relationship status. Currently partnered women were asked about experiences, starting with the current relationship. Women not in a partnership, were asked about their most recent former partner. In

 $^{^{19}}$ Households were selected so that the survey was nationally representative and with a 5% margin of error and a 95% level of significance, at the provincial level.

these, the main questions were geared to describe the relationships and identified what caused strife in the couples. The last portion of the survey asked about attitudes and beliefs regarding gender norms and intimate partner violence.

Given the sensitive nature of the topics covered in the survey, surveyors were rigorously trained. All surveyors were female and they were instructed to handle different possible scenarios throughout the survey, specially handling sensitive topics, someone interrupting, or the interviewees' partners arriving to the household. There were content warnings at the beginning of the survey as well as the assurance of confidentiality, all surveys were done in private (ENVIGMU, 2011).

Ecuador is a country in which multi-generational households are very common and there are many family structures.²⁰ An interesting feature of the data is that for each woman surveyed, their relationship to the "head of the household" is identified.²¹ The women are noted as one of the following: head of household, spouse of head of household, child of head of household, spouse of child of head of household, grandchild of head of household, parent or parent in-law of head of household, or other relatives of the head of household, maid, other non-relatives. Only 12% of surveyed women were identified as the head of the household. Figure 2.1 shows examples of household structures from which one woman is surveyed.

A criterion for sample selection is relationship status. For the main analyses, women who are widowed as well as never married or legally coupled are excluded from the sample. Widowed women are later re-introduced in the sample for sensitivity checks. Women who do not remember the age at which they had their first child or when they were first coupled are dropped (317 (1.93%) and 269 (1.64%) of the original sample of women, respectively). With these restrictions, the estimating sample consists of 12,202 women, 10,801 of whom are married or common-law and 1,401 of whom are divorced or separated.

There is an important caveat with respect to the differences between the group of married and common-law women and the group of separated and divorced women. Presumably, violence is one of the reasons why women may be separated or divorced.

 $^{^{20}}$ For instance, the mean household size in the data is of 5.28 people. According to the Inglehart et al. (2014), 30% of individuals live with their parents.

 $^{^{21}}$ In this case, the head of the household is defined as the person, who resides in the home and is recognized by the other members in the household as their head due to the larger responsibility in decision making, prestige, seniority, economic reasons or social and cultural norms (ENVIGMU, 2011).



Figure 2.1: Examples of Women Surveyed in Relation to Head of Household in Different Household Compositions

These two groups of women are different,²² their characteristics are very different, and although I looked for a plausible instrumental variable, nothing was available. A complication is that for women who are still coupled, the *current* protectiveness of resources, beliefs held by women and their childhood experience inform the *current* situation of violence in their relationships. For the separated and divorced women, the survey measures the extent to which a *previous* experience of violence associated with a *previous* division of assets. Thus, for previously coupled women, most key variables refer to the past, only the beliefs they hold are current. Even control variables, like length of relationship/marriage have different meanings; one is for an ongoing relationship and the other is for a relationship that has ended.²³ In both

 $^{^{22}}$ The results independent sample t-tests on the means of the intimate partner violence variables reject the null hypothesis that groups are the same and that the sample means are equal. This leads to the conclusion that the difference in means is statistically significantly between women who are married and common-law with the women who are separated or divorced, for all the violence variables.

 $^{^{23}}$ Specifically, the timing of the questions asked are different. For all women who are currently in a relationship, i.e., for the married and common-law women, questions about the relationship are asked in the present. For instance,

cases, samples are selected samples, and given the timing of events, no solution is possible. Similarly, and most importantly, for the separated and divorced women who faced violence, presumably, violence could have been one of the reasons why they ended their relationship and/or initiating the process of separation could have itself increased violence.

The initial exclusion and later inclusion of widowed women allows an examination of intimate partner violence in a way that is not entirely confounded by selection into these groups (remaining coupled or choosing to uncouple). Although there might be associations between socio-economic status and death that might also be associated with violence, for widowed women, their husbands' death is what ends their marriage rather than the woman's decision. The group of widowed women is thus used to test the sensitivity of results to the selection issue.

Dependent Variables The analysis will follow the literature by focusing on the experience of violence as a binary occurrence, i.e., exploring whether an individual has experienced a certain type of violence or not (Bloch and Rao, 2002; Friedemann-Sanchez and Svec, n.d.; Weitzman, 2014) depending on their ownership of resources, childhood experiences and beliefs. There are four main dependent variables that are defined and measured as follows:²⁴

(i.) *Physical Violence:* includes all acts of force that cause physical harm, pain or suffering to the person on whom the acts are inflicted, regardless of the medium used or its consequences. It is a binary variable equal to one if the individual's partner (or most recent former partner) has ever either pushed or pulled from their hair, hit with their hands or an object, attempted to choke, attacked with a knife or switch-blade, fired a weapon or kicked the woman during the duration of their relationship. The following questions were asked to the women and used to create this variable:

• "Now, I'd like to ask you if you have lived or experienced difficult situations

ownership of resources, intimate partner violence, refer to current interactions with their partners. For separated, divorced and widowed women, the timing of the relationship questions refer to the time when the women were coupled with their former partners. In terms of the definitions of the variables in this analysis, all ownership variables pertain to *when the couple is/was still together*. Therefore, that a divorced woman has ownership of a vehicle refers to her having owned a vehicle at the time of her previous relationship that has ended. For a married woman, it refers to currently owning a vehicle with her spouse.

 $^{^{24}}$ Note that the questions from the survey that are used for these variables have been translated from Spanish.
or problems with your current (or former for post-coupled women) spouse or partner throughout your life together. Since your relationship with your (former) spouse or partner started:

- have they pushed you or have they pulled your hair? Yes/No.
- have they hit you with their hands or with an object? Yes/No.
- have they tried to strangle or to choke/asphyxiate you? Yes/No.
- have they attacked you with a knife or switch-blade? Yes/No.
- have they fired a weapon at you? Yes/No.
- have they kicked you? Yes/No."

(ii.) *Psychological Violence:* includes all acts or omissions that cause harm, pain, emotional distress, psychological alterations, or diminish self-esteem of the abused. It is a binary variable equal to one if the individual's partner (or most recent former partner) has ever either insulted, offended or humiliated, threatened to hit, locked up or prohibited visits, threatened with a weapon, threatened with killing the woman, their children or themselves during the duration of their relationship. With the same preamble as above, the questions that are asked to the women that inform this variable are as follows:

- "Since your relationship with your (former) spouse or partner started:
 - have they insulted, offended or humiliated you? Yes/No.
 - have they threatened to hit you? Yes/No.
 - have they locked you up or prohibited you from going out or from you receiving visits? Yes/No.
 - have they threatened you with a weapon (knife, switch-blade, gun or rifle)?
 Yes/No.
 - have they threatened to kill you, himself or the children? Yes/No."

(iii.) *Sexual Violence:* refers to impositions on the exercise of a person's sexuality to whom sexual relations or practices with the aggressor or a third party are forced upon through the use of physical force, intimidation, threats or any other coercive means. It is a binary variable equal to one indicating if the individual's partner (or most

recent former partner) has ever either demanded sex forcibly, forced to perform sexual acts against the will of the woman or used force to have sexual relations during the duration of their relationship. As above, the following questions were used to define this variable:

- "Since your relationship with your (former) spouse or partner started:
 - have they demanded to have sexual relations although you did not want to using force? Yes/No.
 - when having (had) sexual relations, have they forced you to perform things that you do (did) not like? Yes/No.
 - have they used force to make you have sexual relations? Yes/No."

(iv.) *Economic Violence:* refers to the transformation, theft, destruction, withholding of objects, personal documents, property rights and economic resources destined to satisfy the necessities of the victim.²⁵ It is a binary variable equal to one if the individual's partner (or most recent former partner) has ever either destroyed, thrown out or hid the woman's belongings or household belongings or appropriated or taken money or goods during the duration of their relationship. As with the other variables, with the same preamble, the following questions were used to create this variable:

- "Since your relationship with your (former) spouse or partner started:
 - have they destroyed, thrown out, or hidden your belongings or household items? Yes/No.
 - have they appropriated or taken from you money or goods (things, terrains, etc.)? Yes/No."

It is important to note that although the questionnaire tries to measure factual instances of experiencing violence, the harm that is inflicted on the person, and the way in which they process it psychologically is entirely idiosyncratic. Whether violence

 $^{^{25}}$ In line with the World Health Organization's forms of intimate partner violence, violent behaviours are grouped into four broad categories: physical violence, psychological, sexual violence and economic violence. The WHO and PAHO (2012) do not make the distinction of 'economic violence' from 'controlling behaviours' and includes restricting access to financial resources in the same group as isolating a partner or monitoring them. Taking advantage of the emphasis that the survey makes on the use of economic means to inflict violence, and incorporating the Ecuadorian's government distinction of 'economic violence,' in this study, 'economic violence' is a stand-alone category.

inflicted injury or harm, or the way in which women process its social significance may not be captured through these measures of violence. For instance, measuring being 'pushed' is presented as a 'yes' or 'no' question, regardless of whether or not the person assimilated it as a positive or negative experience and depending on what the person understands as 'push.' Holding the force of the movement equal, for some women a horizontal shift by their husband's arm could be a 'push,' for others it may be a 'nudge,' not strong enough to process it as a push, for others it may carry no meaning. Likewise, depending on the social acceptance of violence, or desensitization to violence, a woman may no longer perceive being called names as a way of being 'insulted, offended or humiliated' whereas another woman may perceive it as the complete opposite. The perception from the individual who experiences violence might affect the reliability and interpretation of not only the response to these question but also their meaning. In a way, women may only report experiencing violence if they thought it was significant or harmful, depending on how salient the experience was to them. The debate on whether socially accepted violence or desensitization to violence is perceived by the individual as harmful is beyond the scope of the chapter and perhaps even beyond the scope of the discipline. The assumption underlying the analysis is that, given the nature and efforts of the survey to provide a safe space for women to respond and talk about their experiences, their answers are as truthful as they could be.

The percentage of women experiencing each type of violence is shown in Figure 2.2 and discussed in the following 'empirical results' section of the chapter.

Economic Resources Given that ownership of assets may provide longer-term well-being and more protection to women than income (Phipps and Woolley, 2008), the main variables used to measure economic resources are binary variables defining ownership of four assets: crops, vehicles, savings or a home. These variables are identified as: assets owned just by the woman, assets owned just by the man, jointly owned or some mixed ownership (e.g., if some assets are owned by one spouse while the rest are co-owned, or any other combination) (Menon et al., 2017). In this case, 'ownership' refers exclusively to having a specific asset under the person's name, i.e.,

the owner.²⁶ The variables are defined as follows, at the time of the partnership:

- i. 'all assets owned by the female;'
- ii. 'all assets owned by the male;'
- iii. 'mixed ownership of assets;'
- iv. 'family owns no assets;'
- v. 'all assets are co-owned' (base).

This aggregation of resources allows an examination of whether women's complete ownership or partial ownership of the resources associated with violence, compared to having co-ownership of all assets (reference category). As robustness checks, these resource variables are disaggregated and the base is changed to having no ownership of assets. Ownership of economic resources by marital status is shown in Figure 2.4.

Including whose name these assets are under allows to both explore absolute access to resources, as one can see which assets a person has ownership of, as well as the relative division of wealth in the household. Women in Ecuador tend to have strong property rights and can accumulate assets individually or as part of their conjugal community. Whatever assets are accumulated as a single individual prior to a marriage or union, remain the property of that individual, even after the union is dissolved. Similarly, any inheritance acquired remains the property of the individual who inherited it (El Comercio, 2011; Civil Code, 2005).²⁷ For assets that were shared or obtained during the union, the legal system in Ecuador entitles both parties to 50% of all shared assets following dissolution, which is especially protective of housewives who did not engage in work outside the home. Most women accumulate their individual assets through inheritances, especially crops. Only 1% of unions formulate and agree to a different legally binding contract or pre-nuptial agreements (Deere et al., 2014).

As a measure to control for access to resources, an indicator variable is included of whether the women answered affirmatively to the question "do you have money for personal expenses?" This variable aids in measuring women's access to the most liquid asset, money. Education, employment status and own income are similarly included

²⁶To code these questions, two-part questions from the survey were used: "Are (were) you or your (former) partner the owner of [land or crops/ a car or vehicle/ savings/ the house you live in]?" "In whose name is (was) it [yours, (former) spouse or partner, both of you]?"

 $^{^{27}}$ Inherited assets remain the property of the individual who inherited it following the dissolution of a legal union, even if the asset is legally registered as a shared asset during the union.

in the analysis as research has found an association between intimate partner violence and the socio-economic status of the women (Aguero, 2013; Bloch and Rao, 2002; Svec and Andic, 2018; Bowlus and Seitz, 2006; Weitzman, 2014; WHO and PAHO, 2012).²⁸ Employment status is divided into three categories: employed, unemployed and not in the labour force. The indicator variable 'employed' denotes those individuals who worked in the previous week or who did not work but had a job in the previous week. 'Unemployed' is an indicator variable for those who did not work in the past week but are looking for a job. Lastly, 'not in the labour force' is an indicator variable equal to one if the individual is retired, a full-time student, a caregiver (including those who stated their job was being a housewife), or is unable to work.²⁹ Education is divided into four groups: 'no education' which refers to an individual did not receive a formal education (base group); 'basic education' which refers to completing Ecuador's general basic education of 10 years; 'medium education' which refers to the three years of 'baccalaureate education' (i.e., the individual would have completed high school); and 'technical or university education' which refers to completing a technical, university or graduate degree.

There is a slight difference in the variables for the coupled group compared to the no longer coupled group. In the case of the married and common-law women, as part of their individual characteristics, an indicator variable denoting whether the woman made 51% or more of the household income is included. This variable can only be added for the married women group and not the separated or divorced women. In the first place, there is an identification issue, as established by Stevenson and Wolfers (2009): women who earn more than 51% of total household income are more likely to face a divorce. Second, though perhaps more important, the measure of contribution to household income refers to the time of the interview, with the members of the household that currently live in that household, not prior to a divorce or separation. Therefore, analysing this variable for separated or divorced women will yield no useful information in terms of bargaining power with their former spouse.

As noted earlier, a limitation of the study is the potential of endogeneity. For

²⁸In all regressions, income is scaled, i.e., own income is calculated by adding earnings, capital income, retirement income, gifts or donations, remittances and the BDH transfer and is divided by one thousand dollars, for scaling purposes.

 $^{^{29}}$ The 'employed' and 'unemployed' are the two main categories stemming from the 'economically active population' and the 'not in the labour force' category is the complement of the 'economically active population' category. The latter, 'not in the labour force' is the base category.

instance, having all resources under the husband's name and experiencing intimate partner violence could depict a 'controlling' marriage not a causal relationship between assets and violence. Using whose name the asset is under provides a measure of actual ownership of resources given that the strength of property rights are arguably not influenced by controlling behaviours. Nonetheless, the possibility of women being under extreme duress with their former, nearly former or current partner forcing them to transfer them all assets to his name, remains. It could be that they need to do this due to threats, so that they can lessen current violence, it could be learned helplessness or even an exit strategy to leave the marriage. Unfortunately, other than using these measures for ownership of resources, given the nature of the data, the potential of endogeneity cannot be fully addressed. Another limitation is that these resources reflect wealth accumulation or savings, as well as absolute affluence of the household. Savings also reflect preferences, discount rates and attitudes towards risk (Phipps and Woolley, 2008). These are not captured directly though any variable.

Attitudes and Beliefs The ENVIGMU survey asks respondents about circumstances in which they believe a husband is justified in hitting his wife. Specifically, if the woman believes that a husband is justified in hitting his wife if:

- i. the wife does not obey her husband;
- ii. the wife does not take proper care of the children;
- iii. the wife is unfaithful; or
- iv. the wife goes out too much.

This variable potentially reflects a woman's family values and own-tolerance of violence. Following Jesmin (2017), an 'Acceptability of Wife Beating Attitudes Index' is created, counting the responses of when the woman believes a husband is justified in hitting his wife. The index ranges from '0' to '4' and if the respondent agrees to all of these statements, they are awarded a '4' and if they agree to none, they are awarded a '0.'³⁰

Similarly a 'Strength of Patriarchal/Traditional Belief Index', is created using women's agreement with the statements:

 $^{^{30}}$ Note, the survey asks women their general belief in justifying a man hitting his wife, it does not explicitly ask them if they believe their own partner is justified in hitting them.

- i. 'a good wife must obey all orders from her husband;'
- ii. 'a wife can choose her own friends even if they are disliked by her husband;'
- iii. 'men should be responsible for all family expenses;'
- iv. 'women have equal rights to work and earn money;'
- v. 'it is a wife's obligation to have sex, even if unwanted.'

If the woman strongly agrees with the statement, they are given a '2' as a value, if they agree, a '1,' and if they do not agree, they are awarded a '0'. This is reversed for statements (ii.) and (iv). The index ranges from 0 (least traditional/patriarchal beliefs) to 10 (most traditional/patriarchal beliefs). Whether either justified violence or holding "equitable" beliefs is more reflective of individually- or socially-held beliefs is purely a conjecture derived from the literature (see Jesmin (2017)). The survey does not ask either of these questions "as a whole" or "for you." Therefore, it is only the respondents' interpretation of the question that results in answering personally, or as a representation of the status quo, which is not information directly available from the survey. Beliefs by marital status are shown in Figures 2.5 and 2.6.

Intergenerational Violence To capture the intergenerational aspect of violence and exposure to violence, as in Bowlus and Seitz (2006), I use information from the survey about whether there is violence in the individual's own family background, and violence in spouse's family background. For personal experience, there is an indicator variable, equal to one, if the woman, as a child, was verbally abused (insulted or offended), if she was physically abused (hit) by members of their household; and/or if she was sexually abused as a minor (kissed or touched against their will; forced to undress or show their intimate parts; forced to look at other's intimate parts; or had their intimate parts touched against their will)³¹ For the spouses' childhood experience with domestic abuse, there are two variables: an indicator variable equal to one if as a child, the spouse was directly abused (physically or verbally) and/or if the spouse's father hit his mother. The second variable is a dummy equal to one if there is no knowledge of spouse's childhood violence background since non-responses or lack of knowledge could be correlated with abuse (Bowlus and Seitz, 2006). Note that the

 $^{^{31}}$ The age distinction between child and minor in this set of questions are due to the nature of the survey. When addressing the family context, the survey asks the questions using the timing "in childhood." When addressing the sexual abuse questions, the survey refers to the timing of these questions as "before you were 18," i.e., when the individuals were minors.

women are the ones who answer to question regarding their own childhood experience as well as their spouses'.

Control Variables A first set of controls describe personal characteristics of the woman. These include: age, number of children, and duration of relationship. Given the potential of collinearity between age, age at first birth and age of first marriage, indicator variables for whether the individual married or had her first child as a minor (below the age of 18) are added instead.

For household characteristics, there are controls for family size and whether the household is multi-generational. As previously mentioned, a multi-generation household is one in which at least three generations live in the household. Similarly, an indicator for other non-family members living in the household is included. This indicator combines any other non-related individual residing in the household and live-in maids (3% of the households in the sample have help living in the household).

The social characteristics included are the individual's cultural background (i.e., indicator variables for Indigenous, Afrodescendent, Mestizo (base) and Other (White, Montubio) ethnicities) and an indicator for whether the individual lives in a rural or urban setting. Lastly, location controls are included as province dummy variables, given that they could capture province-level economic opportunities as well as local norms and expectations (Weitzman, 2014; Garcia-Moreno et al., 2005).

Econometric Method The main specification is:

$$IPV_{ij} = \alpha + \theta \text{ ResourceOwnership}_{ij} + \omega \text{ ChildhoodExperiences}_{ij} + \rho \text{ Beliefs}_i + \delta IC_i + \gamma HC_{ij} + \varphi SC_i + X'_i\beta + \varepsilon_i$$

Where, IPV_{ij} is one of the violence measures for a woman i, during their match j;³² ResourceOwnership_{ij} is the set of indicator measures describing the assets ownership for woman i during match j (as described above); *ChildhoodExperiences_{ij}* is the set of indicator variables depicting the domestic violence experience by the woman i and her spouse in match j when they were children; *Beliefs_i*: is the set of indexes describing

 $^{^{32}}$ The unit of observation, are the women who are in a particular type of match. What is being studied is the violence experienced by a woman *i*, while she was in match *j*.

woman's beliefs; IC_i is the vector of individual characteristics of the woman; HC_{ij} is the vector of household characteristics (including length of relationship) depending on the type of match j woman i is in; SC_i is the vector of cultural characteristics for woman i and X'_i is the set of (location) controls, as described above.

Regressions are run for the four different definitions of intimate partner violence using this specification. The analysis is done initially for the full sample of women and subsequently, separately, by whether the women are married/common-law or separated/divorced. I estimate a linear model by ordinary least squares (OLS) and I estimate Probit models, with and without controls. The results presented below are the marginal effects obtained from the Probit regressions with controls for province. Standard errors shown are robust and clustered at the primary sampling units.³³ In all figures and tables results are weighted using the household weights provided by the National Institute of Statistics and Censuses (INEC from its name in Spanish).

2.4 Empirical Results

For married and common-law women, intimate partner violence experience and asset ownership, for instance, pertain to their current relationship. On the other hand, for the divorced and separated women, variables describe the relationship that they have left so intimate partner violence and asset ownership reflect the environment in which they lived before they left the relationship. The only set of independent variables that relate to the time of the survey for both sets of women, are the variables that describe the women's beliefs.³⁴ Widowed women are excluded at this point.

Descriptive Analysis Figure 2.2 shows the proportion of women ever experiencing intimate partner violence. The most common type of violence experienced by Ecuadorian women aged 15 years or above is psychological violence: 60% of the separated and divorced women) and 34% of married and common-law women report experiencing

 $^{^{33}}$ In the case of the ENVIGMU, the primary sampling units are clusters of 30-60 occupied houses that share similar socio-economic characteristics.

 $^{^{34}}$ As previously mentioned, endogeneity may be problematic here. For instance, it could be that women who left their relationship, to justify their (past) divorce or separation, if they were abused, say (currently) that they do not think wife-beating is justifiable, and thus, they left. Whereas, the women who remain in a relationship and are hurt may say that it is acceptable for a man to beat their spouse to self-justify staying. The example, and endogeneity issue extends to all other forms of intimate partner violence. Due to data limitations, I am unable to fully address this concern.

psychological violence in their past and current relationship, respectively. This is followed by physical violence, where, 53% of post-coupled women experienced physical violence in their last relationship and 31% of coupled women experienced it in their current relationship. Sexual violence comes in third place, 29% of post-coupled women and 8% of coupled women having experienced violence in their past and current relationship, respectively. Economic violence is the least common type of violence, though 27% of post-coupled women in their last relationship and 7% of coupled women in their current relationship have experienced economic violence.

Figure 2.3 shows a similar pattern, post-coupled women have higher incidences of violence, even when controlling for years in the relationship, or, 'potential exposure' to violence, i.e., by dividing the indicator variable that the woman ever experienced violence by the length of the match. As in Figure 2.2, the most common type of violence women are exposed to is psychological followed by physical.

Figure 2.4 depicts asset ownership. Again, for coupled women this is asset ownership during their current relationship and for post-coupled women, this was asset ownership during their last relationship. Each bar in Figure 2.4 represents the average proportion of women who have a particular ownership of resources, who have access to personal resources or who received the BDH transfer and are identified as poor. Though half of the divorced and separated women are entirely asset poor, more of them own all assets than do married or common-law women. Nearly a third of coupled women have no assets, for a fifth of them, their spouses own all assets and close to a third co-own all assets with their spouses. For both coupled and post coupled women, around 70% of the women have access to personal resources. A third of all women are poor and receive the BDH government transfer.

Figures 2.5 and 2.6 show the sample means for own current beliefs. To reiterate, these are the beliefs the women themselves (as opposed to their spouses) hold currently, at the time of the survey. The Acceptability of Wife Beating Index shows that separated or divorced women have lower acceptance rates of abuse than do coupled women, however, most women (73% of coupled and 80% of post-coupled) women do not agree with any of the statements justifying wife beating. 12% of coupled and 9% of post-coupled agree that a husband is justified in hitting his wife in one of the circumstances while 4% of coupled and 2% of post-coupled agree that wife beating is

acceptable in all four instances. Similarly, the Strength of Patriarchal or Traditional Beliefs Index seems to point to previously coupled women being less traditional in beliefs than currently coupled women.

Figure 2.7 shows the average incidence of domestic violence in the woman's and her spouse's family background. Domestic violence in childhood is high for all women. The proportion of women who faced violence as children is higher for separated and divorced women than for married and common-law women, 60% and 52%, respectively. The proportion of women whose spouse experienced violence in childhood is slightly higher for the married and common-law women than for the divorced and separated women, 43% and 40% respectively - though, notably, post-coupled women are less likely to know their former spouses' family background of domestic violence.

The means for the control variables are presented in Table 2.1. Currently coupled women are younger, less educated, less engaged in the labour market (and earn less income) than previously coupled women. Households of separated and divorced women have lower household income, are slightly smaller in size, more urban and only 19% have a male household head, compared to 94% of the married and common-law women. Both groups of women have on average three children, gave birth for the first time at the age of 20 (around half of them were mothers as teenagers) and became first partnered around the same age. The average length of a marriage is 21.55 years for the currently partnered women and was 11.77 years for the currently separated or divorced women.³⁵

Main Results The results discussed will focus on the tables of results for Married and Common-Law women and Separated and Divorced women, tables 2.3a. and 2.4a., respectively. All else equal, women who are currently coupled are less likely to report experiencing violence than the post-coupled women.

Married and Common-Law Women: In general, for married and common-law women, economic resources do not seem to be as important as the women's attitudes and beliefs or intergenerational experience. The largest associations between violence and resources come from having access to money for personal expenses. On average,

 $^{^{35}}$ Figures for the different aggregated and disaggregated measures are presented in the Appendix.

having access to money for personal expenses is associated with lower incidences of violence, ranging from - 7.9 percentage points (p.p.) lower for psychological violence to -2.2 percentage points lower for economic violence. Compared to the baseline, co-owning all assets, individual ownership of assets or a mixture of ownership of assets are associated with higher probabilities of all kinds of violence. The magnitude of the coefficients is larger if women own all assets in the household. On average, owning all assets is associated with 6.5 p.p., 10.6 p.p., 5.5 p.p., and 5.4 p.p. higher incidence of physical, psychological, sexual and economic violence for married or common-law women. Asset poverty is only associated with higher incidences of psychological violence.

Earning more than 51% of household income is associated with a 4.2 percentage point lower incidence of physical violence only. On average, a woman who is more accepting of wife beating beliefs is associated with a higher incidence of physical and psychological violence, 2.5 p.p. and 2.0 p.p., respectively. Strength of patriarchal beliefs is correlated with lower incidences of all types of violence. A one-point increase in the strength of patriarchal or traditional beliefs index is associated with a -11.9 p.p. lower incidence of physical violence, a -13.3 p.p. lower likelihood of psychological violence, a -6.0 p.p. lower association of sexual violence and a -5.3 p.p. lower incidence of economic violence.

In terms of magnitude, the largest statistically significant variables are the childhood experiences of violence. Having domestic violence in the women's family of origin is associated with an increased incidence of physical, psychological, sexual and economic violence by 18.9 p.p., 17.9 p.p., 4.8 p.p. and 5.0 p.p., respectively, on average. Having a spouse who experienced domestic violence as a child is associated with an increased incidence of physical, psychological, sexual and economic violence by 18.4 p.p., 20.9 p.p., 5.5 p.p. and 4.5 p.p., respectively, on average. Not knowing their spouses childhood experience with violence is positively associated with physical and psychological violence.

Age, university education and having non-family members in the household are associated with lower probabilities of some types of violence. Identifying as Indigenous or as Afrodescendent, being employed, number of children and relationship length are associated with higher incidences of violence, on average, for coupled women. Separated and Divorced Women: Results for separated and divorced women highlight that having had access to money for personal expenses in their past relationships was much more protective from violence than it is for the married and common-law women in their current relationships. However, the largest negative associations with violence for previously coupled women come from the women holding patriarchal or traditional beliefs. Intergenerational violence also has large positive associations with violence in their past relationship for the divorced and separated women, especially regarding their former spouses' history of domestic violence. All independent variables, with the exception of the belief variables ('acceptability of wife beating index' and 'strength of patriarchal/traditional beliefs'), refer to the post-coupled women's past relationship (belief variables reflect the women's current beliefs).³⁶ All control variables are either time-independent or reflect the women's current state and situation, including her current (at the time of the survey) household's conditions.

Although the direction of the coefficients for both the currently coupled and the formerly coupled are the same for access to money for personal expenses (negative), women being more accepting of patriarchal and traditional beliefs (negative), their own domestic violence experience as children and their (former) spouses' domestic violence background (positive), the magnitudes of the coefficients are considerably larger for the separated or divorced women.

Regarding economic resources, on average, for previously coupled women, having had money for personal expenses in their previous relationship was protective of violence, with associations of lower incidences of violence of -21.7 p.p, -22.5 p.p., -16.8 p.p. and -13.9 p.p., for physical, psychological, sexual and economic violence, respectively. In terms of the ownership of resources, compared to the baseline, coowning all assets, sole ownership of assets is not associated with higher probability of violence, not for women owning all assets, nor for men owning all assets. Compared to co-owning all assets, having some mixed ownership of assets (e.g., the woman owns the house but the spouse owns the car and they both own crops) is associated with lower physical violence by -23.6 p.p., on average. Being asset poor, compared to co-owning assets is associated with lower probabilities of physical and psychological violence by

 $^{^{36}}$ Again, it is worth noting that current beliefs, may have been affected by women's past experience of violence.

-19.1 and -18.4 p.p., respectively.

Current acceptance of wife beating beliefs has no association with violence for separated or divorced women. Strength of patriarchal and traditional beliefs is associated with a lower incidence of violence, in all cases, -30.9 p.p., -38.9 p.p., -27.7 p.p., -30.0 p.p., for physical, psychological, sexual and economic violence respectively.³⁷ Having domestic violence in the women's own family background is associated with an increased incidence of physical, psychological, sexual and economic violence by 13.6 p.p., 15.6 p.p., 9.7 p.p. and 6.9 p.p., respectively, on average, at the time of the women's last relationship. Having a former spouse who experienced domestic violence as a child is associated with an increased incidence of physical, psychological, psychological, psychological, psychological, psychological, sexual and economic violence by 21.1 p.p., 18.1 p.p., 20.6 p.p. and 10.8 p.p., respectively, on average. Unknown violence in their former partners' childhood has no statistically significant associations with violence.

In terms of the controls, the statistically significant variables for the previously coupled women pertain to: size of family being positively associated with the likelihood of physical violence; having basic education (compared to no education) and number of children being positively associated with psychological violence; being employed size of household and living in an urban area are positively correlated with sexual violence while having non-family members living in the household is negatively associated with sexual violence; number of children is positively associated with economic violence, on average.

Robustness Checks

Disaggregation of Variables: First, in order to test whether key results were dependent on the sets of variables included, the variables and indexes are disaggregated. For ownership of resources, the variables are disaggregated by whether most assets are owned by the females, or the males, or co-owned. Specifically, the set of variables describing ownership of resources is organized into independent ownership of resources by the woman and by her (former) partner, joint ownership or lack of ownership, during the duration of the relationship. There are four assets: land and crops, cars

 $^{^{37}}$ Again, whether these beliefs changed or were strengthened after the separation or divorce remains untested and potential endogeneity remains an issue.

and vehicles, savings and homes. The variables are defined as follows, at the time of the partnership:

- i. 'woman has ownership of [asset];'
- ii. 'spouse has ownership of [asset];'
- iii. 'household has no ownership of [asset];'
- iv. 'woman has co-ownership of [asset];' (base)

Disaggregated ownership of assets by marital status is shown in Fig. A.1 to A.3.

For married or common-law women, owning a house is positively associated with psychological, sexual and economic violence, 6.7 p.p., 10.1 p.p., 4.2 p.p. and 2.7 p.p., respectively. If a spouse has individual ownership of a car or vehicle, this is associated with 8.6 p.p. higher likelihood of physical violence. Households who are asset poor, in terms of land and crops, compared to households who co-own assets, have lower incidences of physical (-4.8 p.p.), psychological (-4.7 p.p.) and sexual violence (-2.2 p.p.). Being asset poor in terms of a car or vehicle is associated with a higher incidence of physical violence by 8.0 p.p., sexual violence by 2.8 p.p. and economic violence by 2.0 p.p., on average. Having no ownership of savings is correlated with a 2.3 p.p. increase in the likelihood of sexual violence and having no ownership of a house is associated with a 4.6 p.p. increase in the likelihood of psychological violence, on average, compared to co-ownership of these assets. It is worth recalling that close to 4/5 of the sample is asset poor when it comes to a owning a car/vehicle, savings, and land or crops and close to 1/2 of the sample is asset poor in terms of owning a house.

Separated or divorced women who had land or crops in their name had a lower incidence of physical violence (-28.7 p.p.) during the course of their past relationship. Individually having owned any other resource has no statistically significant correlation to likelihood of violence. If a former partner had individual ownership of land or crops during their relationship, there is an association of -37.3 p.p. and -39.1 p.p. lower likelihood of physical and psychological violence, respectively, compared to having had co-ownership of assets. If the household was asset poor in terms of land or crops during the former relationship, on average, there is a negative correlation of -20.9 p.p. and -22.7 p.p. to psychological and sexual violence, respectively. See tables A.1 to A.2.

The Strength of Patriarchal/Traditional Belief Index and the Acceptability of Wife Beating Attitudes Index are also disaggregated. The variable that seems to be driving most of the effect in the aggregate index of beliefs is the woman's agreement with 'a good wife must obey all orders from her husband.' It is negatively associated with all types of violence for all women. Other than that, departure from patriarchal beliefs is associated with higher likelihoods of experiencing violence. For instance, for married and common-law women, if they believe they can choose their own friends even if their spouses do not approve of them, there is a higher likelihood of sexual and economic violence. In the case of divorced and separated, believing that a wife can choose her own friends is associated with higher incidences of psychological violence while the belief of equal rights to work and earn money is positively associated with physical and economic violence.³⁸ Interestingly, a patriarchal belief that seems to be associated with retaliation from men is when married and common-law women believe that men should be responsible for all family expenses. In the case of coupled women, this belief is associated with higher likelihoods of physical and psychological violence (for post-coupled women, this variable is insignificant for all types of violence). Perhaps, even though this belief aligns with patriarchal precepts, the onus does not fall on the women, it demands a lot of work from the man which may not be welcomed. Similarly, it could put a lot of stress on the males, and this financial worry may result in violence.

In the case of wife beating acceptability, for the married and common-law women, most beliefs have positive associations with at least one type of violence. The only exception is when women believe that 'a husband is justified in hitting his wife if she is disobedient', with that belief being negatively associated with sexual and economic violence. Again, as with the other set of beliefs, results here seem to imply that when a woman does not confront male dominance in her marriage or union, there is a lower likelihood that she experiences violence. For divorced and separated women all disaggregated measures of beliefs regarding acceptance of wife beating are insignificant. See tables A.3 to A.4.

For the violence in childhood measures, the variables are disaggregated as follows:

i. if the individual was verbally abused (insulted or offended) as a child by members of their household;

 $^{^{38}}$ As before, note that these are current beliefs and the experience of violence for the divorced and coupled was during the time of their last relationship so current beliefs might be coloured by their past relationship experiences, including the violence the women might have encountered.

- ii. if the individual was physically abused (hit) as a child by members of their household;
- iii. if the individual was sexually abused (kissed or touched against their will; forced to undress or show their intimate parts; forced to look at other's intimated parts; or had their intimate parts touched against their will) as a minor;
- iv. if spouse had violence in their family background;
- v. no knowledge of spouse's childhood violence background.

For currently coupled women, all types of violence in childhood are positively associated with increased likelihoods of all types of violence in adulthood. The only exception is no statistically significant association between being physically abused as a child with sexual violence in adulthood. For previously coupled women, most types of violence in childhood are positively associated with increased likelihoods of all types of violence in adulthood. The exception is physical abuse as a child, which is positively associated with psychological violence only. For both groups of women, being sexually abused as a child appears to be the most important part of the association between childhood domestic violence and adult intimate partner violence. See tables A.5 to A.6.

Sequential Addition of Variables: Second, the sets of independent variables (i.e., ownership of resources, beliefs and childhood experience) are added in different order exploring sequential addition mediation (Gelbach, 2016).

For married and common-law women, after the sequential addition of variables, the main specification with all sets of variables nearly doubles the pseudo R^2 in most types of violence, the only exception is economic violence. For physical and psychological violence, the magnitude of having money for personal expenses decreases by around 1 p.p. after all variables are included and the association with all asset ownership disappears completely. For sexual violence, the magnitude of having money for personal expenses and sole ownership of resources decrease by 1 p.p. and 2 p.p., each, respectively after all variables are included. See Tables A.7 to A.10.

For separated and divorced women, in all cases of violence, the sequential addition of covariates seem to slightly change the magnitude of coefficients. In all cases, the magnitude for women having money for personal expenses goes down by about 3 p.p. on average when adding all covariates. In the case of women having had ownership of land or crops (for physical violence), of the spouse having had ownership of land or crops (for physical and psychological violence), or if the household had no ownership of land and crops (for psychological and sexual violence) and when the spouse owned a car or vehicle (for economic violence), the inclusion of all three sets of variables increase the magnitude of the coefficients, though the increase is vary, ranging from 0.4 p.p. (for sexual violence and no ownership of land/crops) to 9.5 p.p. (for psychological violence and spouse's ownership of land/crops). For the variables surrounding beliefs, the index for acceptability of wife beating is always insignificant (for all types of violence) and the variable for strength of patriarchal/traditional beliefs increases in magnitude when all covariates are added. Similarly, for all types of violence, the inclusion of all sets of variables increase the pseudo R^2 . Altering the order in which the covariates are added (not shown, provided upon request) does not dramatically change results. These results point to most significant coefficients being relatively stable when different sets of covariates are added sequentially. See Tables A.11 to A.14.

Widowed Women: A third component in the robustness checks is the analysis of widowed women. Women enter the sample of widowed women not by choice, but because their spouse dies. Although their spouse's death might be correlated with socio-economic status, there is presumably no choice made by the woman. That is, widowed women do not select themselves into a sample as the divorced and separated women might. If models estimated for widowed women are similar to those estimated for divorced or separated women, this might provide some reassurance against concerns about selection. Initially, the main specification is run on the sample of 980 widowed women. Further, the sample is restricted to women between the ages of 45-65³⁹ years to make meaningful comparisons, which yields 3,918 women, of which 3,038 are married or common-law, 523 are separated or divorced and 357 are widowed.⁴⁰ Widowed women aged 45-65 have experienced high rates of intimate partner violence: 46% have experienced physical violence, 48% psychological violence, 19% sexual violence and 15% economic violence. See Figure A.8. It is worth noting that the smaller

 $^{^{39}}$ Most importantly, older women are excluded given the possibility of dementia or Alzheimer's disease and a host of confounding factors that might be associated with experiencing violence.

 $^{^{40}}$ Results presented are for the women in the 45-65 year old sample. The results for the main specification and for all robustness checks on the sample of 980 widowed women are not presented, and are available upon request.

samples reduce precision of estimates and that given that the women are older and were partnered with men who died earlier than other men (perhaps due to riskier jobs, or lower health status, or engagement in crime), they are still not an entirely representative sample.

By including widowed women in the analysis, ownership of resources continues to appear protective of violence. All results are shown in Tables A.15 to A.19 with a summary of results in 2.5. For widowed women, in most cases (the only exception being all assets owned by the spouse and sexual violence), asset ownership is negatively associated with all types of violence, compared to having no ownership of assets, Tables A.16 to A.19. It is worth noting, however, how large the coefficients are for asset poverty and violence for widowed women. For instance, when having "all assets are co-owned" as the base category, (Table 2.5), if their family owned no assets, widowed women were 51.6 p.p. 40.7 p.p. 38.6 p.p. and 48.2 p.p. more likely to experience physical, psychological, sexual and economic violence, respectively. This is not the case for either the coupled of post coupled women (see Table 2.5). Arguably, widowed women are more financially insecure than all the other women in this sample.⁴¹ Results suggest that for widowed women, asset poverty is the strongest determinant of violence. As is the case for the coupled and post-coupled women in the all-ages sample, when all assets are owned by the male, there is a positive association with violence, in the case of the widowed, with sexual and economic violence.

Comparing across groups of women, there are few similarities between widowed and post-coupled women. There are no clear patterns of associations with attitudes/beliefs though violence in childhood is associated with higher psychological and sexual violence. Given that widowed women are different in that they lack self-selection into staying in their relationship or leaving it and the fact that they are also older, the insignificance of the belief variables could be explained in two ways. On the one hand, it could be the case that the strength and significance in these variables of the main set of regressions is only due to women forming or strengthening their beliefs due to self-selection, i.e., their beliefs are determined by their decision of staying or leaving their partnership. For instance, for married women, both in the main and the 45-65 year old samples, acceptability of wife beating is positively associated with physical and psychological

 $^{^{41}}$ Widowed women's own monthly income is higher (around \$15 USD) than all other women that age, though their household monthly income is much lower (\$259 USD) than all the other women

violence while for the separated/divorced and widowed these variables are insignificant. It is not that widowed women experience less violence (as a whole, they experience even higher incidences of violence than married/common-law women but less than divorced/separated) so it could be that in the case of married/common-law women, their acceptance of wife-beating and them remaining in a potentially abusive marriage could be a way to normalize abuse and/or there being cognitive dissonance. As it is the case for divorced and separated women, for women who are widowed there are no associations with believing it is acceptable for a husband to beat his wife perhaps since they no longer need to tolerate any abuse. On the other hand, it could be that, being older, accepting wife beating was 'of the times' for the cohort and it carry no meaning or real consequence to them as it could've been the norm to be more tolerant to abuse. Nonetheless, the age difference that the married/common- law and separated/divorced have with the widowed is on average slightly over four years, not enough of an age gap to think of generational differences (and cohort effects are explored, with no suggestion of there being any). Further, these beliefs are statistically significant for two types of violence for married women, which does not fully support this second potential option. Therefore, the story that widowed women tell appears to be one where intimate partner violence is strongly associated with resources and poverty as well as the intergenerational transmissions of abuse.

Exposure to Violence: Fourth, the dependent variables are modified to take into account the length of a relationship (and thus the potential exposure to violence) that a woman has (or might have had). The main specification controls for the length of partnership.⁴² However, it could be that someone who is 55 years of age and has been married for 30 years would have a longer potential exposure to violence than someone who is 25 and has been married for less than a year.⁴³ To explore incidence

 $^{^{42}}$ The potential length of partnership being collinear with age is recognized. As previously mentioned, various non-linear forms of age where explored and they were always insignificant.

 $^{^{43}}$ Similarly, there is the potential of having cohort effects that are different from age effects. For instance it could be that older women experience more violence because they have longer exposure to violence or because violence has escalated or because societal norms are more accepting of violence. To test for this, regressions were run where dummy variables for age groups (Age 15-24 years; Age 25-34 years; Age 35-44 years; Age 45-54 years; Age 55-64 years and Age 65+ (base)) were included using a piece-wise linear approximation. For all cases of violence, for all groups of women, married and common-law, separated and divorced and widowed, the age group variables were statistically insignificant, the only exception being physical violence for married women for which there seems to be a decline in the cohort specific experience of violence as the younger cohorts have a lower probability of physical violence compared to women who are 65 years of age or older.

of violence as a function of duration of a match, a new variable is defined, *Exposure* to [Type of] Violence. In this case, the binary observation of all forms of violence, physical, psychological, sexual and economic violence, is divided by the duration of a match.⁴⁴ For instance, if a married women experienced physical violence (=1) over her 5 year marriage, this new variable would be 0.20 (1/5).⁴⁵ In this case, the model aims to capture if there was ever violence, divided by the number of years for which the person was exposed to the potential of violence. Similarly, in order to take into account that all variables are left censored at zero, with women who never experience different types of violence, the main regression is run using a Tobit.

Tables A.20 to A.23 look at the *potential exposure to violence* for the currently coupled and the previously coupled women. All results appear to be consistent with the results from the main specification. Given that all of these regressions take into account the length of the relationship, it seems that results are robust and are not sensitive to the duration of a relationship.⁴⁶ Tables A.20 to A.21 present the OLS estimates while A.22 to A.23 present the Tobit estimates with left censoring at zero for women who have not experienced violence. Again, results appear robust as ownership of resources seem to be protective of violence, beliefs and domestic violence histories are consistent with the main results for both coupled and post-coupled women. When taking into account the left censoring for the separated and divorced, the variables denoting having money for personal expenses and those denoting violence in childhood are statistically significant again.

Other Considerations: Various income measures are explored to test the sensitivity of the model to a specific definition of the variable, income (raw) these include household income standardized, log of income as well as using a poverty measure.⁴⁷ Results are

 $^{^{44}}$ In the survey, the women who were in the match for under a year were coded as having a length of their relationship equal to zero, '0.' So that this does not inflate the coefficients, for all women who are married less than a year, their length of relationship is coded as six months or 0.5 years.

 $^{^{45}}$ Looking at exposure to violence could resemble hazard models in the labour economics literature, e.g., hazard models of duration of unemployment spells or of being in social assistance programs. Although it could be plausible to try and find the duration of the marriage through these models, in this chapter duration of marriage is taken as given, and controlled for the years in the relationship.

 $^{^{46}}$ An exception being the OLS estimates for separated and divorced women, with having access to money for personal expenses being statistically insignificant for all types of violence, though this is not the case for these women with Tobit estimates.

 $^{^{47}{\}rm The}$ poverty measure is whether the household receives the BDH transfer, equivalent, at the time of the survey, to \$35 USD per month.

not sensitive to the income measure used. Similarly, interaction terms are used to explore whether the estimated effects differ by the age and education of the female. Arguably, it could be that the opportunity cost of leaving an abusive relationship increases with age (i.e., assets accumulate over the life course; so perhaps it is less likely of an issue in their relationships for the young women who own assets, as it could be the case for older women who might have accumulated more assets) or depending on women's education. Nonetheless, when interacting assets with age groups or when interacting assets with education, most of the interaction terms are statistically insignificant.⁴⁸

In order to simplify presentation, for the robustness checks, tables are divided only into married and common law and separated and divorced women (i.e., the tables for 'all women' are excluded). All results for these robustness checks are presented in the Appendix.

2.5 Discussion

Women in Ecuador are repeatedly exposed to violence. Exposure starts as children, with over half of all women having had experienced domestic violence in their childhood and close to half, around 40%, of all male spouses also having had domestic violence in their family background. Violence later in life remains, on average 46.20% of all women (40.5% coupled and 65.8% of previously coupled women) have experienced some type of violence as adults. Consistently, results point to the largest associations with physical, psychological, sexual and economic violence coming from women's own patriarchal and traditional beliefs and from their own or their (former) partner's domestic violence background. Compared to economic resources, and the focus placed

⁴⁸When looking at the interaction terms of asset variables and women's age, the only interaction terms that are statistically significant are men owning all assets interacted with age groups. In this case, compared to the baselines of co-ownership of all assets and the age group 15-24, all assets being owned by the male is negatively associated with the probability of experiencing violence, especially when looking at older age groups (55+) and at sexual violence. Results suggest that the positive impact of men owning all assets is concentrated on older men, which is in line with the idea that assets are accumulated over a lifetime and also in accordance with the life stage by which assets accumulate in Ecuador (see Fig. 2.8). In the case of interactions between assets and education, the only statistically significant variables are the interaction terms of women owning all assets and having higher levels of education (completing high school and above). Compared to the baselines (no education and co-ownership of all assets), women owning all assets and having higher levels of education is positively associated with higher probabilities of violence. Both results support the idea that married women may face backlash when going against cultural norms (e.g., being well-positioned resource-wise) and may be protected when abiding by them (e.g., the man being the sole owner of resources, especially so for older generations).

on them in the economics literature, the intergenerational aspect of violence and the belief/attitudinal aspect of violence are strikingly larger.

Overall, the findings are somewhat mixed when it comes to asset ownership. Compared to the baseline of having co-ownership of assets, any other type of ownership is associated with higher violence for the married and common law women. For the divorced and separated, mixed ownership of assets is associated with lower incidences of physical violence and asset poverty is associated with lower probabilities of physical and psychological violence, compared to co-owning all assets.

Looking at the disaggregation of ownership of assets, for all women, having land or crops in the woman's name seems somewhat protective of intimate partner violence, while (for married women only) having a house in her name is positively associated with intimate partner violence. Land or crops is the only asset in which women's sole ownership of it is associated negatively with violence.⁴⁹ For married and commonlaw women, being asset poor in terms of a car or vehicle is associated with higher probabilities of all types of violence, except psychological. Given that four in every five households do not own a car, this correlation affects a large majority of the women in the sample.

Having savings in a woman's name increases the likelihood of sexual violence for married women only. Perhaps this reflects an intense form of 'discrete' male backlash. Men could for instance, seeing that their wives hold a safety net outside of the home, resort to attempting to get control via sexual means. However, married women's own saving capacity is not very large, 5.39% of married women hold savings in their own name.⁵⁰ Perhaps it could be that it is so against the norm to have a safety net on her own that their spouses react in such a negative way to 'restore' control. Since women's ownership of a home is associated with an increased incidence of all types of violence for married women, it could also be a reflection of 'male backlash' potentially due to more relative power shifted towards the wife as having the same assets in the man's name or being asset poor in terms of owning a house have virtually no impact

 $^{^{49}}$ However, this large association affects only a very small group of people as only 4.56% of coupled women and 6.71% of previously coupled women own(ed) land or crops during their relationships.

⁵⁰Although there is no data available on the amount of savings a woman has, women who have savings in their names have higher monthly incomes. The average monthly income for women who own savings is \$300.23 USD per month, while for women who do not own savings their monthly income is \$141.41 USD, on average. Going by the family basket being \$548.63 USD in 2011, neither group of women would've had much saving capacity. Compared to the \$264.00 USD per month minimum wage, women who own savings have incomes of about \$36 USD higher than the minimum wage, on average.

on intimate partner violence. Given that this is the asset that women are most likely to own on their own (a little over one in every ten married women have the house in her name), it could also be the most common asset that men are able to retaliate against with negative effect and potential backlash being the largest observed.

Interestingly, for the currently coupled, not all kinds of relative bargaining power shifted towards women affects them negatively with backlash. The measure of contribution to household income, indicating that women earn 51% or more of the household income is slightly protective of violence, having a negative association with physical violence of 4.2% for married women.

Access to money for personal expenses is negatively associated with incidence of violence for all women, but strongly so for the separated and the divorced. A reason for this may be that ownership of assets requires accumulating wealth, which is out of reach for most people in Ecuador. Just in terms of saving capacity, the proportion of previously coupled women who owned savings during their last relationship is 8.1%, a stark contrast to the 58.8% of Ecuadorians that the government reports. In fact, for the main sample, only 18.9% of households own savings.

It could also be due to the fact that wealth accumulation takes time and women are still relatively young (on average, women are 41.46 years old for coupled and 43.33 for previously coupled) and the main sample analysed excludes the widowed women who potentially had more time to accumulate assets and own them at the time of the survey.⁵¹ On average, only one half, 52.85%, of all unions own a house. By looking at asset ownership by age (see Figure 2.8), most households are able to own their house eventually. By the time women are 55-64 years old, 72.6% of households own a house. Unfortunately, there is no information available on the value of these houses for the households in the survey. Additionally, a house is pretty much the only asset that unions get to own.⁵² Close to 80% of all households are entirely asset poor when it comes to land and crops, savings and cars and vehicles. In fact, 32.5%, close to a third of all households, are entirely asset poor and have no ownership of any asset.

Likewise, money is the most liquid asset. Therefore, when it comes to escaping a

 $^{^{51} \}rm Widowed$ women though are in worse economic situations than their married/common-law or divorced/separated counterparts.

 $^{^{52}}$ The second most commonly owned asset would be owning land and crops. Peak ownership of land/crops happens when women are above 65 years of age where 35.9% of unions own crops or land. Other than that, peak ownership of vehicles happens when women are 45-54 where 24.5% of marital estates include a car or vehicle. The last peak in ownership would be for savings, where 22.8% of unions have savings and it occurs when women are 25-34 years old.

situation of violence, in the bargaining framework, threat points may be more sensitive to more liquid assets than not. It is presumably easier to take a wad of money (or perhaps a debit and a credit card for the more affluent) and flee, than it is to use or sell a crop or a house as means of protection. Changes to inheritance laws in India suggest that when women have improved inheritance rights, there is a positive impact on women's security but not on their autonomy (Amaral, 2017). Perhaps these findings are suggestive of a reverse scenario, where access to personal resources and crops (which have the potential to supply the owner with at a basic level food) provide autonomy but not security since sole ownership of other more durable assets have no or detrimental effects when it comes to violence. Assets are hard to quickly liquidate in value and particularly for crops, there is a seasonal component to them and even their value may depend on annual yields. Maybe it is the case that to flee violence women require more (short-term) autonomy than (long-term) security and perhaps the reason that this is the case is because this is not a really affluent population. As such, women may have to make do with what little they have to leave an abusive marriage or relationship.

In terms of beliefs, strongly adhering to patriarchal or traditional beliefs is negatively associated with all types of violence for both coupled and post-coupled women. The belief that appears to be the most protective of violence is women believing that a good wife must obey all orders from her husband (this is the case for both currently and previously coupled women). Although it may seem counter-intuitive at first, if a woman believes a good wife 'obeys all orders,' she is probably more likely to, in fact, obey her husband's orders and requests and thus, avoid confrontations and perhaps, consequently, avoid violence. Similarly, women's sense of identity may be strongly defined by 'obeying.' Simply looking at naming conventions following a marriage, if women choose to 'take on' their husband's name, they do not just take their last name as it is the case in some western societies, e.g., Women's [First Name] [Last Name], changes to [Women's First Name] [Husband's Last Name]. Rather, their name changes to 'of' their husband's last name, e.g., it changes to [Women's First Name] of [Husband's Last Name].⁵³ The underlying sense of ownership that remains through

⁵³For instance, say the woman's name is Susana Chiriboga and her husband's name is Fernando Peñaherrera. Following their marriage, the convention would be to change her name to Susana Chiriboga 'of' Peñaherrera or simply Susana 'of' Peñaherrera. In Spanish: Susana Chiriboga de Penaherrera or Susana de Peñaherrera, respectively.

the naming convention may be reflective of women's sense of identity as 'belonging to' their husbands or of men's 'entitlement' over their wives. Therefore, 'obeying' may simply be what is expected of women as a vestigial remain from a perceived entitlement in the relationship.

The same would be true for feeling obliged to have sexual relations even if unwanted, the other protective belief for coupled women for both psychological and economic violence (it is statistically insignificant for previously coupled women). Acting on that belief could be protective and "allow" control to "return" to the man and re-establish or maintain harmony in the household. Further, violent individuals tend to be less confident in their ability to resolve their problems in ways that do not involve violence (Orpinas, 1999). By 'obeying' or 'indulging' their husbands, women may "resolve" requests or conflicts that way, through obedience and sex, avoiding violence.

Interestingly, although in line with traditional norms, for married women, believing that a husband should be responsible for all family expenses is also positively associated with higher incidences of physical and psychological violence. So the caveat might be, that patriarchal or traditional beliefs can be protective of violence, as long as they are convenient for the males. This perhaps alludes to current wives demanding economic resources from their spouses and it being received with abuse from their spouse.⁵⁴ Intriguingly, believing in financial dependence has no statistically significant associations for divorced or separated women.

Women's belief that 'women have equal rights to work and earn money' is positively related with psychological violence for coupled women which could indicate backlash from women voicing their opinions, and with physical and economic violence for post-coupled women. These results support the pioneering hypothesis. Women's belief in equality of work may still result in abuse from their husbands as it could be that the societal norms or that the male population have not caught up with women believing in equality. The same can be the case when it comes to the independence of separated and divorce women. Previously coupled women believe in associating with the friends they choose to, even if disliked by their spouses.⁵⁵ Autonomy in their social networks,

 $^{^{54}}$ On the one hand, it could be that a wife being completely financially dependent on a husband is a notion rejected by husbands, which leads to violence. It could also be that men resent that they are responsible for all expenses, and as women spend more money (on themselves, or their children, see Lundberg and Pollak (1996); Phipps and Burton (1998)) resentment grows and escalates to violence.

 $^{^{55}}$ Again, it is worth recalling that these are the associations between current beliefs with past experiences of violence.

thus, seems to have been confronted with psychological backlash. These findings seem indicative of supporting the literature that women with more liberal ideas tends to be associated with having had experienced violence (Jewkes, 2002). In line with the results from the World Values Survey, where 60% of people surveyed believe that if the mother works, their children suffer and where 59% of people believe being a housewife is satisfactory (Inglehart et al., 2014), the results support the premise that Ecuadorian society at large is traditional and patriarchal if women themselves hold these kind of beliefs. Especially because these beliefs may be strongly ingrained in Ecuadorian women. Therefore, disrupting the status quo and breaking from patriarchal beliefs may be hard to do especially when they seem so closely linked to women's own identity and norms.

Alternatively, increases in the 'acceptability of wife beating' index is associated with higher incidences of both physical and psychological violence for married women only. This might be an indication of how tolerant the coupled women are to intimate partner abuse. Aggregated, this variable is not statistically different from zero for the divorced and separated nor for the widowed women. The disaggregation of this variable provides further evidence supporting the idea that women's "obedience" is likely protective of violence. When married and common-law women believe that a husband is justified in hitting their wives when disobedient, there is a negative association with sexual and economic violence. Similarly, the backlash that seems to exist from women's autonomy is also present for the coupled women. If married or common-law women believe a husband is justified to hit his wife if they go out too much, there is an increased likelihood of physical and sexual violence. In general, for the coupled women it seems that believing that violence against them is justified is associated with higher likelihoods of violence. Unfortunately, in the absence of a causal mechanism, it is unclear in which direction the acceptance and normalization of violence and the experience of violence affect each other. However, that there is an association for the married and common-law in and of itself is important, especially due to the implications that this may also have for future generations if children witness or grow up in a violent household.

The experiences of domestic violence as a child and the individual's spouse's experience of domestic violence in childhood have the largest associations with all types of violence. For all women, and especially so for the separated and divorced, the magnitudes of the coefficients are larger for the variables denoting their spouses' experience than the variables denoting their own, i.e., they are more likely to be (to have been) a victim of intimate partner violence if their spouse experienced or witnessed violence as a child than them being themselves victims or witnesses of domestic abuse during their own childhoods. This finding is consistent with the literature. Further, across women, the divorced and separated group have higher magnitudes for both types of domestic violence backgrounds than do married or common-law women. On average, there is a stronger association in the intergenerational transmission of violence and violence in their past relationship for the divorced and separated women than for the married and common-law women in their current relationship. Abuse is strongly associated with divorce and men observing domestic violence are likely to repeat those behaviours by abusing their wives (Bowlus and Seitz, 2006; Pollak, 2004). The importance of childhood experiences is supported by the sample of widowed women. Without exception, for all types of women, violence in childhood whether in their own family or in their spouse's family is strongly correlated with abuse. Consequently, there is strong evidence suggesting that violence perpetuates itself across generations.

As previously mentioned, the coupled and post-coupled women may self-select into their respective groups while widowed women, to a large extent lack self-selection. The decision to leave or stay in a marriage could depend on whether women experience violence or not. For the case of the widowed, their husband 'leaves,' not them, due to their death. In this case, comparing the main results with the widows' results, evidence is suggestive that it is mostly the absence of assets that is related to violence and that the spouse's childhood experience of violence is strongly associated with physical, psychological and sexual violence. The women's own childhood experience is positively associated with sexual violence. The only belief variable that remains significant with the widowed is the negative association between women's strength of patriarchal/traditional beliefs and sexual violence. In addition to the fact that for coupled women acceptability of wife-beating is associated with physical and psychological violence, whereas this is not the case for the post-coupled or widowed, the danger of remaining in a violent marriage is heightened as it may normalize the violence not only for these women, but also for their children, both males and females alike. If these children learn from their parents how to socialize with their partners or their own future children, they may not have the resources to stop the propagation of violence in their own relationships.

It is important to recognize that the narrative or belief that accepts and normalizes abuse could be a means of protection that women use to address their cognitive dissonance in staying in an abusive marriage. Especially if violence has been normalized intergenerationally, it could be conceived, at a society level, simply as a 'cultural norm.' Corporal punishment to children remains common, spanking, pinching, flicking at children are normal occurrences in 'child rearing.⁵⁶ Thus, for some families, if violence is 'normal' socially, a spouse inflicting violence may not be reason enough to end a relationship, especially in a country that is 73% Catholic, very traditional and ending a marriage is still taboo. As the sayings go, "better to be dead than divorced" or "though he hits, though he kills, my husband he is."⁵⁷ Even in the absence of the religion factor, and even the absence of violence being widely accepted socially, women may see a divorce as a failed marriage instead of an escape of violence and the fear of 'failure' may trump the 'pain' of abuse. The danger is that violence that was perhaps invisible (e.g., psychological) could eventually escalate into femicide, and it is truly 'death that does them part.' Further, social norms establish the ways in which social pressure is exerted, if the social pressure is tilted towards 'remaining married' as the social costs of divorce or separation are too high, higher than the cost of violence and women may change or disassociate what is seemingly 'good' or 'better' for them and stay in violent relationships.

2.6 Conclusion

Intimate partner violence is largely widespread worldwide and especially so in Ecuador. Six in ten women in Ecuador will experience violence in their lifetime (MJDHC, 2018). The purpose of this chapter was to provide a comprehensive view of intimate partner violence in Ecuador. This chapter does so by exploring the relationship between having assets in a woman's name at the time of their relationship,

 $^{^{56}}$ One in three, or 33% of all children receive corporal punishment if they are not 'obedient' or if they are not 'behaving' (El Comercio, 2018).

⁵⁷In Spanish: "primero muerta que divorciada" and "aunque pegue, aunque mate, marido es."

their beliefs and attitudes towards wife-beating and patriarchal/traditional norms and their histories of domestic violence with the likelihood of them experiencing physical, psychological, sexual and economic intimate partner violence. The chapter contributes to scholarship by integrating the main ongoing themes regarding the relationship between economic resources and intimate partner violence, providing the analysis a thorough and multidisciplinary lens that also includes social norms and beliefs as well as the intergenerational transmission of behaviours.

Results show that conditional on having assets, these assets being in the women's names are only somewhat protective of intimate partner violence. The most protective resource for women appears to be access to money for personal expenses. It seems that, in the bargaining framework, threat points for Ecuadorians are more sensitive to the most liquid asset which is further substantiated by the fact that this is not an exceedingly affluent population. Similarly, for Ecuador, when it comes to ownership of assets and enforcing property rights, the results suggest the law is not biased towards or against any of the genders. There is no evidence of any systematic relationship, positive or negative between ownership of assets and economic abuse from an intimate partner.

It is worth highlighting that most economic models tend to focus on improving women's outside options in case a relationship fails. Usually, the way in which this is presented and studied is in terms of the economic resources and assets available to women like transfers, earnings, or labour market conditions as well as laws, services and institutions. All of these provisions for women are important in their own right and they may even allow opportunities for clear policies to be put in place. However, in the case of Ecuador, in the context of intimate partner violence, social norms and beliefs as well as the intergenerational transmission of violence really loom large and have the largest associations with violence. Women's 'traditional' beliefs supporting obedience to husbands are protective of all types of violence. Similarly, having domestic violence in both a woman's own family background and in her (former) partner's family background is one of the strongest predictors of experiencing intimate partner violence. In fact, that the spouse was a victim (direct exposure to domestic violence) or a witness (indirect exposure to intimate partner violence) of a violent family environment is associated with an even higher likelihood of all types of intimate partner violence. This is particularly relevant for policy implications. Pollak (2004) argues that there is an amplification effect to implementing policies in the short-run aimed at the incidence of intimate partner violence, given the intergenerational effects of intimate partner violence. Due to the permanence of these behaviours across generations, aiming to reduce them today could further reduce the incidence of intimate partner violence for future generations. It is thus imperative to identify children who are experiencing domestic violence and individuals who have been abused in childhood to recognize their vulnerability for any sort of intervention. Therefore, it is not only necessary to protect women in abusive marriages but also to prevent future cycles of violence from occurring from children who have experienced or witnessed abuse.

An important limitation of this study is that, similar to previous research (Cools and Kotsadam, 2017), the data and the analysis do not include information on the male's (or society's at large) beliefs associated with tolerance and acceptance intimate partner violence. In this chapter, one can only assume that own-beliefs regarding social norms and tolerance of wife-beating are a reflection of the gendered or macro-level beliefs. Similarly, it is worth noting that when it comes to intimate partner violence, under-reporting is usual and can be due to various factors including "fear, shame or denial" (Bowlus and Seitz, 2006, p. 1116). Given the sensitive nature of the questions, there is a tangible likelihood that there was under-reporting and the associations may be larger than what the chapter presents. Another limitation is that the survey is planned as repeated cross-sections and so far, I was unable to identify a policy change or exogenous change that could allow for an instrument or a natural experiment. Ideally, the same women would be followed to examine any change that could happen, before and after a marriage, before and after abuse, the evolution of beliefs, etc. It is thus, imperative to continue gathering information on both domestic and intimate partner violence as this only gives a picture and to fully understand and address the problem, it would be extremely beneficial to have a more detailed, historical accounts of the same women. Therefore, the research frontiers lay in the improvement and investment in more data and in the continuation of multidisciplinary, intersectional and integrated research on intimate partner violence.

	Wo	All Married & Separated & Vomen Common Law Divorced		rated & orced		
	Mean	St. Dev.	Mean	St. Dev.	Mean	St. Dev.
	(1)	(2)	(3)	(4)	(5)	(6)
Relationship Status: Coupled	0.85	0.36	1.00	0.00	0.00	0.00
Woman's Ago	41 75	14 54	11.00	14.56	13 33	1/1 30
Woman's Edu Loval: Basic Education	41.75	0.50	41.40 0.55	0.50	40.00	0.50
Woman's Edu. Level: Madium Education	0.00	0.00	0.00	0.50	0.44 0.27	0.30
Woman's Edu Loval: Technic or Univ. Education	0.24 0.17	0.40	0.25	0.42	0.27	0.44
Woman's Activity: Employed	0.17	0.30	0.10 0.37	0.37	0.25	0.42
Woman's Activity: Unomployed	0.41	0.45	0.57	0.40	0.02	0.49
Woman's Activity: Onemployed Woman's Activity: Not in Labour Force	0.11	0.51	0.11 0.52	0.51	0.03	0.28
Woman's Ethnic Identity: Mestiza	0.40	0.50 0.45	0.52 0.73	0.50 0.45	0.23 0.72	0.40
Woman's Ethnic Identity: Indigena	0.07	0.10	0.10	0.10	0.06	0.10
Woman's Ethnic Identity: Afrodescendent	0.07	0.20 0.25	0.07	0.20 0.25	0.00	0.21
Woman's Ethnic Identity: Other Ethnicity	0.08	0.20 0.27	0.08	0.20 0.27	0.06	0.28
Woman's Monthly Income (From All Sources)	149.33	315.76	135.31	319.85	227.00	279.55
Household Receives BDH Transfer	0.33	0.47	0.33	0.47	0.29	0.45
Household Income	637.70	679.48	649.41	692.18	572.80	600.18
Household Equivalent Income	305.15	339.30	309.69	347.89	280.61	287.22
Income Prop. to Household Income	24.93	30.07	20.00	24.74	52.27	40.52
Woman Earns 51% or More of Household Income	-	-	0.11	0.31	-	-
Woman Has Children	0.96	0.20	0.96	0.20	0.97	0.18
Woman's Number of Children	3.02	1.71	3.06	1.72	2.77	1.65
Age at First Birth	20.49	4.34	20.51	4.32	20.42	4.46
Woman Had First Child as Teenager	0.46	0.50	0.46	0.50	0.49	0.50
Age When First Legally Partnered	19.90	4.71	19.93	4.71	19.76	4.71
Woman Partnered as Teenager	0.55	0.50	0.55	0.50	0.56	0.50
Woman's Marriage/Relationship Length	20.05	14.28	21.55	14.44	11.77	9.94
Size of Family in Household	4.68	2.04	4.74	1.98	4.34	2.31
Household is Multi-Generational	0.29	0.45	0.26	0.44	0.44	0.50
Non-Family Members Live in Household	0.03	0.18	0.03	0.18	0.03	0.18
Male Head of Household	0.83	0.38	0.94	0.24	0.19	0.39
Household in Urban Area	0.72	0.45	0.70	0.46	0.84	0.37
	N =	12,202	N =	10,801	N =	1,401

 ${\bf Table \ 2.1: \ Descriptive \ Statistics: \ Means \ Values \ by \ Marital \ Status}$

 Table 2.2: Percentage of Women Who Have Experienced Violence in Current or Past Relationship by Age Group

	Physical	Psychological	Sexual	Economic	
	(1)	(2)	(3)	(4)	
Age 15 to 24	30.77	33.36	6.70	7.40	
age 25 to 34	34.03	36.98	8.84	9.11	
Age 35 to 44	35.06	39.51	11.56	9.95	
Age 45 to 54	39.51	41.53	14.43	9.57	
Age 55 to 64	40.45	42.42	15.35	11.33	
Age 65+	40.58	42.94	12.39	9.29	



Figure 2.2: Ever Experienced Intimate Partner Violence by Marital Status

Note: For post-coupled women, this refers to violence experienced during previous relationship. **Source:** National Survey on Family Relations and Gendered Violence Against Women

Figure 2.3: Ever Experienced Intimate Partner Violence Divided by Years in Relationship (Exposure to Violence) by Marital Status



Note: Exposure to violence is defined as experiencing a type of violence, (0/1), divided by the duration of the match.
 For post-coupled women, this refers to violence experienced during previous relationship.
 Source: National Survey on Family Relations and Gendered Violence Against Women



Figure 2.4: Ownership of Economic Resources by Marital Status (Aggregate Measures)

Source: National Survey on Family Relations and Gendered Violence Against Women



Figure 2.5: Strength of Acceptability of Wife Beating (Index) by Marital Status

Note: Each value in the index is the count of times in which a female agreed with a statement justifying wife beating (if a wife is disobedient; takes improper care of children; is unfaithful; goes out too much), i.e., a value of '0/4' refers to not agreeing with any of the statements justifying wife beating while a value of '4/4' refers to agreeing with all four statements justifying wife beating.

Source: National Survey on Family Relations and Gendered Violence Against Women



Figure 2.6: Strength of Patriarchal/Traditional Beliefs (Index) by Marital Status

Note: Each value in the index is the strength of the agreement to patriarchal/traditional values (a good wife obeys; a wide cannot choose her friends, men are financially responsible, women do not have equal rights to work; it is wife's obligation to have sex with their husbands). For each of the statements, women agree with the statement, receiving a value of '0' for least agreement, '1' for agreeing, and '2' for strongly agreeing with the statement. Each category represents the total count of strength of agreement for each women regarding those values. Source: National Survey on Family Relations and Gendered Violence Against Women



Figure 2.7: Domestic Violence in Family Background (Aggregate Measures) by Marital Status

Source: National Survey on Family Relations and Gendered Violence Against Women



Figure 2.8: Ownership of Assets in Marital Estate by Age of Women

Source: National Survey on Family Relations and Gendered Violence Against Women
	Physical	Psychological	Sexual	Economic
	Violence	Violence	Violence	Violence
	(1)	(2)	(3)	(4)
Woman Has Money for Personal Expenses	-0.053***	-0.079***	-0.037***	-0.022***
All Assets Owned by Female	0.065**	0.106***	0.055***	0.054***
All Assets Owned by Male	[0.032] 0.050**	0.070***	[0.019] 0.024^{*}	0.023**
Mixed Ownership of Assets	[0.024] 0.043^{*}	0.023	[0.015] 0.030** [0.012]	[0.011] 0.023*
Family Owns No Assets	[0.025] 0.024	[0.024] 0.047**	[0.013] 0.009	[0.012] 0.011 [0.010]
Woman's Monthly Income (Scaled)	[0.021] -0.004	-0.010	$\begin{bmatrix} 0.010 \end{bmatrix}$ 0.003	0.006
Woman Earns 51% or More of Household Income	[0.030]	[0.028]	[0.010]	$\begin{bmatrix} 0.009 \end{bmatrix}$
	-0.042*	-0.009	-0.006	0.012
Woman's Acceptability of Wife Beating Index	[0.022]	[0.021]	[0.011]	[0.011]
	0.025***	0.020***	0.003	0.003
Woman's Strength of Patriarchal/Traditional Beliefs	[0.007]	[0.007]	[0.003]	[0.003]
	-0.119***	-0.133***	-0.060***	-0.053***
Violence in Women's Family Background	[0.045]	[0.045]	[0.021]	[0.016]
	0.189***	0.179***	0.048***	0.050***
Violence in Spouse's Family Background	[0.016]	[0.016]	[0.008]	[0.007]
	0.184^{***}	0.209***	0.055***	0.045***
Unknown Violence in Spouse's Background	[0.019]	[0.017]	[0.009]	[0.008]
	0.062***	0.079***	0.005	-0.001
Woman's Age	[0.017]	[0.017]	[0.008]	[0.006]
	-0.007***	-0.005**	-0.001	-0.003**
Woman's Ethnic Identity: Indigena	[0.003]	[0.002]	[0.001]	[0.001]
	0.065**	0.039	0.040***	-0.001
Woman's Ethnic Identity: Afrodescendent	[0.032]	[0.035]	[0.015]	[0.013]
	0.043	0.119***	0.014	0.043*
Woman's Ethnic Identity: Other Ethnicity	[0.037]	[0.037]	[0.020]	[0.023]
	-0.040*	-0.004	-0.007	-0.017**
Woman's Edu. Level: Basic Education	[0.023] 0.002	[0.025] 0.050	$\begin{bmatrix} 0.011 \end{bmatrix} \\ 0.002 \end{bmatrix}$	[0.009] 0.000
Woman's Edu. Level: Medium Education	[0.027] 0.004	[0.031] 0.046	[0.013] -0.009	$\begin{bmatrix} 0.012 \end{bmatrix} \\ 0.002 \end{bmatrix}$
Woman's Edu. Level: Technical or University Education	[0.035]	[0.039]	[0.016]	[0.013]
	-0.100***	0.001	-0.008	-0.007
Woman's Activity: Employed	[0.037]	[0.042]	[0.019]	[0.015]
	0.054^{***}	0.046^{**}	0.016^*	0.032^{***}
Woman's Activity: Unemployed	[0.019]	[0.020]	[0.009]	[0.009]
	0.045	-0.010	-0.007	0.030^{**}
	[0.030]	[0.029]	[0.012]	[0.014]
Observations	10.801	10.801	10.801	10.801
Province Controls	Yes	Yes	Yes	Yes
Pseudo R^2	0.146	0.123	0.120	0.117

 Table 2.3a.:
 Results, Married and Common-Law Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Intimate Partner Violence in Current Partnership

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. All variables refer to the woman's current partnership. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence in partnership. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has No Money for Personal Expenses;' 'All Assets Are Co-Owned;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Physical Violence	Psychological Violence	Sexual Violence	Economic Violence
	(1)	(2)	(3)	(4)
Woman's Number of Children	0.023***	0.022***	0.011***	0.007***
Warman Had First Child on Teansmon	[0.000]	[0.007]	[0.003]	[0.002]
woman nad First Child as Teenager	[0.041	0.027	[0,010]	-0.009
Warran Dartmanad as Taspagen	[0.020]	[0.021]	0.010	[0.009]
woman Farthered as Teenager	-0.005	-0.027	-0.014	-0.008
Woman's Marriage / Polationship Longth	0.0020]	[0.020]	0.013	0.002**
woman's warnage/ Relationship Length	[0.009]	[0.002]	0.002	[0.003
Size of Femily in Household	[0.003]	[0.003]	0.001	0.001
Size of Family in Household	0.008	0.005	-0.001	-0.001
Household is Multi Constitional	[0.005]	[0.000]	0.002	[0.002]
Household is Multi-Generational	-0.015	-0.000	0.002	-0.002
Mala Haad of Hausahald	[0.021]	[0.022]	0.010	[0.008]
Male nead of nousehold	0.001	0.050	-0.005	0.015
Non Family Mombons Live in Household	[0.035]	[0.035]	[0.018]	[0.010]
Non-Family Members Live in Household	-0.117	-0.109	-0.019	-0.019
	[0.038]	[0.040]	[0.021]	[0.011]
Household in Urban Area	$0.047^{(0,0,0)}$	0.023	0.008	0.005
	[0.016]	[0.019]	[0.008]	[0.007]
	10.001	10.001	10.001	10.001
Observations	10,801	10,801	10,801	10,801
Province Controls	Yes	Yes	Yes	Yes
Pseudo R^2	0.146	0.123	0.120	0.117

 Table 2.3b.:
 Results, Married and Common-Law Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Intimate Partner Violence in Current Partnership (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. All variables refer to the woman's current partnership. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence in partnership. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has No Money for Personal Expenses;' 'All Assets Are Co-Owned;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Had Money for Personal Expenses	-0.217***	-0.225***	-0.168***	-0.139***
All Assets Owned by Female	[0.047] -0.091 [0.084]	[0.041] -0.122 [0.096]	[0.042] -0.075 [0.069]	$\begin{bmatrix} 0.040 \end{bmatrix}$ 0.048 $\begin{bmatrix} 0.060 \end{bmatrix}$
All Assets Owned by Male	-0.121	0.010	[0.009] 0.052 [0.086]	0.093
Mixed Ownership of Assets	-0.236***	-0.182 [0.115]	-0.012	[0.004] 0.004 [0.082]
Family Owns No Assets	-0.191** [0.077]	-0.184** [0.078]	-0.110	-0.032 [0.057]
Woman's Monthly Income (Scaled)+	0.072	0.013	-0.095 [0.073]	-0.113
Woman's Acceptability of Wife Beating Index+	0.028 [0.025]	0.020 [0.025]	-0.027 [0.020]	0.017 [0.020]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.309** [0.143]	-0.389*** [0.142]	-0.277** [0.120]	-0.300** [0.127]
Violence in Women's Family Background	0.136^{***} [0.043]	0.156^{***} [0.042]	0.097***	0.069^{**}
Violence in Spouse's Family Background	0.211^{***} [0.046]	0.181^{***} [0.053]	0.206^{***} [0.041]	0.108^{**} [0.042]
Unknown Violence in Spouse's Background	0.059 [0.052]	0.020	0.053 [0.042]	-0.042 [0.039]
Woman's Age	0.000	-0.003	$\begin{bmatrix} 0.012 \\ 0.001 \\ \begin{bmatrix} 0.002 \end{bmatrix}$	-0.003 [0.002]
Woman's Ethnic Identity: Indigena	0.038 [0.092]	0.065	0.058 [0.089]	-0.015 [0.088]
Woman's Ethnic Identity: Afrodescendent	-0.004 [0.100]	-0.056 [0.105]	0.013	-0.031 [0.053]
Woman's Ethnic Identity: Other Ethnicity	-0.074 [0.070]	0.041 [0.056]	0.043 [0.060]	0.024 [0.056]
Woman's Edu. Level: Basic Education+	0.089 [0.097]	0.171^{*} [0.099]	-0.004 [0.084]	0.129 [0.084]
Woman's Edu. Level: Medium Education+	-0.065 [0.113]	0.028	-0.104 [0.083]	0.091 [0.103]
Woman's Edu. Level: Technical or University Education+	-0.134 [0.122]	-0.001 [0.125]	-0.066 [0.103]	0.137 [0.122]
Woman's Activity: Employed+	0.033 [0.052]	0.044 [0.050]	0.090**	-0.016 [0.047]
Woman's Activity: Unemployed+	0.069 [0.084]	0.067 [0.074]	[0.071] [0.080]	-0.072 [0.061]
Observations Province Controls	1,401 Yes	1,401 Yes	1,401 Yes	1,401 Yes
Pseudo R ²	0.200	0.203	0.205	0.152

 Table 2.4a.: Results, Separated and Divorced Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Intimate Partner Violence in Past Partnership

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence in partnership. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Had No Money for Personal Expenses;' 'All Assets Are Co-Owned;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Physical	Psychological	Sexual	Economic
	Violence (1)	Violence (2)	Violence (3)	Violence (4)
	(1)	(2)	(0)	(4)
Woman's Number of Children+	-0.002	0.041**	0.025	0.029*
	[0.020]	[0.019]	[0.016]	[0.016]
Woman Had First Child as Teenager	0.115^{*}	0.001	0.087^{*}	0.010
	[0.060]	[0.062]	[0.049]	[0.051]
Woman Partnered as Teenager	0.034	0.038	0.042	0.090^{*}
	[0.055]	[0.059]	[0.048]	[0.051]
Woman's Marriage/Relationship Length	0.003	0.000	0.001	0.001
	[0.003]	[0.003]	[0.002]	[0.002]
Size of Family in Household+	0.036**	0.020	0.023*	-0.008
	[0.014]	[0.014]	[0.013]	[0.012]
Household is Multi-Generational+	-0.089	-0.055	-0.071	0.000
	[0.060]	[0.058]	[0.049]	[0.049]
Male Head of Household+	0.003	-0.050	0.093	-0.004
	[0.070]	[0.079]	[0.064]	[0.056]
Non-Family Members Live in Household+	-0.095	-0.002	-0.167**	0.028
	[0.116]	[0.130]	[0.070]	[0.116]
Household in Urban Area+	0.075	0.025	0.119***	0.033
	[0.062]	[0.059]	[0.037]	[0.046]
Observations	1,401	1,401	1,401	1,401
Province Controls	Yes	Yes	Yes	Yes
Pseudo R^2	0.200	0.203	0.205	0.152

 Table 2.4b.: Results, Separated and Divorced Women: Marginal Effects from Probit Estimates for the Probability of Experiencing Intimate Partner Violence in Past Partnership (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence in partnership. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Had No Money for Personal Expenses;' 'All Assets Are Co-Owned;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Panel A.	Married/Co	mmon Law	v Women	Panel	B. Separated,	/Divorced	Women	Panel C.	Widowed Wo	omen (Ag	e: 45-65)
	Physical (1)	Type of V Psychological (2)	<i>iolence</i> Sexual (3)	Economic (4)	Physical (5)	<i>Type of V</i> Psychological (6)	<i>iolence</i> Sexual (7)	Economic (8)	Physical (9)	Type of Vi Psychological (10)	olence Sexual (11)	Economic (12)
Woman Has/Had Money for Personal Expenses	-0.053***	-0.079***	-0.037***	-0.022***	-0.217***	-0.225***	-0.168***	-0.139***		-0.170*		
All Assets Owned by Female All Assets Owned by Male	0.065^{**} 0.050^{**}	0.106^{***} 0.070^{***}	0.055^{**} 0.024^{*}	0.054^{***} 0.023^{**}							0.261^{**}	0.182^{**} 0.263^{*}
Mixed Ownership of Assets	0.043^{*}	0.043^{*}	0.030^{**}	0.023^{*}	-0.236^{***}							
Family Owns No Assets		0.047^{**}			-0.191^{**}	-0.184^{**}			0.516^{***}	0.407^{***}	0.386^{***}	0.482^{***}
Woman's Monthly Income (Scaled)+											0.110^{*}	-0.113*
Woman Earns 51% or More of Household Income+	-0.042^{*}											
Woman's Acceptability of Wife Beating Index+	0.025^{***}	0.020^{***}										
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.119^{***}	-0.133^{***}	-0.060***	-0.053***	-0.309**	-0.389^{***}	-0.277^{**}	-0.300^{**}			-0.212^{*}	
Violence in Women's Family Background	0.189^{***}	0.179^{***}	0.048^{***}	0.050^{***}	0.136^{***}	0.156^{***}	0.097^{***}	0.069^{**}		0.141^{*}		
Violence in Spouse's Family Background	0.184^{***}	0.209^{***}	0.055^{***}	0.045^{***}	0.211^{***}	0.181^{***}	0.206^{***}	0.108^{**}		0.172^{*}	0.169^{***}	
Unknown Violence in Spouse's Background	0.062^{***}	0.079^{***}										-0.107^{***}

Statistically Significant Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence in Current or Past Relationship
 Table 2.5: Summary of Results:

Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence in partnership. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has/Had No Money for Personal Expenses;' 'All Assets Are Co-Owned;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education,' 'Woman's Activity: Out of Labour Force;' 'Household in Rural Area,' 'Province: Pichincha.' qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced. Physical,

Chapter 3

Seen and Not Heard: Household Chores and the Well-Being of Children in Ecuador

3.1 Introduction

While a lot of attention has been given to the effects of child labour in the context of market activities, the literature remains thin when it comes to studying household work as part of child labour, especially in terms of how housework might affect children's current well-being (ILO, 2017; Dammert et al., 2018).¹ Household chores or household work are domestic tasks that are used for consumption in the household (ILO, 2017).

¹In this chapter it is worth noting the different ways in which children can divide their time, e.g., leisure, play, chores, market work and education. There is subjectivity in how children would understand or conceptualize these activities and that children could simultaneously combine household chores with play. For instance, a parent may request their child to take care of their elderly grandparent though taking care of the grandparent could involve playing cards with them which could also be regarded as play. There is nuance surrounding the definitions of leisure, play and work both in how the children conceptualize them and in how they can be studied. For the purpose of this analysis, the definitions used are those detailed in the Empirical Framework which rely on the conceptualization employed in the survey used. In what follows, domestic work, household work, housework, and chores denote the unpaid work that children perform at home. Market work refers to the work that children do in the labour market, whether they are employed by their parents (e.g., on a farm or for their own business) or employed by a third party. Even though children are "employed" they may or may not receive a wage, which is why it would be misleading to refer to it as paid work. However, domestic work broadly refers to chores done at home (excluding farming or tending to the family business) while market work refers to the children being engaged in economic activities (regardless of whether they are paid a wage or not).

Broadly, these tasks include caring for other household members, cleaning, getting groceries or water and cooking, as well as household maintenance (ILO, 2017). In most contexts, child labour refers to work in economic activities that children under the age of 18 perform, illegally (ILO, 2017). The work goes against regulations if it is in violation of minimum age requirements or if it exceeds the amount allowed, which depends on the child's age. Definitions of 'child labour' vary widely across nations, and there is even more variability on whether or not housework is included.

Chores are usually excluded from definitions of child labour given the "noneconomic" nature of these activities, following standard guidelines for measuring economic activity (ILO, 2017). In some cases, 'light work,' working in only some economic activities or performing domestic chores for less than two hours a day, is allowed i.e., children can perform a certain amount of hours in household work with it not considered to be child labour (ILO, 2017). The variability in definitions and exclusion of household work from child labour is problematic for several reasons. The number of children performing domestic chores is large and their contributions to their households are not meaningless. Somewhat overlooked, 800 million children between the ages of 5 and 17 years are involved in some form of housework weekly (ILO, 2017). A large number of these children (37.5-39.6%) work less than 14 hours per week in their homes and in all cases, at every age, there are important differences in gender, as girls are more likely to perform household chores than boys (ILO, 2017; Bruckauf and Rees, 2017; Gershuny and Sullivan, 2014; Webbink et al., 2012). "Globally, girls aged 5-14 spend 550 million hours every day on household chores, 160 million more hours than boys their age spend" (UNICEF, 2016, p. 10). The amount of work that children contribute to total household work is not minimal and varies from around 4%-12% (Blair, 1992a; Goldscheider and Waite, 1991; Gershuny and Sullivan, 2014).

Overlooking the unpaid work children do at home feeds into an underlying assumption that this type of work may be helpful at best, and necessary at worst, when, in fact, it may be harmful. Whether housework is beneficial to children has not been tested in the context of self-assessed well-being and previous work suggests that household work performed by children is not universally harmless (Zapata et al., 2011; French, 2010; Ragui et al., 2010; Dinku et al., 2019; Sharmin, 2015). This chapter seeks to explore the effect of a child's involvement in housework on their self-assessed well-being. Specifically, it seeks to answer how happy children self-report themselves to be, conditional on the time they spend on household work, i.e., do children perceive the unpaid work they do at home as detrimental to their happiness? A second question this chapter seeks to answer is: does the activity or specific chore matter? i.e., are there differences between cooking and cleaning? Is care-taking more detrimental than shopping for groceries? Similarly, when compared to other members in the household, is the share of the total work performed in domestic activities important in determining children's happiness, i.e., if the children take on a larger share of total household work, is their self-assessed happiness affected? It could be that the children regard being assigned a larger share of chores as 'unfair,' especially relative to their siblings which could in turn affect their perceived happiness. Lastly, this chapter explores the different threshold of hours of housework and its relationship to children's well-being. Is there a certain threshold hours where children start perceiving doing chores as detrimental to their well-being?

Results show that the domestic activities that Ecuadorian children aged 8-17 perform in their home are negatively associated with their self-assessed well-being. This finding holds both at the extensive and intensive margin of household work. On average, close to 86% of children say that they are happy most of the time. Performing chores is associated with an average -4.0 p.p. lower probability in children reporting as happy. An increase in domestic work of 10 hours per week is associated with a -2.1p.p. lower probability of children saying they are happy. Not all chores are equal and there are differences in the type of chores children perform. For instance, care-taking and doing laundry are negatively associated with girls' self-assessed well-being while shopping for groceries is positively correlated with the probability that girls say they are happy. The share of work that children perform in their home, relative to both the household work from all other members and from all other children in the household is also important. Boys appear to be significantly more affected when they take on a larger share of the total work done from all members while both girls and boys are negatively affected when they take on a larger share of the household work performed by children, i.e., when they are doing more work than their siblings. Although the decrease in happiness associated with household work (-2.1 p.p.) may seem small given the mean level of happiness, the magnitude is very close to that of the decrease

associated with performing work in the labour market (-2.9 p.p.).

While a lot of attention is placed on the work children perform in economic activities, from the children's perspective, the work they perform at home is almost as detrimental, especially for younger children. Although legal thresholds for 'allowing' household work to be performed are usually kept at 14 hours per week (thus, ignoring this amount of hours in housework from child labour definitions), children engaging in any positive number of chores already perceive it as negative to their well-being. Working up to one hour per day (i.e., (0-7] hours per week) is negatively associated with boys' well-being (-4.8 p.p.). Overall, these findings bring attention to the importance of listening to children. While domestic activities are often assumed to be beneficial, by studying children's responses and their perceived well-being, household work is almost as detrimental as the work performed in the labour market, especially in a context where the average child performs at least an hour of chores daily and a tenth of all household work in their homes.

This chapter is structured in the following way, Section II provides a conceptual framework and reviews the literature, Section III discusses children and household work in the context of Ecuador. Section VI structures the empirical framework and describes the data. Section X presents the results.

3.2 Conceptual Framework & Literature Review

Household Work Children provide substantial contributions to their household in the form of domestic work. Early research using data on American households in the mid-1960s to the late-1980s by Goldscheider and Waite (1991), shows that the contribution of children and adolescents was considerable, totalling 15% of all household work (Goldscheider and Waite, 1991).² In 1988, Blair (1992a) shows that American children from married or cohabiting households work on average, seven hours a week, equivalent to 12% of total household work. Since then, studies have substantiated the claim that children's contributions to domestic work are not negligible (Gershuny and Sullivan, 2014; Webbink et al., 2012).³

 $^{^{2}}$ Goldscheider and Waite (1991) used the Young Men and Young Women and Mature Women samples of the National Longitudinal Surveys of Labour Market Experience.

 $^{^{3}}$ By distinguishing between household types, using a sample of UK households from 2000-2001 and focusing on couples with children, Gershuny and Sullivan (2014) find that the contribution to housework from ranges from 4%

It has been further demonstrated that children's contribution starts early in life. By looking at evidence from twelve high income countries,⁴ Bruckauf and Rees (2017) find that by the age of eight, more than half of all children (52%) already contribute to their household every day.⁵ In the United States, most children (85% of boys and 81% of girls) are required to perform chores on a regular basis, as early as when they are two years of age. By ages 6-17, 88% of children perform at least one household chore on a regular basis (Blair, 1992a). Analysing sixteen countries,⁶ 30% and 11% of children in Africa and Asia, respectively, between the ages of 8-13, do over 15 hours of housework (Webbink et al., 2012). In countries with data available on chores performed by girls, the United Nations Children's Fund (UNICEF) estimates that 64% of girls (ages 5-14) help with cleaning and cooking meals (UNICEF, 2016).

Generally, adults are the ones who control the nature and amount of work that children do (Bruckauf and Rees, 2017). The participation of children in household labour is usually seen as ubiquitous and is regularly assumed as beneficial for children to learn valuable skills and become responsible while performing domestic tasks (Blair, 1992a; UNICEF, 2016). Child socialization happens when parents assign chores to children as an educational or instructional tool to help promote their responsibility and autonomy. The theory of socialization has a competing explanation, a pragmatic take on the requirements of a household: sourcing children for housework occurs when structural constraints alter the demand for household labour in the household, i.e., when parents are constrained in their time (Blair, 1992a). Evidence seems to support that children work to meet the labour demands of the household (Blair, 1992a).

^{- 9%} depending on the child's gender, age and sibling configuration. In this case, the lower bound in participation corresponds to male only children between the ages of 8-13 and the upper bound corresponds to teenage girls, 14-17 years old who have younger siblings. One exception is Bonke (2010), who using the European Community Household Panel Survey from 1998, finds that in 1,328 Danish children, children's contribution to domestic work is only modest, compares to the total domestic work performed in households. The peak of contributions appears in girls at age 14 with a contribution of three hours per week. However, the author argues this is to be expected because of patterns witnessed in adults; Danish parents spend substantially less hours on domestic work than their Nordic or American counterparts. Another finding is that domestic work is gendered as boys participate less than girls in performing chores (Bonke, 2010).

⁴The countries are Estonia, Finland, Germany, Israel, Malta, Norway, Poland, Romania, Republic of Korea, Spain, Turkey and the United Kingdom and the analysis is performed using the 2013-2014 wave of the International Survey of Children's Well-Being.

 $^{{}^{5}}$ The results also show that the daily contribution to housework from the younger, eight year old children is higher than older, twelve year old children. This is interpreted as indication that later in their lives, children shift away from domestic work towards market work.

⁶The countries are Bangladesh, Burundi, Central African Republic, Cote D'Ivoire, Gambia, Ghana, Guinea Bissau, Sierra Leone, Togo, Malawi, Mauritania, Somalia, Syria, Thailand, Vietnam and Yemen and the analysis is performed using the 2005-2006 round of UNICEF's Multiple Indicator Cluster Surveys.

The decision of parents to engage children in chores is likely also to be influenced by family and societal expectations and social norms (Bruckauf and Rees, 2017; Blair, 1992a; Wikle, 2014). These social norms can make household work gendered as the parental demand for domestic work could be sex-typed (Bonke, 2010). If parents expect boys to have a career in the labour market, they could consequently expect boys to be less engaged in household work and more involved in their investments in schooling, doing homework, sports and socializing (Bonke, 2010).⁷ In fact, girls from more traditional backgrounds tend to do more housework following the behaviour of their mothers (Bruckauf and Rees, 2017; Gimenez-Nadal et al., 2017).

Attitudes towards the work that goes on in a household seem to be transmitted across generations. Gimenez-Nadal et al. (2017) argue that the mechanism of this transmission is three-fold: through the intergenerational transmission of preferences, having parental role models and through imitation. The intergenerational transmission of preferences relies on the idea that parents have a strong influence in the formation of their children's preferences (Gimenez-Nadal et al., 2017; Our Watch, 2018). Cunningham (2001) shows that what children perceive as an 'ideal allocation of housework' when they are eighteen years old, is predicted by their mothers' gender role attitudes when the children are very young, and by how parents divide their housework when the child is an adolescent.

Identity has been shown to be extremely important in determining behaviour within a household (Akerlof & Kranton, 2000; Cunningham, 2001). If there are gender norms dividing work within a home, and these are part of an individual's gender identity, then deviating from these behaviours results in social costs to the individual,⁸ resulting in the individuals behaving according to the social norms to avoid guilt or social backlash. Parents may become role models to their children so that they can abide by these social norms or children may simply imitate what their parents do (Gimenez-Nadal et al., 2017; Cunningham, 2001) to prepare for their adulthood (Wikle, 2014; Ragui et al., 2010). In this context, "the 'doing' gender thesis also

 $^{^{7}}$ It is important to note that there is evidence contradicting these expectations. Using Canadian data, Fortin et al. (2015) study the predominance of girls at the top of the GPA distribution, i.e., a gender differential in academic achievement. Results show that post-secondary educational expectations are the most important factor in explaining the gender differential. The authors also note that parents in Canada have higher educational expectations for their daughters than for their sons.

⁸For instance, if for women their role and identity is that of Suzy Homemaker, then they may feel guilty if their house is not clean or if there is no pie fresh out of the oven cooling on the windowsill.

applies to children" (Bonke, 2010, p. 4).

In general, there is a wide consensus that girls do more work than boys when it comes to household labour. Worldwide, it is estimated that girls spend 50% more time on domestic chores than boys (UNICEF, 2016). Exploring the twelve countries mentioned above, Bruckauf and Rees (2017) show that girls do more daily housework than boys. Looking at children in the United Kingdom, Gershuny and Sullivan (2014) find that for the two age groups studied (ages 8-13 and 14-17), girls are more likely to work at home by cooking, cleaning or taking care of clothes and that participation rates are higher for the older group of girls. In Africa and Asia,⁹ girls are not only more involved in housework (boys tend to be involved in farm/family business related work), but also, the percentage of girls who work long hours is larger than the percentage of boys working long hours (Webbink et al., 2012). Similar patterns are evident in Italy (Busetta et al., 2019), Brazil (French, 2010), Bolivia (Zapata et al., 2011), Egypt (Ragui et al., 2010), the United States, (Wikle, 2014; Blair, 1992a; Cunningham, 2001), Germany and Spain (Gimenez-Nadal et al., 2017) and Denmark (Bonke, 2010) where being an older child and number of children increases time doing chores.¹⁰

These differences in time spent doing housework between boys and girls seem to grow as children age. Looking at global averages, when girls are 5 to 9 years of age, they spend 30% more time helping in their homes than do their male peers of the same age. By ages 10 to 14 they spend 50% more of their time helping at home than do boys (UNICEF, 2016).¹¹ The participation of female children remains lower than that of the adult women in the household (Wikle, 2014). Nonetheless, the fact that the disparity in time spent at home grows as children age is worrisome as it may lead to these disparities being perpetuated through women's lives. As Bonke (2010) establishes, "the different amounts of time girls and boys spend on housework may contribute to the not fully documented unequal distribution of time spent by mothers

⁹Bangladesh, Burundi, Central African Republic, Cote D'Ivoire, Gambia, Ghana, Guinea Bissau, Sierra Leone, Togo, Malawi, Mauritania, Somalia, Syria, Thailand, Vietnam, and Yemen (Webbink et al., 2012).

¹⁰A few nuances exist when it comes to these patterns. The gender differences are reversed by changes in household composition and depending on the behaviour modelled by parents. For instance, teenage boys in single-mother households contribute more than female teenagers in dual-parent households (Goldscheider and Waite, 1991). Looking at children in Spain and Germany, Gimenez-Nadal et al. (2017) show that the effect of fathers' contributions to housework determines differences in participation in children. "If children observe that fathers devote more time to household labour, it can influence their behaviour, while mother's housework has no such influence, since it is perceived as work that must be done by women" (Gimenez-Nadal et al., 2017, p.1162).

¹¹In the United States, for example, Wikle (2014) finds that the gender differential in time dedicated to chores begins for most girls at age eight and for the girls who work, it increasingly rises as children age through adolescence.

and fathers [in household production]" (Bonke, 2010, p. 2).

As previously mentioned, the discrepancies in time spent in chores by girls and boys, could be simply attributed to children growing into the roles that are pre-established. "Gender roles ... are shared beliefs that apply to individuals on the basis of their socially identified sex which are the basis of the division of labour in most societies. In Western societies, the home sphere, and the household chores as part of this sphere, it is assumed to be in charge of women" (Cerrato and Cifre, 2018, p. 2-3).¹² There are tasks that tend to be perceived as feminine including shopping for the home, cleaning, laundering, taking care of children, assisting children while masculine tasks include home and auto repair (Cerrato and Cifre, 2018; UNICEF, 2016; Cunningham, 2001; Wikle, 2014; Bonke, 2010; Blair, 1992a). Additionally, since these tasks are traditionally assigned to women, men do not feel obligated to be involved in the home in the way women do since the tasks are perceived as entirely voluntary or a hobby (Cerrato and Cifre, 2018, p. 3). Tasks that 'fall' under women's purview, like cooking or feeding taking care of the children are tasks that are unlikely or impossible to be deferred, unlike those of men, like moving the lawn or washing a car. Consequently, even in the face of equal (or more) market work this specialization of women in household chores leads to less free time for them, a new form of inequality.¹³

A caveat worth noting is that the way in which individuals classify activities as leisure or household work is not clear cut. For instance, in adults, if a wife really enjoys rearing their children or cooking does it qualify as leisure or as household production (Phipps et al., 2001)? Or if children are cooking or doing chores with their parents or with their siblings, could it be considered socializing or enjoying family time (Bruckauf and Rees, 2017)? Perhaps if they are providing care for a grandparent, if that time is used playing cards, would it fall under play or domestic work or both? In general, for adults, it seems that time spent in paid work is the least enjoyed activity (Byson

¹²It is well documented that adult women work more in the home than do men in high-income and middle- and low-income countries (Blair, 1992a; Wikle, 2014; Gimenez-Nadal et al., 2017; Gimenez-Nadal and Sevilla-Sanz, 2011). Women not only spend comparatively more time in household work, but they also spend more time on routine and time-intensive work. On the other hand, men tend to do work that is less time-intensive and more sporadic (Gimenez-Nadal et al., 2017).

¹³The difference in the amount of unpaid work at home for adults remains, even when women do comparable (or more) paid market work or earn more than their spouses (Bertrand et al., 2015; Bruckauf and Rees, 2017; Stevenson and Wolfers, 2009; Gimenez-Nadal and Sevilla-Sanz, 2011). What is referred to as the "second shift" in the literature establishes that even though females have increased their labour market participation and work hours, a compensated decrease in their household work has not been observed (Gimenez-Nadal and Sevilla-Sanz, 2011; Stevenson and Wolfers, 2009; Phipps et al., 2001). Women have a 'second shift' of housework at home after a shift in the labour market or vice-versa.

and MacKerron, 2015).¹⁴ For activities at home, "housework has been reported to be among the less enjoyable daily activities" (Gimenez-Nadal et al., 2017, p.1172). Even though for some individuals it might be enjoyable to perform chores, in terms of momentary well-being, it seems that it generally has negative effects.¹⁵

Children's Work and Well-Being When it comes to children's work and their well-being, most of the literature has explored the effects of market work and very few studies have looked directly at household work (French, 2010; Ragui et al., 2010; Zapata et al., 2011; Dinku et al., 2019; Sharmin, 2015). One of the reasons that this might be the case is that in most cases, assisting parents with household chores or in the family enterprise or farm to help support the household is expected and usually acknowledged as a positive activity that contributes to the well-being of children by allowing them to acquire helpful skills, a sense of responsibility, competence, self-efficiency and maturity and even develop their identity as helpful members to their families and society (Blair, 1992a; Koller et al., 2014). This is particularly true if the work is performed in the children's own homes, in the company or supervision of their family members, under reasonable conditions and hours (UNICEF, 2016).

As it does with adults, working, whether in the market or at home, inevitably diminishes the time children have for leisure, and arguably for most children, it impacts the time they have for educational activities (even if it does not impact the child's ability to go to school, children may have less time to study or do homework, for instance). Over the years, most of the concern has been on children's engagement in market activities (ILO, 2017; Beegle et al., 2009; Bourdillon, 2013; Dammert et al., 2018). Working prevents children from going to school and investing in their human capital, children who work will consequently receive lower future earnings due to this diminished time allocated to education (Udry, 2004). The human capital consequences

¹⁴In the United Kingdom, paid work is ranked lower than all (39) activities reported, with the only exception being that the individual is sick in bed. Economists have generally argued that work derives well-being, especially due to the negative effects of unemployment (Byson and MacKerron, 2015). These assessments however, are in the context of evaluative, global assessments of life or eudaimonic well-being, related to meaning, purpose, and worth rather than hedonic, or moment-to-moment levels of experienced utility.

¹⁵Studies suggest that the disutility from household work is derived not only from general dislike from the work but also from inegalitarian distributions of work (Gimenez-Nadal et al., 2017). In adults, these inequalities in the division of work or leisure time, generally translate to inequalities in well-being (as measured by satisfaction with time) (Gimenez-Nadal and Sevilla-Sanz, 2011; Phipps et al., 2001; Kulik, 2018; Cerrato and Cifre, 2018). In some cases, however, if there is a gender gap in household work, individuals are not negatively affected if the differences align with what is expected of them through social norms (Cerrato and Cifre, 2018).

from child work matter not only to the individual, but also to society, as a vicious circle of poverty may be created, hindering the development of society as a whole (Udry, 2004). In addition, child labour is associated with several risk factors like "hazardousness, insalubrity and hardship, ... diseases and accidents [and] damages the dignity of children and adolescents" (Koller et al., 2014, p. 4), including injuries, addiction, fatigue, malnutrition, sexually transmitted infections, abuse, amongst many others (Koller et al., 2014; Dinku et al., 2019).

When it comes to the well-being of children when performing household chores, the United Nations Children's Fund claims:

"The types of chores commonly undertaken by girls - preparing food, cleaning and caring for others - not only set the stage for unequal burdens later in life but can also limit girls' outlook and potential while they are still young. The gendered distribution of chores can socialize girls into thinking that such domestic duties are the only roles girls and women are suited for, curtailing their dreams and narrowing their ambitions. Household chores are usually not valued by the family and community the way income-earning activities are, rendering the contributions of girls less visible and less valuable, and having lasting effects on their self-esteem and sense of self-worth.

Time spent on chores also limits girls' chances to enjoy the pleasures and leisure of childhood, including time to play, build social networks and focus on their education. In addition, caring for other children or family members who are sick or elderly imposes adult responsibilities on girls while they are still children themselves. Other chores can inadvertently place girls at risk of violence - for instance, while travelling to or from their homes to carry water or collect firewood" (UNICEF, 2016, p. 11).

Empirical work exploring the consequences of housework for children support this claim.¹⁶ When looking at educational attainment, all forms of child work - including domestic work and work for family enterprises negatively affect participation in school¹⁷ (Webbink et al., 2012).

Examining girls' contribution to housework within a household, Ragui et al. (2010) explore the effects of domestic work on schooling in Egypt. A large percentage of parents in Egypt (47%) place a low value on education and many children, especially

 $^{^{16}}$ For a review of the literature regarding children's work in general and well-being, see Bourdillon (2013).

 $^{^{17}}$ This is especially so in low- and middle-income countries, where many children grow into occupations that value labour market experience over a formal education. For instance, in an agricultural or family enterprise, where children are trained by their parents; or in household spheres where girls are getting prepared for their marriage (Webbink et al., 2012).

girls, remain out of school and out of work. Ragui et al. (2010) highlight that arbitrary decisions are made to determine what constitutes work and what does not when it comes to examining child work. Unpaid activities, like working in the family's business or preparing food for a market stall are measured as child work due to the labour market nature of the work. However, similar activities, when they are conducted inside the household for internal consumption, are not regarded as child labour. "While such distinctions may make sense in the context of national accounts or labour force statistics, they may result in biases when trying to understand the phenomena of child labour and schooling" (Ragui et al., 2010, p. 81). Including domestic work and care of children to the definition of child labour, Ragui et al. (2010) study the relationship between the need for a daughter to contribute to household work and the parents' decision to have her drop out or not send her to school. Results show that an increase in the probability of work of 10% is associated with a 6% decrease in the probability that a girl attends school. "Many girls who work would have been in school had they not been expected to work fourteen or more hours per week" (Ragui et al., 2010, p. 117). Questioning whether a child would forego school to "simply" perform chores, (Ragui et al., 2010) argue that a household's need for housework is large and has an inelastic nature, (similar to what Phipps et al. (2001) find in adult women), and should not be underestimated, especially for poorer households.¹⁸

French (2010) studies the implications of labour market work and household work for education, life satisfaction and health of Brazilian children in 7th and 8th grade from two public schools. Examining 461 children in the city of Franca, a leather-shoe exporting town that heavily relies on child labour (independent of bans), French (2010) finds that both household chores and labour market work affect the well-being of

¹⁸An important methodological aspect of Ragui et al. (2010) is that the survey used made an exception for informed consent when it came to responses of children under the age of 15. Children could respond or their parents could respond for them. In 80% of the cases, parents answered the survey on behalf of their children, and in the remaining 20%, children responded themselves. In the case of market work, when comparing children's (6-14 year old) reports to proxy (parental) reports, there are no statistically significant differences in the activity rates for the children suggesting no systematic under-reporting of market work performed by children. This is not the case, however, for girls' domestic work, where there is a substantial difference and under-reporting from the parents of the work performed in the home. Although the actual differences are not reported in the study, authors hypothesize that the under-reporting might be due to discrepancies in perceptions between children and their parents on what comprises domestic work (Ragui et al., 2010). It has been similarly shown in the work of anthropologist Pamela Reynolds that adults do not necessarily categorize activities that children do as 'work.' For instance, adults do not include child care in their definitions of work, while children perceive it as such. From her work in Zimbabwe, her observations estimate that for girls under the age of 10, 56% of their time goes into taking care of younger children which is largely unrecognised by adults (Reynolds, (1991) as cited in Levinson (2000)).

children.¹⁹ The negative effects that domestic work has on well-being are larger for girls than for boys. Effects are also sensitive to pay levels, for instance, the higher the pay the lower the grades and the higher the absences for boys.²⁰

Zapata et al. (2011) directly explore the role that gender has on child labour and child education in Bolivia. Domestic tasks are included in the definition of child work noting that domestic work is not gender neutral and can demand significant time from the children. The authors measure simultaneously whether a child (aged 7-14) is in school or works at least 15 hours per week.²¹ If the definition of child labour was only based on labour market activities, results show that boys are more likely to work than girls and that girls are less likely to enrol in school, a novel finding in the Latin American context.²² Using a definition of child work that includes household chores, girls have a higher likelihood to be working and out of school than boys, a higher probability of combining work and school and a lower likelihood of being exclusively in school and out of work.²³

A point that is highlighted is that most empirical work assumes that market work is more harmful to children (in terms of their educational outcomes) than domestic

¹⁹In this case, well-being is measured by how children fare in three indices. *Health* is an indicator constructed from frequently experiencing: headaches, stomach aches, feelings of dizziness, unusual fatigue, feelings of sadness, and nervousness. The children can agree completely, agree in part, disagree in part or disagree completely with the frequency of experiencing the above. For *life satisfaction*, the measure is constructed by the extent to which children are satisfied with: the support of their family, their time for leisure, the space and comfort of their house, their health, their friends, the opportunities in Franca and the quality of their clothing. Children can say they are very satisfied, satisfied, dissatisfied, or very dissatisfied with the above. For *education*, the author uses the children's final grades in Portuguese and Mathematics and the average number of absences in both classes during the year-long marking periods (French, 2010). Results show that there is gendered specialization; working in the shoe industry or commercial sector is negatively related to health but has no association with school attendance or grades; domestic work has a negative association with health and grades; and caring for children is negatively related to the children's health and life satisfaction.

²⁰Another important outcome in the study is that there is evidence supporting the second shift argument for girls "unsurprisingly, employed girls do more work [within the household] than boys ... among younger (11-13) employed children girls' greater domestic work is nearly offset by fewer hours in employment so total work hours for each group are similar (30.0 for girls, 29.2 for boys)" (p. 75). Among older children (14-16), boys reduce the amount of work they do at home, reducing their total work load (to 28.4 hours per week) while girls increase employment hours and domestic work hours to a total of 37.4 weekly hours (9 more hours than boys) (French, 2010).

 $^{^{21}}$ The study also focuses on gender and on the Indigenous population in Bolivia, where almost a third (30%) of the population is classified as Indigenous according to their mother tongue.

 $^{^{22}}$ Education enrolment is defined as children's affirmative response to "did you enrol in a primary or secondary grade or college during this year?" For market work, the authors use children's responses to, "did you work last week?" Two measures of child work are used: a binary variable denoting that the child worked 15 hours or more in market activities and another binary variable if the child worked 15 hours or more in market activities or in domestic activities. For domestic work they use children's responses to: "during the previous week, did you carry out any of the following activities within your household? Take care of children and/or elderly family members; cook and clean the household; wash and/or iron clothes; perform minor household repairs; shop for food; chop and carry firewood; carry water from external water source; organize and maintain neatness" (Zapata et al., 2011, p. 598).

 $^{^{23}{\}rm The}$ results are larger for Indigenous girls who are 82% more likely to just work and 16% less likely to be just in school compared to non-Indigenous girls. In general, Indigenous children have a 60% higher probability of being out of school and working.

work. Zapata et al. (2011) classify children into two groups: i. low-domestic (highmarket) intensity and ii. high-domestic (low-market) intensity, where less than 50% of children's hours are in domestic work and 50% or more hours are in domestic work, respectively, to examine whether outcomes change depending on where the children spend most of their time (again, 'working' is still defined by working 15 or more hours per week, this part of the analysis explores the structure of those hours).²⁴ Results show that 73% of children spend a greater portion of their time doing household work, and if the work a child does is high-domestic intensity, girls have a 51% higher likelihood of being out of school and working and are 26% less likely to only study, compared to boys. If work is low-domestic (high-market) intensity, girls and boys are as likely to be working and out of school. Zapata et al. (2011) also explore the number of hours dedicated to both, and find that high-domestic intensity children work an average of 17 hours a week, compared to 23 hours a week that children spend in the high-market intensity category, suggesting that, although less time-intensive, more time at home is associated with lower likelihoods of enrolment.

Exploring intensity of domestic work, Dinku et al. (2019) analyse the effect that domestic chores have on two outcomes, 'school time' and 'body mass index for age' for 12-13 year old children in Ethiopia.²⁵ The authors study children's use of time by noting how children place 24 beans in 8 different pots interpreting this as how children divide their daily hours. Then, according to labour laws for children's work, two categories are defined, 'harmful level of chores' denoting children who report over four hours per day in household work and 'harmful-work' denoting children engaged in two hours or more in market work. Results suggest that excessive involvement in chores results in large negative effects on children's body mass index and with 13% less time in school (the same order of magnitude as the effect of participating in harmful work).

Although most official definitions of child labour assume domestic work is beneficial or deemed as 'light work,' evidence contradicts this seemingly innocuous assumption. In some cases, the work performed by children in their homes is excessive and could

 $^{^{24}}$ Explicitly, low-domestic intensity is defined as children spending less than 50% of their total hours worked in domestic activities. High-domestic intensity is defined as children spending 50% or more of their total hours in domestic work. The authors divide the sample by low-intensity of domestic work and high-intensity of domestic work and run their empirical specification on the sub-samples (Zapata et al., 2011).

 $^{^{25}}$ Dinku et al. (2019) measure education time as the children's time spent at school and studying time outside of school minus the commuting time that the children spend travelling. The health measure is the child's body mass index relative to their age using the World Health Organization guidelines (Dinku et al., 2019, p. 1264).

become a direct source of stress and even suffering for children due to an 'instrumental parentification' as they may fully take on the roles and responsibilities of their parents (Koller et al., 2014; Zapata et al., 2011; Dinku et al., 2019).

Without ignoring the merits of each of these papers, none of the previous studies address directly how the children perceive their own well-being. In the context of children's, when it comes to domestic work, well-being, as assessed by the children themselves, has not been directly studied. Several authors have explored it indirectly through indices of life satisfaction (asking them about support from their family, time for leisure, the comfort of their house, their health, their friends, labour opportunities and the quality of their clothing - though not directly asking them about their happiness) (French, 2010), health outcomes (French, 2010; Dinku et al., 2019), or educational outcomes (Ragui et al., 2010; French, 2010; Zapata et al., 2011; Dinku et al., 2019). As (French, 2010) notes that "equally important, the implications of work for children's welfare often depend on how they understand and interpret the experience" (p. 64). Similarly, when exploring children's housework and well-being, most of the research focuses on work in farms or family businesses or generally, subsistence work (Bourdillon, 2013) or it combines domestic work with market work or subsistence work for measures of child labour (Ragui et al., 2010).

This chapter expands the current literature in the following ways. First, I add a gender dimension to the existing research on household work and child labour, a component often overlooked in the research (Dammert et al., 2018). Although previous work (Gimenez-Nadal and Sevilla-Sanz, 2011; Phipps et al., 2001; Hajdu and Hajdu, 2018; Kulik, 2018; Cerrato and Cifre, 2018) has highlighted how inequalities in housework impact the well-being of adults, especially women, little attention has been given to girls. Second, I focus my empirical work directly on the association between household work and well-being of children as perceived by children's own assessments of both time and well-being.²⁶ This is important since there are large differences between parents and children in reported domestic work hours (Ragui et al., 2010; Levinson, 2000) and it is quite rare to have children reports of their own happiness. Most measures of child well-being are obtained indirectly (either through life satisfaction indices, health or education measures or through parents' perception

 $^{^{26}}$ For what remains of the chapter, given that the data explores whether children feel happy, unless otherwise specified, a child's 'happiness' and their 'well-being' are used interchangeably.

of their children's well-being).²⁷ Reliable self-assessments of children's well-being begin at age eight and children are considered very capable of assessing their own well-being (Burton and Phipps, 2010). As Casas (2011) notes, "which members of the population can best inform us of well-being in different areas of life? The quality-of-life perspective suggested that the point of view of "experts" (practitioners, researchers, or whoever) may not be enough to understand complex social realities. It is important to understand the point of view of all social agents involved in a given phenomenon being studied... the point of view of the youngest generations is also important to understand our societies, particularly those aspects of social life involving or affecting them" (p. 559).²⁸ A third contribution to the literature is that household work takes a significant portion of children's time, especially in the case of young girls, and studying its relation to well-being is an important issue.²⁹

3.3 Children and Household Work in Ecuador

Housework or domestic chores are not included in the official definition of 'child labour' under Ecuadorian legislation.³⁰ Household work usually consists of cooking and preparing meals, cleaning, conducting minor repairs, caring for others, washing and ironing clothes, buying groceries, accompanying family members in their commute, amongst other tasks (ILO, 2017). Technically, for accounting purposes, these tasks are deemed "non-economic" forms of production and are excluded from most national accounts in most nations following the United Nations System of National Accounts (UNSNA) with these activities going unmeasured for a very long time. Consequently,

²⁷Fattore et al. (2017) argue that "Embedded in what children told us are challenges to adultcentric discourses on child well- being. These challenges affirm that not only are children authoritative about their own well-being, additionally, because of their location, structurally, in the social order, they see things differently from adults" (p. 46). Similarly, looking at health and education measures is in a way done to protect children's future, and as such, are measures of well-*becoming*, rather than current, present accounts of well-*being* (Fattore et al., 2017).

 $^{^{28}}$ See Burton and Phipps (2010), Fattore et al. (2017) and Casas (2011) for an overview of children assessing their well-being with measures like the ones used in this analysis.

 $^{^{29}}$ As argued by Levinson (2000), "banning child labour" from paid market work has effectively moved children to informal work and household work, which is often characterized by its social invisibility. For a review of how Economics has failed to recognize children's agency, why children should be treated as agents, and the need to study their well-being "in ways that are true to girls' and boys' experiences," see Levinson (2000).

³⁰According to the Ecuadorian Childhood and Adolescence Code, the minimum age for work is fifteen years old. Children above the age of fifteen are allowed to work if the work day does not exceed six hours or five days a week, i.e., does not exceed 30 hours per week, as long as the work does not limit the child's right to an education or is not performed under conditions that endanger the child's health or safety (Children and Adolescents Code, 2002; INEC, 2012c). 'Child labour' refers to the work that children perform illegally and 8.56% of children were child labourers in 2012. This amounts to approximately 359,597 out of the 4,199,296 children in Ecuador (INEC, 2012a).

household work is not included in the definitions of child labour and is often overlooked.

Household work is not gender neutral, most home production in Ecuador comes from women. In an effort to conduct gender-informed policy, Ecuador started measuring unpaid household work in 2011.³¹ Through this initiative, the government estimated that in 2012, unpaid household work would have contributed to around 18.8% of GDP (gross value added) (CSTNRH, 2019). In terms of household consumption expenditure, it was estimated that for total household final consumption expenditure in 2012, unpaid work in the home resulted in "savings" of 33%, i.e., for \$100 USD of household final consumption expenditure, households "saved" \$33 USD in household work from the worked produced by members of their home instead of external purveyors of services. When looking at age groups, although this information is only available for 2015, children aged 12-17 contributed to 6.6% of all unpaid work, individuals aged 18-29, 29.2%, aged 30-44, 34.5%; aged 45-64, 22.8% and adults aged 65 years or more, 7.0%, respectively. Of the 6.6% contribution from children to total unpaid work, girls produced 67.3% of all unpaid labour from children and boys produced 32.7% (CSTNRH, 2019).³²

3.4 Data and Empirical Framework

Data To examine gender differences in the well-being of children, I use microdata from the 2012 National Child Labour Survey (ENTI, for its name in Spanish, Encuesta Nacional de Trabajo Infantil) in Ecuador. The survey was conducted to understand and assess the children and adolescents between the ages of 5-17 who work in Ecuador. The survey targeted all provinces except the Galapagos Islands. Of the 31,687 households surveyed, 146,814 individuals completed the survey. Of these, 51,233 were children

³¹The government of Ecuador began calculating a satellite account for unpaid housework. The account measures 66 unpaid activities that are generally excluded from the Ecuadorian System of National Accounts. These activities are divided into three main groups, i. unpaid care to household members (care of children, care and support of people with a disability and any care related to health); ii. unpaid domestic work in own household (cooking, shopping, management, maintenance and organization of the household, construction and repairs, care of clothing and tailoring); and iii. unpaid work in other homes, community and volunteer work (charity work, community support and other volunteering activities). To calculate the gross value added from these activities, and its economic value, the number of hours dedicated to these tasks was multiplied by the average compensation those activities would receive had transactions occurred in the market (CSTNRH, 2019).

 $^{^{32}}$ Although not empirically shown family structures are close-knit in Ecuador. Children remain in their parents' home until they are approximately 25 or 26 years old or until they get married, which happens, on average, when men are 32 years old and women are 29 years old (Metro Ecuador, 2017; Rodriguez, 2017). Similarly, households are multi-generational as parents or in-laws of the heads of household are frequently found to live in the same household.

between 5 to 17 years of age.

The survey was divided into seven different sections. Surveyors were instructed to ask the head of household or their spouse to be the 'informant' or person most knowledgeable.³³ The person most knowledgeable (PMK) answered the first six sections of the survey providing information on all members of the household about their demographic characteristics, occupational characteristics and labour including both market and domestic work.

The last module, however, was entirely directed to the children of the household. Children between the ages of five to seventeen answered 48 questions regarding their educational attainment, activities in the past week, including time dedicated to market work and household work, job conditions, reasons for working, uses of earnings, health consequences of to work, and whether they consider themselves to be happy children.

Since children were surveyed, surveyors were explicitly trained to ensure the safety of children as they might have been in vulnerable situations and they wanted to limit or eliminate exposure to risk or abuse of children following their responses.³⁴ Surveyors were instructed to explain that the methodology required the last section be answered directly by the children. They required consent to direct questions to children and have the questions answered solely by the children, if in the presence of an informant, or to be left alone with the children. Respondents were informed of the confidentiality of the survey and guaranteed it throughout, assured secrecy. Although surveyors were to ensure no one else would listen in during the interview and it is noted in the survey's documentation that all responses were confidential, this is not the case for most children (71.77% of the estimating sample), as responses were not confidential from other family members given that, for these children, their parents/guardians were present. Surveyors make note of whether the child's parents (or whoever was

 $^{^{33}}$ Ideally, the head of household or the spouse of the head of the household were the informants as, in most cases, they would have been the parents of the child. Surveyors were asked to exert their own judgment and when parents were not in the household, the informant was meant to be of legal age (above the age of 18), habitual resident of the household who had good knowledge of the activities of household members (INEC, 2012b).

³⁴The recommendations for surveyors to address the security and vulnerability of children included the following: kind and respectful conversation, creating an atmosphere of trust and showing empathy, support and understanding; informing all members the length of the survey and asking if they have time to answer without being interrupted or if they would rather do it in a different place or time (surveyors were in each survey area for three days); ensuring the surroundings are appropriate for the family and the children so that they can answer calmly and without exposure to uncomfortable situations, if that is not the case, agreeing to a different time or day; explaining the topics discussed are important and listening, paying attention and avoiding interruptions; as surveyors themselves could be at risk, keeping a cellphone with at least one phone number to call in case or need or maintaining contact with other surveyors in the area was suggested (INEC, 2012b, p. 8).

responsible for the child) were present at the time of their interview (INEC, 2012b).³⁵ All children in the household are asked these questions, therefore, there may be more than one child interviewed per household.

The survey was intended to be the first cross-section of a recurring (every two years) survey. Unfortunately, the data from 2014 was never published and the survey was discontinued shortly after. As such, an important disadvantage is that there is only a single cross-section of children, measured in 2012. Because of this, there is no way to track changes within a household, especially in terms of whether a household falls in or out of poverty, a stay-at-home mother decides to partake in the labour market, a divorce or death of a parent occurs, all of which could affect both the children's domestic duties and their well-being. Results thus, cannot be interpreted as causal effects as they are associations between variables.

The main advantage of the National Child Labour Survey, for the purposes of this research, is that in addition to a thorough battery of labour related questions, children were asked directly, "In general, would you say you feel a happy child/youth: most of the time? just in some instances? almost never?"³⁶ Given that the children are directly asked a question regarding their general, perceived happiness, there is no need to rely on proxy-based responses in this study. The options presented to the children, i.e., most of the time, some instances, almost never, prompt a general assessment of life and well-being, a eudamonic perception of their happiness rather than a hedonic, moment-to-moment assessment (Byson and MacKerron, 2015). Although advantageous, this measure is not perfect in terms of allowing variability as it only allows three categories as possible responses.

Another advantage is the fact that weekly hours are included for both market labour and domestic labour. Having the work done by children measured on a weekly basis is important given the inelastic nature of domestic work (Ragui et al., 2010; Phipps et al., 2001). It is uncertain, however, whether the weekly activities, "in the past week," are accurately representative of what life is like year-round for a child.

³⁵It is worth noting that in the preamble of asking questions to the children, surveyors were asked to reiterate the objective of the survey: "updating data on child labour to abolish, relaying information on the age of children and adolescents and the nature of their work" (INEC, 2012b, p. 11). Given that the objective stated abolishing child labour, this might have resulted in measurement error with respondents under-reporting work, which in turn would potentially bias estimates downward.

 $^{^{36}\}mathrm{The}$ questionnaire also allows for non- response, though it codes non-response and "do not know" in the same category.

The criteria for sample selection are as follows. As children can make meaningful assessments on their well-being from the age of eight (Burton and Phipps, 2010), the sample is restricted to children between the ages of 8 and 17. Children from dual-parent households are studied, children from single-parent households are excluded. Households from a single mother, or children who live with their grandparents, for instance, have arguably very different family structures which could have been impacted by death, divorce, separation, and might be associated with child happiness in very different ways than it would for children with both parents in the house. Therefore, the focus is on children who live with both of their parents.

The sample is also restricted to households where only children under the age of 18 reside. There is a lot of richness in terms of how multi-generational or diverse households are in Ecuador. Many individuals remain in their household home into their adulthood (their late twenties or early thirties) or until they get married (Metro Ecuador, 2017; Rodriguez, 2017). There could be members like a grandmother or an older brother living in the same household who could both supply labour to perform chores and increase the demand of housework (e.g., a grandmother could help with the cooking or an older brother could be an extra plate on the table). To make meaningful comparisons, households are restricted to only minors living in the same household. Children who have a disability are excluded. With these exclusions, the estimating sample includes 14,810 children.

Dependent Variables The focus of this study is to understand whether the unpaid domestic work that Ecuadorian children between the ages of 8-17 perform at their homes affects their well-being. From the survey responses, child well-being is defined as whether children consider themselves to be happy. The question is categorical, ordered by the frequency with which children experience happiness. The measure of happiness is defined as follows:

Happy: a binary variable equal to one if the child states that they are happy "most of the time" and equal to zero if the child states they are happy "just in certain occasions," or "almost never."³⁷

 $^{^{37}}$ If a child states they "do not know" if they are happy or do not respond, both answers are coded in the same category and no distinction can be made between them. The 45 children (0.30% of the estimating sample) who fall

Independent Variables The main explanatory variable of interest is household hours, the total amount of hours that a child spent in the following activities:

- i. *house cleaning* sweeping, dusting or making the bed(s);
- ii. grocery shopping in markets or supermarkets;
- iii. *clothing cleaning and repair* laundering, ironing, sewing or fixing clothes;
- iv. food preparation cooking breakfast, dinner or lunch;
- v. care of other household members care of other children, elderly, or those who are ill;
- vi. *help with homework* helping other family members with schoolwork;
- vii. household repair repairing any equipment for the home.

To answer the question 'does unpaid domestic work affect the well-being of Ecuadorian children?' household work is studied both at the intensive and extensive margin with the total number of hours performed in all of these activities and with a binary variable equal to one if the child does any amount of household work, zero otherwise.

Similarly, given that there might be differences in the level of happiness that each activity brings (Byson and MacKerron, 2015; Turcotte, 2013), weekly hours spent on each of the seven activities: cleaning, grocery shopping, doing laundry, cooking, caring for others, helping with schoolwork and repairs are defined separately to test whether the association with happiness and household work depends on the activity performed by the children.

Even though it is anticipated that increases in own hours of work will have an impact on the child's perceived happiness, it is also likely that inequalities in the division of domestic work in the household could also have an impact in the well-being of children. Children may perceive doing more work than other family members as unfair and be unhappy doing a larger share of chores. This is especially the case for girls who are more likely to perform a larger share of chores than boys. To test this hypothesis, the share of total household work performed by the child is also defined and tested.³⁸ Similarly, to compare the share of housework across siblings, rather

in this group are dropped.

 $^{^{38}}$ To avoid potential multicollinearity, the link between well-being and *household hours* or *share of total household* work performed by child are tested separately. A caveat with regards to the share of household hours is that children between the ages of 8-17 are able to report how many hours of household work they perform in a week. The informant or person most knowledgeable (PMK) provides information on household work performed by all children as well as all other members in the household. Therefore, when it comes to the hours of household work a child performs, there

than all members of the household, child's share of children's total household hours is defined, denoting the share that a child takes on from all the housework that is performed by the children in the home.³⁹

If children work in the labour market, the variable 'labour market hours' denotes the hours of work they perform in economic activities. For this measure, the survey asks the child whether they worked for at least one hour last week, whether they performed any economic activities⁴⁰ in the last week and whether they did not work last week but had a job. If the child affirmatively answered any of the previous questions, they are asked the amount of hours that they dedicated to these labour market activities, which is the number of hours used for this variable.⁴¹

A child's self-perceived well-being is likely affected by whether or not they attend school, so education is controlled for. Ideally, the hours spent in school and school related activities would be included (as well as sleep) to tally all hours in a child's day or week.⁴² The survey does not ask the children how much time they spend in school or on school-related activities. The children are asked "are you attending school, currently?" Education is measured with a binary variable denoting whether the child

are both self-reported hours as well as the hours that the PMK reports, though these are not the same all the time. In 2,956 cases, children report less hours of domestic work than the informant reports (the average difference is -4.23 hours). In 7,783 cases, the PMK and the children report the same number of hours worked in chores. In 4,071 cases, children report working more hours of household work than their PMK reports (the average difference is 4.00 hours of domestic work). Therefore, when adding the total hours of household work performed by all members of the household, both measures could be used, one being adding the hours for all members in the household as reported by the PMK. The other measure being using the hours that the children report for themselves and adding those to the hours that the PMK reports for all other household members, so that for this measure, the total hours of household work performed in a household adds what the children (8-17 y.o.) report for their work to what the PMK reports for everyone else in the family, i.e., using the hours that the children report for themselves, and adding them to what the informant reports for all other adults, calculating total household work hours for the entire household. When calculating *share of total household work performed by a child*, it can be done using: *child's self-reported hours*/total household hours *only*).

 $^{^{39}\}mathrm{To}$ reiterate, all children in the same home are siblings.

 $^{^{40}}$ These activities include: tending to their own business, manufacturing a product, carrying out work for an income, providing a service, helping in a family business, helping in a family member's workplace, agricultural work or care of animals, student-related work, or working for another family.

⁴¹It is worth noting that in this case, I do not use the 'official' definition of child labour, which is defined by legislature (and depends on age, amount of work performed, days in which the work is done and working conditions), rather, I only denote whether the child works outside of the home in economic activities and calculate the total amount of hours in said activities. Importantly, the answers to these questions depend on the interpretation of 'work.' However, given that the children are first asked whether they worked, followed by whether they performed any of the aforementioned activities which is then followed by whether they had a job even if they did not work the previous week, I assume that the focus of the survey, and hopefully the understanding of these questions are on economic activities. When exploring the extensive margin of household work, if the child works in the labour market, rather than including their labour market hours, a binary variable identifying whether the child works or not is included.

 $^{^{42}}$ The focus of the study is precisely the potential effects of household work on well-being, implicitly testing how domestic duties impact leisure time. One cannot assume that all children spend the mandated 35 and 45 hours a week that is established by the government as weekly hour loads for "basic education" and "baccalaureate education," respectively given that their homework load could directly have an effect on the time available for them not only to attend classes, but to be focused for the entire school day (S.R.O. No. 754, 2012). The same applies to hours of sleep.

currently attends school or not.⁴³

Control Variables For individual characteristics of the child, a set of dummy variables for age are included (ages 8-11 (base), 12-15 and 16-17).⁴⁴ There is a control for oldest child in the household, as being a first born is often associated with more contributions to housework (French, 2010; Zapata et al., 2011).⁴⁵ The child's cultural background (i.e., indicator variables for whether the PMK identifies the child as Indigenous, Afro-descendant, Mestizo or Other (including: White, Montubio and Other, unspecified), the base category is Mestizo) is also included to account for differences due to ethnicity (Zapata et al., 2011).

For household characteristics, work status of the parents is included by identifying whether the household is a dual-earner household: the head of household and their spouse are employed and earned a wage in the previous week.⁴⁶ Similarly, household equivalent income is included in the analysis due to its importance in the determination of well-being (Clark et al., 2008).⁴⁷ Household equivalent income is calculated by adding the income of all members in the household and dividing by the square root of household size (i.e., the number of members in the household).⁴⁸ As such, household equivalent income is income from all sources, before taxes⁴⁹ equivalized.⁵⁰ As a measure of socio-economic status, mother's and father's highest level of education attained is included as: primary education, secondary education, or university, the base category is less than primary education.

To identify who lives in the household in terms of children, there are indicator

 $^{^{43}}$ Although there is potential of having multicollinearity between household hours, labour market hours and education, the fact education is mandatory, labour restrictions impose bans on child labour and given the inelastic nature of household work, give enough reason to leave all three variables in the regressions, as having them is a priority in the study.

 $^{^{44}\}mathrm{Self}\textsc{-}\mathrm{awareness}$ and tee nage angst may affect children as they get older and as such well-being may change as child ages.

 $^{^{45}}$ An interaction term for being female and first born (*female* × *firstborn*) was also included, though it was statistically insignificant in all regressions.

 $^{^{46}}$ Including directly employment status of the parents, or hours of work may be problematic due to endogeneity.

 $^{^{47}}$ For a review of the literature regarding income and subjective well-being, see Clark et al. (2008).

⁴⁸Own-income is calculated by adding: earnings from the individual's main and secondary occupations; capital/interest earnings from saving accounts, loans, stocks, rental income from real state, land and machinery; transfers from retirement, widowhood, orphanage, disability, or divorce pensions, or severances; income from gifts or donations (e.g., lottery winnings); income from remittances and income from the human development transfer (BDH).

 $^{^{49}}$ Personal income in Ecuador is untaxed up to \$10,800 USD/year. After \$10,800 USD/year income is taxed at 5% and the rates increase progressively up to 35% for income above \$110,190 USD/year.

 $^{^{50}}$ There is potential of collinearity with household equivalent income and other household variables, such as the parents being dual earners, their education attainment, fertility (household size) and especially, receiving transfers. However, given the importance with well-being, as detailed in Clark et al. (2008), it is kept in the estimation.

variables denoting the presence of children under the age of five, other girls between the ages 5-17, other boys between the ages 5-17.⁵¹ There is an indicator variable for whether there is someone with a disability in the household. It is assumed that household composition is determined exogenously in the short term (Zapata et al., 2011). An indicator variable denoting whether the gender of head of household is female is included. Lastly, location controls are included by an indicator variable determining whether the household is in an urban setting as well as province dummy variables.

Econometric Specification The main specification is:

$$\begin{aligned} Happiness_i &= \alpha + \theta \text{ HouseholdWork}_i + \omega \text{ LabourMarketWork }_i \\ &+ \rho \text{ AttendsSchool}_i + \delta IC_i + \gamma HC_i + X'_i\beta + \varepsilon_i \end{aligned}$$

Where, $Happiness_i$: is the well-being measure for child *i*; $HouseholdWork_i$ is one of the measures indicating domestic work performed by child *i*; $LabourMarketWork_i$ denotes child *i*'s work in the labour market; $AttendsSchool_i$ is an indicator variable for whether the child currently attends school; IC_i is the vector of individual characteristics of the child; HC_i is the vector of child *i*'s household characteristics; and X'_i is the set of (location) controls, as described above.

Regressions are run using this specification. Given the nature of the dependent variables, probit and ordered probit models are estimated, with and without controls. The unit of observation is the child. The analysis is done initially for the full sample of children and then separately by gender. The variables that are denoted as 'scaled' are divided by ten, (e.g., for household hours (scaled) should be interpreted as 'an increase of 10 household hours is associated with ...'). The results presented below are the marginal effects obtained from the regressions with controls for province. Standard errors shown are robust and clustered at the household level (as there are multiple children surveyed per household). All results are weighed using household weights provided by the National Institute of Statistics and Censuses (INEC).

 $^{^{51}}$ Other' refers to additional children within that age group, i.e., siblings in those ages who live in the home. These variables were also disaggregated as: girls under the age of five, girls between the ages 5-7, girls between the ages 8-14, girls between the ages 15-17, boys under the age of five, boys between the ages 5-7, boys between the ages 8-14 and boys between the ages 15-17.

3.5 Results

In general, girls are more likely to perform chores and spend longer hours on chores (see Figure 3.1). In general, the more average daily/weekly hours performed by the children, the fewer children report being happy.⁵²

Table 3.1 presents the means of variables. On average, 86.5% of children report being happy most of the time and there is no statistically significant difference between boys and girls. Most of the children (81.6%) perform domestic chores, though more girls (84.2%) do chores compared to boys (79.2%). Similarly, girls work more hours in domestic activities (8.4 hours per week) than boys (6.9 hours per week). The largest difference in hours comes from cooking, were girls cook on average 1.12 hours per week more than boys do. The only activity where boys spend more time in than girls is household repairs, though the difference is small, 0.04 hours per week. Though boys spend on average 1.28 hours working in the labour market more than girls, when looking at the total hours of work from both household chores and labour market work, girls spend on average 9.8 hours per week working, compared to 7.9 hours per week for boys, a difference of 1.63 more hours per week.

Girls are also more likely to take on a larger share of total household work. They perform on average 11.4% of the total household hours on chores while boys only perform 7.8%; girls perform 48% of all household chores done by children in the household while boys perform 41.3%, a statistically significant difference.

The main results are presented in Table 3.2. As a reminder, the variables denoting hours worked are scaled, a unit increase in this case is of ten hours per week. On average, for Ecuadorian children aged 8-17, an increase of 10 hours per week in household chores is associated with a lower probability of -2.1 percentage points that the child says they are happy. When compared to hours worked in the labour market, an increase of 10 hours per week in labour market activities is associated with a -2.9 p.p. decrease, on average, in the probability of children assessing their well-being positively in terms of happiness. A caveat worth noting is that when children respond in front of their parents or PMK (parents are present in 77% of the cases), children report to be happier by 3.3 p.p., on average.

 $^{^{52}}$ The only exception to this is boys who report working more than 21 hours per week but given that there are only 261 boys in this group it could be that the exception is due to the small sample size.

Tables 3.3a. and 3.3b. present the results disaggregated by chore performed. For girls, going shopping to get groceries is associated with higher levels of happiness, an increase in 10 hours of shopping per week is associated with a 9.4 p.p. increase in the probability that the child says they are happy. In contrast, an increase in 10 hours of doing laundry and 10 hours of care-taking are associated with a -6.3 p.p., and a -4.1 p.p. decrease, respectively, in the probability that children say they are happy. When disaggregating the activities, none of them are statistically significant for boys.

Table 3.4a. looks at the share of household work performed by children. Columns (1)-(3) look at the share of hours that children performed compared to the hours in household work performed by all members of the household (e.g., if the household includes parents and three siblings, the share of child *i* is based on the hours in chores of all six members). Columns (4)-(6) look at the share of hours that children performed compared to the hours in household chores performed by the children in the household (e.g., if two children live in the household the share of child *i* is compared to the total hours in chores performed by themselves and the other child). For boys, an increase of 1% in the share of total household worked performed by all members is associated with a -16.5 p.p. lower probability in their happiness assessment. For girls, the share of total household work from all members is statistically insignificant. Looking only at the work performed by children only, (i.e., excluding the work from their parents), an increase in 1% in the share of total household work performed by children is associated with a -6.6 p.p. decrease in the probability that girls say that they are happy.

Most of the legal restrictions establish a fourteen hours per week threshold as the maximum allowable time for children to engage in household work activities before this work can be considered 'child labour.' To test whether associations between household work and well-being are sensitive to different thresholds of domestic work and whether the effects are sensitive to the 14-hour international threshold, dummy variables are defined for the following thresholds of household work: the child does (0-7] hours of household work per week, (7-14] hours of household work per week, (14-21] hours of household work per week, and 21 hours of household work or more per week, identifying working an average of 1, 2, and 3 hours per day (Ragui et al., 2010). For boys, working (0,7] hours in chores is associated with a -4.8 p.p. lower probability

that they say they are happy, (7,14] with a -5.7 p.p. lower probability, (14,21] with a -15.3 p.p. lower probability of saying they are happy (the 21 or more hours is statistically insignificant). For girls, working (7,14] hours per week is associated with a -6.7 p.p. lower probability they say they are happy and working more that 21 hours per week is associated with a -11.2 p.p. lower probability of saying they are happy. See Table 3.5.

In all specifications, household equivalent income, age, and ethnicity are not statistically significant. For the parental education variables, that a father went to university, compared to a father with no education, is associated with a higher probability that boys say that they are happy. The more children in the household the lower the probability that boys assess themselves as happy children.

Robustness Checks To explore non-linear effects of household work (i.e., that possibly, performing some housework is okay and perhaps even beneficial, but that performing chores for a long time may be burdensome for children), various functional forms are explored.

To explore whether there is a diminishing nature to performing chores at home and the associations with happiness of the children a quadratic of the child's total hours in household chores is introduced. The squared term is statistically insignificant, see Table B.1.

To test whether it is the binary nature of domestic work (i.e., the extensive margin) that affects the well-being of the children rather than the hours of chores performed (i.e., the intensive margin) unpaid domestic work, market work and school attendance are introduced as binary variables, i.e., 'child does household chores' = 1 if the child performs any type of chore weekly; 'child works in the labour market' = 1 if child worked in the labour market during the past week; 'child attends school' =1 if the child attended school during the past week. Table B.2 presents the results of including only binary variables. Performing domestic chores is associated with a -4.5 p.p. and a -3.3 p.p. lower probability that boys and girls say they are happy. The magnitudes of the marginal effects of doing housework are about half of the magnitudes of the effect of the extensive margin for work in the labour market (-10 p.p. for boys and -6.5 p.p. for girls). These findings suggest that chores are negatively associated with children's

well-being both in the intensive and extensive margins.

Table B.3 summarizes the main results when dividing the sample into children aged 8-14 and 15-17. As it is the case with the fourteen-hour threshold, most 'child labour' definitions 'allow' some work to be performed after children turn 15, i.e., if the children who are 15 years of age and over work below a certain hour limit, this work is legally allowed and thus, not considered child labour. To look at whether this age limit should be taken into consideration for work at home, I explore both the extensive and intensive margin of performing household work as well as the different thresholds of hours worked (panels (i.), (ii.) and (iii.), respectively, in Table B.3). Columns (1)-(3) look at the sample of children aged 8-14 and columns (4)-(6) look at the sample of children aged 15-17.

In all cases, for both the intensive and extensive margin and when using thresholds as independent variables, for the sample of children aged 8-14, housework is negatively associated with children's well-being. For the 15-17 year old sample, when looking at the intensive margin an increase in household work of 10 hours is associated with a decrease in the probability that girls say they are happy of -1.9 p.p., while it is statistically insignificant for boys (labour market hours are statistically significant for boys (not girls) and an increase of 10 hours in labour market activities is associated with a decrease in boys' well-being of -2.5 p.p., on average). The extensive margin and the threshold variables are also statistically insignificant for children above fifteen years old.

It is worth noting that the sample of 15-17 year old children consists of only of 3,232 children while the 8-14 year old sample consists of 11,578 children which could be why the results are no longer significant for the older children. It is also worth noting that, on average, 15 to 17 year old boys perform 7.78 hours of household work while girls perform 13.53 hours of household chores (boys also work on average 7.19 hours in the labour market while girls work 3.78 hours in the labour market). Girls work almost twice the amount of housework hours than boys and boys work almost twice the amount of house hours than work which is why there might be a gendered difference in the intensive margin variables.

3.6 Discussion and Conclusion

The work that Ecuadorian children aged 8 to 17 years old perform at home appears to be as detrimental as the work that children perform in the labour market. The mean level of happiness for children is relatively high, 86% of children self-report to be happy most of the time. An increase of 10 hours per week in domestic chores is associated with a -2.1 p.p. decrease in the probability that both boys and girls say their happy. Compared to work performed in the labour market, an increase of 10 hours in market work is associated with a -2.9 p.p.in the probability of children self-assessing as happy. While most of the research on child labour focuses on the impact that activities outside the home, in economic activities in the labour market, in children's own perception and assessments, the work they perform in their own home is almost as negative to their happiness as the work performed in the market. The negative effects of household work on children's self-assessed happiness are present in both the extensive and intensive margins. It is not just the fact that they perform chores, the number of hours in chores are also associated with lower probabilities that children in Ecuador say they are happy.

The gendered aspect of chores is present in Ecuadorian children. While girls spend more hours cleaning, shopping for groceries, doing laundry, cooking, caring for others, and helping others with their school work, the only activity in which boys spend more time doing than girls is household repair, a finding consistent with previous literature (Cerrato and Cifre, 2018; UNICEF, 2016; Cunningham, 2001; Wikle, 2014; Bonke, 2010; Blair, 1992a).

When looking at the different activities children perform at home, not all chores are equal. Some activities seem to have no effect on children's perceived well-being (i.e., cleaning, cooking, helping with schoolwork or household repairs) while shopping for groceries is associated with higher probabilities of happiness. For girls, an increase of 10 hours per week going shopping is associated with a 9.4 p.p. increase in the probability girls say they are happy. Perhaps it is the socializing of going out to get groceries, perhaps it is the enjoyment of purchasing items, why sometimes 'retail therapy' is colloquially perceived as a way to boost someone's mood or perhaps it could be that having the ability to purchase and shop household goods provides a sense of stability and thus, affects girls' well-being.⁵³ Another finding that is consistent with the literature is the negative associations between well-being and care-taking. For girls, an increase in 10 hours of care-taking is associated with a decrease in the probability that they say they are happy of 4.1 p.p., on average. For policies addressing the perceived happiness of children, this might be an area where children, particularly girls, would benefit with programs that provide respite for caretakers. Boys do not particularly seem affected when looking at the breakdown of chores, perhaps due to the fact that the amount of time per individual chore is much lower than what girls spend; meaning that, for boys, perhaps it is a matter of volume not type of chore.

When looking at how work is divided in the household, when compared to household activity performed by other members, boys seem to be negatively affected by taking on larger shares of chores. An increase in the share of total household work of 1% (i.e., 1% in the share of work from all household members) has a negative association with boys' happiness of -16.5 p.p., on average. In this case, going to school for boys is associated with a 15.3 p.p. increase in the probability that they self-assess as happy. Taking on a larger share of the household work done appears to be very negative to boys as the magnitude of the association is larger than the magnitude for the school variable. It could be perhaps that maybe it goes against their identity and shows a dislike in performing more 'feminine' activities. It could be that their perception of 'fairness' is affected. It could also be that perhaps the expectation of boys performing a large share of household chores is so low that when this expectation changes, boys perceive it as very negative.

When looking at only the activities performed by children (the share each child takes from the work performed by children in the household, i.e., relative to their siblings), both boys and girls have negative associations with the share they take on and their perceived happiness, though again, boys more so than girls. Interestingly, even though girls perform a larger share of work from children (48.8% on average, compared to 41.3% for boys), boys are more negatively affected (i.e., -6 p.p. lower probability of happiness for boys, compared to -4.2 p.p. for girls).

Importantly, the fourteen-hour threshold that is commonly used when taking into account domestic chores into definitions of child labour seems very arbitrary. Doing

⁵³Note that this does not imply that these girls are better off as I am controlling for family income.

any type of work affects children's happiness negatively. Performing up to seven hours of chores a week (less than one hour per day) has a negative effect for boys, though is insignificant and seemingly irrelevant for girls. Performing up to twenty one hours of chores per week is also negatively associated with happiness for boys but not for girls while doing over 21 hours is only negatively associated for girls.

Even though it seems that girls have a higher tolerance of conducting more housework before it negatively affects their well-being, this might be a result of early socialization or parentification of girls. If one were to look at the importance of these thresholds for definitions child labour, it is unlikely that legal standards would be adjusted by gender to take into account these differences (e.g., for boys, the threshold of hours for child labour is any household work done whereas for girls, since there are no negative associations with up to seven hours per week, child labour is anything above seven weekly hours). Rather, these findings could support the argument that, instead, the threshold should be lowered for all children, using the effect on boys as the baseline and the expectations for girls to do more work that boys is contested. As such, the negative effect of housework should be examined more broadly and perhaps the threshold should be as low as the one for market work in an aim to include both how it affects children's happiness and also to dismantle potential expectations for girls being socialized to do more chores than boys.

Another important finding that relates to how 'child labour' is defined and whether or not housework should be included in the legal framework of child labour is how, when dividing the sample of children by age, the negative effects of household work are robust for younger (8-14 year old) children. The analysis of extensive and intensive margin variables as well different thresholds for housework shows that there is a negative association with chores and children's well-being. Although there is a small sample for children aged 15-17, for older girls, a negative association with household work and well-being remains when looking at the intensive margin. Girls carry a larger burden of housework and the effect this has on their well-being should not be overlooked. As ubiquitous as chores might be, results suggest that legislators and policy makers should not disregard the work performed at home given the negative impact it can have on children.

Lastly, by far, the largest effect on self-assessed well-being of children comes from

their school attendance, especially for boys. On average, attending school is associated with a 13.6 p.p. higher probability of a child saying they are happy (15 p.p. for boys and 12.1 p.p. for girls). Ensuring that children go to school and remain in school seems to be an important avenue to ensure their well-being.

To conclude, this chapter provides evidence that the unpaid chore work that children perform in their home is negatively associated with their well-being. This finding is true both at the extensive and intensive margin. Moreover, not all chores have the same effect on children's well-being. The brunt of work that children take seems to be more important for boys than for girls. Importantly, the analysis conducted in this chapter was done solely using children's responses. While policies suggest that domestic chores could be beneficial to children's development of their skills, character and sense of responsibility, listening to what children say, through their own voices, points to this not being the case. Most countries have laws against child labour (only defined as market work) and multi-government agencies call for the elimination of child labour (ILO, 2017). However, while children's market work is eradicated little attention is given to the work that they perform in their own home, which children perceive almost as detrimental as the former though chores are not regulated. Domestic chores need to be studied more thoroughly so that policies to address their negative impact can be tested an implemented, respecting children's agency and awareness of their own well-being.


Figure 3.1: Proportion of Children Working in Chores by Gender

Source: National Survey of Child Labour



Figure 3.2: Child's Responses of Feeling 'Mostly Happy' by Ranges of Hours Worked

Source: National Survey of Child Labour

Table 3.1: Descriptive Statistics: Means Values for Children Aged 8-17 Years Old, Living in Two-Parent Households by Gender

	Female Mean	St. Dev	Male C Mean	St Dev	Diff. Girl-Boy	
	(1)	(2)	(3)	(4)	(5)	
	0.000	0.000	0.000	0.044	0.0000	
Mostly Happy (Dummy $= 1$ if shild is happy most of the time)	0.868	0.338	0.863	0.344	0.0033	
Occasionally/Rarely Happy	0.127	0.333	0.133	0.340	-0.0036	
(Dummy = 1 if child is happy occasionally/rarely)	0.221		0.200	010-00		
Well-Being $(0 = rarely, 1 = occasionally, 2 = mostly happy)$	1.863	0.358	1.858	0.361	0.0029	
Child Works in Domestic Activities	0.842	0.365	0.792	0.406	0.0543***	
Child's Weekly Hours Spent Cleaning	2.176	2.462	1.631	1.966	0.4957^{***}	
Child's Weekly Hours Spent Getting Groceries	0.714	1.469	0.685	1.421	0.0742***	
Child's Weekly Hours Spent Doing Laundry	1.475	2.133	0.810	1.464	0.6429***	
Child's Weekly Hours Spent Cooking	1.023	3.220 2.760	0.598	1.550	0.4922***	
Child's Weekly Hours Spent Caring for Others	1.004	5.709 2.362	0.962	2.070	0.4255	
Child's Weekly Hours Spent in Repairs	0.041	0.450	0.072	0.471	-0.0438***	
Child Does (0,7] Hours of Household Chores Per Week	0.461	0.498	0.532	0.499	-0.0918***	
Child Does (1,14] Hours of Household Chores Per Week	0.193	0.395	0.170	0.375	0.0440****	
Child Does More Than 21 Hours of Household Chores Per Week	0.097	0.290 0.287	0.035	0.227	0.0470	
Child's Total Hours of Household Chores	8.445	10.048	5.626	6.897	2.9084***	
Child's Share of Total Household Hours of Chores	0.114	0.107	0.078	0.076	0.0373***	
Child's Share of Children's Total Household Hours of Chores	0.488	0.346	0.413	0.347	0.0763***	
Total Household Hours of Chores, All Members	71.393	38.285	70.157	36.716	2.0976***	
Total Household Hours of Chores, Children	18.158	20.206	15.470	18.431	2.9317***	
Child Works in Market Activities	0.086	0.280	0.136	0.343	-0.0568***	
Child's Total Hours in Labour Market Activities	1.331	5.687	2.317	7.722	-1.2827***	
Child's Total Hours of Household & Labour Market Work	9.777	12.099	7.943	10.671	1.6257***	
Child Attends School	0.970	0.170	0.961	0.194	0.0070**	
Boy	0.000	0.000	1.000	0.000	-	
Age	11.792	2.767	11.897	2.795	-0.0861*	
Age Group: 8 to 11	0.501	0.500	0.474	0.499	0.0223^{***}	
Age Group: 12 to 15	0.366	0.482	0.390	0.488	-0.0160**	
Age Group: 15 to 17	0.132	0.339	0.136	0.343	-0.0063	
Oldest Child	0.558	0.497	0.561	0.496	-0.0019	
Identifies as: Afro descendent	0.081	0.272	0.077	0.207	0.004	
Identifies as: Mestizo	0.045	0.208	0.040 0.792	0.210 0.406	0.0013	
Identifies as: Other Ethnicity	0.065	0.246	0.085	0.278	-0.0055	
Dual-Earner Household	0.673	0.469	0.656	0.475	-0.0003	
Mother's Educ.: Less Than Primary	0.051	0.220	0.054	0.225	-0.0035	
Father's Educ.: Less Than Primary	0.035	0.185	0.032	0.175	0.0027	
Mother's Educ.: Primary	0.457	0.498	0.451	0.498	-0.0037	
Mother's Educ.: Secondary	0.351	0.477	0.359	0.480	-0.0003	
Mother's Educ.: University	0.141	0.348	0.136	0.343	0.0076	
Father's Educ.: Primary	0.400	0.499 0.476	0.480 0.356	0.300	-0.0175**	
Father's Educ : University	0.348 0.151	0.470	0.330 0.132	0.479	0.0078	
Household Equivalent Income	281.860	285.560	277.208	285.186	3.0761	
Household Size	4.961	1.324	4.978	1.356	-0.0183	
House has Children Below Age 5	0.295	0.456	0.301	0.459	-0.0042	
House has Other Boys Aged 5-17	0.563	0.496	0.563	0.496	-0.0026	
House has Other Girls Aged 5-17	0.551	0.497	0.542	0.498	0.0037	
Household Has a Member With A Disability	0.046	0.209	0.056	0.229	-0.0066*	
Head of Household is Female	0.014	0.118	0.013	0.111	-0.0004	
Parent or PMK Present During Child's Responses	$0.580 \\ 0.769$	0.494 0.421	0.578	0.494 0.420	-0.0011 -0.0057	
Observations	N =	7,294	N =	7,516	N = 14,810	

 $\label{eq:bernetic} \begin{array}{c} \mbox{Observations} \\ \hline \mbox{*** $p{<}0.01, ** $p{<}0.05, * $p{<}0.1$ denote levels of significance.} \end{array}$

	All Children (1)	Male Children (2)	Female Children (3)
Child's Total Hours of Household Chores (Scaled)	-0.021***	-0.021**	-0.021***
	[0.005]	[0.009]	[0.007]
Child's Total Hours in Labour Market Activities (Scaled)	-0.029*** [0.006]	-0.033*** [0.007]	-0.022** [0.009]
Child Attends School	0.136***	0.150***	0.121**
	[0.038]	[0.052]	[0.053]
Age Group: 12 to 15	-0.000	0.016	-0.017
Age Group: 15 to 17	-0.001	0.012	-0.014
	[0.017]	[0.022]	[0.025]
Oldest Child	-0.005	-0.010	0.000
Identifies as: Indigenous	0.010	-0.005	0.009
	[0.016]	[0.020]	[0.019]
Identifies as: Afro-descendant	-0.037	-0.055	-0.012
	[0.029]	[0.038]	[0.038]
Identifies as: Other Ethnicity	-0.026	-0.017	-0.038
Boy	[0.026]	[0.033]	[0.034]
Боу	-0.007	-	-
Dual-Earner Household	-0.011	0.009	-0.034**
	[0.012]	[0.015]	[0.015]
Mother's Educ.: Primary	0.008	0.035	-0.023
	[0.021]	[0.025]	[0.026]
Mother's Educ.: Secondary	0.028	0.066**	-0.015
	[0.023]	[0.027]	[0.030]
Motner's Educ.: University	0.022	0.029	0.018
Father's Educ - Primary	0.028	0.004]	$\begin{bmatrix} 0.054 \end{bmatrix}$ 0.027
rance's Educ I filinary	[0.028]	[0.033]	[0.032]
Father's Educ.: Secondary	0.016	0.001	0.022
v	[0.030]	[0.036]	[0.035]
Father's Educ.: University	0.060^{**}	0.073^{**}	0.040
	[0.027]	[0.029]	[0.036]
Household Equivalent Income (Scaled)	-0.014	-0.038	0.014
II	[0.027]	[0.034]	[0.040]
nousenoid Size	0.000	[0.018]	-0.005
House has Children Below Age 5	-0.022	-0.033*	-0.011
nouse has emiliaten below rige e	[0.015]	[0.019]	[0.019]
House has Other Boys Aged 5-17	-0.022*	-0.040**	0.001
	[0.013]	[0.018]	[0.020]
House has Other Girls Aged 5-17	-0.015	-0.039**	0.012
Haushald Har a Marshar With A Disability	[0.012]	[0.017]	[0.019]
Household has a Member with A Disability	-0.027	-0.020	-0.020
Head of Household is Female	0.018	0.004	0.033
	[0.038]	[0.052]	[0.040]
Urban Setting	-0.015	-0.004	-0.025*
	[0.014]	[0.018]	[0.015]
Parent or PMK Present During Child's Responses	0.033***	0.033**	0.030*
	[0.013]	[0.016]	[0.016]
Observations	1/ 810	7 516	7 204
Province Controls	Yes	Yes	Yes
Log Likelihood	-5558	-2833	-2678
Pseudo B^2	0.0499	0.0581	0.0576

Table 3.2: Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Total Hours of Household Work (Self-Reported)

	All Children	Male Children	Female Children
	(1)	(2)	(3)
Child's Total Hours of Household Cleaning (Scaled)	-0.007	0.009	-0.024
Child's Total Hours of Household Getting Groceries (Scaled)	$\begin{bmatrix} 0.024 \end{bmatrix}$ 0.055 $\begin{bmatrix} 0.027 \end{bmatrix}$	$\begin{bmatrix} 0.036 \end{bmatrix}$ 0.018	[0.031] 0.094^{*}
Child's Total Hours of Household Doing Laundry (Scaled)	[0.037] -0.066** [0.031]	[0.040] -0.076 [0.048]	[0.055] -0.063* [0.037]
Child's Total Hours of Household Cooking (Scaled)	-0.024	-0.028	-0.027
Child's Total Hours of Household Caring of Others (Scaled)	[0.017] -0.040*** [0.013]	[0.035] -0.034 [0.021]	[0.018] -0.041*** [0.016]
Child's Total Hours of Household Helping with Schoolwork (Scaled)	0.016	0.004	0.030
Child's Total Hours of Household Repairs (Scaled)	[0.023] -0.066 [0.074]	[0.028] -0.148 [0.000]	$\begin{bmatrix} 0.030 \end{bmatrix}$ 0.021 $\begin{bmatrix} 0.102 \end{bmatrix}$
Child's Total Hours in Labour Market Activities (Scaled)	-0.029*** [0.006]	-0.032*** [0.007]	-0.022**
Child Attends School	0.130***	0.152***	0.109**
Age Group: 12 to 15	[0.038] -0.001	[0.052] 0.016	[0.051] -0.017
Age Group: 15 to 17	-0.001	[0.013] 0.014	[0.014] -0.012
Oldest Child	[0.017] -0.006	[0.021] -0.011	[0.024] -0.002
Identifies as: Indigenous	[0.009] 0.005	[0.014] -0.005	$\begin{bmatrix} 0.014 \end{bmatrix} \\ 0.011 \end{bmatrix}$
Identifies as: Afro-descendant	[0.016] -0.039	[0.020] -0.059	[0.019] -0.010
Identifies as: Other Ethnicity	[0.030] -0.025	[0.038] -0.016	[0.037] -0.033
Boy	[0.026] -0.010	[0.033] -	[0.034]
Dual Farner Household	[0.009]	0.010	0.034**
Duar Land Household	[0.012]	[0.015]	[0.015]
Mother's Educ.: Primary	0.005	0.035	-0.028
Mother's Educ.: Secondary	[0.021] 0.025	0.066**	-0.025
	[0.023]	[0.027]	[0.030]
Mother's Educ.: University	0.018	0.028	0.011
Father's Educ.: Primary	[0.028] 0.018	[0.035] 0.000	[0.035] 0.029
Father's Educ.: Secondary	[0.028] 0.016	[0.033] -0.002	[0.032] 0.025
Pathan's Educe Hairanite	[0.030]	[0.036]	[0.034]
Father's Educ.: University	$[0.060^{3.3}]$	[0.072***	0.040
Household Equivalent Income (Scaled)	-0.014	-0.037	0.015
-	[0.027]	[0.033]	[0.039]
Observations	14,810	7,516	7,294
Province Controls	Yes	Yes	Yes
Log Likelihood P^2	-5542	-2828	-2663

Table 3.3a.:	Marginal	Effects from	Probit 1	Estimates	for the	Probability	of Feelin	ıg li	ke a Hap	py Child	by Gender
	of Child;	Independent	Variab	le: Total	Hours	of Household	l Work	by 1	Different	Activity	Performed
	(Self-Rep	orted)									

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, *p<0.1 denote levels of significance. Sample consists of respondents in the ENTI ages 8-17, living with both parents. The dependent variable, *happiness*, is an indicator variables equal to one if child feels happy most of the time. Estimates correspond to the marginal effects for the probability of the child self-assessing as happy. The base categories are: 'Child Doesn't Attend School;' 'Age: 8-11;' 'Not Oldest Child;' 'Child's Ethnicity: Mestizo;' 'Non-Dual Earners;' 'Less than Primary Educ.;' 'No Additional Members'; 'Male Head of Household;' 'Household in Rural Area;' 'PMK not Present.' 'Province: Pichincha' omitted.

	All Children	Male Children	Female Children
	(1)	(2)	(3)
Household Size	0.007	0.018**	-0.005
	[0.006]	[0.007]	[0.008]
House has Children Below Age 5	-0.019	-0.032*	-0.008
	[0.014]	[0.019]	[0.018]
House has Other Boys Aged 5-17	-0.022*	-0.041**	-0.001
	[0.013]	[0.017]	[0.019]
House has Other Girls Aged 5-17	-0.017	-0.040**	0.009
\mathbf{W}_{1}	[0.012]	[0.017]	[0.019]
Household has a Member with A Disability	-0.025	-0.025	-0.022
II. I. CII	[0.022]	[0.028]	[0.028]
Head of Household is Female	0.018	0.003	0.035
	[0.037]	[0.052]	[0.039]
Urban Setting	-0.015	-0.004	-0.026*
	[0.014]	[0.018]	[0.015]
Parent or PMK Present During Child's Responses	0.034^{***}	0.033**	0.031^{*}
	[0.013]	[0.016]	[0.016]
Observations	14,810	7,516	7,294
Province Controls	Yes	Yes	Yes
Log Likelihood	-5542	-2828	-2663
Pseudo R^2	0.0526	0.0601	0.0630

Table 3.3	o.: Marginal	Effects	from	Probit	Estimates	for the	e Prob	abi	lity of	Feel	ing like	a Happy	Child by
	Gender o	of Child;	Indep	pendent	Variable:	Total	Hours	of l	Househ	old	Work by	^v Different	Activity
	Performe	d (Self-H	Report	ed) (co	nt.)								

 Table 3.4a.:
 Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable:

 Share of Household Work Performed by Children (Self-Reported)

	Share of Total Chores I All Members of Househo			Share of Chores by All Children in Househ		
	All Children	Male Children	Female Children	All Children	Male Children	Female Children
	(1)	(2)	(3)	(4)	(5)	(6)
Child's Share of Total Household Hours of Chores	-0.128**	-0.165*	-0.109	-	-	-
Total Household Hours of Chores, All Members (Scaled)	-0.005*** [0.002]	-0.004** [0.002]	-0.006***	-	-	-
Child's Share of Children's Total Household Hours of Chores	-	-	-	-0.053*** [0.017]	-0.066***	-0.042*
Total Household Hours of Chores, Children (Scaled)	-	-	-	-0.007**	-0.005	-0.008**
Child's Total Hours in Labour Market Activities (Scaled)	-0.028***	-0.032***	-0.021**	-0.029***	-0.033*** [0.007]	-0.021**
Child Attends School	0.139***	0.153***	0.128**	0.139***	0.148***	0.130**
Age Group: 12 to 15	-0.002	0.017	-0.020	-0.001	0.016	-0.019
Age Group: 15 to 17	-0.003	0.013	-0.015]	[0.010] -0.005	0.008	-0.014
Oldest Child	[0.018] -0.007	[0.022] -0.011	-0.002	[0.017] 0.000	-0.000	[0.025] 0.002
Identifies as: Indigenous	[0.009] 0.004	[0.014] -0.005	[0.014] 0.009	[0.010] 0.006	[0.014] -0.003	[0.015] 0.011
Identifies as: Afro-descendant	[0.016] -0.037	[0.020] -0.055	[0.019] -0.013	[0.015] -0.035	[0.020] -0.055	[0.019] -0.010
Identifies as: Other Ethnicity	[0.030] -0.026	[0.038] -0.017	[0.038] -0.039	[0.029] -0.028	[0.038] -0.020	[0.038] -0.040
Boy	[0.026] -0.006 [0.009]	[0.033] -	[0.034] -	[0.026] -0.006 [0.009]	[0.033] -	[0.034] -
Dual-Earner Household	-0.012	0.009	-0.036^{**}	-0.010 [0.012]	0.011	-0.034** [0.015]
Mother's Educ.: Primary	0.009	0.036	-0.021	0.010	0.037	-0.021
Mother's Educ.: Secondary	0.030	0.068***	-0.012	0.029	0.068**	-0.015
Mother's Educ.: University	0.023	0.031	0.020	0.023	0.033	0.017
Father's Educ.: Primary	0.018	0.002	0.028	0.016	0.000	0.026
Father's Educ.: Secondary	0.017	[0.035] 0.002 [0.026]	0.022	0.014	-0.003	0.021
Father's Educ.: University	[0.030] 0.061** [0.026]	[0.036] 0.074*** [0.020]	[0.035] 0.041 [0.026]	[0.030] 0.059** [0.027]	[0.036] 0.070** [0.020]	[0.035] 0.041 [0.026]
Household Equivalent Income (Scaled)	[0.020] -0.015 [0.027]	[0.029] -0.041 [0.034]	[0.036] 0.016 [0.038]	[0.027] -0.014 [0.027]	[0.029] -0.037 [0.034]	[0.030] 0.015 [0.040]
Observations	14,810	7,516	7,294	14,810	7,516	7,294
Province Controls Log Likelihood	Yes -5549	Yes -2828	Yes -2672	Yes -5555	Yes -2826	Yes -2681
Pseudo R^2	0.0514	0.0598	0.0597	0.0505	0.0606	0.0567

	Share o All Men	f Total Cl bers of H	hores by lousehold	Shaı All Chil	re of Chore dren in He	es by ousehold
	All Children	Male Children	Female Children	All Children	Male Children	Female Children
	(1)	(2)	(3)	(4)	(5)	(6)
Household Size	0.008	0.019**	-0.004	0.006	0.017**	-0.005
House has Children Below Age 5	[0.006] -0.021	[0.008] -0.033*	[0.008] -0.011	[0.006] -0.022	[0.008] -0.033*	[0.008] -0.013
House has Other Boys Aged 5-17	[0.015] -0.021*	[0.019] -0.040**	$\begin{bmatrix} 0.019 \end{bmatrix} \\ 0.003 \end{bmatrix}$	[0.015] -0.026**	[0.019] -0.047***	[0.019] -0.003
House has Other Girls Aged 5-17	[0.013] -0.014	[0.017] -0.038**	[0.019] 0.014	[0.013] -0.021*	[0.017] -0.047***	[0.020] 0.009
Household Has a Member With A Disability	[0.012] -0.027	[0.017] -0.024	[0.018] -0.026	[0.012] -0.026	[0.017] -0.022	[0.019] -0.027
Head of Household is Female	$\begin{bmatrix} 0.022 \end{bmatrix} \\ 0.017 \end{bmatrix}$	[0.028] 0.004	[0.029] 0.031	$\begin{bmatrix} 0.022 \end{bmatrix} \\ 0.020 \end{bmatrix}$	[0.028] 0.005	$\begin{bmatrix} 0.030 \end{bmatrix} \\ 0.036 \end{bmatrix}$
Urban Setting	[0.038] -0.015	[0.052] -0.004	[0.040] -0.026*	[0.037] -0.014	[0.051] -0.003	[0.038] -0.024
Parent or PMK Present During Child's Responses	$[0.013] \\ 0.032^{***}$	[0.017] 0.033^{**}	[0.015] 0.028^*	$[0.014] \\ 0.033^{***}$	[0.018] 0.034^{**}	[0.015] 0.029^*
	[0.012]	[0.016]	[0.015]	[0.013]	[0.016]	[0.016]
Observations	$14,\!810$	7,516	7,294	14,810	7,516	7,294
Province Controls	Yes	Yes	Yes	Yes	Yes	Yes
Log Likelihood	-5549	-2828	-2672	-5555	-2826	-2681
Pseudo R^2	0.0514	0.0598	0.0597	0.0505	0.0606	0.0567

 Table 3.4b.:
 Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable:
 Share of Household Work Performed by Children (Self-Reported) (cont.)

	All Children (1)	Male Children (2)	Female Children (3)
Child Does (0,7] Hours of Household Chores Per Week	-0.035**	-0.048***	-0.020
Child Does (7,14] Hours of Household Chores Per Week	[0.015] -0.063***	[0.018] -0.057**	[0.022] -0.067**
Child Does (14 21] Hours of Household Chores Per Week	[0.019] -0.093***	[0.026] -0.153***	[0.029] -0.057
	[0.028]	[0.043]	[0.035]
Child Does More Than 21 Hours of Household Chores Per Week	-0.101*** [0.030]	-0.049 [0.041]	-0.112*** [0.039]
Child's Total Hours in Labour Market Activities (Scaled)	-0.029***	-0.033***	-0.020**
Child Attends School	0.142^{***}	0.152^{***}	[0.009] 0.131^{**}
Age Group: 12 to 15	[0.038]	[0.052]	[0.053]
Age 010up. 12 to 10	[0.010]	[0.013]	[0.014]
Age Group: 15 to 17	0.002	0.012	-0.007
Oldest Child	-0.005	-0.010	0.001
	[0.010]	[0.014]	[0.014]
Identifies as: Indigenous	0.006	-0.003	0.013
Identifies as: Afro descendent	[0.015]	[0.020]	[0.018]
Identifies as. Ano-destendant	[0.029]	[0.037]	[0.037]
Identifies as: Other Ethnicity	-0.027	-0.016	-0.039
	[0.026]	[0.033]	[0.034]
Boy	-0.008	-	-
Dual-Earner Household	-0.009	0.011	-0.032**
	[0.012]	[0.015]	[0.014]
Mother's Educ.: Primary	0.008	0.034	-0.022
Mother's Educ - Secondary	[0.021] 0.028	[0.025] 0.065**	0.026
Mother's Educ. Secondary	[0.023]	[0.026]	[0.030]
Mother's Educ.: University	0.021	0.029	0.016
Fathar's Educ , Drimany	[0.028]	[0.034]	[0.034]
rather's Educ.: Frimary	[0.028]	[0.003]	[0.030]
Father's Educ.: Secondary	0.018	0.001	0.025
	[0.030]	[0.036]	[0.034]
Father's Educ.: University	0.061**	0.073***	0.043
Household Equivalent Income (Scaled)	-0.016	-0.042	0.012
	[0.027]	[0.033]	[0.040]
Household Size	0.006	0.018**	-0.006
House has Children Below Age 5	[0.006] _0.022	[0.007] -0.036*	[0.008]
House has emiliten below fige 5	[0.014]	[0.019]	[0.018]
House has Other Boys Aged 5-17	-0.021	-0.039**	0.002
Hauss has Other Cirls Ared 5 17	[0.013]	[0.017]	[0.020]
House has Other Ghris Aged 5-17	-0.013	[0.017]	[0.013]
Household Has a Member With A Disability	-0.027	-0.025	-0.026
	[0.022]	[0.028]	[0.029]
Head of Household is Female	0.020	0.005	0.032
Urban Setting	-0.014	-0.002	-0.024
	[0.014]	[0.018]	[0.015]
Parent or PMK Present During Child's Responses	0.033***	0.035**	0.030*
	[0.013]	[0.016]	[0.016]
Observations	14,810	7,516	7,294
Province Controls	Yes	Yes	Yes
Pseudo K ²	0.0519	0.0636	0.0601

 Table 3.5:
 Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Thresholds of Total Hours of Household Work (Self-Reported)

Chapter 4

The Road Not Taken or The One Most Travelled By: The Intergenerational Transmission of Child Labour in Ecuador

4.1 Introduction

Child labour has been widely discussed in the policy arena over the last few decades.¹ In most cases, child labour refers to children illegally working while being under the minimum age requirement (in most countries, 14 years of age) or children above the minimum age requirement who exceed the amount of work they are allowed to perform (usually 30 hours of work in economic activities² per week) (ILO No. 138, 1973; ILO

¹A note on definitions. It is important to mention that there is no clear global consensus on what 'child labour' is. Although the ILO has resolutions outlining minimum age requirements or the worst forms of child work, each country has its own way of defining child labour (ILO, 2017; Dammert et al., 2018). This chapter focuses on exploring different definitions of child labour when analysing Ecuadorian households' decision to send children to work and/or school. As a benchmark, given the importance of its use in policy, I start by defining 'work' as child labour according to Ecuadorian legislation. That is, children who are illegally working violating minimum age requirements, exceeding hour limits, whose work interferes with their schooling or who work in dangerous conditions. I then examine the heterogeneity in the type of work and work conditions for Ecuadorian children by using different measures of 'work' reflecting this heterogeneity.

 $^{^{2}}$ Again, there is variation in what countries define as children employed in economic activities, and thus, child labour.

No. 182, 1999; ILO, 2017). From the bans and restrictions imposed by numerous countries, there seems to be a widely accepted understanding that child labour is in most, if not all respects, harmful. Yet, despite all bans, child labour has not been eliminated and is pervasive worldwide. According to the latest estimates of the International Labour Organization (ILO), as of 2017, there were 152 million children working, and 73 million of these children engaged in hazardous labour. Regional estimates of prevalence vary from 2.9% in Arab States to 19.6% in Africa and affects girls and boys differently (of the 152 million children, 58% are boys and 42% girls). Of all the children who work, it is estimated that 32% are out of school and for the remaining 68%, their educational attainment is hindered (ILO, 2017).

Most policies to address child labour stem from concerns about human capital implications. Working prevents children from going to school, which will decrease their future earnings (Emerson and Souza, 2003; Udry, 2004). There is a vicious cycle associated with child labour, where poor families send their children to work and children grow to become poor adults predicted by the diminished investment in their education (Emerson and Souza, 2003; Udry, 2004). Theoretically, there is the premise that child labour is partly a result from a vicious cycle of poverty. Empirically, it is important to understand the mechanisms of the intergenerational transmission, especially due to policy implications. On the one hand, it could be that the mechanism is exclusively an economic one, where child labour is transmitted from parents to children due to a budget constraint resulting from poverty. Presumably, if there is a dire need to sustain the household, the quality and potentially dangerous characteristics of the work would not be as important as the need to send the children to work. In this case, improving the economic standing of parents would be beneficial to children and child labour could be diminished or even eradicated.

On the other hand, it could also be that the mechanism for the intergenerational transmission of work is partly an attitudinal or behavioural one (i.e., that if parents were themselves child labourers, this might have an effect beyond the consequences of having had worked to sustain the household for current family income). Child labour may be perpetuated due to the beliefs, values, expectations and potentially rites of passage that are passed on from parent to child (e.g., "I learned a lot from being a child labourer" or "I had to work as a child and so should you"). If parents

send children to work to develop attitudes and abilities, then, it is likely that children would not be engaged in work that would harm them. The transmission mechanism and the way in which parents make this evaluation of sending their children to work would depend on how they perceive and value formal education, returns to formal education, as well as how dire their current household needs may be. In terms of policy, if the goal is banning child labour due to the human capital argument and if the behavioural mechanism is important, even if poverty is eliminated, child labour would remain.

This chapter contributes to the literature on child labour by exploring the effects of parental work histories of being child labour themselves on the family's school-work decision for their children (aged 5 - 17 years old). It does so while taking into account the heterogeneity in the type of work (e.g., market vs. domestic work; paid vs. unpaid work) and the working conditions (light vs. heavy work; safe vs. unsafe - dangerous, unhealthy and abusive - work) of children in Ecuador. Taking advantage of a detailed survey, where both children and parents are surveyed within a household, I can identify the age of first employment for both the children and parents for individuals who are currently working. The dataset also includes information on the households' current work experiences, incomes and poverty status.

Exploring the mechanisms of the intergenerational transmission of child labour and separating the behavioural and attitudinal argument from the budget constraint argument is important for policy prescriptions. If for instance, paid work in the labour market occurs as a consequence of poverty, where, children endure unfavourable work conditions in order to receive the pecuniary returns, it can be regulated (or more heavily regulated), even if imperfectly so. Notice, however, that child labour in the form of unpaid household work, is not subject to government regulation, it is often unaccounted for, yet a valuable activity for households.³ It could be that labour performed by children, however, is perpetuated intergenerationally because of the attitudinal and behavioural motives rather than the budget constraint motive. It is thus, not just important to understand which mechanism is at play, but also to understand which mechanism applies to which kind of work for the creation of effective

³Children engaging in unpaid domestic work could be a way of alleviating poverty within a household. A child performing unpaid chores in the household could free up parents' time from these chores so that they can participate in paid market activities. In doing so, household income could increase, improving the well-being of the household.

policies that would benefit children.

This chapter seeks to address the following questions: (a.) what is the effect of having a parent who worked as a child on the likelihood of their own children (ages 5-17) engaging in child labour or going to school? Specifically, if a parent started working before the minimum age requirement (15 years old), are their children more likely to engage in child labour and less likely to go to school after controlling for current family poverty? (b.) are there differences in the intergenerational transmission of child labour depending on the type of work in which the child engages (i.e., market vs. domestic work, paid vs. unpaid work, light vs. heavy work, safe vs. unsafe work)? (c.) are there gender differences in the strength and nature of the intergenerational transmission process?

The decision to send children to work or to school is assumed to be simultaneously determined by the household. By modelling the household's school-work decision jointly, I am able to take into account correlations in the error terms of the school and work equations. For instance, if there are individual or household unobserved characteristics (e.g., like a child's unobserved ability or keen nature to learn and seek experiences) that jointly determine the school-work decision, modelling these decisions separately may produce biased estimates. The experiences of going to school and working vary, many children work while they are enrolled in school, other children do not work or go to school. Some children only work and do not go to school and others dedicate all their time to school and do not work (ILO, 2017, 2018; Zapata et al., 2011), thus, modelling the school-work decision jointly seems most appropriate.

Results suggest that there is an intergenerational effect of child labour for Ecuadorian children beyond the effects of parental child labour on current family income. Children of parents who were child labourers are more likely to combine school and work and more likely to work and forego school when defining 'child work' according to Ecuadorian legislation (i.e., children aged fourteen or younger working at least an hour a week in market activities (both with and without pay) or children between the ages of 15-17 whose work day is over six hours per day, who work over five days per week, whose work limits their right to an education or whose work is under dangerous conditions). Specifically, if a child's mother worked as a child herself, the probability that the child combines school and work is 2.0 percentage points higher for boys and 1.7 p.p. higher for girls. For boys only, if their mother worked as a child, they have a 0.8 p.p. higher probability to only work, without going to school. For children of fathers who started work below the age of 15, the probability of working and going to school is 1.9 p.p. higher for boys and 1.7 p.p. higher for girls, on average, compared to the children of fathers who were not child labourers.⁴ The intergenerational transmission of child labour depends on the type of work the children perform, and is not gender neutral. Thus, policies solely aimed at reducing or eliminating child work through poverty alleviation may reduce the worst cases of child labour (e.g., those in hazardous or abusive environments), but poverty alleviation alone may not eliminate child labour given the intergenerational links that remain between children and child labourer parents.

What remains of the chapter is organized as follows. Section II provides a literature review. Section III discusses the legal and contextual framework of child labour and education in Ecuador. Section IV outlines the data and the definitions of variables used in the analysis. Section V presents the empirical strategy used. Section VI presents the results and Section VII concludes.

4.2 Relevant Literature

Child labour can be treated as a necessary best response that a household chooses to ensure their survival when in desperate circumstances (Basu & Van, 1998; Edmonds, 2007; Emerson and Souza, 2003; Udry, 2004). Working as a child is considered to be detrimental as it prevents children from investing in their human capital, mainly due to work interfering with or diminishing their time in school or for school activities. While the benefits of child labour are immediate, either through increased income or consumption, or through freeing up time resources for other family members, the main costs of child labour - reduced human capital investments - are realized in the future.⁵

It is worth noting that in most cases where the human capital argument is used against child labour, there is little nuance to the discussion of consequences

 $^{^{4}}$ Results are quite robust to more flexible measures of 'child labour' that do not incorporate as many restrictions as the official definition does.

⁵The losses from lower educational attainment may not only be lower future earnings when these children grow and enter the labour market, they may also extend to other dimensions, e.g., future health (Udry, 2004; Currie and Moretti, 2003; Currie, 2009). Similarly, the losses are not just individual, but in aggregate, there could be a less productive workforce in society that would hinder society's development (Udry, 2004).

stemming from the work children perform. Working could arguably allow children to learn valuable skills, promote their sense of purpose and develop their autonomy. Empirically, returns to child labour have been documented. Studying children from Vietnam, Beegle et al. (2009) explore the consequences of child labour five years after the children work. Results show that working during the ages of 8-13 results in decreased chances of being enrolled in school and lower grade completion five years later. However, when looking at their labour outcomes, children who worked in the past are significantly more likely to work for wages and be earning a higher wage, i.e., the returns to work experience appears to be higher than the returns to schooling (Beegle et al., 2009). The negative effects from forgoing investing in human capital through education seem to be offset by the benefits of having earlier work experience, at least in the short run. Even if it is not paid market work, whether it is working at the family farm or taking care of younger siblings, one could make the case that early exposure to work could be at least somewhat beneficial to children.

There are several theoretical arguments about why parents send their children to work.⁶ A child working now increases family income either through their work or through freeing up time for their parents so that the parents can work (Edmonds, 2008; Ali, 2019; Emerson and Souza, 2003).⁷ Theoretical discussions on the supply of child labour assume a unitary household, usually an overlapping generations model with a household that lives for two periods, consisting of an (altruistic) adult parent and a child. The decision-maker has preferences over current consumption and the child's future well-being, produced from human capital investments in education (the educational attainment as children is what determines wage earnings in adulthood). Children face a time constraint, limiting total hours of work and school. Sometimes leisure is included in the time constraint, other times leisure and play are assumed to be part of the child's education (Edmonds, 2007). Domestic work, work in farms or family enterprises or work for wages can all be modelled under 'work' (Edmonds and

⁶It is important to note that the decision that the child works, without delving into the potential future consequences for children, starts with a problem of agency. In most theoretical models, it is the parents (or unitary parent) who make(s) the decision to send the child to work or to school. Even though the household can immediately reap the benefits of the child's work, it is the child who experiences a reduction in their future welfare from not being able to invest in their education and accumulate human capital. Parents could potentially compensate their children for their work, through a bequest in the future, for instance. However, there is usually no way to enforce such compensations and particularly, for poor families, parents may never have the means to accumulate enough savings for such endowments (Basu & Van, 1998; Udry, 2004).

⁷For instance, if a child stays at home to conduct all unpaid household chores, the child's parent who was responsible for completing all unpaid homemaking could now go to the labour market and work for a wage.

Schady, 2012). There can be alternative lucrative uses of time beyond the work-school choice, like petty crime (Gonzalez and Rosales, 2017). Total family earnings include the wages of the adult as well as the wages of the children (if children engage in paid work). The family's budget constraint restricts consumption to be less than or equal to the household's total income that comes from adult and child earnings, with assumptions made to make the constraint hold with equality. Liquidity constraints are usually implicit, households cannot borrow against the future earnings of their children. Households maximize utility subject to their budget constraint and the children's time constraint which results in an optimal choice for schooling and education. Multiple equilibria can be found where households send their children to school and/or to work depending on the assumptions made over the returns to schooling and preferences over current consumption (Basu & Van, 1998; Emerson and Souza, 2003; Edmonds, 2008; Edmonds and Schady, 2012; Udry, 2004).⁸

The emphasis of such models is usually on the role that poverty plays in household decisions. The crucial assumption tends to be that a child works if and only if everyone in the family falls below a fixed subsistence level without the child's income (Basu & Van, 1998), or if the benefit the household would enjoy in the future from investing in human capital are outweighed by the contributions from the child's work in current consumption (Emerson and Souza, 2003). Thus, the argument of child labour being a 'best response' to dire situations follows as households are only assumed to send children to work if it helps them survive. Nevertheless, even if child labour is a best response in extreme circumstances, this does not mean it is desirable socially. The intergenerational link, where poor parents decide on the human capital accumulation of their children, in turn may lead to those children becoming poor adults. If the parents themselves started working as children, this is a multi-generational poverty perpetual cycle.

A concern in economics has been equality of opportunity and social mobility (D'Addio, 2007; Corak, 2006; Corak and Piraino, 2011; Corak, 2013). Economists have explored the strength of intergenerational correlations through the transmission of "abilities, traits, behaviours, and outcomes from parents to their children," i.e., "intergenerational transmissions" (Lochner, 2008, p. 1). Particularly, in the presence

 $^{^{8}}$ If the model is a general equilibrium model, assumptions need to be made regarding firms and the model is solved as a standard labour market problem, see Basu & Van (1998).

of low intergenerational mobility, individuals' lives may reflect characteristics and realities for which they are not responsible, an example of which would be child labour. In this case, children's lives resemble their parents' and social mobility is low. This could lead to children entering a cycle of intergenerational deprivation (D'Addio, 2007). Breaking this cycle could make it possible for children to make prosperous transitions throughout their lives and allowing everyone to have an equal playing field through equality of opportunity.

In high-income countries, there has been evidence of a strong link between poverty and inequality and intergenerational transmission of mobility or work histories.⁹ D'Addio (2007) highlights the importance of understanding the mechanisms of the intergenerational links across generations. Learning the extent to which children may experience disadvantage from the disadvantages that their parents experienced, can have strong ramifications for policy implications that could intervene by allowing children the "best possible start to their lives." However, there is no consensus about the desirability of policies particularly because of the difficulty in identifying which characteristics in family background matter most in affecting the future outcomes of children (D'Addio, 2007).

In a cross-country comparison of intergenerational mobility, Corak (2006) explores how the long-run labour market success or failure of children is related to that of their parents. Analysing the degree of generational mobility of earnings in a society, i.e., the generational income elasticity, Corak (2006) shows that there is significant variation across countries in the extent to which children inherit their parental advantages. Some countries like France, the United States and the United Kingdom document strong generational persistence, while in other countries like Canada, Finland, Norway and Denmark, family economic status is not as strongly associated with children's earnings success in adulthood.¹⁰ The results are similar in many OECD countries,

⁹D'Addio (2007) provides an extensive survey of research in OECD countries exploring intergenerational mobility in income, education, occupations and personality traits. The survey also includes discussions on the importance of mobility under the lens of equality of opportunity and why it may be desirable and beneficial for societies. The Organisation for Economic Co-operation and Development (OECD) members are: Austria, Australia, Belgium, Canada, Chile, Colombia, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Iceland, Ireland, Israel, Italy, Japan, Korea, Latvia, Lithuania, Luxembourg, Mexico, the Netherlands, New Zealand, Norway, Poland, Portugal, Slovak Republic, Slovenia, Spain, Sweden, Switzerland, Turkey, the United Kingdom and the United States.

¹⁰Specifically, Corak (2006) finds that in the United States, for instance, half of all children born to low-income parents become low-income earners as adults. In Canada, for one in three children who were born to low income families, their low-income status persists as adults. Studying Canadian men, Corak and Piraino (2011) examine the intergenerational transmission of employers, specifically the main employer that fathers had when their children were adolescents becoming the child's main employer when children grow to become adults. Using a large administrative

mobility is usually lower at extremes, the top and bottom of the distribution, though early investments on children are helpful and education is a major contributor to mobility (D'Addio, 2007).

Poverty and the intergenerational transmission of child labour and deprivation are at the core of the discussion for low- and lower-middle-income countries. As Udry (2004) posits, the mutual reinforcement of child labour and poverty and the human capital implications are particularly important. Though the theoretical literature highlights the role that poverty has on child labour incidence, the empirical literature provides conflicting evidence.¹¹ While some scholars find evidence that poverty is the main reason that parents send children to work (Schady and Araujo, 2006; Edmonds and Schady, 2012; Fallon and Tzannatos, 1998; Grootaert and Patrinos, 1999; Edmonds and Pavcnik, 2002; Cockburn, 2001) others find that child labour also occurs when income is high or that there is an income effect with improving households' economic conditions (Bhalotra and Heady, 2003; Ray, 2000; Kruger, 2007; Soares et al., 2012) and that there is an intergenerational effect beyond what can be attributed to income (Emerson and Souza, 2003). Sorting out the intergenerational transmission mechanism is particularly important for policy. If the goal is to eliminate child labour, the efficacy of policies that aim to stop children from work by alleviating poverty (e.g., cash transfers, increasing access to credit) may be hindered if poverty is not the only or primary reason why families send children to work.

How child labour is defined, depending on the conditions in which children work and the type of work the children perform, is crucial for distinguishing between poverty or behavioural/attitudinal determinants. Ali (2019) argues that when studying the effects of poverty on child labour, most studies treat child labourers as one homogeneous group which ignores the differences that exist in the working conditions of children such as work intensity, hazard exposure and employer which in turn may also reflect heterogeneity in parental preferences or perceptions on returns to schooling or work. Ali (2019) defines sub-populations of working children depending on their working

dataset, results show that, at some point in time, 41% of men are employed with an employer their fathers worked for and that 5.6% of sons at age 33 have the same employer their father did two decades earlier (9.4%, conditional on the firm still being in business). These intergenerational associations are stronger at the top of the earnings distribution (Corak and Piraino, 2011).

 $^{^{11}}$ For a review of the literature on child labour, see (Basu, 1999; Edmonds, 2007, 2008; Fors, 2012). For a discussion on how the theoretical literature on child labour and the emphasis on the role of poverty conflicts with empirical findings, see Ali (2019).

conditions and conducts a series of bivariate probit models on the child labour and school attendance decision. The criteria used to differentiate favourable and unfavourable working conditions are: type of employer (market work vs. unpaid family work), work intensity (light economic activity vs. the definition of child labour from the ILO), age at first job (working as an adolescent vs. working before the age of 12). In addition to the intergenerational effect that is transmitted from parental income to child labour, a control variable that is equal to one if the head of the household worked as a child is included.¹²

Analysing 54,669¹³ Egyptian children, Ali (2019) finds that with a broad definition of child labour (i.e.,children working at least one hour in the previous week), as income increases, children's probability of work decreases and children are more likely to go to school, other factors held constant. However, when exploring the heterogeneity of child's work, the effect of parental income is minimal on work that is not likely to harm children (work in the family business, jobs that are not physical or hazardous). On the other hand, higher parental income decreases the likelihood that children work in unfavourable work conditions (working in the labour market, jobs that are physically demanding and hazardous). In light of these findings, interventions that would increase the income of poor households with children working in unfavourable conditions would likely reduce this type of child labour. However, these interventions would have little effect on households who send their children to work for reasons other than the pecuniary returns from work.

Zapata et al. (2011) also explore differences in the definition of child work and consider the role that gender and ethnicity have on the work-school trade-off. The authors incorporate domestic chores to a second definition of child labour to understand the extent to which domestic activities displace schooling. Child labour is first defined as, a child (7-14 y.o.) working 15 or more hours a week in market activities and second, a child working 15 or more hours in market and domestic activities combined.¹⁴ Using

 $^{^{12}}$ Coefficients or marginal effects of this variable are not reported though 19% of non-working children had household heads who worked as children as did 41% of working children.

 $^{^{13}}$ The starting sample is of 66,922 children aged 5-17. The restrictions on the sample are children who are sons or daughters to the head of the household, where at least one parent is present. The estimations include an indicator variable for the household being a single-parent household and another indicator denoting whether the mother is absent. The sample is further restricted to children who are too young to attend school as it wouldn't enter as a choice for the household and for whom there is no missing data on monthly income or school status.

 $^{^{14}}$ The authors use the survey question "did you work last week?" and "if not, were you absent due to sickness, vacation, labour strike, adverse weather, etc." to denote whether a child worked in market activities. Domestic work

a bivariate probit model, when including domestic chores into the definition of child work, girls are 23% more likely than boys to be out of school and working. In fact, for the 5,277 children in Bolivia that are examined, when the work is of high-domestic intensity (i.e.,50% or more of total hours worked are dedicated to domestic chores), girls are 51% more likely than boys to be working and out of school (Zapata et al., 2011). In contrast, when the definition of child work is restricted to market activities and excludes domestic chores, the opposite occurs, boys are more likely to work than girls. Outcomes for children differ depending on the type of work performed.

Exploring whether the child labour status of parents affects the probability of their own children being child labourers, Emerson and Souza (2003) investigate whether there is a child labour trap in Brazil. The main assumption is that financial need creates the link between parents and their children though they also examine whether the intergenerational link remains beyond what can be transmitted when income is held constant. The model presented is an overlapping generations model of the household's child labour decision that allows for a poverty trap.¹⁵ Empirically, 28,805 Brazilian children from the ages of 10 to 14 are studied.¹⁶ A child is defined as a child labourer if they worked any positive hours per week in the labour market (the threshold is increased to 20 hours per week as a robustness check, and results hold). Parents are said to have been child labourers if the age at which they started working is below 14 years. Two main models are estimated: a probit model using the child labour

is defined as children carrying out any of the following activities in the household: taking care of children and/or elderly family members, cooking and cleaning, washing and/or ironing clothes, performing minor household repairs, grocery shopping, chopping and carrying firewood, carrying water from an external water source and organizing and maintaining neatness (Zapata et al., 2011, p. 598).

¹⁵The model builds on the seminal work of Basu & Van (1998). In Basu & Van (1998), the model relies on two key axioms. The first, a luxury axiom, treats children's non-work as a luxury good where there might be cases in which households cannot afford this good. However, as soon as the household's income rises to a certain level, households are able to acquire it. The second axiom is the substitution axiom, where firms view child and adult labour as substitutable. Empirically, these hypotheses have not always held, see Ray (2000), where both hypotheses are rejected in a study comparing child labour in Pakistan and Peru (the luxury axiom is rejected in both countries, the substitution axiom in Pakistan). In Emerson and Souza (2003) instead of the luxury axiom, it is assumed that the head of the household makes the decision to send the child to work and this decision is made only if the child's current contributions to consumption outweigh the benefit the household would enjoy from future consumption if the child remains in school. Following assumptions on the household's preferences and the technology converting education in childhood to human capital, an analytical solution is presented where there is a critical value of human capital above/below which parents will send their children to study/work full-time, i.e., there are two steady state equilibria, one at full education and one at no education. If the household places more weight on their current consumption, the higher the likelihood that they will also choose no education. The higher the returns to schooling, the more likely the household will choose education over sending the child to work.

 $^{^{16}}$ The original sample of 36,975 is restricted by having complete work histories for both mothers and fathers, singleheaded households are excluded as well as households whose age difference between the head of the household and oldest child is less than 14 years.

indicator for children regressed on their parent's child labour status and a vector of controls; and a Cox proportional hazard model of the year difference between the child's birth and age of entering the labour force on the parent's child labour indicator variable with the vector of controls.¹⁷ If parents were child labourers, children are more likely to be child labourers themselves. Given that the model predicts that if the child labour trap is the only determinant of intergenerational persistence of entering the workforce as a child, then the parental child labour effect would disappear once family wealth (or a proxy) is included as a control. Using parents' years of schooling as a proxy for family wealth, the parental effect of child labour remains (the effect of parental schooling itself is negative and statistically significant; the more educated the parents, the lower likelihoods that their children work).¹⁸ Results suggest that the persistence of child labour seems to remain beyond the effect of family income or parental educational attainment. Additionally, the earlier an individual starts to work, the lower their earnings in adulthood. Overall, the human capital gains that a child would obtain from educational investments outweigh those that could come from experiences in the labour market.

In Ecuador, specifically, both child labour and the intergenerational persistence of poverty have been examined in the context of cash transfers. Schady and Araujo (2006) explore the unconditional cash transfer, "Bono de Desarrollo Humano" (BDH, 'human development transfer')¹⁹ and its impact on school enrolment and child work. The cash transfer positively impacted schooling and it negatively impacted child work (Schady and Araujo, 2006). Edmonds and Schady (2012) study the BDH transfer

 $^{^{17}}$ Emerson and Souza (2003) estimate three additional models (a sequential probit, a multinomial logit and a bivariate probit model) to account for the joint decision structure of child labour and school attendance. The main results hold for all models.

 $^{^{18}}$ To enhance the proxy of permanent family income, Emerson and Souza (2003) explore the effect of the educational attainment of their grandparents, finding no impact on child labour status of the children. The impact of schooling seems to operate through occurring through the child's parents' education only.

¹⁹The "Bono de Desarrollo Humano" (BDH) is (now) a conditional cash transfer. In 2012 it provided households in poverty with \$35 (currently, it provides \$50 to households in extreme poverty). Originally, it was instituted in 1998 (due to the country undergoing a financial crisis with high inflation, extremely devaluation of the currency which resulted in dollarization, a bank run and high unemployment). It has been explicitly means tested for poor households (in the lowest two quintiles) and when instituted, due to lack of monitoring or enforcing capacity it was virtually unconditional. Following changes in 2003-2013, to receive the transfer each member of the household below the age of 6 and pregnant women were required a bi-monthly visit to a doctor. It also required that children between the ages of 6-16 regularly attend public school (only four absences per two months were allowed). There was a 'proof of attendance' process where schools would provide stamps to a 'verification card' confirming that the children attended school. If households failed to comply to this schooling requirement, there was a \$6 USD penalty. In 2013, the conditions of the transfer became stricter, both in health and education requirements and it further stipulated that households must commit to refrain children under the age of 15 from working and begging (however, there were no mechanisms in place to verify compliance of the latter commitment).

and the effect it has on the decision of Ecuadorian families to send their children to work. Using random assignment of the BDH to certain families as a source of variation,²⁰ Edmonds and Schady (2012) find that poor families use this income to delay the children's entry to the labour market. Focusing on 1,833 children who were 11 to 16 years of age at the baseline, child labour declines with receiving the BDH even though the size of the transfer does not offset the loss of earnings from the children. Particularly, the \$15 USD transfer that accounted for approximately 7% of expenditures of the households studied but less than 20% of median earnings from the work of children (\$80 USD), resulted in a 41% reduction of paid work for all children (a 78% reduction for children who were already enrolled in school) and a 34%reduction in unpaid activities. Similarly, for the children who attend school, there is also a shift from paid to unpaid activities presumably due to the rigidity of schedules in paid work. The higher income households achieve from receiving the transfer allows them to feel that they can continue to afford their children's schooling. Edmonds and Schady (2012) highlight the important role that poverty plays in deciding whether to send children to work or not.

Araujo et al. (2017) study the long-term (10 years) effects of the BDH in helping households escape intergenerational poverty traps. The identification mechanism is using the fact that there was randomization in assigning children under age six into early or late treatment groups (the former receiving \$1,200 USD in transfers, compared to \$625 for the latter in the period analysed) and that there was a cut-off for households to become eligible (making children just eligible or just ineligible). Comparing the children in the early treatment group to the ones in the late treatment group, there are no differences in language, mathematics, attention, or fluency tests, nor in behavioural outcomes. Further, focusing on the children who were eligible for transfers (at the age of entering and completing secondary school), they are slightly more likely to complete secondary school compared to the just ineligible children and there is no effect on increased employment of these (now) adults (and not due to higher enrolment in tertiary education). Overall, the long-term effects of the BDH appear to be modest. In Ecuador, it seems that programs, like this unconditional cash

 $^{^{20}}$ Based on scores of the means test, households were deemed eligible. During the re-structuring of the program in 2003, for new eligible households, the roll-out of the program contained a randomized component in four of the 24 provinces. Within selected parishes (territorial division), a sample of households was randomly selected for evaluation through a lottery and more households were eligible than those selected through the lottery.

transfer, are not sufficient to break the cycle of inter-generational poverty where the children of poor households (who are liquidity-constrained and do not invest in the human capital of their children) are not able to escape poverty in adulthood. It could also be that it is still too early to tell and ten years is not enough time to evidence the impacts of the investments.²¹

An important distinction in the literature is the way in which the decision to send children to work or school is modelled. Grootaert and Patrinos (1999) compare the determinants of child labour in Cote D'Ivoire, Colombia, Bolivia and the Philippines and approach the allocation of child's time as a sequential decision making process where assumptions are made regarding which options are preferred by the households. The authors note that whether a simultaneous or sequential approach is taken depends on how the decision process is viewed, whether it is assumed that the decision maker in the households considers the options available to them simultaneously or sequentially, in a hierarchical manner (Grootaert and Patrinos, 1999). The hierarchy presented for ordering of alternatives is: school, wage work, home enterprise work and no work.²² In many cases, the impact of poverty on child labour has been studied using specifications that include child labour as a dependent variable, usually as an indicator of whether the child works or not (or number of hours to look at the intensive margin) (Edmonds, 2007; Edmonds and Schady, 2012; Ali, 2019; Udry, 2004; Beegle et al., 2004; Bhalotra and Heady, 2003). Emerson and Souza (2003) estimates a probit model using an indicator variable for child labour as the dependent variable, a Cox proportional hazard model of the difference in years between birth and the age children start to work, a sequential probit, a multinomial logit and a bivariate probit model to take into account the decision structure. Kruger (2007) estimates the school and child labour decision separately with univariate independent probit regressions for child

²¹Paxson and Schady (2010) find that the BDH improves the physical, cognitive and behavioural development of young (3-7 year old) children. Given the importance of early investments in childhood, (Almond and Currie, 2010), it could be that it is just too early to bear witness to the effects, if any.

 $^{^{22}}$ Grootaert and Patrinos (1999) argue that there is little dispute with schooling. However, if the quality of education or the returns to education in a country are low, then potentially under the same criteria (welfare of the child and income contribution), education can be may not always be the preferred alternative. Similarly, domestic chores are not included which could be highly beneficial for a household if it liberates an adult (who can then earn adult wages) from domestic work to labour market work. Further, wage work may be informally conducted, meaning that children may earn less than adults. In this case, if the household owns a business or a farm, where work is required and a non-family member may be hired to conduct such work, an alternative would be to 'hire' the children 'saving' the (adult) wages that would've been paid to the non-family member, which could potentially be larger savings than earnings that a child would make informally. Further, the hierarchy does not include petty crime that could be found to be lucrative for a household (Gonzalez and Rosales, 2017).

labour and for school participation. Ali (2019) and Zapata et al. (2011) both treat the school-work decision as simultaneously determined and estimate a bivariate probit.

Despite the ongoing efforts to understand child labour, the intergenerational effect of families' child labour histories is not fully understood. I explore the implications of parents having worked as children on different kinds of child labour (i.e., work as it is defined by Ecuadorian legislation; work in market activities (outside the household or in the family's farm/enterprise) vs. work in domestic activities (chores like cleaning, shopping, laundry, cooking, care-taking, house repair); paid (where children receive a wage) vs. unpaid work (where children are nor remunerated); light (less than 14 hours per week of domestic chores for children under the age of 15 and less than 30 hours per week of work for children between 15-17 years) vs. heavy work (over 14 hours per week of chores and 30 hours per week of work); safe (working conditions are favourable) vs. unsafe work (where the children are exposed to dangerous materials or chemicals, their health has been affected negatively or if they have received abuse)). Previous work has taken into account both child labour status of the parent and for heterogeneity in working conditions though not in detail simultaneously.

While Ali (2019) accounts for differences in the type of work that the child engages in, especially distinguishing favourable from unfavourable work conditions, there is little attention paid to whether the children's parents were child labourers themselves. An indicator variable denoting whether the head of the household worked as a child is included though there is no further study of when the parent started working nor whether it was the mother or the father of the child. There are also no controls for the economic conditions that the parents might have faced as children that led to them entering the work force when young. Further, the study does not account for differences in household composition. While there is a control for single-parent households or female-led households, these households are potentially very different from dual-earner households or multi-generational households. Zapata et al. (2011) include domestic chores as part of their definition for child labour and they analyse children separately depending on work intensity (towards domestic chores or towards market work). The inclusion of domestic chores to the definition of child work points to the implicit assumption that there might be a behavioural aspect to the type of work children perform and that gendered roles may inform the probability of

children working. Nonetheless, parental work histories are not included to understand the intergenerational persistence of child labour. Emerson and Souza (2003) most closely resembles this chapter's research question as the authors examine the child labour trap accounting for parental work histories (indicator variables of whether the parents worked under the age of 14) and their educational attainment. While Emerson and Souza (2003) find that there is in fact an intergenerational link in child labour incidence, their analysis focuses on children aged 10-14 from a 1996 household survey. It does not take into account younger children, older adolescents, or heterogeneity in working conditions. Lastly, I study the school-work decision as it being simultaneously determined by the household.

4.3 Legal and Contextual Framework

Ecuador's Childhood and Adolescence Code establishes the minimum age for work at fifteen years old.²³ The minimum age requirement applies to all types of work, though exceptions are provided for this clause when the type of work is traditionally related to ancestral practices.²⁴ In concordance with the Childhood and Adolescence Code, the Labour Code further specifies restrictions for children. Children above fifteen are allowed to work as long as a work day does not exceed six hours, children only work up to five days a week (excluding weekends), their work will be organized in a way that it does not limit the child's right to an education and the work is not performed in dangerous²⁵ conditions (S.R.O. No. 167, 2012; Children and Adolescents Code,

²³In line with most of the international community, in 1988 Ecuador adopted the International Labour Organization's (ILO) Declaration on Fundamental Principles and Rights at Work. One of the declaration's core principles is the effective abolition of child labour. The Declaration calls attention to a set of eight labour standards, two of which are directly related to child labour and are drafted in ILO conventions. Convention No. 138, the Minimum Age Convention was created in 1973, recommends member countries to set the minimum age for work to be 14 years (ILO No. 138, 1973). Convention No. 182, the Convention for the Elimination of the Worst Forms of Child Labour identifies four broad types of worst forms of labour: practices related to slavery or forced work, those related to prostitution and pornography, those related to illicit activity and finally, the ones that are "likely to harm the health, safety or morals of children." (ILO No. 182, 1999, np).

²⁴While article 82 of the Childhood and Adolescence Code specifies that the minimum age requirement applies "to all forms of work, including domestic service," when calculating the hours worked in the reference week, "household chores" are not included as work by statisticians (INEC, 2012e). While it could be understood from the code that "domestic services" refer to only services performed in households different from where the child is (e.g., a cleaning service or childcare), this is not explicitly specified in the Code, and is assumed by how the National Institute of Statistics and Censuses calculates and codes child labour. In all official statistics, unpaid household work performed in the child's own home is not included as part of their 'official' work load, i.e., domestic chores at home are excluded from the official child labour calculations.

 $^{^{25}}$ Dangerous work is defined as any job that endangers a child's safety or health due to the nature of the work or the conditions in which the work is carried out (INEC, 2012e).

2002; INEC, 2012e). All work that complies with the Codes' regulations is allowed, otherwise, it is illegal and the work performed by the children who are illegally working is referred to as 'child labour.' Under this framework, Ecuador's official definition of child labour includes:

- all children ages 14 or younger who work at least one hour during the reference week in economic activities both paid and unpaid (work which is in violation of the minimum age requirement);
- all children between the ages of 15 to 17 who exceed 30 hours a week or work on weekends; whose work interferes with schooling or who work in dangerous conditions (work that exceeds or is in violation of the conditions for permissible work for adolescents).

4.4 Data

To study the work-school decision of Ecuadorian households, I use microdata from the 2012 National Child Labour Survey (ENTI, for its name in Spanish, Encuesta Nacional de Trabajo Infantil). Surveying 31,687 households and 146,814 individuals in Ecuador, the survey was conducted to understand and assess children and adolescents between the ages of 5-17 who work in Ecuador. The survey targeted all provinces except the Galapagos Islands. Of the 31,687 households surveyed, 51,223 children between 5-17 years of age completed the survey. All children in the household were surveyed so there may be more than one child interviewed per household.

For the purposes of this research, the main advantage of the National Child Labour survey is the detail with which the working experiences and histories of households are recorded. Particularly, all household members who are currently working (in the previous week) or who have a job but did not work the previous week, are asked, "at what age did you start working for the first time in your life?" From this question, parental work histories, specifically the age at which they first started working can be derived. Although advantageous, this measure is not perfect. The question is only asked to household members who worked in the previous week or who are currently working. If the person is unemployed or out of the labour force (even though they have had a job or they started work as children), the question is not asked to them.²⁶

Although the survey asks the age at which an individual starts working, there are no details on which activity individuals first engaged in, the type of work it was, or whether there were pecuniary returns or not. It could be that this question is biased in favour of paid labour and is interpreted as the first time an individual worked for pay outside their home (rather than, say, the first time they did unpaid work in economic activities such as the family's business or farm). If answers are biased towards wage labour there could potentially be a downward bias to the effects of parents working as children on their own children as unpaid forms of work would not be taken into account in parental work histories. Similarly, for the children, the survey relies on self-reports of the hours that they work and the working conditions in which they perform these activities. Although it is advantageous to have children's self-reports to acknowledge their own experience, the surveyors do not keep tally of the hours nor do they independently assess the working conditions. As such, there might be measurement error and the results are as reliable as the reports are.²⁷

Another important disadvantage is that the survey was intended to be the first cross-section of a recurring (every two years) survey. Data from 2014 was never published and the survey was discontinued shortly after. Thus, there is only a single cross-section of children, measured in 2012.²⁸

The criteria for sample selection are as follows. Children from single-parent or absent-parent households are excluded, the focus is on children who live with both parents. The dual-parent household restriction does not exclude multi-generational households.²⁹ While it restricts the sample to children who live with both parents,

 $^{^{26}}$ Since there could be household members (specifically, parents) who worked as children and are not accounted for, but there are no members who did not start to work in childhood but are considered child labourers in their work histories, estimates would likely be biased downwards and provide conservative estimates for the effect of the intergenerational transmission of child work. However, it could be that because parents worked as children and because of this, they did not invest enough in their human capital as children, resulting in them now being unemployed or out of the labour force (and thus, missing their age of first job). To test this, the sample is later restricted to children of dual earner households. In this case, even though it is a selected sample, there is information on the age of first job for both parents.

²⁷It is worth noting that for hours worked particularly, a person most knowledgeable (PMK, usually a parent) also reports the number of hours children work. Although there are differences between the children's reports and the parents' reports on the number of hours, these do not appear to be systematic and results are robust to when the PMK-reported hours are used (not presented but available upon request).

 $^{^{28}}$ Data limitations prevent tracking changes within a household, especially in terms of whether a household falls in or out of poverty, a stay-at-home mother decides to enter labour market, a divorce or death of a member occurs, all of which could affect the household's school-work decision. Results thus, cannot be interpreted as causal effects as they are associations between variables.

 $^{^{29}}$ Of the 51,233 children who completed the survey, 16,070 (31.37%) children do not live in dual-parent households

other family members, like grandparents could be part of the household. Children's whose age difference with their parents is less than 12 years are further excluded.³⁰ The initial estimating sample is 35,085 children and when the sample is further restricted to parents with complete work histories (i.e., dual-earner parents), the estimating sample becomes 24,327 children.

Dependent Variables The focus of this chapter is to understand the intergenerational persistence of child labour in Ecuador. In analysing child labour, the household's decision to send a child to school and/or to work is assumed to be made jointly. To model the school-work choice jointly, child labour and school attendance are defined as follows:

- a. *School*: a binary variable equal to one if the child states that they are currently attending school.
- b. Official Child Labour: a binary variable equal to one if the work performed by the child would constitute 'child labour' according to Ecuador's regulations, i.e., a child aged 14 or younger working at least one hour during reference week in market activities, children between 15 17 years old who exceed what is legally allowed (children whose work day is more than six hours per day, who work over five days per week, whose work limits the child's right to education and whose work is in dangerous conditions).³¹

The empirical question focuses on the impact of parents' child labour experiences on child labour understanding that the effect might depend on the type of work that the child engages in. Using the 'official definition of child labour' allows to establish a benchmark to define the work children do.³² However, the official definition of child

and are removed from the sample.

 $^{^{30}}$ Although the age difference with parents is restricted to 12 years, with all other restrictions the minimum age difference in the estimating sample is 16 years, with either parent.

³¹This is slightly different than the official definition used in government reports and there are very nuanced discrepancies. Art. 137 and Art. 150 of the Labour Code establishes further restrictions on work performed by minors, mainly prohibiting minors working at night and on weekends or holidays, respectively. While INEC calculates the 30 hour threshold using hours worked on Monday-Friday, it does not take into account work performed on the weekends nor work performed at night. Similarly, when coding the 'official child labour' measure, the National Institute of Statistics and Censuses uses responses from both the child and the PMK to denote whether the child worked or not. In the analysis, I only use the responses from the children.

 $^{^{32}}$ This is especially important when it comes to policy implications as the government uses this definition for measuring purposes as well as policies to target and eradicate this type of work.

labour does not capture the heterogeneity in the work children perform and, in some cases, does not even treat some activities as 'work' given that it only measures work that children are carrying out illegally.

To have a clearer picture of the type of activities children participate in, the 'child labour' variable is disaggregated in four different ways. The first, differentiates the work children perform in 'market' and in 'domestic' activities (Edmonds and Schady, 2012; Grootaert and Patrinos, 1999). The second, divides child labour by whether the children who engage in the labour market earn a positive wage or not (Ali, 2019; Edmonds and Schady, 2012; Grootaert and Patrinos, 1999). Third, child labour is separated by the intensity with which the children work (Ali, 2019; Zapata et al., 2011). Fourth, unsafe working conditions are distinguished, i.e., if a child works under dangerous work conditions, if the work they perform harms their health or if their working situation is abusive (Ali, 2019).

- (i.) 'Market' vs. 'Domestic Activities:'
 - Market activities refer to any work performed outside the household (whether paid or unpaid) or in a family farm or enterprise (a store or a kiosk).
 - Domestic activities refer to any work performed by the child inside the home, including: house cleaning sweeping, dusting or making the bed(s); grocery shopping in markets or supermarkets; clothing cleaning and repair
 laundering, ironing, sewing or fixing clothes; food preparation cooking breakfast, dinner or lunch; care of other household members care of other children, elderly, or those who are ill; help with homework helping other family members with schoolwork; household repair repairing any equipment for the home.
- (ii.) 'Paid' vs. 'Unpaid' Market Work:
 - *Paid work* refers to work performed by the children for which they receive a positive monetary return in the form of wages.
 - Unpaid market work is work performed by the children in a family farm or enterprise (store or kiosk) that is not remunerated. This work excludes domestic chores performed in the household.

(iii.) Work Intensity:

- Light work refers to work that is performed and legally allowed for the children depending on their age, i.e., a maximum of two hours per day (or 14 per week) of domestic chores for children under the age of 15 and a maximum of five hours per day of work (or 30 per week) for children between the ages of 15-17.
- Heavy work is work performed by children exceeding what is deemed appropriate for their age, i.e., over 14 hours a week of domestic chores for children under the age of 15 and over 30 hours a week of work for children between the ages of 15-17.
- Note: since the 'official child labour' variable addresses work that is illegal (i.e., children 14 or younger who work at least an hour in the labour market or children 15-17 years old whose work limits their education or is performed in dangerous conditions), these variables only focus on the activities that are permitted for the age group but the intensity with which these are performed, in terms of hours worked, is higher than what is permissible.
- (iv.) Unsafe Working Conditions:
 - Dangerous exposure refers to work that is performed by children where they are exposed to: dust or gases; fire, gas or flames; loud noises or vibrations; extreme heat or cold; dangerous instruments (e.g., knives); underground work; work in heights; work in water, lakes or rivers; work in darkness or excess confinement; insufficient ventilation; chemicals (e.g., pesticides, glues, etc.); explosives or other processes or conditions that are dangerous to the child's health or safety.
 - Unhealthy work is work that children perform and has caused them: superficial injuries or wounds; broken bones; dislocations or sprains; burns, scalding or frost bite; respiratory problems; eye problems; skin problems; digestive issues; fever; exhaustion or other health problems.
 - Abusive work is work performed by children where they have faced the following types of abuse: being yelled at; received insults; was hit or physically hurt; experienced sexual abuse (e.g., fondling or performing unwanted acts); other forms of abuse.

In each case, the 'official child labour' variable is replaced by a specific type of 'child labour,' a binary variable equal to one if the child's work falls under one of the criteria described above with the base case always being children who do not work in any type of labour.

Independent Variables The main variables of interest are the child labour variables for the parents. As described above, parental child labour histories are defined using the question "at what age did you start working for the first time in your life?" If a parent states that they started working before the age of fifteen they are considered to have been child labourers, as follows:

- *Mother Worked as a Child* is a binary variable equal to one if the mother of the child first started working when they were fourteen, or younger.
- Father Worked as a Child is a binary variable equal to one if the father of the child first started working when they were fourteen, or younger.³³

Given that the question is asked to individuals currently working, it is possible that a parent worked as a child but because they are not currently in the labour force, this is not taken into account. Since non-response could be correlated with child labour histories (e.g., because the parent worked as a child, they could not invest as much time in schooling and thus in their human capital acquisition and as adults have irregular work histories, are discouraged workers, they are unemployed, etc.), an indicator variable is added, denoting if there is no knowledge of parent's age of first work. This is done through the variables 'mother not in the labour force' and 'father not in the labour force,' though, a parent not being in the labour force has meaning in addition to non-response to the child labour question.³⁴

 $^{^{33}}$ As previously mentioned, this variable relies on what parents perceive or define as 'work' and the age at which they first performed these activities. Since there is no information other than age for the parents' earliest work experience, this definition of child labour is solely based on the minimum age requirement. There is no information on the number of hours they worked in their first job or the kind of activities the job entailed. To test the sensitivity of the threshold, this measure is changed to below the age of 10 and is also replaced by three different dummy variables: [Mother/Father] started work at [5-9/10-14/15-17] equal to one if the parent first worked at the ages of 5-9, 10-14 and 15-17, respectively; and a continuous variable for the age when they started working.

³⁴Although there is less information available about parental child labour (e.g., the type of work parents engaged in or the working conditions) than there is for the children, I do check for the intergenerational transmission of work using exactly the same definition for parent and child, i.e., using the age for when they started working and defining a binary variable equal to one of they started working before meeting the minimum age requirement. See Robustness Checks.

In addition to their child labour experience, the parents' educational attainment, household income or a control for whether the household received the "Bono de Desarrollo Humano," (BDH) transfer are included.³⁵ If the intergenerational persistence of child labour is solely transmitted through the household's budget constraint, then the effect of parental child labour should vanish after controlling for family wealth or permanent income (Emerson and Souza, 2003). Although not a perfect measure, the education of parents can serve as a proxy for permanent income for the family. Thus, mother's and father's highest level education attained is included as: 'primary education,' 'secondary education' or 'university' as dummy variables indicating the education of the parents with the base category being 'less than primary education.'

Control Variables A first set of controls describe the child. These include: age, gender, and ethnicity (i.e., Indigenous, Afro-descendant, Mestizo or Other (including: White, Montubio and Other, unspecified), the base category is Mestizo) of the child, whether the child is the oldest child, often associated with more contributions to work (Zapata et al., 2011).

The following variables are included to control for household characteristics. Age and employment status for father and mother are included. For household composition, a set of controls identifying who lives in the household are added denoting both age and gender of other children and other adults in the household. It is assumed that household composition is determined exogenously in the short term (Zapata et al., 2011). An indicator variable for whether the household is in an urban setting is included. Geographic controls are added for location-specific characteristics. The data is divided into administrative units, 'cantons,' a division that is below 'province' but above 'parishes,' comparable to municipalities. Including canton-level dummy variables allows to control for permanent differences across these geographical locations, e.g., availability of schools, local labour market conditions (e.g., adult and child wages, unemployment rates) that could affect the demand and supply of child labour (Zapata

³⁵As a robustness check, instead of the BDH measure, I include current family income (including income of all members except children) and household expenditures. Own income includes earnings from the individual's main and secondary occupations; capital/interest earnings from saving accounts, loans, stocks, rental income from real state, land and machinery; transfers from retirement, widowhood, orphanage, disability, or divorce pensions, or severances; income from gifts or donations (e.g., lottery winnings); income from remittances and income from 'development transfers.' Note, in the sensitivity checks, either 'household income' or 'household is BDH recipient' is included, not both simultaneously.

et al., 2011).³⁶

4.5 Empirical Strategy

This chapter seeks to address the following questions: a. what is the effect of having a parent who worked as a child on the likelihood of their children engaging in child labour or going to school? Specifically, if a parent started working before the minimum age requirement, are their children more likely to engage in child labour and less likely to go to school after controlling for poverty? b. are there differences in the intergenerational transmission of child labour depending on the type of work the child engages in? and c. are there gender differences for the strength of the correlation in the intergenerational transmission of child labour where boys and girls are affected differently depending on the type of work and the gender of the parent who worked before they were legally allowed to?

In line with the theoretical literature, the present study focuses on the supplyside determinants of child labour. To study the household's decision of children's participation in school or working, the following latent variable model is defined. Child labour, C_{ijk}^* and school participation, S_{ijk}^* are two continuous latent variables, functions of vectors of individual characteristics, X_{ij} , household characteristics, Z_j , local characteristics, P_k , and random error:

$$\begin{split} C_{ijk}^* &= \alpha_0 + \alpha_1 Mother \ Worked \ as \ Child_i + \alpha_2 Father \ Worked \ as \ Child_i + \\ &+ X_{ij}' \alpha_3 + Z_j' \alpha_4 + P_k' \alpha_5 + \varepsilon_{ijk} \\ S_{ijk}^* &= \beta_0 + \beta_1 Mother \ Worked \ as \ Child_i + \beta_2 Father \ Worked \ as \ Child_i + \\ &+ X_{ij}' \beta_3 + Z_j' \beta_4 + P_k' \beta_5 + \eta_{ijk} \end{split}$$

 C_{ijk}^* and S_{ijk}^* are not directly observed, it is only observed if the latent variables are positive. C_{ijk} , child labour, is a binary variable taking the value of one, $C_{ijk} = 1$, if child *i* from household *j* in region *k* works during that week (according to the

³⁶School quality may determine the quality of the investment in education and with the labour market characteristics, returns to school can be captured. The data has no information on schools or labour market characteristics. Looking at 2011-2012 administrative records the Ministry of Education has on number of schools (30,070), number of teachers (238,0374) per school and numbers of students enrolled (4,433,538) per school, each canton has on average 135 schools, 1072 teachers, and 19,969 students. Per each school, per canton, there are 16.37 teachers per student and 114.07 students in each school, on average (Ministerio de Educacion, 2012).

definitions above), it is equal to zero otherwise. S_{ijk} , is a binary variable denoting school participation and is equal to one, $S_{ijk} = 1$, if child *i* from household *j* in region *k* currently attends school, zero otherwise.

$$C_{ijk} = \begin{cases} 1 \text{ if } C^*_{ijk} > 0; \\ 0 \text{ otherwise.} \end{cases}$$
$$S_{ijk} = \begin{cases} 1 \text{ if } S^*_{ijk} > 0; \\ 0 \text{ otherwise.} \end{cases}$$

It is assumed that the error terms follow a bivariate normal distribution and are correlated with a correlation coefficient $\rho \neq 0.^{37}$

$$\begin{bmatrix} \varepsilon_{ijk} \\ \eta_{ijk} \end{bmatrix} \sim \mathcal{N} \left(\begin{bmatrix} 0 \\ 0 \end{bmatrix}, \begin{bmatrix} 1 & \rho \\ \rho & 1 \end{bmatrix} \right).$$

The following bivariate probit model is estimated:

$$\begin{aligned} Pr(C_{ijk} = 1) &= Pr(C_{ijk}^* > 0) \\ &= Pr(\alpha_0 + \alpha_1 Mother \ Worked \ as \ Child_i + \alpha_2 Father \ Worked \ as \ Child_i + \\ &+ \alpha_3 Mother's \ Education_i + \alpha_4 Father's \ Education_i + \\ &+ X_{ij}'\alpha_5 + Z_j'\alpha_6 + P_k'\alpha_7 + \varepsilon_{ijk} \ge 0) \\ Pr(S_{ijk} = 1) &= Pr(S_{ijk}^* > 0) \\ &= Pr(\beta_0 + \beta_1 Mother \ Worked \ as \ Child_i + \beta_2 Father \ Worked \ as \ Child_i + \\ &+ \beta_3 Mother's \ Education_i + \beta_4 Father's \ Education_i + \\ &+ X_{ij}'\beta_5 + Z_j'\beta_6 + P_k'\beta_67 + \eta_{ijk} \ge 0) \end{aligned}$$

Where, the observed variables, C_{ijk} and S_{ijk} correspond to the latent variables for child labour and school attendance, C_{ijk}^* and S_{ijk}^* , respectively. *Mother Worked As Child_i* is an indicator variable for child *i* if their mother started working at an age below the

³⁷An advantage of estimating a bivariate probit over a univariate one, is that this joint model for two binary outcomes allow outcomes to be correlated. Estimating the child labour and school decision jointly with a bivariate probit can account for unobservable characteristics that affect both decisions (Ali, 2019).

minimum age requirement; Father Worked As a Child_i is an indicator variable denoting that child i's father began working before the minimum age requirement. Mother's Education_i and Father's Education_i denote the highest level of education (indicator variables for primary education, secondary education or university education with a base category of less than primary education) child i's mother and father achieved, respectively. X'_{ij} is a vector of individual characteristics of child i in household j, including age, gender, an indicator for whether they are the oldest child and child i's ethnicity, child i's parents' ages are included as well as indicator variables for child i's father and or mother not being in the labour force. Z'_j is the vector characteristics for household j including variables denoting household j's composition, as well as an indicator variable for whether the household is in an urban setting and an indicator variable for whether household j received the BDH transfer that month;³⁸ P'_j is the vector of (location) controls, as described above, for each k canton.

The analysis is conducted initially for all children using the 'official definition of child labour.' Explicitly, for a general analysis, the bivariate probit model is estimated, where the child labour variable takes the value of one, $C_{ijk} = 1$, if the child is a child labourer according to Ecuadorian legislation.³⁹ To explore if there are differences in size or significance of estimated correlates of intergenerational transmission depending on the type of work conducted by the children, several definitions of child labour are used.⁴⁰ The types of child labour considered, i.e., $C_{ijk} = 1$, distinguish: (i). location and nature of work, (market vs. domestic activities); (ii). monetary returns, (paid vs. unpaid market work); (iii). work intensity, (light vs. heavy work); and (iv). safety of working conditions (safe vs. unsafe work).⁴¹ A different bivariate probit regression is run for each definition of child labour. In all cases, the control group for the child labour variable, $C_{ijk} = 0$, are the children who do not work at all (i.e., who have had no engagement in either activities outside their home or in their home, children who

³⁸Alternatively, this variable is substituted by household income, where the income earned by all minors is excluded. ³⁹As a reminder, this means that the binary variable is equal to one if a child between the ages of 5 and 17 worked at least one hour during reference week and it excludes children between 15 and 17 years old who are legally permitted to work (children whose work day is less than 6 hours per day, who work up to 5 days per week, whose work does not limit the child's right to education and whose work is not under dangerous conditions).

 $^{^{40}}$ The way in which child labour is defined depending on type of work and potential harm to a child follows that presented in Ali (2019).

⁴¹For instance, in the case that distinguishes market from domestic work, a bivariate probit regression is run using *'market activities'* as the definition of 'child labour,' i.e., $C_{ijk} = 1$ if child works in market activities and another bivariate probit regression is run using *'domestic activities'* as the definition of 'child labour,' i.e., $C_{ijk} = 1$ if child works in domestic activities.

have zeros in their child labour variable).⁴² Disaggregating the children into these categories of child labour is done to explore the different effects of parental child labour histories on the work-school decision. Specifically, it enables exploring for which types of work there is an intergenerational effect.

The results presented below are selected average marginal effects obtained from the different bivariate probit regressions with controls for cantons. The analysis is done initially for the full sample of children and subsequently, separately, by gender of the child, results for the latter are presented. Standard errors shown are robust and clustered at the household level. In all figures and tables results are weighted using the household weights provided by the National Institute of Statistics and Censuses (INEC from its name in Spanish).

4.6 Results

Descriptive Analysis Most children in the sample attend school, 95.5%. Using the official definition of child labour, 8.9% of Ecuadorian children aged 5-17 years currently work illegally and boys (10.8%) work more than girls (6.9%).⁴³ While the same is true in market activities, 11.3% of boys work outside their home or in their family farm or enterprise (store or kiosk) compared to 7.1% of girls. For household work, 76.2% of girls perform domestic chores compared to 69.1% of boys. Very few children (2.7% of the sample) earn a wage while 6.2% of all children perform unpaid work. Most children (46.9%) have light workloads performing less work than what is deemed appropriate for their age, while 5.7% have heavy workloads. Girls work more hours than boys when it comes to hours spent doing domestic tasks in their home and market activities. See Table 4.1.

For children who work, the average age at which their mother started working is 10.3 years, while the average age at which their fathers started working is 12 years.

 $^{^{42}}$ For example, this means that when comparing children who receive monetary returns to those who do not, two regressions are run. The first, is a bivariate regression for school and children who work and earn a wage, the base group for child labour being children who do not work. The second regression is for school and children who work and do not earn a wage, with the base group for child labour also being children who do not work. This way, the comparison group for child labour $C_{ijk} = 0$ is always the same: children who do not work at all. I avoid including other groups (e.g., children who work in an unsafe environment but do not earn a wage) in the comparison group of $C_{ijk} = 0$. Specifically, for all previous definitions, $C_{ijk} = 0$ if children's total hours in any economic activity or in any domestic activity are zero. This amounts to 7,244 children.

 $^{^{43}}$ This is very similar to what is reported by the government as net attendance rates for general basic education are 95.6% (Antamba, 2015) and given that the official report on child labour states that 8.56% of children work.

For children who do not work, the mean age at which their mother started working is 8 years, and 14 years for their father. For children who are currently working according to the official definition of child work, half of mothers and almost three quarters of fathers worked before the age of 15 (compared to one fifth and almost half, respectively, for non-working children). Children who work have less educated parents, are older, are poorer (more of them receive the BDH transfer), have lower household income, are households with more members and live in more rural settings. See Table 4.2.

Main Results Tables 4.3a. and 4.3b. present the average marginal effects⁴⁴ of selected variables from bivariate probit regressions on the official definition of child labour and school attendance for all children.⁴⁵ Tables 4.4a. and 4.4b. present the same results but for children whose parents work, i.e., there are complete child labour histories for all mothers and fathers. Columns represent the possible school-work outcomes. Columns (1) and (3) show the probability that the children work and attend school (Work=1; School=1) for male and female children, respectively. Columns (2) and (4) show the probability that children only work and do not go to school (Work=1;School=0), for boys and girls, respectively.⁴⁶

The first panel, panel (a) in Table 4.3a., presents the effects for when 'child work' is defined as the official definition of child labour and the subsequent panels (b-f)

⁴⁴Average marginal effects should be interpreted as follows. Suppose we are looking at the first variable, 'mother started working before the age of 15', and first column, that a child works and goes to school. In this case, when computing average marginal effects for the first observation, the person is treated as though their mother did not work as a child, regardless of whether that is the case or not. The probability that a child (if their mother had not worked under 15 years) is working and going to school is computed. The same is done subsequently by treating the child as though their mother did work when they were themselves children and calculating the probability again, of that child working and going to school. The difference of these two probabilities is calculated and would be the marginal effect of having a mother who was a child labourer for the first child observed. The process is repeated for all children in the sample and then the average of all marginal effects is computed, which is what is presented on the table, in this case, the average marginal effect of a mother starting to work before the age of 15 (Williams, 2012).

 $^{^{45}}$ The full set of results are presented in tables C.4.a. - C.4.b.. To put these effects in perspective, the bivariate predicted probabilities for each outcome have been calculated for two children: a child who lives in the canton Guayaquil, the largest city in the country, the economic capital, located in the coast and for a child who lives in the canton Cotopaxi a rural city centre in the highlands. Predicted probabilities in this case would be the probability of the outcome at the means of covariates for each canton. See Table C.7.

 $^{^{46}}$ In the Appendix, Tables C.1.a. and C.1.b. present all outcomes for all children and Tables C.2.a.-C.3.b. show all outcomes by gender. In this case, each column represents the possible school-work outcomes: the probability that the children work and attend school (Work=1; School=1) , the probability that children only work (Work=1;School=0) and the probability that they do not work and only attend school (Work=0; School=1). The probability that the children do not work and are not in school (Work=0; School=0) should be addressed with caution. As in Zapata et al. (2011), the theoretical framework establishes that children can go to work or school or combine both, arguably there is no motivation for children to remain idle. From the original sample of 51,233 children, 1,575 children (about 3%) are idle when work excludes hours in chores. If hours dedicated to domestic chores are taken into account, 457 children (0.89%) are idle.
present the effects for the different levels of disaggregation. For boys, if the child's mother worked when she herself was a child, i.e., the intergenerational transmission of child labour from the mother's side is associated, on average, with a 2.0 percentage points higher probability that the child goes to work and school and with 0.8 p.p. higher probability that the child only works. For girls, a mother working as a child is associated with a 1.7 p.p. higher probability of combining work and school, and has no effect on girls only working. On average, the probability of a child working and going to school, is 1.9 p.p. higher for boys and 1.7 p.p. higher for girls, if their father worked as a child than it is for those whose parents started working after the age of 15.

The effect of poverty is as predicted by theory. If household receives the BDH transfer, it is associated with a 1.9 p.p. higher probability that boys work and go to school, with a 0.8 p.p. higher probability that boys just work and with a 1.2 p.p. higher probability that the girls combine work and school. In terms of more permanent income as proxied by education (as discussed in Emerson and Souza (2003)), the more educated parents are (compared to parents who did not have an education), the lower the probabilities of a child combining school and work, doing neither, and just working (with no schooling) and the higher the probability that a child is just going to school, on average. The results for education and all other covariates for the regression on the official definition of child labour can be found in Tables C.4.a.-C.4.b. in the Appendix.

To explore if there are differences in the size or significance of the estimated correlates of intergenerational transmission depending on the type of work conducted by the children, child labour is disaggregated into different types of work. Disaggregating the kind of work a child performs into 'more' serious types of child labour versus 'less' serious types of child labour is done to understand whether the same processes (e.g., behavioural or attitudinal motivations or poverty) are at work for all kinds of child labour.

The first level of disaggregation is work outside the household or in the family's farm or enterprise (store or kiosk) and work that is deemed domestic and performed at home. As with all other levels of disaggregation, the comparison group for the working children are children who only attend school. They do not work outside nor inside their home, they perform zero hours of work per week. When looking at market activities having a mother who worked as a child is associated with a 2.5 p.p. higher probability

of combining work and school only for girls and has no statistically significant effect for other outcomes for either gender. Children have higher probabilities of combining work and market activities of 3.5 p.p. and 4.6 p.p., for boys and girls respectively, if their father worked as a child himself. Receiving the BDH transfer is associated with a 1.7 p.p. higher probability that boys only work and do not go to school, with a 2.7 p.p. higher probability that girls combine market activities with school. There does not seem to be an intergenerational effect of parental work histories for domestic chores for girls, which could reflect parents not reporting domestic work as 'child labour.' For boys, if a mother worked as a child, it is associated with a 0.9 p.p. increase in the probability male children only work. The average effects of parental work histories is that a father being a child labourer is associated with a -0.8 p.p. lower probability of just working in domestic chores for boys and a 2.5 p.p. higher probability that girls combine domestic chores with school. However, if the household is poor, boys have a 0.9 p.p. higher probability of performing chores without being in school, and a higher probability, 3.7 p.p. of combining work and school for girls. See panel (b) in Table 4.3a..

The results from the paid and unpaid work disaggregation are presented in panel (c) of Table 4.3a.. In this case, paid work is any work for which the child receives a wage while unpaid work is any economic activity performed including work in the family farm or enterprise, for which the child received no remuneration but excluding chores. The mother's work history seems to have no effect on any outcome in paid work (the extremely small sample sizes, in this case, might reduce the precision of estimates) or unpaid work for boys. It is, however, associated with a higher probability, 1.1. p.p. of combing school and paid work and a 1.9 p.p. of girls combining school and unpaid work. That a child's father was a child labourer is associated with an average 1.9 p.p. higher probability of working and being in school when work is paid for boys and with a 1.0 p.p. higher probability for girls. For unpaid work, a father working as a child is associated with higher probabilities of combining unpaid work and school, 1.9 p.p. for boys and 4.2 p.p. for girls. The poverty variable is associated with a higher probability (1.6 p.p.) of just being in paid work (and no school) for boys and high a higher probability (0.8 p.p.) that girls combine unpaid work and school.

Disaggregating light and excessive work is presented in panel (d) in Table 4.3b..

For light work, there seems to be no intergenerational transmission of child labour. For heavy loads of work, that the child's mother worked as a child herself is associated with increased probabilities of boys (1.4 p.p.) and girls (0.9 p.p.) only working, on average. If the child's father was a child labourer there is an average 3.0 p.p. higher likelihood that boys combine school and excessive work, and a 5.3 p.p. higher likelihood that girls do so. In terms of light work, being a recipient of the BDH transfer is associated with a 0.5 p.p. increase in the probability that boys only work and with a 4.6 p.p. increase in the probability that girls combine light work and school. For heavy workloads, being a recipient of the BDH transfer has no statistically significant effects with the exception of girls being 6.1 p.p. more likely to combine heavy workloads and school.

Lastly, panel (e) in Table 4.3b. presents the disaggregation between safe and unsafe work conditions. Safe working conditions are those that are free of dangerous exposures to materials or chemicals, where the health of the child is not negatively affected by injuries or health problems and where there is no verbal, physical or sexual abuse while unfavourable conditions are the opposite. When work is defined as safe, a father working as a child is associated with an increased probability of working in favourable conditions and going to school (2.6 p.p.) for girls only. Poverty is also associated with a higher probability of combining school and safe work for girls only, 3.7 p.p., on average. For unsafe working conditions, having parents who were child labourers, as opposed to not, is associated, on average with higher likelihoods (2.8) p.p. for a mother and 2.5 p.p. for a father for boys and 2.0 p.p. and 2.4 p.p. for girls, respectively) of working in and unsafe environment and going to school. Having a mother who worked as a child is associated with a 1.3 p.p. higher likelihood that boys only work and do not attend school. Poverty (as measured by the BDH transfer) has no statistically significant effect for the school-work outcomes when work is defined as dangerous, unhealthy and abusive other than a higher likelihood (2.7 p.p.) that girls combine school and work, on average.

Tables 4.4a.-4.4b. show the average marginal effects of parental work histories for children whose parents are both in the labour force, and as such, have complete work histories. As above, the first panel (a) presents the effects for when 'child work' is defined as the official definition of child labour and the subsequent panels (b-f) present the effects for the different levels of disaggregation. The marginal effects for this group of children, when using the official definition of child labour are very similar, though slightly larger, on average. When child work is disaggregated between market and domestic work, the same occurs for market activities, estimates are larger by decimals. For domestic work, a father being a child labourer is associated with a -1.0. p.p. lower probability of just working in domestic chores for boys. However, if the household is poor, boys have a 1.4 p.p. higher probability of performing chores without being in school, no effects for girls. Given that in this case, boys would be out of school and not work, or be out of school and work only performing household it is interesting to find a negative association between fathers and sons but not fathers and daughters as it seems to indicate how chores may be perceived as 'feminine' and father who worked as a child may not want their sons to leave school to be idle or to work in domestic chores.

In the case of paid work, the effect of a child labourer mother loses significance. Having a father who worked as a child is associated with a higher probability of working in paid work and going to school for both boys and girls, on average (and a higher probability of only working for girls). For unpaid work, the effects for this set of children are the same as above. A mother having been a child labourer is associated with a higher probability (2.6 p.p.), that girls combine school and unpaid work, while a father child worker is associated with higher probabilities of combining unpaid and school for both boys and girls (3.4 and 3.3 p.p., respectively). Receiving the BDH transfer is associated with a higher probability (4.4 p.p.) that girls combine school and unpaid work. In terms of heavy and light work, the results for children of dual earners are the same as above, the estimates are slightly larger for this set of children. The intergenerational effects of child labour for unsafe working conditions mirror the main results while the effects for safe work are no longer significant.

In all regressions, the Wald test of the null hypothesis of no correlation in error terms, ($\rho = 0$) is rejected. This provides evidence supporting the joint estimation of the school-work decision.

Figures 4.4 to 4.12 present results of adjusted predictions for different combinations of parental child labour backgrounds at different ages. In all cases, for all definitions of work, the probability that a child combines work and school increases with age. For the official definition of child labour 4.4, when looking at both boys and girls combining school and work, there is a marked difference between children whose parents were not child labourers compared to children whose parents were child labour. For market activities, paid and unpaid work, Fig. 4.5, Fig. 4.7, and Fig. 4.8, for boys, the father effect is almost as large as having both parents as child labourers. For domestic work, Fig. 4.6, light work, Fig 4.9, it seems that this stronger paternal effect is present for both girls and boys but slightly more marked in earlier years. For heavy workloads, the father effect is larger than that of both child labouring parents, for both boys and girls, see Fig. 4.10. The Appendix presents Fig C.2 to C.10 which include combinations with the BDH variable to include poverty in the comparison.

Robustness Checks As sensitivity checks, potential problems in carefully identifying the effect of the intergenerational transmission of child labour are addressed as follows.

Given that the 'official definition' of child labour excludes certain forms of work as it only measures work that children perform illegally, two other benchmark measures for child work are included.⁴⁷ First, I define the concept of 'work' as doing any positive amount of work in the labour market. In this case, the variable 'Work' equals one if the child works one hour or more in market activities, zero otherwise. This allows me to include a more comprehensive and standard measure of work that does not exclude 'legally allowed' work as the official child labour measure does. Second, I define the concept of work for the child in the exact same way as for the parents, depending on the age of their first job. In this case, the variable 'Work' equals one if the child started engaging in labour market activities before they turned fifteen.⁴⁸ This variable allows me to compare the same 'child labourer' definition in both the parents and the children variables. Results are quite robust to changing the 'work' variable to these benchmarks and are quite similar to those using the 'official definition' in the main specifications, see Tables C.5 and C.6.

As is the case of the results that employ the 'official definition' of child labour as a benchmark, when using the 'working one hour or more' variable in the sample

 $^{^{47}\}mathrm{Note}$ that in redefining the dependent variable for work, the school variable remains the same as in the main specification.

 $^{^{48}}$ For some children who have positive labour market hours, there is no data on the age of their first job. For these children only (i.e., children who have a positive number of hours worked in labour market activities but do not have the age of when they first starting working), I use their current age as 'age of first job' and use that to determine whether they started working before the age of fifteen.

of dual-parent households (Table C.5 panel (i.)), the intergenerational effect of child labour from the mother remains for boys in the probability of working and not going to school, beyond what can be explained by poverty, though it disappears when looking at the sample of dual-earner households (panel (ii.)). As before, if a mother was a child labourer, boys are more likely to combine school and work (2.1 p.p.) and more likely to only work and not go to school (0.8 p.p.), on average, when using this definition of work. When using 'starting to work before the age of fifteen' as the definition of work, the same results (with slightly different magnitudes) can be found. The only difference when using this definition is that for the sample of dual-parent households (Table C.6 panel (i.)), the intergenerational effect of a child labourer mother extends to girls. For these girls, having a child labourer mother is associated with a 0.2 p.p. increase in the probability that girls start working before the minimum age and do not go to school.

To ensure that the results are not driven by the poverty measure used (household receives the BDH transfer), household income (excluding that of the children) is included. As previously mentioned, the BDH receipt measure is binary measure denoting the poorest households in Ecuador.⁴⁹ While the binary of being 'poor' or not is informative, it could also be that the mechanism of intergenerational transmission of child labour that operates through the budget constraint, is not a dichotomous one, but a gradual and continuous one, for which the income variable would be informative. Neither measure is perfect in the attempt to address potential endogeneity from reverse causality and excluding income from child earnings from the household income calculation may not rid of it completely. For instance, if a child's labour (in this case, suppose it is work conducted at the home in domestic chores) allows a mother to work outside of the home for pay, which would have not been possible without the work performed by the child, then the mother would've not entered the labour market and would have not been able to contribute to household earnings that way. Nonetheless, given the explanatory usefulness that both variables have, they are kept in the analysis (Emerson and Souza, 2003; Edmonds, 2007; Ali, 2019). As alternatives, monthly average household expenditures are included (Ray, 2000; Grootaert and Patrinos, 1999; Edmonds and Schady, 2012) and ownership of assets (whether the

⁴⁹Notably, it was not only poor households who received the transfer nor were all poor people in the lowest two income quintiles recipients the transfer, but it is a good measure of households in poverty in Ecuador (see Araujo et al. (2017); Schady and Araujo (2006); Edmonds and Schady (2012)).

dwelling is: fully paid for (base); or if household rents dwelling; whether the household owns arable land; whether the household owns livestock and whether the household owns a vehicle) are used instead of household income.

Results hold when including household income, expenditures or assets instead of the BDH measure, see Tables C.8 and C.9. Using the official definition of child labour and market and domestic definitions of work, the intergenerational effects of child work are very close to the ones resulting from estimations using the BDH measure (they are slightly different in the decimals of percentage points). Household income and household expenditures seem to favour educational attainment (lower associations of children working and being in school or just working and higher associations of children just in school), on average. In terms of assets, compared to having a house that is fully paid for, other tenancy types do not appear significant. That the household owns arable land is associated with a higher probability of the children working and going to school, lower probability that they do not attend school and do not work, lower probabilities of the children just going to school for the official definition of work, market and domestic work (for the official definition only, it is also associated with a higher likelihood that the child only works), on average. Owning a vehicle is associated with lower probabilities of outcomes where the children are out of school. Owning livestock is associated with higher probabilities of combining school and work and only working and a lower probability of being just in school using the official definition of child labour; it is also associated with higher likelihoods of working and going to school and lower likelihood of just working when using the market activities definition of work.

There are two other potential sources of endogeneity, time-variant and timeinvariant sources. For the latter, the community-level time invariant characteristics are controlled for using the canton variables.⁵⁰ For household-level time invariant characteristics that would affect child labour, a control variable is included denoting whether the child worked in the previous year. If households' circumstances or

 $^{^{50}}$ An interesting result (not presented) is that in some types of work, when controlling for cantons rather than having location controls at the province-level, the BDH measure loses significance, i.e., in some cases poverty is significant for some outcomes when province, not canton controls are included. Controlling for economic conditions, school quality and community-level characteristics at a more granular level allows to better distinguish household poverty from regional poverty in the sample.

preferences related to sending children to work or not are time invariant in the shortterm (i.e., in the last year), the child's working status in the year prior would account for most observable and unobservable differences across households. See Table C.10. The results for paternal child histories and the official definition of child labour and market activities hold. The maternal effect disappears as well as the poverty effect from being a BDH recipient. Unfortunately, information on the previous year's work experience is only available for current labour workers. As such, this only captures changes in the children that are currently working, not children who were working and no longer are. It seems that perhaps the measure may be capturing sustained poverty that results in the household's need to send children to work and that we only see the effect for the households who are still in dire circumstances.

Time-variant sources of endogeneity could be omitting variables affecting the child labour decision for both the child and the parent (or the parent's income or wealth). Given that the driver of the link between parental work histories and child labour could be due to spurious correlation or a reflection problem, controls for local growth rates are included. It is possible that the correlation between child and parental child labour is largely due to both the parents and their child facing similar economic opportunities or circumstances as opposed to parental work histories influencing the child's decision to work. To separate the intergenerational link from the potential reflection problem, time, trends for labour force participation in 1960s-1980s and in 1990s-2010s are included by incorporating provincial average of labour force participation to control local employment opportunities. It is worth noting that this is an imperfect attempt in doing so. Labour force participation may not truly control for similar economic opportunities as well as local unemployment rates, GDP per person might. However, it is the only measure that is available at a provincial level at the time that parents were children themselves, 1960s-1980s. Results from these regressions are consistent with all previous findings. See Table C.11.

Although the growth variables aim to take into account the economic conditions that both the parents and the children faces, and the child's working history may reflect household circumstances in the previous year, I also am able to explicitly control for exogenous shocks to the household in the last year. The economic conditions for when the parents were children are not taken into account, but the household is directly asked whether they received a negative shock, whether their income fell and whether they had to borrow in the last 12 months. Specifically, the household is asked whether a natural disaster, epidemic, firm closures due to recessions or fall in agricultural prices negatively affected the family in the last 12 months. The household is also asked if in the last 12 months there was a decrease in income due to a job loss, the family business closing, a disease or death in the household, due to the head of the household leaving, due to a fire or robbery, due to a land conflict, due to loss of support (unspecified), due to a fall in commodity prices, due to a loss of harvest or livestock or due to another reason, unspecified. Similarly, the household is asked if a member of the family had to take out a loan or borrow money in the last 12 months. The advantage of these variables is that they are asked to all households. Although both control for time-variant shocks to the household that could affect their income and consequently the household's child labour decision, controlling for these shocks would account for most of the potential endogeneity from previously unobserved events that may send children to work that would also affect household income. Similarly, the variable denoting loans and borrowing money allows to test the intergenerational transmission of child labour while taking into account potential credit barriers that the household faced, the argument that is usually key in the decision to send children to work.

When including the negative exogenous shock measure and when controlling for whether the household took out a loan or borrowed money in the last year, results are the exact same for the official definition of child labour, market activities and domestic activities definitions of work (see Table C.12) and consistent⁵¹ in all other work definitions (unreported). All results include a control for the household experiencing a negative income shock in the last 12 months.⁵² The effects from poverty (from the BDH measure) hold and remain significant in the same cases as in the main specification. If a household experiences a negative shock the probability of work and school increases by 1.1 p.p. and 1.8 p.p. when work is defined as the child labour officially and market work, respectively; it is also associated with a decrease (-1.2 p.p.) in the probability that the child only goes to school and is not an official child

 $^{^{51}}$ In the case of paid work and excessive work the magnitudes of the intergenerational average effects are larger.

 $^{^{52}}$ Without the control, the intergenerational effect of having parents working as children are consistent for the same school-work outcomes as in the main specifications (in some cases the magnitudes are slightly larger, by 0.1 percentage points).

labourer and with a lower probability (-0.5 p.p.) that the child is idle when work is defined as market work. On average, taking out a loan or borrowing has no effect on probabilities of the different work-school combinations when work is the official definition; it is associated with an increase (2.5 p.p.) in the probability of combining school and market work and a decrease (-1.9 p.p.) in the probability of just going to school without any market activities.

Results are sensitive to the age in which parents first started to work. For instance, when the cut-off is lowered to age 10, i.e., the indicator variables for parental histories denote their parents started working when they were younger, results are consistent, though the magnitudes of the average marginal effects are higher, see Table C.13. The younger parents work, the stronger the intergenerational transmission. The more conservative estimates are presented in the main results.

4.7 Discussion and Conclusion

In general, the intergenerational transmission of child labour provides an avenue to explore the intergenerational links that occur through behavioural or attitudinal aspects since there is presumably intergenerational transmission of economic status. Being in poverty (as measured by being a recipient of the BDH transfer) is in part a measure of the budget constraint that households face and in part, a measure of intergenerational poverty for those parents who were child labourers. Overall, results suggest that there is an intergenerational transmission of child labour, even after controlling for poverty. The direction and magnitude of the effects depend on the type of work the children perform. In the case of the official definition of child labour, that parents worked as children is associated with higher probabilities in working combined with schooling and higher probabilities of working foregoing school. On average, that a mother was a child labourer is associated with a 2.0 percentage point higher probability of working and going to school and a 0.8 p.p. higher probability that boys only work. For girls, a mother child labourer is associated with a 1.7 p.p. higher probability that girls combine school and work. That a father worked as a child himself is associated on average with a 1.9 p.p. higher probability of combining

school and work for boys, a 1.7 p.p. higher probability for girls.⁵³

For domestic activities, the intergenerational channel is weaker though there seems to be a gendered aspect to it. For boys, a child labourer mother increases the probability that they only work, but a child labourer father decreases the probability that they only work in domestic chores. There is no maternal effect in combining work and domestic chores. Only for girls, if their father worked as a child himself, there is a higher probability of working and being in school. It is interesting to find that mothers' work histories have no effect on any of the work-school outcomes for girls when work is defined as domestic. Having a child labourer mother has no impact on how the school-work a girl combination for work performed at home.

On the other hand, when fathers were child labourers, their boys are less likely to just work at home (no school, just domestic work). It could be that the 'feminine' association to domestic chores is something that fathers perceive and they are averse to sending their boys to only work at home, forgoing an education; this aversion is not present for girls. Given the importance of identity and roles in what work is performed based on gender (Akerlof & Kranton, 2000; Fortin, 2005; Zapata et al., 2011), it is interesting to find the negative effect between fathers and sons and no effect from mothers to daughters. For female children, this could mean that either the behavioural or attitudinal channel of the intergenerational transmission of work is not present when it comes to chores, or it could be that performing chores is so normalized and expected of children that their parent's work histories has no significant impact on whether the children do it or not. It could also be that when the parents answered the question 'when did you first started working,' their answer reflects market rather than domestic work or when they first were hired by someone. This is supported by the findings for light work or work in safe working conditions.

In the disaggregation of paid and unpaid work, both 'work' definitions are regarded as economic activities. Even after controlling for poverty, an intergenerational effect of parents having worked as children remains through the child's father experience. It seems that for fathers, that their children develop a work ethic is important, though not at the expense of school, (unless it is to help in the family's farm or business for girls and when both parents are already working). Although a lot of the discussion on

 $^{^{53}\}mathrm{Notably},$ results are robust to using more 'flexible' definitions of 'child labour.'

child work focuses on unpaid work, usually in a farm or in the family's enterprises, boys are more likely to perform this type of work (ILO, 2017; Edmonds and Pavenik, 2002). However, when it comes to the intergenerational link, the link in the case of Ecuadorian children is present in girls, not in boys, i.e., the intergenerational transmission of child labour when work is defined as unpaid work, is marked in girls. While boys may be more prone to perform this type of work in most cases, the perpetuation of this type of work happens in girls. If the intergenerational link is interpreted as attitudinal or behavioural (which, given the evidence and how the result remains after various tests to take into account poverty, economic conditions and shocks to the family), this is important as behavioural expectations seem to be only imposed on girls while boys seem to have more freedom from these intergenerational linkages in work that is unpaid and the girls receive no financial remuneration for their work.

While there is intergenerational persistence when work is defined as unsafe (working in dangerous, unhealthy and abusive conditions) for both boys and girls, poverty has no impact on the school-work choice for boys but it does for girls. Poverty plays no effect on outcomes for boys that involve dangerous unhealthy and abusive work which points to perhaps parental altruism in poorer households towards boys, and a more expendable attitude towards girls. For boys, even though the attitudinal aspect of the intergenerational transmission is work-biased, poverty is not enough to keep children in unfavourable working environments. Notably, working conditions are only observed for children who currently work, so it could also be that in altruistic households, pull the children from work after any of these conditions materialize (e.g., if the child comes home to inform their parents they were yelled at work). However, the altruism in poverty seems to be overridden by the character-development motivation (if that is part of the intergenerational transmission) as results point to a 'I did it, you can do it' or 'that is what working is like' attitude from the intergenerational link. Children who are poor are not more likely to sustain work in detrimental working conditions, but children of parents who worked when they themselves are children are. The hypothesis that households in very dire conditions would have their children work in any type of work as long as it sustains the household is rejected for dangerous, unhealthy and abusive work.

All sensitivity checks provide evidence of robust findings. Results hold when the

household faced adverse negative shocks as well as when the households took out a loan or borrowed money. If the reason why households send children to work is to sustain negative income shocks or because they cannot access credit, then the intergenerational effects would disappear and be capture entirely by the shock, falls in income or credit access measures. This is not the case, however, all intergenerational effects remain.

Evidence from Ecuadorian children suggests that there is an intergenerational transmission of child labour beyond what can be explained by a potential budget constraint. The intergenerational link depends on the type of work the children perform and the link does not dissipate when poverty or shocks to the household are taken into consideration and it remains when controlling for parents education. Gender differences suggest that the intergenerational link is perhaps an attitudinal or behavioural one that favours gaining experiencing from working.

Given that most policies are based on the idea that poverty alleviation will help eliminate child labour, evidence from this study suggests that this may not be the case. Although beneficial in their own right, policies improving household economic conditions will only partially reduce the incidence of child labour in Ecuador for the households who do use a child's work to sustain the household. However, the attitudinal or behavioural link remains, independent of poverty status or shocks to the household. The study is not able to determine the long-term effects of the children's increased probabilities of combining school and work or decreased probabilities of being just in school for the children of child labourers. However, if these decreased investments in time devoted to school or school attainment are negative, then the negative effects will be perpetuated beyond policies addressing household poverty or parental education. If, however, parents' lived experience point to high returns from working, as is the case in Beegle et al. (2009), where adults who worked as children have higher probabilities of working for a wage and higher living standards, parents might push their children to increase their work participation combining it with school or even at its expense if the returns from work in the future are potentially larger than those from education.

Results suggest that this might be the case in Ecuador, the intergenerational transmission of child work favours working beyond what can be explained by poverty, shocks to the household and borrowing constrains (especially given that the poverty effect disappears but the intergenerational effect remains even in unfavourable working conditions). Importantly, a better understanding of parental attitudes, beliefs and behaviours is required as well as their perception of the returns to work if child work is to be eliminated. Additionally, the returns to working in childhood could be further explored in Ecuador to understand the intergenerational effect from child labour. These avenues of research are left as exciting possibilities in the research frontier.

	All Children		Female Children		Male Children	
	Mean	St. Dev.	Mean	St. Dev.	Mean	St. Dev.
	(1)	(2)	(3)	(4)	(5)	(6)
Child Attends School	0.955	0.208	0.960	0.197	0.950	0.217
Child Works, Official Definition	0.089	0.285	0.069	0.253	0.108	0.311
Child Only Attends School, No Work	0.257	0.437	0.230	0.421	0.283	0.450
Idle Child: No Work, No School	0.009	0.097	0.006	0.079	0.013	0.111
Child Works, Market Activities	0.092	0.290	0.071	0.257	0.113	0.316
Child Works, Domestic Activities	0.725	0.446	0.762	0.426	0.691	0.462
Child Works, Paid Work	0.027	0.162	0.014	0.116	0.040	0.195
Child Works, Unpaid Work for Family	0.062	0.242	0.055	0.229	0.069	0.253
Child Works, Light Work	0.469	0.499	0.473	0.499	0.464	0.499
Child Works, Heavy Work	0.057	0.231	0.068	0.251	0.046	0.210
Child Works, Unsafe Work	0.051	0.221	0.037	0.190	0.065	0.246
Child Works, Safe Work	0.041	0.199	0.034	0.180	0.048	0.215
Child's Total Hours of Household & Labour Market Work	7.466	11.298	8.294	11.792	6.676	10.747
	N=	35,085	N=	17,176	N=	17,909

 Table 4.1: Descriptive Statistics: Types of Work by Gender of Child

Figure 4.1: Proportion of Children Working According to Official Definition of Child Labour by Parental Work Histories & Poverty



Source: National Survey of Child Labour



Figure 4.2: Proportion of Children Working According to Official Definition of Child Labour by Parental Education

Source: National Survey of Child Labour



Figure 4.3: Proportion of Children Only Going to School by Parental Work Histories & Poverty

Source: National Survey of Child Labour

	All Children Mean St. Dev		Non-Working Children (Official Definition) Mean St. Dev		Working Children (Official Definition) Mean St. Dev	
	(1)	(2)	(3)	(4)	(5)	(6)
Are Methon Stanted Working	♀ 210	0.226	Q 195	0.206	10 207	6.056
Age Nother Statted Working	12.060	9.220	0.120	9.390	11.007	0.950
Methon Started Working	13.900	4.007	14.152 0.170	4.007	0.552	4.010
Father Started Working Defore 15	0.215	0.409	0.179	0.564	0.002	0.497
Father Started Working Delore 15	0.310	0.300	0.400	0.000	0.734	0.442
Household Receives DDH Transfer	0.459	0.490	0.410	0.495	0.069	0.405
(E al alian Children's Eaguines)	004.937	045.440	680.590	039.441	505.052	449.338
(Excluding Children's Earnings)	0.069	0.040	0.059	0.004	0.169	0.070
Mother's Educ.: Less Than Primary	0.063	0.242	0.053	0.224	0.163	0.370
Mother's Educ.: Primary	0.467	0.499	0.450	0.497	0.646	0.478
Mother's Educ.: Secondary	0.342	0.475	0.360	0.480	0.165	0.371
Mother's Educ.: University	0.128	0.334	0.138	0.345	0.026	0.159
Father's Educ.: Less Than Primary	0.042	0.201	0.037	0.188	0.096	0.294
Father's Educ.: Primary	0.486	0.500	0.467	0.499	0.675	0.468
Father's Educ.: Secondary	0.346	0.476	0.360	0.480	0.206	0.404
Father's Educ.: University	0.126	0.332	0.136	0.343	0.023	0.151
Mother not in Labour Force	0.469	0.499	0.495	0.500	0.202	0.401
Father not in Labour Force	0.021	0.143	0.022	0.146	0.014	0.117
Boy	0.512	0.500	0.501	0.500	0.622	0.485
Age	10.725	3.706	10.470	3.666	13.325	3.051
Oldest Child	0.348	0.476	0.348	0.476	0.349	0.477
Identifies as: Indigenous	0.081	0.272	0.062	0.241	0.270	0.444
Identifies as: Afro-descendant	0.048	0.214	0.049	0.215	0.040	0.195
Identifies as: Mestizo	0.798	0.401	0.815	0.388	0.629	0.483
Identifies as: Other Ethnicity	0.073	0.260	0.074	0.262	0.061	0.240
Has a Disability	0.018	0.132	0.018	0.133	0.014	0.119
Mother's Age	37.735	7.971	37.413	7.911	41.019	7.837
Father's Age	41.773	9.058	41.482	9.004	44.747	9.080
Household Size	5.621	1.847	5.559	1.810	6.251	2.080
House has Children Below Age 5	0.316	0.465	0.317	0.465	0.308	0.462
House has Other Children Aged 5-7	0.266	0.442	0.260	0.439	0.326	0.469
House has Other Children Aged 8-14	0.599	0.490	0.586	0.493	0.737	0.441
House has Other Children Aged 15-17	0.287	0.452	0.278	0.448	0.376	0.484
House has Other Boys Aged 5-17	0.530	0.499	0.517	0.500	0.654	0.476
House has Other Girls Aged 5-17	0.521	0.500	0.511	0.500	0.622	0.485
House Has Daughter of Head of Household Aged 18+	0.172	0.378	0.170	0.376	0.192	0.394
House Has Son of Head of Household Aged 18+	0.209	0.406	0.205	0.404	0.249	0.433
House Has Other Female Relatives Aged 18+	0.049	0.216	0.051	0.221	0.025	0.157
House Has Other Male Relatives Aged 18+	0.058	0.234	0.059	0.235	0.051	0.220
Child's Grandmother Lives in Household	0.065	0.247	0.068	0.252	0.035	0.185
Child's Grandfather Lives in Household	0.042	0.200	0.044	0.204	0.023	0.151
Urban Setting	0.590	0.492	0.623	0.485	0.254	0.435
Household Average Monthly Expenditures	382.017	351.417	392.160	359.115	278.420	236.308
Household Experienced Negative Shock in Last Year	0.176	0.381	0.172	0.377	0.226	0.418
Household's Income Fell in Last Year	0.272	0.445	0.265	0.442	0.343	0.475
Household Accessed Borrowing or Credit in Last Year	0.265	0.441	0.263	0.440	0.287	0.453
	N =	35,085	N =	30,907	N =	= 4,178

 Table 4.2: Descriptive Statistics: Means of Values by Whether Child is Child Labourer According to the Ecuadorian Legislation

	Male Children		Female Children		
	Work=1;School=1 (1)	Work=1;School=0 (2)	Work=1;School=1 (3)	Work=1;School=0 (4)	
a. Official Definition Child Labour					
Mother Started Working Before $15 = 1$	0.020**	0.008**	0.017^{***}	0.003	
Father Started Working Before $15 = 1$	0.019*** [0.006]	-0.002	0.017*** [0.005]	0.002	
Household Receives BDH Transfer = 1	$\begin{array}{c} 0.019^{***}\\ 0.007\end{array}$ n=1	[0.003] 0.008*** [0.003] 7,909	[0.005] 0.012^{**} [0.005] n=1	0.002 [0.001] 7,176	
b. Market Activities vs. Domestic	Activities				
i. Market Activities					
Mother Started Working Before $15 = 1$	0.011 [0.016]	0.011 [0.008]	0.025** [0.011]	0.002 [0.006]	
Father Started Working Before $15 = 1$	0.035^{***} [0.012]	-0.006 [0.007]	0.046*** [0.012]	0.001 [0.006]	
Household Receives BDH Transfer = 1	$0.015 \\ [0.014]$	0.017^{**} [0.007]	0.027^{**} [0.011]	-0.000 [0.005]	
ii Domostio Activities	n='	7,002	n=5	5,151	
Mother Started Working Before $15 = 1$	0.011	0.009*	0.002	-0.001	
Father Started Working Before $15 = 1$	[0.018] 0.009	[0.005] -0.008**	[0.018] 0.025**	[0.004] -0.000	
Household Receives BDH Transfer = 1	[0.014] 0.014	[0.004] 0.009**	[0.012] 0.037***	[0.004] 0.002	
	[0.015] n=1	[0.004] 7,625	[0.013] n=1	[0.004] 7,114	
c. Unpaid vs. Paid Work					
i. Paid Work					
Mother Started Working Before $15 = 1$	-0.004 [0.009]	0.007 [0.008]	0.011^{*} [0.006]	0.005 [0.004]	
Father Started Working Before $15 = 1$	0.019** [0.008]	-0.001 [0.006]	0.010** [0.005]	0.006 [0.004]	
Household Receives BDH Transfer = 1	0.018* [0.011]	0.016** [0.007]	0.008* [0.005]	0.000 [0.003]	
	n=	5,010	n=3	3,613	
Mother Started Working Before $15 = 1$	0.007	0.003	0.019*	0.002	
Father Started Working Before $15 = 1$	[0.013] 0.019*	[0.003] -0.002	[0.011] 0.042*** [0.012]	[0.004] -0.000	
Household Receives BDH Transfer = 1	[0.010] 0.016	[0.003] 0.004 [0.002]	[0.012] 0.027**	[0.003] -0.002	
	[0.012] n=0	[0.003] 3,136	[0.011] n=4	[0.004] 1,792	
Income Shock Control	Yes	Yes	Yes	Yes	

 Table 4.3a.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, All Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendent,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the 'canton' in which the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

	Male Children		Female Children		
	Work=1;School=1 (1)	Work=1;School=0 (2)	Work=1;School=1 (3)	Work=1;School=0 (4)	
d. Light Work vs. Heavy Work					
i. Light Work					
Mother Started Working Before $15 = 1$	0.016	-0.003	0.003	-0.000	
	[0.023]	[0.003]	[0.024]	[0.002]	
Father Started Working Before $15 = 1$	0.000	-0.002	0.018	0.001	
	[0.017]	[0.002]	[0.016]	[0.002]	
Household Receives BDH Transfer $= 1$	0.025	0.005^{*}	0.046**	-0.000	
	[0.018]	[0.002]	[0.018]	[0.002]	
	n=1	2,985	n=1	1,698	
ii. Heavy Work					
Mother Started Working Before $15 = 1$	-0.018	0.014**	-0.014	0.009*	
	[0.013]	[0.007]	[0.021]	[0.005]	
Father Started Working Before $15 = 1$	0.030***	-0.001	0.053***	0.004	
	[0.011]	[0.005]	[0.016]	[0.004]	
Household Receives BDH Transfer $= 1$	0.009	0.004	0.061***	-0.001	
	[0.013]	[0.005]	[0.018]	[0.003]	
	n=5	5,179	n=4	4,556	
e. Safe vs. Unsafe Working Condit	ions				
i. Safe Working Conditions					
Mother Started Working Before $15 = 1$	0.006	0.003	-0.002	-0.001	
5	[0.019]	[0.004]	[0.019]	[0.004]	
Father Started Working Before $15 = 1$	0.006	-0.005	0.026**	-0.002	
0	[0.015]	[0.003]	[0.013]	[0.003]	
Household Receives BDH Transfer $= 1$	0.020	0.005	0.037***	0.003	
	[0.016]	[0.003]	[0.014]	[0.003]	
	n=1	6.253	n=1	6.131	
<i>ii.</i> Unsafe Working Conditions		-)		-) -	
Mother Started Working Before $15 = 1$	0.028**	0.013^{*}	0.020***	0.004	
0	[0.014]	[0.007]	[0.007]	[0.004]	
Father Started Working Before $15 = 1$	0.025***	-0.005	0.024***	0.005	
	[0.009]	[0.005]	[0.006]	[0.004]	
Household Receives BDH Transfer $= 1$	-0.002	0.008	0.027***	-0.001	
	[0.010]	[0.006]	[0.008]	[0.004]	
	n=5	5,890	n=4	1,353	
Income Shock Control	Yes	Yes	Yes	Yes	
Location Controls	Yes	Yes	Yes	Yes	

 Table 4.3b.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, All Children (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the 'canton' in which the household is in.

	Male Children Work=1;School=1 Work=1;Schoo (1) (2)		Female Children) Work=1;School=1 Work=1;School=1 (3) (4)		
a. Official Definition Child Labour					
Mother Started Working Before $15 = 1$	0.022**	0.006	0.020***	0.002	
Father Started Working Before $15 = 1$	[0.011] 0.023***	[0.004] -0.003	[0.007] 0.015**	[0.002] 0.002	
Household Receives BDH Transfer = 1	[0.009] 0.021* [0.012]	[0.004] 0.012*** [0.004]	[0.007] 0.024***	[0.002] 0.005** [0.002]	
	[0.012] n=1	[0.004]	[0.008] n=1	[0.002] 1,898	
b. Market Activities vs. Domestic	Activities				
i. Market Activities					
Mother Started Working Before $15 = 1$	0.007	0.009	0.032**	-0.002	
Father Started Working Before $15 = 1$	0.047***	-0.006	$\begin{bmatrix} 0.014 \end{bmatrix}$ 0.043***	0.007	
Tather Started Working Delore 15 – 1	[0.015]	[0.010]	[0.014]	[0.006]	
Household Receives BDH Transfer $= 1$	-0.003	0.021**	0.033**	0.013**	
	[0.020]	[0.010]	[0.015]	[0.007]	
	n=	5,090	n=3	3,649	
ii. Domestic Activities					
Mother Started Working Before $15 = 1$	0.024	0.007	0.010	-0.002	
	[0.019]	[0.006]	[0.019]	[0.005]	
Father Started Working Before $15 = 1$	0.009	-0.010*	0.011	0.000	
Household Dessives PDH Transfer - 1	[0.015]	[0.005]	[0.014]	[0.004]	
Household Receives DDH Transfer = 1	-0.007	0.014	0.014	0.007	
	[0.019] n=1	[0.000]	[0.013] n=1	[0.000] 1,849	
c. Unpaid vs. Paid Work					
i. Paid Work					
Mother Started Working Before $15 = 1$	-0.012	-0.000	0.005	0.003	
	[0.009]	[0.010]	[0.006]	[0.003]	
Father Started Working Before $15 = 1$	0.019**	0.001	0.024***	0.008***	
	[0.009]	[0.008]	[0.007]	[0.003]	
Household Receives BDH Transfer = 1	0.017	0.017*	-0.007	0.003	
	[0.012]	[0.010]	[0.007]	[0.005]	
ii Unnaid Work	11—	5,594	11-2	2,040	
Mother Started Working Before $15 = 1$	0.004	0.001	0.026*	-0.002	
inother started forming before to	[0.018]	[0.005]	[0.014]	[0.006]	
Father Started Working Before $15 = 1$	0.034**	0.001	0.033**	0.003	
5	[0.014]	[0.005]	[0.013]	[0.005]	
Household Receives BDH Transfer = 1	0.010	0.007	0.044***	0.006	
	[0.019]	[0.005]	[0.016]	[0.006]	
	n=	4,399	n=3	3,351	
Income Shock Control	Yes	Yes	Yes	Yes	
Location Controls	Voc	Vos	Vor	Vor	

 Table 4.4a.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Children of Dual Earner Parents

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

	Male Children		Female Children		
	Work=1;School=1 (1)	Work=1;School=0 (2)	Work=1;School=1 (3)	Work=1;School=0 (4)	
d. Light Work vs. Heavy Work					
i. Light Work					
Mother Started Working Before $15 = 1$	0.038	-0.005	0.014	-0.002	
	[0.024]	[0.004]	[0.026]	[0.003]	
Father Started Working Before $15 = 1$	-0.005	-0.001	-0.000	0.000	
Household Passives PDH Transfer - 1	[0.018]	[0.003]	[0.019]	[0.002]	
Household Receives DDH Hansler = 1	[0.024]	[0.003	[0.026]	[0.002	
	[0.024] n=8	8.778	[0.020] n=7	[0.000]	
ii. Heavy Work		.,		,	
Mother Started Working Before $15 = 1$	-0.015	0.018**	-0.021	0.007	
	[0.015]	[0.009]	[0.022]	[0.005]	
Father Started Working Before $15 = 1$	0.032^{**}	0.002	0.068^{***}	0.009^{**}	
	[0.013]	[0.007]	[0.020]	[0.004]	
Household Receives BDH Transfer $= 1$	0.004	0.006	0.035	-0.001	
	[0.017]	[0.009]	[0.023]	[0.006]	
	n=a	3,474	n=a	3,022	
e. Safe vs. Unsafe Working Condit	ions				
i. Safe Working Conditions					
Mother Started Working Before $15 = 1$	0.022	0.001	0.009	-0.002	
0	[0.020]	[0.005]	[0.019]	[0.005]	
Father Started Working Before $15 = 1$	0.004	-0.005	0.012	-0.000	
	[0.016]	[0.004]	[0.015]	[0.004]	
Household Receives BDH Transfer $= 1$	0.005	0.005	0.017	0.003	
	[0.020]	[0.005]	[0.020]	[0.006]	
	n=1	1,049	n=1	1,018	
ii. Unsafe Working Conditions	0.096**	0.010	0.000***	0.000	
Mother Started working Before $15 = 1$	0.030	0.012	0.028	0.002	
Father Started Working Defers 15 - 1	[0.010]	[0.009]	[0.009]	[0.005]	
Father Started working before $13 = 1$	[0.011]	-0.003	[0.021	[0.004]	
Household Beceives BDH Transfer — 1	-0.016	0.008	0.019*	0 004	
	[0.016]	[0.009]	[0.011]	[0.005]	
	n=4	4,143	n=2	2,970	
Income Shock Control	Voc	Voc	Voc	Voc	
Location Controls	Yes	Yes	Yes	Yes	

 Table 4.4b.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Children of Dual Earner Parents (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the 'canton' in which the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.



Figure 4.4: Average Marginal Effects at Different Ages for All Children Child Labour: Official Definition



Figure 4.5: Average Marginal Effects at Different Ages for All Children Child Labour: Work in Market Activities



Figure 4.6: Average Marginal Effects at Different Ages for All Children Child Labour: Work in Domestic Activities



Figure 4.7: Average Marginal Effects at Different Ages for All Children Child Labour: Paid Work



Figure 4.8: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work



Figure 4.9: Average Marginal Effects at Different Ages for All Children Child Labour: Light Work







Figure 4.11: Average Marginal Effects at Different Ages for All Children Child Labour: Safe Work



Figure 4.12: Average Marginal Effects at Different Ages for All Children Child Labour: Unsafe Work

Chapter 5

Conclusion

The purpose of this dissertation is to understand intimate partner violence children's well-being and child labour in Ecuador. At present, there has not been a unified, multidimensional approach to intimate partner violence or child labour that takes into account gender norms and behavioural or attitudinal motivations or a focus on children's well-being in the context of household work from the children's perspective. Using rich datasets from Ecuador, I explore the relationship between intimate partner violence and women's ownership of economic resources, their beliefs and attitudes as well as their own and their (former) spouses' childhood experiences of violence (Chapter 2). I also examine children's well-being, particularly, the relationship between the household work and children's self-assessed happiness (Chapter 3) and the intergenerational transmission of different types of child labour (Chapter 4).

In Chapter 2, I explore the impact of economic resources, beliefs and attitudes towards wife-beating and patriarchal/traditional norms and childhood violence on the probability of coupled and post-coupled women experiencing physical, psychological, sexual and economic violence. I find that ownership of assets is somewhat protective of intimate partner violence. While the most liquid asset, access to money for personal expenses, is highly protective of violence for all women, ownership of houses, crops, vehicles and savings offer mixed results. There is evidence of 'male backlash,' when women own certain assets it is associated with higher incidences of violence. Importantly, holding traditional beliefs encouraging obedience to spouses is quite protective of violence. There is also a pervasive intergenerational component to violence, the largest associations with intimate partner violence in adulthood pertain to childhood experiences of violence both in the women's home of origin and her spouse's home of origin. I address the limitations of the study with regards to endogeneity by looking at widowed women and find that for the widowed, poverty seems to have strong associations with violence. Similarly, I find that results are robust to taking into account the length of the relationships. Given the nature of the outcomes, it is likely that women may under-report their experiences of intimate partner violence.

In Chapter 3, I examine the relationship between household work performed by children aged 8-17 and their self-assessed well-being. I find that, with a mean level of happiness of 86% of children reporting they are happy most of the time, an increase in 10 hours of weekly chores is associated with around a 2 percentage point decrease in children's self-assessed happiness. Though it may seem like a small decrease in magnitude given the mean level of happiness, the decrease is similar to that associated with an increase in 10 hours of market work (-2.9 p.p.). The negative effect of domestic chores on children's well-being is present in both the intensive and extensive margin of performing household work. The effects depend on the type of chore performed and are not gender neutral. Activities like shopping are associated with higher probabilities of well-being while care taking has negative associations. When looking at the share of household work performed by children, both compared to the all household members and compared to their siblings, I find that the effect of a larger brunt of the work performed is more important for boys than it is for girls. Lastly, when thinking of how household chores are excluded from definitions of child labour, results suggest that given the negative impact of chores on children's self-assessed well-being (especially for younger children and given that the negative effects are present even when conduction less than seven hours per week), this exclusion should be reconsidered.

In Chapter 4, I investigate the intergenerational transmission of child labour. I aim to separate the intergenerational links that occur due to behavioural or attitudinal channels, beyond what can be transmitted through poverty. I find that, overall, having a parent who worked as a child (before meeting the minimum age requirement) is associated with higher likelihoods that their children combine school and work and that their children only work and do not go to school, when child labour is defined according to the Ecuadorian legislation. Results suggest that the intergenerational transmission of child labour depends on the type of work that children do, on which parent was a child labourer and the gender of the child. Further, findings suggest that poverty elimination will not be enough to eradicate child labour as there seems to be a persistent attitudinal/behavioural link in the transmission of child work, even when poverty and economic shocks to the household are controlled for.

References

- Aguero, J. (2013). Causal Estimates of the Intangible Costs of Violence Against Women in Latin America and the Caribbean. IBD Working Paper Series, 414.
- Aizer, A. (2010). The Gender Wage Gap and Domestic Violence. American Economic Review, 100, 1847-1859.
- Akerlof, G., & Kranton, R. (2000). Economics and identity. The Quarterly Journal of Economics, 105 (3), 715-753.
- Akerlof, G., & Kranton, R. (2010). Identity economics : How our identities shape our work, wages, and well-being. Princeton: Princeton University Press.
- Ali, F. (2019). In the Same Boat, but not Equals: The Heterogeneous Effects of Parental Income on Child Labour. The Journal of Development Studies, 55(5), 845-858.
- Alpizar, F., & Gsottbauer, E. (2015). Reputation and household recycling practices: Field experiments in Costa Rica. *Ecological Economics*, 120, 366-375.
- Almond, D., & Currie, J. (2010). Human Capital Development Before Age Five. NBER Working Paper Series, 15827.
- Amato, P. (1996). Explaining the Intergenerational Transmission of Divorce. Journal of Marriage and Family, 58(3), 628-640.
- Amaral, S. (2017) Do Improved Property Rights Decrease Violence Against Women in India? (No. 2017-13). ISER Working Paper Series. http://dx.doi.org/10.2139/ ssrn.2504579.
- Amin, S., Quayes, S., & Rives, J. (2004). Poverty and Other Determinants of Child Labor in Bangladesh. Southern Economic Journal, 70(4), 876-892.
- Amuedo-Dorantes, C., Bonke, J., & Grossbard, S. (2010). Income Pooling and Household Division of Labor: Evidence from Danish Couples. *IZA Discussion Paper Series* No. 5418.

- Andresen, S., & Fegter, S. (2011). Children Growing Up in Poverty and Their Ideas on What Constitutes a Good Life: Childhood Studies in Germany. *Child Indicators Research*, 4(1), 1-19. doi:10.1007/s12187-010-9073-3
- Anderson, S. (2007). The Economics of Dowry and Brideprice. Journal of Economic Perspectives, 21(4), 151-174.
- Angelucci, M. (2008). Love on the Rocks: Domestic Violence and Alcohol Abuse in Rural Mexico, The B.E. Journal of Economic Analysis & Policy, 8(1), 1-43.
- Antamba Chacua, L. (2015). Estadistica Educativa Reporte de Indicadores. Ministerio de Educacion Ecuador.
- Araujo, C., Bosch Mossi, M., & Schady, N. (2017). Can Cash Transfers Help Households Escape an Inter-Generational Poverty Trap? IDB Working Paper Series No. 767
- Atkinson, M. P., Greenstein, T. N., & Lang, M. M. (2005). For women, breadwinning can be dangerous: Gendered resource theory and wife abuse. *Journal of Marriage* and Family, 67(5), 1137-1148.
- Aumentan denuncias de hombres maltratados (2011, February 10). La Hora. Retrieved from: https://lahora.com.ec/noticia/1101092867/aumentan-denuncias-de-hombres-maltratados.
- Bauer, T., & Sinning, M. (2010). Blinder-Oaxaca decomposition for Tobit models. Applied Economics, 42(12), 1569-1575.
- Basu, K. (1999). Child Labor: Cause, Consequence, and Cure, with Remarks on International Labor Standards. *Journal Of Economic Literature*, 37(3), 1083.
- Basu, K. & Van P. H. (1998). The Economics of Child Labour. The American Economic Review, 88(3), 412-427.
- Becker, G. (1962). Investment in Human Capital: A Theoretical Analysis. The Journal of Political Economy, 70(5, Part 2), 9-49.
- Becker, G. (1965). The Theory of the Allocation of Time. *The Economic Journal*, 75 (299), 493-517.
- Becker, G. (1976). Altruism, Egoism, and Genetic Fitness: Economics and Sociobiology. Journal of Economic Literature, 14(3), 817-826.
- Becker, G. S. (1981). A treatise on the family. Cambridge, Mass: Harvard University Press.
- Beegle, K., Dehejia, R., & Gatti, R. (2004). The Education, Labour Market and Health Consequences of Child Labour. IDEAS Working Paper Series from RePEc, IDEAS Working Paper Series from RePEc, 2004.

- Beegle, K., Dehejia, R., & Gatti, R. (2009). Why Should We Care about Child Labor? The Education, Labor Market, and Health Consequences of Child Labor. *The Journal of Human Resources*, 44(4), 871-889.
- Berenger V. & Verdier, A. (2016). Child Labour and Schooling in South Sudan and Sudan: Is There a Gender Preference? African Development Review, 28(2), 177-190.
- Bertrand, M., Kamenica, E., & Pan, J. (2015). Gender Identity and Relative Income within Households. *The Quarterly Journal of Economics*, 130(2), 571-614.
- Bhalotra, S., & Heady, C. (2003). Child Farm Labor: The Wealth Paradox. The World Bank Economic Review, 17(2), 197-227.
- Black, S., Devereux, P., & Salvanes, K. (2005). Why the Apple Doesn't Fall Far: Understanding Intergenerational Transmission of Human Capital. American Economic Review, 95(1): 437-449.
- Blair, S.L. (1992) (a). Children's Participation in Household Labour: Child Socialization Versus the Need for Household Labor. Journal of Youth and Adolescence, 21(2), 241-258.
- Blair, S.L. (1992) (b). The Sex-Typing of Children's Household Labor: Parental Influence on Daughters' and Sons' Housework. Youth & Society, 24(2), 178-203.
- Blau, F., & Winkler, A. E. (2014). The Economics of Women, Men, and Work (Seventh ed.). New York, NY: Oxford University Press. Chapter #3.
- Block, F., & Rao, V. (2002). Terror as a Bargaining Instrument: A Case Study of Dowry Violence in Rural India. The American Economic Review, 92(4), 1029-2049.
- Bobonis, J., Gonzalez-Brenes, M., & Castro, R. (2013). Public Transfers and Domestic Violence: The Roles of Private Information and Spousal Control. American Economic Journal: Economic Policy, 5(1), 179-205.
- Bonke, J. (2010). Children's Housework Are Girls More Active than Boys? International Journal of Time Use Research, 17(1), 1-16.
- Bourdillon M. (2013) In: Ben-Aryeh, A., Frones, I., Casas, F., & Korbin, J.E (eds) Handbook of child well-being : Theories, methods and policies in global perspective. Springer, Dordrecht.
- Bowlus A., & Seitz, S. (2006). Domestic Violence, Employment and Divorce. International Economic Review, 47(4), 1113-1149.
- Brown D., Deardorff A., and Stern, R. (2002). The Determinants of Child Labor: Theory and Evidence. *Research Seminar in International Economics* (Discussion Paper No. 486). Ann Arbor, Michigan: School of Public Policy.
- Bruckauf, Z. &, Rees, G. (2017). Children's Participation in Housework: Is there a case of gender stereotyping? *Innocenti Research Brief* (17), 1-3.
- Burton, P., Lethebridge L., & Phipps, S. (2008). Children with disabilities and chronic conditions and the longer-term parental health. *The Journal of Socio-Economics*, 37, 1168-1186.
- Burton P., & Phipps S. (2010) In Children's Voices. In: Kamerman S.B., Phipps S., Ben-Arieh A. (eds) From Child Welfare to Child Well-Being. Children's Well-Being: Indicators and Research, vol 1. Dordrecht: Springer.
- Burton, P., & Phipps, S. (2011). Families, Time, and Well-Being in Canada. Canadian Public Policy, 37(3), 395-423.
- Burton P., & Phipps, S. (2017). The Economic Well-Being of Canadian Children. Canadian Public Policy, 43,(4), 299-330.
- Busetta, G., Campolo, M.G., Di Pino, A. (2019). Children's Use of Time and Well-Being in Italy. *Child Indicators Research*, 12(3), 821-845.
- Byson, A., & MacKerron G. (2015). Are You Happy While You Work? The Economic Journal, (127)(599), 106-125.
- Camacho, G. (2014). La Violencia de Genero Contra Las Mujeres en el Ecuador. Quito, Ecuador.
- Carrion Mena, F. (2003). "Violencia intra-familiar" in *Ecuador, Seguridad Ciudadana* y Violencia. Quito, Ecuador.
- Carrion Mena, F. (2007). "Los costos de la violencia en Ecuador" in Ciudad Segura, Flacso: Quito, Ecuador. Retrieved from: http://works.bepress.com/fernando_ carrion/107/
- Case, A., Lubotsky, D., & Paxson, C. (2002). Economic Status and Health in Childhood: The Origins of the Gradient. American Economic Review, 92(5): 1308-1334.
- Casas, F. (2011). Subjective Social Indicators and Child and Adolescent Well-being. Child Indicators Research, 4(4), 555-575.
- Censo de Poblacion y Vivienda (CPV), 2010. Instituto Nacional de Estadisticas y Censos (INEC). 2010. Quito, Ecuador: INEC.
- Cerrato, J., & Cifre, E. (2018) Gender Inequality in Household Chores and Work-Family Conflict. Frontiers in Psychology, 9, 1330.
- Clark, A., Frijters, P., & Shields, M. (2008). Relative Income, Happiness, and Utility: An Explanation for the Easterlin Paradox and Other Puzzles. *Journal of Economic Literature*, 46(1), 95-144.

- Cockburn, J. (2001), "Child labour versus education: Poverty constraints or income opportunities?" Working Paper.
- Codigo Civil (Civil Code), (2005). Quito: National Congress.
- Codigo de la Ninez y Adolescencia (Children and Adolescents Code), (2002). Quito: National Congress.
- Codigo Organico Integral Penal (Penal Code) (COIP), (2014). Quito: Ministerio de Justicia, Derechos Humanos y Cultos.
- Coleman, D. H., & Straus, M. A. (1986). Marital power, conflict, and violence in a nationally representative sample of American couples. Violence and Victims, 1(2), 141-157.
- Como es el proceso de divorcio en el Ecuador (2018, March 6). *Metro Ecuador*. Retrieved from: https://www.metroecuador.com.ec/ec/noticias/2018/03/06/ proceso-divorcio-ecuador.html
- Como se hace el calculo de las pensiones alimenticias (2017, January 31). *El Telegrafo*. Retrieved from: https://www.eltelegrafo.com.ec/noticias/sociedad/ 6/que-debo-conocer-sobre-las-pensiones-alimenticias
- Como y cuando se otorga pension de alimentacion (2016, Octubre 10). *Metro Ecuador*. Retrieved from: https://www.metroecuador.com.ec/ec/noticias/2016/10/10/ se-otorga-pension-alimentacion.html.
- Cools S., & Kotsadam A. (2017). Resources and Intimate Partner Violence in Sub-Saharan Africa. World Development, 95, 211-230.
- Corak, M. (2006) Do poor children become poor adults? Lessons from a cross country comparison of generational earnings mobility, IZA Discussion Papers, No. 1993.
- Corak, M. (2013). Income Inequality, Equality of Opportunity, and Intergenerational Mobility. *Journal of Economic Perspectives*, 27 (3): 79-102.
- Corak, M., & Piraino, P. (2011). The Intergenerational Transmission of Employers. Journal of Labor Economics 29(1): 37-68.
- Corte Constitucional aprueba matrimonio civil igualitario en Ecuador (2019, June 12). *El Comercio*. Retrieved from: https://www.elcomercio.com/actualidad/corte-constitucional-aprobacion-matrimonio-igualitario.html
- Cuentas Satelite del Trabajo No Remunerado de los Hogares (CSTNRH) 2007-2015. Boletin tecnico No. 01-2019. Instituto Nacional de Estadisticas y Censos (INEC).
- Cunningham, M. (2001). The Influence of Parental Attitudes and Behaviors on Children's Attitudes toward Gender and Household Labor in Early Adulthood. Journal of Marriage and Family, 63(1), 111-122.

- Currie, J. (2009). Healthy, Wealthy and Wise: Socioeconomic Status, Poor Health in Childhood and Human Capital Development. *Journal of Economic Literature*, 47(1), 87-122.
- Currie, J., & Moretti, E. (2003). Mother's education and the intergenerational transmission of human capital: evidence from college openings. *Quarterly Journal of Economics*, 118(4), 1495-1532.
- Currie, J., & Moretti. E. (2007). Biology as Destiny? Short- and Long-Run Determinants of Intergenerational Transmission of Birth Weight. *Journal of Labor Economics*, 25(2), 231-264
- Currie, J., Mueller-Smith, M., & Rossin-Slater, M. (2018). Violence while in Utero: The Impact of Assaults During Pregnancy on Birth Outcomes. *NBER Working Paper Series*, 24802.
- D'Addio, A. (2007) Intergenerational transmission of disadvantage mobility or immobility across generations? A review of evidence for OECD countries. *OECD Social, Employment and Migration Working Paper* 52.
- D'Agostino, A., Giusti, C. & Potsi, A. (2018). Gender and Children's Wellbeing: Four Mediterranean Countries in Perspective. *Child Indicators Research*, 11(5), 1649-1676.
- Dammert, A., De Hoop, J., Mvukiyehe, E., & Rosati, F. (2018). Effects of public policy on child labor: Current knowledge, gaps, and implications for program design. *World Development*, 110, 104-123.
- Deere C., Twyman J., & Contreras, J. (2014). Gender, marital status, and asset accumulation in Ecuador: a look at patrimonial violence *Eutopia*, 5, 93-119.
- Dessy, S. (2000). A defense of compulsive measures against child labour. Journal of Development Economics, 62(1), 261-275.
- Dickie, K., Holder, M., & Binfet, J. (2017). Happiness in Children. *Childhood Studies* - Oxford Bibliographies.
- Diener, E., Oishi, S., & Lucas, R. (2015). National Accounts of Subjective Well-Being. American Psychologist, 70(3), 234-242.
- Dinku, Y., Fielding, D., & Genc, M. (2019). Counting the uncounted: The consequences of children's domestic chores for health and education in Ethiopia. *Review of Development Economics*, 23(3), 1260-1281.
- Ecuador, segundo America Latina embarazo adolescente pais en en 24). ElUniverso. Retrieved http://www. (2012,February from: eluniverso.com/2012/02/24/1/1384/mas-17-jovenes-ecuador-son-madres-\ solo-venezuela-lo-supera-america\-latina.html.

- Ecuador 3160 de maltrato hombres registra casos а en 10 6).ElTelegrafo. (2014.)December Retrieved meses from: https://www.eltelegrafo.com.ec/noticias/justicia/1/ ecuador-registra-3-160-casos-de-maltrato-a-hombres-en-10-meses.
- Edmonds, E. (2007). Child Labor. IDEAS Working Paper Series from RePEc.
- Edmonds, E. (2008). Defining child labour: A review of the definitions of child labour in policy research (Working Paper). Retrieved from: http://www.ilo.org/ipecinfo/product/viewProduct.do?productId=11247
- Edmonds, E. & Pavcnik, N. (2002), Does Globalization Increase Child Labor? Evidence from Vietnam, Working Paper, Dartmouth University.
- Edmonds, E., & Schady, N. (2012). Poverty Alleviation and Child Labour. American Economic Journal: Economic Policy, 4(4): 100-24.
- El regano y el golpe aun se usan para la crianza de hijos en Ecuador (2018, Mayo 23). *El Comercio*. Retrieved from: https://www.elcomercio.com/actualidad/crianza-ninos-violencia-padresdefamilia-ecuador.html.
- Ellsberg, M., & Heise, L. (2005). Researching Violence Against Women: A Practical Guide for Researchers and Activists. Washington DC, United States: World Health Organization, PATH.
- Adolescente Embarazo У Fecundidad (2012,March 01).Ministerio Salud (MSP).dePublica Retrieved from: http://instituciones. msp.gob.ec/misalud/index.php?option=com content&view=article& id=668:embarazo-adolescente-y-fecundidad-en\-cifras&catid=51: mi-salud-al-dia&Itemid=242.
- Emerson, P., & Knabb, S. (2006). Opportunity, Inequality and the Intergenerational Transmission of Child Labour. *Economica*, 73(291), 413-434.
- Emerson, P., & Souza, A. (2003). Is There a Child Labor Trap? Intergenerational Persistence of Child Labor in Brazil. *Economic Development and Cultural Change*, 51(2), 375-398.
- Encuesta Nacional de Empleo, Desempleo y Subempleo (ENEMDU), 2011. Instituto Nacional de Estadisticas y Censos (INEC). 2011. Quito, Ecuador: INEC.
- Encuesta Nacional de Empleo, Desempleo y Subempleo Modulo Ambiental. (EN-EMDU), 2012. Instituto Nacional de Estadisticas y Censos (INEC). 2012. Quito, Ecuador: INEC.
- Encuesta Nacional de Empleo, Desempleo y Subempleo (ENEMDU), 2015. Instituto Nacional de Estadisticas y Censos (INEC). 2015. Quito, Ecuador: INEC.

- Encuesta Nacional de Ingresos y Gastos en Hogares Urbanos y Rurales (ENIGHUR), 2011- 2012. Instituto Nacional de Estadisticas y Censos (INEC). 2012. Quito, Ecuador: INEC.
- Encuesta Nacional de Relaciones Familiares y Violencia de Genero Contra las Mujeres (ENVIGMU), November, 2011. Instituto Nacional de Estadisticas y Censos (INEC). 2011. Quito, Ecuador: INEC.
- Ernst, M. (2002). "Ser mujer, un factor de riesgo" in Carrion Mena, F. ed. Segurindad ciudadana, espejismo o realidad?. Quito, Ecuador. Retrieved from: http://works. bepress.com/fernando_carrion/107/
- Fallon , P. & Tzannatos, Z. (1998). "Child Labor: Issues and Directions for the World Bank," World Bank.
- Farmer, A. & Tiefenthaler, J. (1997). An economic analysis of domestic violence. *Review of Social Economy*, 55(3), 337-358.
- Fattore, T., Mason, J., Watson, E., & SpringerLink. (2017). Children's Understandings of Well-being Towards a Child Standpoint (Children's Well-Being: Indicators and Research, 14). Dordrecht: Springer Netherlands : Imprint: Springer.
- Fernandez, J., Quiroga, M.A., Escorial, S., & Privado, J. (2016). The Gendered Division of Housework. *Psicothema*, 28(2), 130-136.
- Ferreira, C., Garcia, K, Macias, L, Perez, A., Tomsich, C. (2013). Mujeres y hombres del ecuador en cifras III. Editorial Ecuador.
- Fiscalia General del Estado (FGE). (2019). Rendicion de Cuentas 2019.
- Fleming, S. (2018, January). Women are "Bossy" and Men are "Decisive" What Gender Stereotypes Really Mean in the Workplace and How to Overcome Them. Cornell University. Ithaca.
- French, J.L. (2010). Children's Labor Market Involvement, Household Work and Welfare: A Brazilian Case Study. *Journal of Business Ethics*, 92(1), 63-78.
- Friedemann-Sanchez, G., & Svec, J. (2012). Intimate Partner Violence in Colombia: Who Is at Risk? Social Forces, 91(2), 663-688.
- Friedemann-Sanchez, G., & Svec, J. Intimate Partner Violence in Central Colombia: Exploring the Relationship Between Bargaining Factors. Working Paper.
- Fors, H. C. (2012). Child labour: A review of recent theory and evidence with policy implications. *Journal of Economic Surveys*, 26(4), 570-593.
- Fortin, N. (2005) Gender role attitudes and the labour-market outcomes of women across OECD countries. Oxford Review of Economic Policy, 21(3), 416-438.

- Fortin, N., Oreopoulos, P., & Phipps, S. (2015). Leaving Boys Behind: Gender Disparities in High Academic Achievement. Journal of Human Resources, 50(3), 549-579.
- Garcia-Moreno, C., Jansen, H., Ellsberg, M., Heise, L., & Watts, C. (2005). WHO Multi-country Study on Women's Health and Domestic Violence against Women. World Health Organization.
- Gelbach, J.B. (2016). When do covariates matter? And which ones, and how much? *Journal of Labour Economics*, 34(2), 509-543.
- Gershuny, J. & Sullivan, O. (2014). Household structure and housework: assessing the contributions of all household members, with a focus on children and youths. *Review of Economics of the Household*, 12(1), 7-27.
- Gimenez-Nadal, J., Molina J.A., & Ortega R. (2017). Like my parents at home? Gender differences in children's housework in Germany and Spain. *Empirical Economics*, 52(4), 1143-1179.
- Gimenez-Nadal, J.I. & Sevilla-Sanz, A. (2011). The Time Crunch Paradox. Social Indicators Research, 102(2), 181-196.
- Goldscheider, F., & Waite, L. J. (1991). New families, no families?: The transformation of the American home (Studies in demography (Berkeley, Calif.); 6). Berkeley: University of California Press.
- Gonzalez, F. & Rosales, I. (2017, Aug.) The Case Against Child Labour Bans. Working Paper.
- Government of Canada, Department of Justice. (14 Feb. 2018). "Research and Statistics Division - Just Facts." Intimate Partner Violence, www.justice.gc.ca/eng/rp-pr/ jr/jf-pf/2017/may01.html..
- Grawe, N., & Mulligan, C. (2002). Economic Interpretations of Intergenerational Correlations. Journal of Economic Perspectives, 16(3), 45-58.
- Grimsrud, B. (2003). Millennium development goals and child labour (English). Understanding Children's Work Project working paper series. Washington, DC: World Bank. http://documents.worldbank.org/curated/en/262041468141580732/Millennium-development-goals-and-child-labour
- Grootaert, C., & Patrinos, Harry Anthony. (1999). The Policy analysis of child labor : A comparative study. New York: St. Martin's Press.
- Grossbard, S. A. (2011). Independent Individual Decision-Makers in Household Models and the New Home Economics. In Molina, J.A. (Ed.), *Household Economic Behaviors*. New York, NY: Springer.

- Haddad, L., J. Hoddinott & H. Alderman, eds., (1997), Intrahousehold Resource Allocation in Developing Countries: Models, Methods and Policy, Baltimore: Johns Hopkins University Press.
- Hajdu G., & Hajdu, T. (2018). Intra-Couple Income Distribution and Subjective Well-Being: The Moderating Effect of Gender Norms. *European Sociological Review*, 34(2), 138-156.
- Heise, L., Pitanguy, J., & Germain, A1994). Violence against women : the hidden health burden. World Bank Discussion Papers no. WDP 255. Washington, D.C.: The World Bank. http://documents.worldbank.org/curated/en/ 489381468740165817/Violence-against-women-the-hidden-health-burden.
- Heise, L., Ellsberg, M.,& Gottemoeller, M.,. (1999). Ending Violence Against Women. Population Reports, 27(4), 1-45.
- Hidrobo, M. & Fernald, L. (2013). Cash Transfers and Domestic Violence. Journal of Health Economics, 32(1), 304-319.
- Hien, V.M2012). Investigating the Wealth Paradox on Child Labor: A Case Study of rural areas in Vietnam. Retrieved from: www.ipd.gu.se/digitalAssets/1373/ 1373706_hien.pdf
- Holder, M. D., Coleman, B., & Sehn, Z. L. (2009). The Contribution of Active and Passive Leisure to Children's Well-Being. *Journal of Health Psychology*, 14(3), 378-386.
- Hoop, J., & Rosati, F. (2012). Does Promoting School Attendance Reduce Child Labour? Evidence from Burkina Faso's BRIGHT Project IZA Discussion Paper No. 6601, 1-43.
- Inglehart, R., C. Haerpfer, A. Moreno, C. Welzel, K. Kizilova, J. Diez-Medrano, M. Lagos, P. Norris, E. Ponarin & B. Puranen et al. (eds.). (2014). World Values Survey (WVS): Round Six Ecuador Datafile Version. Madrid: JD Systems Institute.
- International Labour Organization (ILO), (1973). Convention Concerning Minimum Age for Admission to Employment (No.138), Geneva, International Labour Organization.
- International Labour Organization (ILO), (1998). "Declaration on Fundamental Principles and Rights at Work." Effective Abolition of Child Labour, n.p.
- International Labour Organization (ILO), (1999). Convention Concerning Prohibition and Immediate Action for the Elimination of the Worst Forms of Child Labour (No.182), (1999), Geneva, International Labour Organization.
- International Labour Organization (ILO), (2017). "Global Estimates of Child Labour: Results and Trends, 2012-2016." ILO Report, www.ilo.org/global/publications/ books/WCMS_575499/lang--en/index.htm.

- International Labour Organization, Geneva, Second Ed., (2018). "Ending child labour by 2025: A review of policies and programmes" *ILO Report*, https://www.ilo. org/ipec/Informationresources/WCMS_653987/lang--en/index.htm
- Instituto Nacional de Estadisticas y Censos (INEC), (2010). Censo Nacional de Poblacion y Vivienda 2010.
- Instituto Nacional de Estadisticas y Censos (INEC), (2011). Statistical Yearbook: Marriages and Divorces: 2011.
- Instituto Nacional de Estadsticas y Censos (INEC), (2012a). Presentacion de la Primera Encuesta Nacional de Relaciones Familiares y Violencia de Genero contra las Mujeres [PDF Document]. Retrieved from: https://www.ecuadorencifras.gob.ec/documentos/web-inec/Estadisticas_ Sociales/sitio_violencia/presentacion.pdf
- Instituto Nacional de Estadisticas y Censos (INEC), (2012b). Bolentin Informativo Canasta Familiar Basica y Canasta Familiar Vital de la Economia Dolarizada.
- Instituto Nacional de Estadisticas y Censos (INEC), (2012c). Encuesta Nacional de Trabajo Infantil.
- Instituto Nacional de Estadosticas y Censos (INEC), (2012d). Manual de Encuesta de Trabajo Infantil [PDF Document]. Retrieved from: https: //www.ecuadorencifras.gob.ec//documentos/web-inec/Estadisticas_ Sociales/Trabajo_Infantil-2012/MANUAL_DE_ENCUESTA_DE_TRABAJO_ INFANTIL 2012.pdf
- Instituto Nacional de Estadosticas y Censos (INEC), (2012e). Presentacion de la Primera Encuesta Nacional de Trabajo Infantil [PDF Document]. Retrieved from: https://www.ecuadorencifras.gob.ec//documentos/web-inec/ Estadisticas_Sociales/Trabajo_Infantil-2012/Presentacion_Trabajo_ Infantil.pdf
- Instituto Nacional de Estadisticas y Censos (INEC), (2015). Historial Estadisticas Economicas 2015.
- Jesmin, S. (2017). Social Determinants of Married Women's Attitudinal Acceptance of Intimate Partner Violence Journal of Interpersonal Violence 32 (21), 3226-3244.
- Jewkes, R2002). Intimate Partner Violence: Causes and Prevention. *The Lancet*, 359(9315), 1423-1429.
- Kelly, H. (1994). "Rule of Thumb" and the Folklaw of the Husband's Stick. *Journal* of Legal Education, 44(3), 341-365.

- Stephanie. "Pope Reforms Catholic Church's Mar-Kirchgaessner, Guardian, Guardian Annulment Process." TheNews riage and 8 2015,www.theguardian.com/world/2015/sep/08/ Media, Sept. pope-radically-reforms-catholic-churchs-marriage-annulment-process.
- Koller, S., Dutra, L., Araujo, N., Nieto, C., Prates, S. (2014). Child Labor: Homes, Streets, Armies, Factories, and Stores. Oxford Handbook of Human Development and Culture: An Interdisciplinary Perspective, 456-470.
- Kulik, K. (2018). Contribution of gender role ideology to explaining emotional wellbeing among working parents. *Community, Work & Family, 23*(3), 305-323
- Kruger, D. (2007). Coffee production effects on child labor and schooling in rural Brazil. *Journal of Development Economics*, 82(2), 448-463.
- La conciliacion, una opcion entre los padres divorciados (2019, February 25). *El Telegrafo*. Retrieved from: http://tinyurl.com/y44joosq
- Levison, D. (2000). Children as Economic Agents. Feminist Economics, 6(1), 125-134.
- Levison, D., Moe, K., & Knaul, M. F. (2001). Youth Education and Work in Mexico. World Development, 29(1), 167-188.
- Lochner, L. (2008). "Intergenerational Transmission," in L. Blume and S. Durlauf (eds.), New Palgrave Dictionary of Economics, 2nd Edition, Palgrave Macmillan.
- Los divorcios crecieron 83.45% en diez anos en Ecuador (2017,June 2).Boletin dePrensa Instituto Nacional deCensos Estadistiycas(INEC).Retrieved from: https://www.ecuadorencifras.gob.ec/ los-divorcios-crecieron-8345-en-diez-anos-en-ecuador/.
- Lodhi, A., Tsegai, D., & Gerber, N. (2011). Determinants of Participation in Child's Education and Alternative Activities in Pakistan, ZEF- Discussion Papers on Development Policy.
- Lundberg, S., & Pollak, R. (1996). Bargaining and Distribution in *Marriage. Journal* of *Economic Perspectives*, 10(4): 139-158.
- Main, G. (2014). Child poverty and children's subjective well-being. Child Indicators Research, 7(3). 451 - 472.
- Martinez, D., Borja, T., Medellin, N., Cueva, P. (2017). Como funciona el Bono de Desarrollo Humano: Mejores practicas en la implementacion de Programas de Transferencias Monetarias Condicionadas en America Latina y el Caribe. *Banco Interamericano de Desarrollo* IBD-TN-1350.
- Mas-Colell, A., Whinston, M. D., & Green, J. R. (1995). *Microeconomic Theory*. New York: Oxford University Press.

- Maya Jariego, I. (2017). "But We Want to Work": The Movement of Child Workers in Peru and the Actions for Reducing Child Labor. American Journal of Community Psychology, 60(3-4), 430-438.
- Menon, N., Van der Meulen Rodgers, Y., Kennedy, A. (2017). Land Reform and Welfare in Vietnam: Why Gender of the Land-Rights Holder Matters. *Journal of International Development*, 29(4), 454-472.
- Metro Ecuador. (2017, May 5). Independencia juvenil, 'sufrimiento para los padres'. Retrieved October 9, 2019, from https://www.metroecuador.com.ec/ ec/estilo-vida/2017/05/05/independencia-juvenil-sufrimiento-padres. html
- Mincer, J1958). Investment in Human Capital and Personal Income Distribution. The Journal of Political Economy, 66(4), 281-302.
- Ministerio de Educacion (2012). Registros Administrativos (2011-2012). [Data file]. Retrieved from: https://educacion.gob.ec/amie/
- Ministerio de Justicia, Derechos Humanos y Cultos (MJDHC). (2018). Proyecto para la prevencion de violencia de genero y fortalecimiento de los servicios de atención a victimas. Quito, Ecuador.
- Munyo, I. & Rossi, M. (2015). The Effects of Real Exchange Rate Fluctuations on the Gender Wage Gap and Domestic Violence in Uruguay. IDB Working Paper Series No. IDB-WP-618, Inter-American Development Bank.
- Nuevas reglas para el matrimonio y el divorcio (2015, April 22). *El Comercio*. Retrieved from: https://www.elcomercio.com/actualidad/nuevas-reglas-matrimonio-divorcio-paternidad.html
- Office for National Statistics (ONS). (2014). Measuring National Well-being: Children's Well-being. Retrieved from: http://www.ons.gov.uk/ons/rel/ wellbeing/measuring-national-well-being/children-s-well-being-2014/ rpt-Children-s-well-being-2014.html
- Organization of Economic Cooperation and Development (OECD) (2013), OECD Guidelines on Measuring Subjective Well-being, OECD Publishing, Paris, https: //doi.org/10.1787/9789264191655-en.
- Orkin, K. (2011). 'See first, think later, then test': How children's perspectives can improve economic research. *The European Journal of Development Research*, 23(5), 774-791.
- Orpinas, P. (1999). Who is violent?: Factors associated with aggressive behaviors Latin America and Spain. *Revista Panamericana de Salud Publica*, 5(4-5), 232-44.
- Osberg, L., & Phipps, S1993). Labour Supply with Quantity Constraints: Estimates from a Large Sample of Canadian Workers. Oxford Economic Papers, 45(2), 269-291.

- Oster, E. (2019). Unobservable selection and coefficient stability: Theory and evidence. Journal of Business & Economic Statistics, 37(2), 187-204.
- Our Watch (2018). Challenging gender stereotypes in the early years: the power of parents. Melbourne, Australia: Our Watch.
- Paxson, C., & Schady, N. (2010). Does Money Matter? The Effects of Cash Transfers on Child Development in Rural Ecuador. *Economic Development and Cultural Change*, 59(1), 187-229.
- Phipps, S. (1999). Economics and the Well-Being of Canadian Children. The Canadian Journal of Economics / Revue Canadienne D'Economique, 32(5), 1135-1163.
- Phipps, S., & Burton, P. (1998). What's Mine is Yours? The Influence of Male and Female Incomes on Patterns of Household Expenditure. *Economica*, 65(260), 599-613.
- Phipps S., Burton P., & Osberg, L. (2001) Time as a Source of Inequality Within Marriage: Are Husbands More Satisfied with Time for Themselves than Wives? *Feminist Economics*, 7(2), 1-21.
- Phipps, S., & Woolley, F. (2008). Control Over Money and the Savings Decisions of Canadian Households. The Journal of Socio-Economics, 37(2), 592-611.
- Pollak, R. (2004). An Intergenerational Model of Domestic Violence. Journal of Population Economics, 17(2), 311-329.
- Psacharopoulos, G1997), Child Labor Versus Educational Attainment: Some Evidence from Latin America, Journal of Population Economics, 10(4), 377-386.
- Que el divorcio no le deje quebrado (2011, February 19). *El Comercio*. Retrieved from: https://www.elcomercio.com/actualidad/negocios/ que-divorcio-no-le-deje-1.html
- Ragui, A., Levison, D. & Zibani, N2010). The Effect of Domestic Work on Girls' Schooling: Evidence from Egypt, *Feminist Economics*, 16(1), 79-128.
- Ray, R. (2000). Analysis of child labour in Peru and Pakistan: A comparative study. Journal of Population Economics, 13(1), 3-19.
- Ray, R2001). Child Labour and Child Schooling in South Asia: A Cross Country Study of their Determinants. IDEAS Working Paper Series from RePEc, IDEAS Working Paper Series from RePEc.
- Rodriguez, A. "Los Jovenes Viven Mas Tiempo En Casa De Sus Padres." *El Telegrafo*, 26 Feb. 2017, tinyurl.com/yc2p3g6n.

- Safranoff, A. & Tiravassi A. (2018). The Intergenerational Transmission of Violence: Testimonials from Prison. Technical Note No. IDB-TN-1410. Washington, DC: Inter-American Development Bank.
- Sanchez, D. (2015). La Tendencia del Abandono Escolar en Ecuador: Periodo 1994-2014. (Drop-out Rates Trends in Ecuador: 1994-2004). Valor Agregado (UDLA) 3, 39-66.
- Schady, N. & Araujo, M. (2006). Cash Transfers, Conditions, School Enrollment and Child Work: Evidence from a Randomized Cash Transfer Experiment in Ecuador. Policy Research Working Paper 3930. World Bank, Policy Research Department, Washington, DC.
- Schady, N., & Araujo, M.C. (2008). Cash Transfers, Conditions, and School Enrollment in Ecuador. *Economia*, 8(2), 43-70.
- Secretaria Tecnica del Plan Toda una Vida (STPTV). 2018. "Intervencion Emblematica, Mision Mujer." https://www.todaunavida.gob.ec/wp-content/uploads/ downloads/2018/12/MisionMujerL4.pdf
- Sharmin, S. (2015). Wellbeing of Child Domestic Workers in Urban Bangladesh. Journal of Indian Research, 3(4), 89-95.
- Sistema Integrado de Conocimiento y Estadistica Social de Ecuador. 2018. "Distribucion porcentual de hogares, segun tenencia de vivienda." http://www. conocimientosocial.gob.ec/pages/EstadisticaSocial/herramientas.jsf
- Sistema de Infomacion de Tendencias Educativas en America Latina (SITEAL). "Education Profile, Ecuador." UNESCO & IIEP. Retrieved from: https://www.siteal.iiep.unesco.org/sites/default/files/sit_informe_ pdfs/dpe_ecuador-_25_09_19.pdf
- Smith, S. G., Zhang, X., Basile, K.C., Merrick, M.T., Wang, J., Kresnow, M., Chen, J. (2018). The National Intimate Partner and Sexual Violence Survey (NISVS): 2015 Data Brief-Updated Release. Atlanta, GA: National Center for Injury Prevention and Control, Centers for Disease Control and Prevention
- Soares, R., Kruger, D., & Berthelon, M. (2012). Household Choices of Child Labor and Schooling: A Simple Model with Application to Brazil. *The Journal of Human Resources*, 47(1), 1-31
- Stratton, L. (2015). The determinants of housework time. IZA World of Labor.
- Stata Corporation. (2003). Stata Reference manual. College Station, TX: Stata Corp.
- Stevenson, B., & Wolfers, J. (2006). Bargaining in the Shadow of the Law: Divorce Laws and Family Distress. Quarterly Journal of Economics, 121(1), 267-288.

- Stevenson, B., & Wolfers, J. (2009) The Paradox of Declining Female Happiness. American Economic Journal: Economic Policy, 1(2), 190-225.
- Sugawara, K. (2010). Intergenerational transfers and fertility: Trade-off between human capital and child labour. *Journal of Macroeconomics*, 32(2), 584-593.
- Suplemento al Registro Oficial S.R.O. No. 167, Codigo de Trabajo, (2012). National Congress.
- Suplemento al Registro Oficial S. R.O. No. 754 Reglamento General a la Ley Organica de Educacion Intercultural, (2012). Presidential Decree No. 1241 [PDF Document]. Retrieved from: http://educacion.gob.ec/ wp-content/uploads/downloads/2015/01/REGLAMENTO-GENERAL-A-LA-LOEI-\ CODIFICADO-A-5-DE-ENERO-DE-2015.pdf
- Sumplemento al Registro Oficial R.O. No. 167, Salario Basico Unificado, (2014). Ministerial Agreement. [PDF Document]. Retrieved from: http://www.cit.org. ec/files/RO-No.-167S-del-22-01-2014.pdf
- Svec, J. & Andic, T. (2018). Cooperative Decision-Making and Intimate Partner Violence in Peru. Population Development Review, 44(1) 63-85.
- Tamayo, C. (1998). Entre la Sombra y la Esperanza, investigacion de impacto de las Comisarias de la Mujer y la Familia, Quito Ecuador.
- Tankard, M. & Iyengard, R. (2018). Economic Policies and Intimate Partner Violence Prevention: Emerging Complexities in the Literature. *Journal of Interpersonal Violence*, 33(21), 3367-3388.
- Tarabah, A., Kurdahi Badr, L., Usta J. (2016). Exposure to Violence and Children's Desensitization Attitudes in Lebanon. *Journal of Interpersonal Violence*, 31(18), 3017-3038.
- Tiefenthaler, J. (2012, February). *Economics of Domestic Violence*. Eclectics Talk Colorado College. Denver.
- Turcotte, M., & Statistics Canada, issuing body. (2013). Family caregiving : What are the consequences? (Article (Statistics Canada). English). Ottawa]: Statistics Canada = Statistique Canada.
- Udry C2004). Child Labour. Retrieved from: http://www.econ.yale.edu/~cru2/pdf/kid.pdf
- United Nations Development Programme (UNDP). "UNSDG: Goal 8: Decent work and economic growth." Target 8.7. Retrieved from: www.undp.org/content/undp/en/home/sustainable-development-goals/ goal-8-decent-work-and-economic-growth.html.

- United Nations Educational Scientific and Cultural Organization (UNESCO), (2012). Global Education Digest 2012 [PDF Document]. Retrieved from: http://www.uis. unesco.org/Education/Documents/ged-2012-en.pdf
- United Nations Educational Scientific and Cultural Organization (UNESCO), (2014). Country Profile: Ecuador. Retrieved from: http://uis.unesco.org/en/country/ ec
- United Nations Educational Scientific and Cultural Organization (UNESCO), (2020).Technical Cooperation Indica-Group on the SDG 4. for Retrieved from: http://tcg.uis.unesco.org/ tors 4-1-4-completion-rate-primary-lower-secondary-upper-secondary/
- United Nations Population Fund (UNFPA); Avenir Health, Johns Hopkins University; Victoria University (Australia). (2020). Impact of the COVID-19 pandemic on family planning and ending gender-based violence, female genital mutilation and child marriage.
- United Nations Children's Fund, Harnessing the Power of Data for Girls: Taking stock and looking ahead to 2030, UNICEF, New York, 2016.
- United Nations Office of Drugs and Crime (UNODC), *Global Study on Homicide 2018* (Vienna, 2018)
- Valle, C. (2018). Atlas de Genero. Instituto Nacional de Estadisticas y Censos (INEC), Quito Ecuador.
- Violencia de genero, la pandemia oculta: organizaciones sociales registran 81 femicidios en Ecuador desde enero hasta el 4 de octubre del 2020 (2020, October 25). El Comercio. Retrieved from: https://www.elcomercio.com/actualidad/ femicidios-violencia-mujeres-mapa-ecuador.html.
- Watson, B., & Osberg, L. (2017). Healing and/or breaking? The mental health implications of repeated economic insecurity. Social Science & Medicine, 188, 119-127.
- Watson, B., & Osberg, L. (2018). Job insecurity and mental health in Canada. Applied Economics, 50(38), 4137-4152.
- Weitzman, A. (2014). Women's and Men's Relative Status and Intimate Partner Violence in India Population and Development Review 40(1):, 55-75.
- World Development Indicators (WDI) (2014). *Ecuador*. Retrieved March 17, 2015, from http://data.worldbank.org/country/ecuador
- World Development Indicators (WDI) (2019). *Ecuador*. Retrieved from http://data. worldbank.org/country/ecuador

- Webbink, E., Smits, J., & De Jong, E. (2012). Hidden Child Labor: Determinants of Housework and Family Business Work of Children in 16 Developing Countries. World Development, 40(3), 631-642.
- World Health Organization (WHO) & Pan American Health Organization (PAHO). (2012). Understanding and addressing violence against women : intimate partner violence. World Health Organization.
- World Health Organization. (2014). Global status report on violence prevention 2014. World Health Organization.
- Wikle, J. (2014). Patterns in housework and childcare among girls and boys. *Journal* of Research on Women and Gender, 5, 17-29.
- Williams, R. (2012) Using the margins command to estimate and interpret adjusted predictions and marginal effects. *The Stata Journal*, 12 (2), 308-331.
- Winkelmann, R. (2005). Subjective Well-Being and the Family: Results from an Ordered Probit Model with Multiple Random Effects. *Empirical Economics*, 30(3), 749-761.
- Zapata, D., Contreras, D., & Kruger, D. (2011). Child Labor and Schooling in Bolivia: Who's Falling Behind? The Roles of Domestic Work, Gender, and Ethnicity. World Development, 39(4), 588-599.

Appendix A

Intimate Partner Violence



Figure A.1: Women's Individual Ownership of Assets by Marital Status (Disaggregated Measures)

Note: For post-coupled women, this refers to assets owned during previous relationship. **Source:** National Survey on Family Relations and Gendered Violence Against Women



Figure A.2: Men's Individual Ownership of Assets by Marital Status (Disaggregated Measures)

Note: For post-coupled women, this refers to assets owned by former partner during previous relationship. Source: National Survey on Family Relations and Gendered Violence Against Women



Figure A.3: Co-ownership of Assets by Marital Status (Disaggregated Measures)

Note: For post-coupled women, this refers to assets co-owned during previous relationship. **Source:** National Survey on Family Relations and Gendered Violence Against Women



Figure A.4: Acceptability of Wife Beating (Disaggregated Measures)

Source: National Survey on Family Relations and Gendered Violence Against Women



Figure A.5: Agreement with Patriarchal/Traditional Beliefs (Disaggregated Measures)

Source: National Survey on Family Relations and Gendered Violence Against Women



Figure A.6: Domestic Violence in Family Background for Women (Disaggregated Measures)

Source: National Survey on Family Relations and Gendered Violence Against Women

Figure A.7: Ever Experienced Intimate Partner Violence Divided by Years in Relationship (Exposure to Violence) by Marital Status



Note: Exposure to violence is defined as experiencing a type of violence, (0/1), divided by the duration of the match.
 For separated, divorced and married women, this refers to violence experienced during previous relationship.
 Source: National Survey on Family Relations and Gendered Violence Against Women



Figure A.8: Ever Experienced Intimate Partner Violence by Marital Status: 45-65 Year Old Women

Note: For post-coupled and widowed women, this refers to violence experienced during previous relationship. Source: National Survey on Family Relations and Gendered Violence Against Women



Figure A.9: Ownership of Assets in Marital Estate by Age of Women (All Women)

Source: National Survey on Family Relations and Gendered Violence Against Women

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Has Money for Personal Expenses	-0.053***	-0.081***	-0 035***	-0.020**
······································	[0.017]	[0.019]	[0.010]	[0.008]
Woman Has Ownership of Land or Crops	-0.098***	-0.116***	-0.024*	-0.011
woman mas o whorship of hand of crops	[0.028]	[0.031]	[0.021]	[0.013]
Woman Has Ownership of Car or Vehicle	0.019	0.085	0.055	0.081*
woman mas ownersnip of ear of venicle	[0.052]	[0.059]	[0.051]	[0.046]
Woman Has Ownership of Savings	0.051	0.068	0.063*	0.028
woman mas ownersnip of Savings	[0.048]	[0.052]	[0.033]	[0.023]
Woman Has Ownership of House	0.067**	0.101***	0.042**	[0.025] 0.027*
woman mas Ownersnip of mouse	[0.021]	[0.020]	[0.042]	[0.015]
Spause Hag Ownership of Land or Crops	[0.031]	[0.029]	0.014	[0.015]
spouse mas Ownersmp of Land of Crops	[0.024]	-0.040	-0.014	0.008
Snowe Use Ormenskin of Con on Vakiala	[0.054]	[0.028]	[0.015]	[0.017]
spouse has Ownership of Car of Venicle	[0.080]	0.000	[0.049]	0.055
	[0.040]	[0.036]	[0.027]	[0.025]
Spouse Has Ownership of Savings	0.044	0.025	0.024	0.001
	[0.044]	[0.041]	[0.023]	[0.014]
Spouse Has Ownership of House	0.032	0.054**	0.015	0.015
	[0.027]	[0.025]	[0.014]	[0.013]
Household Has No Ownership of Land or Crops	-0.048**	-0.047**	-0.022**	-0.003
	[0.022]	[0.022]	[0.011]	[0.009]
Household Has No Ownership of Car or Vehicle	0.080^{***}	0.043	0.028^{***}	0.020^{*}
	[0.028]	[0.031]	[0.010]	[0.011]
Household Has No Ownership of Savings	-0.018	-0.026	0.023^{**}	0.009
	[0.033]	[0.035]	[0.012]	[0.010]
Household Has No Ownership of House	0.010	0.046^{**}	0.002	-0.001
	[0.020]	[0.020]	[0.010]	[0.009]
Household Equivalent Income (Scaled)	-0.004	-0.018	-0.011	-0.010
	[0.026]	[0.026]	[0.013]	[0.013]
Woman Earns 51% or More of Household Income	-0.044**	-0.014	-0.008	0.011
	[0.021]	[0.021]	[0.010]	[0.010]
Woman's Acceptability of Wife Beating Index	0.025***	0.021***	0.003	0.002
* ° 0	[0.007]	[0.007]	[0.003]	[0.003]
Woman's Strength of Patriarchal/Traditional Beliefs	-0.122***	-0.133***	-0.062***	-0.053***
0 1	[0.045]	[0.045]	[0.021]	[0.016]
Violence in Women's Family Background	0.188***	0.177***	0.047***	0.050***
	[0.015]	[0.016]	[0.007]	[0.007]
Violence in Spouse's Family Background	0.184***	0.211***	0.055***	0.045***
(isiense in opsuse s raining Baenground	[0.019]	[0.017]	[0 009]	[0 008]
Unknown Violence in Spouse's Background	0.063***	0.083***	0.005	-0.001
Summer and Alexander and Alexandre and Alexandre	[0.017]	[0.017]	[0.008]	[0.001]
Observations	10.801	10.801	10.801	10.801
Controls Included	Ves	Ves	Ves	Ves
Pseudo R^2	0.150	0.126	0.125	0.119

 Table A.1: Disaggregation of Ownership Variables for Married and Common-Law Women:

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Had Money for Personal Expenses	-0.212***	-0.216***	-0.177***	-0.140***
Woman Had Ownership of Land or Crops	[0.045] -0.287** [0.126]	[0.040] -0.221 [0.162]	[0.042] -0.093	$\begin{bmatrix} 0.039 \end{bmatrix}$ 0.095 [0.110]
Woman Had Ownership of Car or Vehicle	[0.120] 0.104 [0.196]	[0.102] -0.252 [0.238]	0.183	$\begin{bmatrix} 0.119 \\ 0.224 \\ \begin{bmatrix} 0.174 \end{bmatrix}$
Woman Had Ownership of Savings	-0.094 [0.180]	0.016	[0.130] 0.073 [0.140]	[0.174] -0.021 [0.107]
Woman Had Ownership of House	0.030 [0.079]	-0.013 [0.078]	-0.098 [0.060]	0.053 [0.072]
Spouse Had Ownership of Land or Crops	-0.373*** [0.107]	-0.391*** [0.132]	-0.025 [0.102]	0.086 [0.125]
Spouse Had Ownership of Car or Vehicle	0.099 [0.150]	0.084 [0.147]	0.062 [0.117]	-0.153*** [0.059]
Spouse Had Ownership of Savings	-0.135 [0.182]	0.175 [0.132]	-0.059 [0.110]	$0.151 \\ [0.184]$
Spouse Had Ownership of House	0.022 [0.114]	0.010 [0.116]	-0.053 [0.075]	0.004 [0.093]
Household Had No Ownership of Car or Vehicle	[0.1138]	[0.087]	[0.104]	[0.009] [0.074] 0.105
Household Had No Ownership of Savings	[0.143]	[0.153] -0.028	[0.014] [0.095] 0.072	[0.103] [0.102] -0.075
Household Had No Ownership of House	[0.171] -0.101	[0.156] -0.085	[0.092] -0.073	[0.107] 0.032
Household Equivalent Income (Scaled)+	$[0.074] \\ 0.103$	[0.070] -0.050	[0.064] -0.057	[0.060] -0.094
Woman's Acceptability of Wife Beating Index+	[0.089] 0.027	[0.093] 0.024	[0.084] -0.026	$\begin{bmatrix} 0.091 \end{bmatrix} \\ 0.024 \end{bmatrix}$
Woman's Strength of Patriarchal/Traditional Beliefs+	[0.024] -0.295**	[0.024] -0.406***	[0.020] -0.229*	[0.019] -0.294**
Violence in Women's Family Background	[0.143] 0.153^{***}	[0.144] 0.166^{***} [0.042]	[0.117] 0.102^{***}	[0.123] 0.079^{**} [0.024]
Violence in Spouse's Family Background	0.220^{***}	0.196^{***} [0.053]	0.208^{***}	0.124^{***}
Unknown Violence in Spouse's Background	0.046 [0.050]	[0.031] [0.051]	[0.041] [0.049] [0.041]	[0.035] -0.040 [0.036]
Observations	1,401	1,401	1,401	1,401
Controls Included Pseudo R^2	Yes 0.209	Yes 0.208	Yes 0.211	Yes 0.166

 Table A.2: Disaggregation of Ownership Variables for Divorced and Separated Women:

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

Table A.3: Disaggregation of Beliefs Variables: Married and Common-Law Women Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Has Money for Personal Expenses	-0.051***	-0.079***	-0.033***	-0.019**
v 1	[0.017]	[0.019]	[0.010]	[0.008]
Woman Has Ownership of Land or Crops	-0.097***	-0.114***	-0.023*	-0.010
	[0.027]	[0.031]	[0.012]	[0.013]
Woman Has Ownership of Car or Vehicle	0.015	0.079	0.052	0.082*
Wollian Hab Ownership of Oar of Venicio	[0.051]	[0.058]	[0.049]	[0.046]
Woman Has Ownership of Savings	0.051	0.068	0.061*	0.029
woman mas o whorship of bavings	[0.049]	[0.052]	[0.032]	[0.023]
Woman Has Ownership of House	0.066**	0.101***	0.043**	0.027*
Wollah Has Owleiship of House	[0.030]	[0 020]	[0.017]	[0.015]
Spouse Hes Ownership of Land or Crops	0.012	0.048*	0.017	0.005
spouse mas ownership of Land of Crops	[0.034]	[0.028]	[0.012]	[0.015]
Spouse Hes Ownership of Car or Vehicle	0.085**	0.053	0.040*	0.034
Spouse mas ownership of oar of venicle	[0.040]	[0.026]	[0.045]	[0.024]
Spauge Heg Ormership of Servings	0.040	0.030	0.027]	0.001
spouse mas ownership or savings	[0.044]	[0.022	[0.023	[0.001
Snource Has Ommenshin of House	[0.044]	[0.042]	[0.023]	[0.014]
spouse has Ownership of house	0.055	0.000	0.017	0.010
	[0.026]	[0.024]	[0.013]	[0.012]
Household Has No Ownership of Land or Crops	-0.049***	-0.049	-0.023***	-0.003
	[0.022]	[0.022]	[0.011]	[0.009]
Household Has No Ownership of Car or Vehicle	0.080***	0.040	0.027***	0.019*
	[0.028]	[0.031]	[0.010]	[0.011]
Household Has No Ownership of Savings	-0.021	-0.028	0.022*	0.008
	[0.034]	[0.035]	[0.012]	[0.010]
Household Has No Ownership of House	0.007	0.044**	0.001	-0.001
	[0.020]	[0.020]	[0.009]	[0.009]
Household Equivalent Income (Scaled)	-0.004	-0.018	-0.010	-0.011
	[0.025]	[0.025]	[0.012]	[0.013]
Belief: Husband Justified in Hitting Wife if Disobedient	0.019	-0.031	-0.033***	-0.022**
	[0.037]	[0.036]	[0.009]	[0.009]
Belief: Husband Justified in Hitting Wife if Improper Care of Children	0.011	0.035	0.013	0.048^{***}
	[0.030]	[0.029]	[0.016]	[0.018]
Belief: Husband Justified in Hitting Wife if Unfaithful	0.027	0.051^{**}	0.001	-0.008
	[0.020]	[0.021]	[0.009]	[0.007]
Belief: Husband Justified in Hitting Wife if Goes Out Too Much	0.054^{**}	0.017	0.025^{*}	0.005
	[0.027]	[0.029]	[0.015]	[0.011]
Agrees: a Good Wife Must Obey All Orders from Husband	-0.096***	-0.076***	-0.024***	-0.019***
	[0.016]	[0.017]	[0.008]	[0.006]
Agrees: Wife Can Choose Friends Even Disliked by Husband	0.017	-0.002	0.013^{**}	0.011^{*}
	[0.017]	[0.016]	[0.006]	[0.006]
Agrees: Men Should be Responsible for All Family Expenses	0.047^{***}	0.039^{**}	0.010	0.008
	[0.018]	[0.017]	[0.007]	[0.007]
Agrees: Women Have Equal Right to Work and Earn Money	0.001	0.063**	0.001	0.003
	[0.026]	[0.027]	[0.013]	[0.013]
Agrees: It is a Wife's Obligation to Have Sex Even if Unwanted	-0.008	-0.047*	-0.002	-0.022***
5	[0.027]	[0.026]	[0.010]	[0.007]
Violence in Women's Family Background	0.185***	0.173***	0.046***	0.049***
	[0.015]	[0.016]	[0.007]	[0.007]
Violence in Spouse's Family Background	0.184***	0.212***	0.055***	0.044***
· · · · · · · · · · · · · · · · · · ·	[0.019]	[0,017]	[0,009]	[0.007]
Unknown Violence in Spouse's Background	0.063***	0.084***	0.006	-0.001
	[0.017]	[0.017]	[0.008]	[0.006]
	[0:011]	[0.011]	[0.000]	[0.000]
Observations	10.801	10.801	10.801	10.801
Controls Included	Yes	Yes	Yes	Yes
Pseudo R^2	0.155	0.131	0.131	0.128

Table A.4: Disaggregation of Beliefs Variables: Separated and Divorced Women

Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Had Money for Personal Expenses	-0.206***	-0.219***	-0.172***	-0.128***
	[0.045]	[0.039]	[0.042]	[0.038]
Woman Had Ownership of Land or Crops	-0.307**	-0.241	-0.129*	0.046
Women Hed Ommenchin of Can on Vehicle	[0.123]	[0.158]	0.076	0.205*
woman had Ownership of Car or Venicle	0.125	-0.280	0.225	0.303
Woman Had Ownership of Savings	-0.065	0.061	0.103	-0.023
woman mad ownersnip of savings	[0.171]	[0.182]	[0.142]	[0.103]
Woman Had Ownership of House	0.046	0.019	-0.083	0.054
*	[0.077]	[0.074]	[0.060]	[0.071]
Spouse Had Ownership of Land or Crops	-0.339***	-0.363***	-0.012	0.105
	[0.115]	[0.137]	[0.104]	[0.125]
Spouse Had Ownership of Car or Vehicle	0.103	0.061	0.068	-0.148**
	[0.149]	[0.149]	[0.119]	[0.059]
Spouse Had Ownership of Savings	-0.081	0.184	-0.015	0.186
Spouse Had Ormership of House	[0.176]	[0.122]	[0.117]	0.025
Spouse flad Ownership of flouse	[0.100]	0.023	-0.081	-0.025
Household Had No Ownership of Land or Crops	-0.165	-0 224***	-0.258**	-0.016
Household Had 140 Ownership of Band of Crops	[0.112]	[0.082]	[0.106]	[0.075]
Household Had No Ownership of Car or Vehicle	-0.020	0.060	0.049	-0.066
1	[0.141]	[0.152]	[0.087]	[0.099]
Household Had No Ownership of Savings	0.021	0.001	0.085	-0.071
	[0.162]	[0.153]	[0.086]	[0.105]
Household Had No Ownership of House	-0.100	-0.077	-0.069	0.033
	[0.073]	[0.068]	[0.062]	[0.058]
Household Equivalent Income (Scaled)+	0.107	-0.051	-0.051	-0.098
	[0.085]	[0.089]	[0.085]	[0.088]
Belief: Husband Justified in Hitting Wife if Disobedient+	-0.041	-0.133	-0.043	-0.062
Belief: Husband Justified in Hitting Wife if Improper Care of Children+	-0.136	-0.071	0.024	0.036
bener. Husband Justified in Hitting when in hiptoper Care of Children+	[0 100]	[0.097]	[0.024	[0.030]
Belief: Husband Justified in Hitting Wife if Unfaithful+	0.068	0.111*	0.040	0.085
	[0.073]	[0.065]	[0.061]	[0.053]
Belief: Husband Justified in Hitting Wife if Goes Out Too Much+	0.169*	0.089	-0.117**	-0.012
	[0.088]	[0.093]	[0.059]	[0.076]
Agrees: a Good Wife Must Obey All Orders from Husband+	-0.165^{***}	-0.158^{***}	-0.158^{***}	-0.119***
	[0.047]	[0.045]	[0.035]	[0.036]
Agrees: Wife Can Choose Friends Even Disliked by Husband+	0.036	0.103**	-0.011	-0.035
America Man Should be Deen ancible for All Femily Furnesses	0.065	[0.043]	[0.039]	[0.039]
Agrees: Men Should be Responsible for All Falliny Expenses+	0.005	0.023	0.050	-0.017
Agrees: Women Have Equal Right to Work and Earn Money+	0.150*	-0.129	0.080	0.156***
ingrees. Women nave Equal regit to Work and Earn Money ([0.090]	[0.089]	[0.063]	[0.039]
Agrees: It is a Wife's Obligation to Have Sex Even if Unwanted+	-0.008	-0.105	0.030	0.007
0	[0.074]	[0.074]	[0.076]	[0.062]
Violence in Women's Family Background	0.152***	0.165***	0.100***	0.080**
	[0.043]	[0.043]	[0.038]	[0.034]
Violence in Spouse's Family Background	0.222^{***}	0.190^{***}	0.205^{***}	0.122^{***}
	[0.046]	[0.052]	[0.040]	[0.039]
Unknown Violence in Spouse's Background	0.052	0.036	0.049	-0.037
	[0.051]	[0.050]	[0.040]	[0.036]
Observations	1.401	1 401	1 /01	1.401
Controls Included	Yes	Yes	Yes	Yes
Pseudo R^2	0.224	0.221	0.228	0.178

	Physical Violence	Psychological Violence	Sexual Violence	Economic Violence
	(1)	(2)	(3)	(4)
Woman Has Money for Personal Expenses	-0.045*** [0.017]	-0.074^{***} [0.019]	-0.030*** [0.009]	-0.014* [0.008]
Woman Has Ownership of Land or Crops	-0.106***	-0.115***	-0.020*	-0.006
Woman Has Ownership of Car or Vehicle	$\begin{bmatrix} 0.029 \end{bmatrix}$ 0.022	0.082	$\begin{bmatrix} 0.012 \end{bmatrix}$ 0.045	0.070
Woman Has Ownership of Savings	$\begin{bmatrix} 0.053 \end{bmatrix}$ 0.071	0.094*	$\begin{bmatrix} 0.047 \end{bmatrix} \\ 0.051 \\ \begin{bmatrix} 0.020 \end{bmatrix}$	0.032
Woman Has Ownership of House	[0.051] 0.064^{**}	[0.053] 0.104^{***}	[0.032] 0.048^{***}	[0.024] 0.027^*
Spouse Has Ownership of Land or Crops	[0.032] 0.019	[0.030] -0.031	[0.018] -0.015	[0.015] 0.011
Spouse Has Ownership of Car or Vehicle	$[0.035] \\ 0.066$	[0.029] 0.051	$[0.012] \\ 0.040$	[0.016] 0.023
Spouse Has Ownership of Savings	$\begin{bmatrix} 0.040 \end{bmatrix} \\ 0.045 \end{bmatrix}$	[0.036] 0.044	[0.026] 0.021	[0.022] 0.003
Spouse Has Ownership of House	$\begin{bmatrix} 0.047 \end{bmatrix}$ 0.024	[0.044] 0.045*	$\begin{bmatrix} 0.024 \end{bmatrix}$	[0.015]
Household Has No Ownership of Lond or Crons	[0.027]	[0.026]	[0.011] [0.014] 0.021**	[0.012]
Household has No Ownership of Land of Crops	-0.047 [0.022]	[0.022]	-0.021 [0.011]	[0.009]
Household Has No Ownership of Car or Vehicle	0.060^{**} [0.028]	[0.031]	$[0.023^{**}]$	[0.014]
Household Has No Ownership of Savings	-0.011 [0.036]	-0.006 [0.034]	0.019 [0.013]	$0.010 \\ [0.010]$
Household Has No Ownership of House	0.001 [0.021]	0.032 [0.022]	0.002 [0.009]	-0.005 [0.008]
Household Equivalent Income (Scaled)	-0.041 [0.028]	-0.031 [0.026]	-0.009 [0.013]	-0.013 [0.013]
Woman's Acceptability of Wife Beating Index	0.021***	0.018**	0.002	0.002
Woman's Strength of Patriarchal/Traditional Beliefs	-0.089* [0.046]	-0.103** [0.045]	-0.051**	-0.049*** [0.017]
As Child: Woman Physically Abused	0.096***	0.077***	0.002	0.023***
As Child: Woman Verbally Abused	0.106***	0.128***	0.042***	0.030***
As Child: Woman Sexually Abused	[0.024] 0.211***	[0.024] 0.165***	[0.010] 0.100***	[0.009] 0.070***
Violence in Spouse's Family Background	[0.030] 0.180^{***}	[0.030] 0.205^{***}	[0.018] 0.052^{***}	[0.015] 0.042^{***}
Unknown Violence in Spouse's Background	$[0.019] \\ 0.071^{***} \\ [0.017]$	$[0.017] \\ 0.087^{***} \\ [0.017]$	$\begin{array}{c} [0.009] \\ 0.008 \\ [0.008] \end{array}$	$[0.007] \\ 0.000 \\ [0.007]$
Observations	10,801	10,801	10,801	10,801
Controls Included Pseudo R^2	Yes 0.160	Yes 0.136	Yes 0.154	Yes 0.135

 Table A.5: Disaggregation of Childhood Violence Variables: Married and Common-Law Women

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

	Physical	Psychological	Sexual	Economic
	Violence	Violence	Violence	Violence
	(1)	(2)	(3)	(4)
Woman Had Money for Personal Fynances	0 195***	0.206***	0 107***	0 150***
Wollian flag Molley for Tersonal Expenses	-0.185	-0.200	[0.045]	-0.150
Woman Had Ownership of Land or Crops	0.047]	0.233	0.043]	0.130
woman may ownership of Land of Crops	-0.219	-0.255	[0.100]	[0.139
Woman Had Ownership of Car or Vehicle	0.049	-0.351	0.214	0.268
woman mad ownersnip of ear of venicle	[0.213]	[0.220]	[0.213]	[0 192]
Woman Had Ownership of Savings	-0.076	0.052	0.027	-0.063
Woman Had Ownership of Savings	[0,190]	[0.196]	[0 137]	[0, 102]
Woman Had Ownership of House	0.017	-0.016	-0.117**	0.054
foliali fiad o miciship of fibabo	[0.081]	[0.083]	[0.057]	[0.075]
Spouse Had Ownership of Land or Crops	-0.390***	-0.417***	-0.024	0.142
Speaker and a second period of a second second	[0.111]	[0.128]	[0.107]	[0.140]
Spouse Had Ownership of Car or Vehicle	0.174	0.074	0.052	-0.165***
»F	[0.140]	[0.151]	[0.114]	[0.059]
Spouse Had Ownership of Savings	-0.178	0.162	-0.086	0.111
I I I I	[0.185]	[0.134]	[0.099]	[0.166]
Spouse Had Ownership of House	0.025	-0.008	-0.050	-0.002
	[0.114]	[0.120]	[0.077]	[0.089]
Household Had No Ownership of Land or Crops	-0.131	-0.227**	-0.226**	0.050
· ·	[0.121]	[0.089]	[0.115]	[0.078]
Household Had No Ownership of Car or Vehicle	-0.102	0.083	-0.011	-0.127
-	[0.148]	[0.154]	[0.103]	[0.111]
Household Had No Ownership of Savings	0.032	-0.025	0.072	-0.087
	[0.181]	[0.156]	[0.092]	[0.114]
Household Had No Ownership of House	-0.093	-0.109	-0.070	0.043
	[0.076]	[0.074]	[0.063]	[0.062]
Household Equivalent Income (Scaled)+	0.106	-0.019	-0.047	-0.086
	[0.097]	[0.098]	[0.089]	[0.096]
Woman's Acceptability of Wife Beating Index+	0.029	0.018	-0.025	0.030
	[0.026]	[0.025]	[0.021]	[0.020]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.386***	-0.437***	-0.269**	-0.323***
	[0.145]	[0.145]	[0.116]	[0.124]
As Child: Woman Physically Abused	0.050	0.093^{*}	0.028	0.004
	[0.047]	[0.049]	[0.047]	[0.038]
As Child: Woman Verbally Abused	0.128**	0.089*	0.112**	0.115***
	[0.055]	[0.051]	[0.048]	[0.041]
As Child: Woman Sexually Abused	0.132**	0.136**	0.223***	0.203***
	[0.052]	[0.055]	[0.050]	[0.054]
Violence in Spouse's Family Background	0.255***	0.201***	0.195***	0.118***
	[0.045]	[0.053]	[0.041]	[0.040]
Unknown Violence in Spouse's Background	0.105**	0.045	0.085**	-0.010
	[0.052]	[0.052]	[0.042]	[0.038]
Observations	1 401	1 401	1 401	1 401
Controls Locked	1,401 Voz	1,401 Voc	1,401 Voc	1,401 Vaa
Pseudo B^2	0.227	0.215	0.255	0.197

 Table A.6: Disaggregation of Childhood Violence Variables: Separated and Divorced Women

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

Table A.7: Sequential Addition of Sets of Variables: Married and Common-Law Women Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Physical Violence

	Set: Resources	Sets: Resources	Sets: Resources, Boliofs & Child Violence
	(1)	(2)	(3)
	0.000***	0.000***	0.050***
Woman Has Money for Personal Expenses	-0.066***	-0.066***	-0.053***
	[0.017]	[0.017]	[0.017]
Woman Has Ownership of Land or Crops	-0.099***	-0.103***	-0.098***
	[0.030]	[0.029]	[0.028]
Woman Has Ownership of Car or Vehicle	0.006	0.007	0.019
	[0.050]	[0.050]	[0.052]
Woman Has Ownership of Savings	0.075	0.073	0.051
	[0.050]	[0.051]	[0.048]
Woman Has Ownership of House	0.067**	0.066**	0.067**
~ ~ ~ ~ ~ ~ ~ ~	[0.031]	[0.031]	[0.031]
Spouse Has Ownership of Land or Crops	0.001	-0.001	0.014
	[0.034]	[0.034]	[0.034]
Spouse Has Ownership of Car or Vehicle	0.090**	0.090**	0.086**
	[0.040]	[0.040]	[0.040]
Spouse Has Ownership of Savings	0.046	0.046	0.044
	[0.043]	[0.042]	[0.044]
Spouse Has Ownership of House	0.037	0.034	0.032
	[0.027]	[0.027]	[0.027]
Household Has No Ownership of Land or Crops	-0.055**	-0.058**	-0.048**
	[0.023]	[0.023]	[0.022]
Household Has No Ownership of Car or Vehicle	0.093***	0.092***	0.080***
	[0.028]	[0.028]	[0.028]
Household Has No Ownership of Savings	-0.027	-0.026	-0.018
	[0.034]	[0.034]	[0.033]
Household Has No Ownership of House	0.012	0.014	0.010
	[0.020]	[0.020]	[0.020]
Household Equivalent Income (Scaled)	-0.020	-0.018	-0.004
	[0.026]	[0.026]	[0.026]
Woman Earns 51% or More of Household Income	-0.042**	-0.043**	-0.044**
	[0.021]	[0.021]	[0.021]
Woman's Acceptability of Wife Beating Index		0.026***	0.025***
		[0.007]	[0.007]
Woman's Strength of Patriarchal/Traditional Beliefs		-0.152***	-0.122***
		[0.045]	[0.045]
Violence in Women's Family Background			0.188***
			[0.015]
Violence in Spouse's Family Background			0.184***
			[0.019]
Unknown Violence in Spouse's Background			0.063***
			[0.017]
Observentions	10.001	10.001	10 001
Observations	10,801	10,801	10,801
Controls Included $D_{\text{cond}} = D^2$	1 es 0.0772	1 es	res 0.150
r seudo R	0.0773	0.0812	0.150

Table A.8: Sequential Addition of Sets of Variables: Married and Common-Law Women Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Psychological Violence

	Set: Resources	Sets: Resources	Sets: Resources,
	(1)	(2)	(3)
	0.000***	0.000***	0.001***
Woman Has Money for Personal Expenses	-0.092***	-0.092***	-0.081***
	[0.018]	[0.018]	[0.019]
Woman Has Ownership of Land or Crops	-0.113***	-0.117***	-0.116***
	[0.032]	[0.032]	[0.031]
Woman Has Ownership of Car or Vehicle	0.063	0.066	0.085
	[0.056]	[0.056]	[0.059]
Woman Has Ownership of Savings	0.091*	0.090*	0.068
	[0.053]	[0.053]	[0.052]
Woman Has Ownership of House	0.099***	0.098***	0.101***
	[0.028]	[0.029]	[0.029]
Spouse Has Ownership of Land or Crops	-0.051*	-0.053*	-0.046*
	[0.029]	[0.028]	[0.028]
Spouse Has Ownership of Car or Vehicle	0.062^{*}	0.062*	0.055
	[0.037]	[0.037]	[0.036]
Spouse Has Ownership of Savings	0.025	0.026	0.025
	[0.040]	[0.040]	[0.041]
Spouse Has Ownership of House	0.053^{**}	0.051^{**}	0.054^{**}
	[0.025]	[0.025]	[0.025]
Household Has No Ownership of Land or Crops	-0.052**	-0.056**	-0.047**
	[0.022]	[0.022]	[0.022]
Household Has No Ownership of Car or Vehicle	0.057^{*}	0.057^{*}	0.043
	[0.032]	[0.032]	[0.031]
Household Has No Ownership of Savings	-0.033	-0.031	-0.026
	[0.035]	[0.035]	[0.035]
Household Has No Ownership of House	0.047^{**}	0.049^{**}	0.046^{**}
	[0.020]	[0.020]	[0.020]
Household Equivalent Income (Scaled)	-0.034	-0.032	-0.018
	[0.025]	[0.025]	[0.026]
Woman Earns 51% or More of Household Income	-0.014	-0.014	-0.014
	[0.022]	[0.022]	[0.021]
Woman's Acceptability of Wife Beating Index		0.022^{***}	0.021^{***}
		[0.007]	[0.007]
Woman's Strength of Patriarchal/Traditional Beliefs		-0.165^{***}	-0.133***
		[0.044]	[0.045]
Violence in Women's Family Background			0.177^{***}
			[0.016]
Violence in Spouse's Family Background			0.211^{***}
			[0.017]
Unknown Violence in Spouse's Background			0.083^{***}
			[0.017]
	10.001	10.001	10.001
Observations	10,801	10,801	10,801
Controls Included	Yes	Yes	Yes
Pseudo R ²	0.0540	0.0575	0.126

Table A.9: Sequential Addition of Sets of Variables: Married and Common-Law Women Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Sexual Violence

	Set: Resources	Sets: Resources & Beliefs	Sets: Resources, Beliefs & Child Violence
	(1)	(2)	(3)
Woman Has Money for Personal Expenses	-0.043***	-0.043***	-0.035***
· ·	[0.011]	[0.011]	[0.010]
Woman Has Ownership of Land or Crops	-0.026*	-0.028**	-0.024*
	[0.015]	[0.014]	[0.012]
Woman Has Ownership of Car or Vehicle	0.056	0.056	0.055
	[0.052]	[0.051]	[0.051]
Woman Has Ownership of Savings	0.076**	0.076**	0.063*
	[0.036]	[0.036]	[0.033]
Woman Has Ownership of House	0.046**	0.046**	0.042**
	[0.018]	[0.018]	[0.017]
Spouse Has Ownership of Land or Crops	-0.019	-0.019	-0.014
	[0.014]	[0.014]	[0.013]
Spouse Has Ownership of Car or Vehicle	0.056*	0.056*	0.049*
	[0.029]	[0.029]	[0.027]
Spouse Has Ownership of Savings	0.029	0.029	0.024
	[0.027]	[0.026]	[0.023]
Spouse Has Ownership of House	0.017	0.017	0.015
	[0.015]	[0.015]	[0.014]
Household Has No Ownership of Land or Crops	-0.026**	-0.028**	-0.022**
Haushald Has Na Ormanskin of Can an Vakiala	[0.012]	[0.012]	[0.011]
Household has No Ownership of Car or Venicle	0.035	0.035	[0.010]
Household Has No Owmership of Sovings	[0.011]	0.022*	[0.010]
Household has No Ownership of Savings	0.021	[0.025]	[0.023
Household Has No Ownership of House	[0.014]	[0.014]	[0.012]
Household has no Ownership of House	[0.002	[0.010]	[0.002
Household Equivalent Income (Scaled)	0.011		0.010
Household Equivalent Income (Scaled)	-0.018	-0.017	-0.011
Woman Farns 51% or More of Household Income	0.004	0.008	0.008
woman Earns 51% of more of nousehold income	-0.008	-0.008	[0 010]
Woman's Acceptability of Wife Beating Index	[0.011]	0.003	0.003
woman's Acceptability of whe beating fidex		[0.003]	[0,003]
Woman's Strength of Patriarchal/Traditional Beliefs		-0.080***	-0.062***
woman's Strength of Fatharenar/ Haditional Deneis		[0.023]	[0 021]
Violence in Women's Family Background		[0.025]	0.047***
Violence in Women's Faining Dackground			[0 007]
Violence in Spouse's Family Background			0.055***
Violence in Spouse's Fanniy Dackground			[0,009]
Unknown Violence in Spouse's Background			0.005
Chanown violence in opplace's Daeaground			[0 008]
			[0.000]
Observations	10.801	10.801	10.801
Controls Included	Yes	Yes	Yes
Pseudo R^2	0.0737	0.0784	0.125

Table A.10: Sequential Addition of Sets of Variables: Married and Common-Law Women Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Economic Violence

	Set: Resources	Sets: Resources	Sets: Resources,
	(1)	(2)	(3)
Woman Has Money for Personal Expenses	-0.027***	-0.028***	-0.020**
	[0.010]	[0.009]	[0.008]
Woman Has Ownership of Land or Crops	-0.011	-0.013	-0.011
	[0.017]	[0.016]	[0.013]
Woman Has Ownership of Car or Vehicle	0.088*	0.088*	0.081*
	[0.046]	[0.045]	[0.046]
Woman Has Ownership of Savings	0.041	0.040	0.028
	[0.028]	[0.028]	[0.023]
Woman Has Ownership of House	0.031*	0.031*	0.027*
	[0.017]	[0.016]	[0.015]
Spouse Has Ownership of Land or Crops	0.003	0.003	0.008
	[0.018]	[0.017]	[0.017]
Spouse Has Ownership of Car or Vehicle	0.046	0.045	0.035
	[0.028]	[0.028]	[0.025]
Spouse Has Ownership of Savings	0.003	0.003	0.001
	[0.016]	[0.016]	[0.014]
Spouse Has Ownership of House	0.017	0.017	0.015
	[0.015]	[0.014]	[0.013]
Household Has No Ownership of Land or Crops	-0.004	-0.005	-0.003
	[0.010]	[0.010]	[0.009]
Household Has No Ownership of Car or Vehicle	0.028**	0.028**	0.020*
	[0.011]	[0.011]	[0.011]
Household has No Ownership of Savings	0.000	0.007	0.009
Hannahald Haa Na Ormanahin af Hanna	[0.012]	[0.012]	[0.010]
Household has No Ownership of House	-0.002	-0.001	-0.001
Household Equivalent Income (Scaled)	[0.010]	[0.010]	[0.009]
Household Equivalent Income (Scaled)	-0.010	-0.010	-0.010
Wemen Ferra 5107 on More of Heusehold Income	[0.010]	[0.010]	[0.013]
woman Earns 51% of more of nousehold income	0.013	0.012	[0 010]
Woman's Accontability of Wife Beating Index	[0.011]	0.003	0.002
woman's Acceptability of whe beating findex		[0.003]	[0.002
Woman's Strongth of Patriarchal/Traditional Boliofs		0.003	0.053***
woman's Strength of Fatharchar/ Haditional Deneis		-0.072	-0.035
Violonco in Womon's Family Background		[0.019]	0.050***
Violence in Women's Fainity Dackground			[0,007]
Violonco in Spouso's Family Background			0.045***
Violence in Spouse's Fainity Dackground			[0,008]
Unknown Violonco in Spouso's Background			0.001
Chknown violence in Spouse's Dackground			[0,006]
			[0.000]
Observations	10 801	10.801	10 801
Controls Included	Yes	Yes	Yes
Pseudo R^2	0.0559	0.0608	0.119

Table A.11: Sequential Addition of Sets of Variables: Separated and Divorced Women

Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Physical Violence

	Set: Resources (1)	Sets: Resources & Beliefs (2)	Sets: Resources, Beliefs & Child. Violence (3)
Woman Had Money for Personal Expenses	-0.233***	-0.235***	-0.212***
Woman Had Ownership of Land or Crops	[0.045]	[0.045]	[0.045]
	-0.257**	-0.254**	- 0.287^{**}
	[0.122]	[0.125]	[0.126]
Woman Had Ownership of Car or Vehicle	$\begin{bmatrix} 0.123 \\ 0.021 \\ \begin{bmatrix} 0.197 \end{bmatrix}$	$\begin{bmatrix} 0.125 \end{bmatrix}$ 0.012 $\begin{bmatrix} 0.198 \end{bmatrix}$	[0.120] 0.104 [0.196]
Woman Had Ownership of Savings	[0.137] -0.017 [0.178]	-0.038 [0.182]	-0.094 [0.180]
Woman Had Ownership of House	0.018	[0.102] [0.022] [0.073]	0.030 [0.079]
Spouse Had Ownership of Land or Crops	-0.306**	-0.309***	-0.373***
	[0.120]	[0.119]	[0.107]
Spouse Had Ownership of Car or Vehicle	0.093	0.078	0.099
	[0.154]	[0.159]	[0.150]
Spouse Had Ownership of Savings	-0.119	-0.124	-0.135
	[0.177]	[0.179]	[0.182]
Spouse Had Ownership of House	-0.005	0.009	0.022
	[0.110]	[0.110]	[0.114]
Household Had No Ownership of Land or Crops	-0.133	-0.129	-0.158
	[0.099]	[0.100]	[0.111]
Household Had No Ownership of Car or Vehicle	-0.056	-0.061	-0.046
	[0.145]	[0.148]	[0.143]
Household Had No Ownership of Savings	0.031	0.009	-0.013
	[0.166]	[0.170]	[0.171]
Household Had No Ownership of House	-0.088	-0.085	-0.101
	[0.071]	[0.071]	[0.074]
Housenoid Equivalent income (Scaled)+	[0.083]	[0.084]	[0.103] [0.089] 0.027
Woman's Acceptability of whe beating index+		[0.024] [0.025] 0.272**	[0.024] 0.205**
Violence in Women's Family Background		[0.136]	[0.143] 0.153***
Violence in Spouse's Family Background			[0.043] 0.220***
Unknown Violence in Spouse's Background			$[0.045] \\ 0.046 \\ [0.050]$
Observations	1,401	1,401	1,401
Controls Included	Yes	Yes	Yes
Pseudo R^2	0.163	0.168	0.209

Table A.12: Sequential Addition of Sets of Variables: Separated and Divorced Women

Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Psychological Violence

	Set: Resources (1)	Sets: Resources & Beliefs (2)	Sets: Resources, Beliefs & Child. Violence (3)
Woman Had Money for Personal Expenses	-0.240***	-0.243***	-0.216***
Woman Had Ownership of Land or Crops	[0.040]	[0.040]	[0.040]
	-0.186	-0.186	-0.221
	[0.150]	[0.153]	[0.162]
Woman Had Ownership of Car or Vehicle	-0.291	-0.311	-0.252
	[0.200]	[0.200]	[0.238]
Woman Had Ownership of Savings	[0.200] 0.071 [0.172]	[0.200] 0.045 [0.182]	0.016 [0.197]
Woman Had Ownership of House	-0.025	-0.017	-0.013
	[0.074]	[0.074]	[0.078]
Spouse Had Ownership of Land or Crops	-0.296**	-0.304**	-0.391***
	[0.143]	[0.141]	[0.132]
Spouse Had Ownership of Car or Vehicle	$0.095 \\ [0.136]$	0.079 [0.143]	0.084 [0.147]
Spouse Had Ownership of Savings	0.181	0.178	0.175
	[0.122]	[0.127]	[0.132]
Spouse Had Ownership of House	-0.018	-0.004	0.010
	[0.118]	[0.118]	[0.116]
Household Had No Ownership of Car or Vehicle	-0.185^{**} [0.078] 0.062	-0.180^{**} [0.081]	-0.209** [0.087] 0.065
Household Had No Ownership of Savings	[0.144]	[0.147]	[0.153]
	0.009	-0.017	-0.028
Household Had No Ownership of House	[0.151]	[0.152]	[0.156]
	-0.073	-0.072	-0.085
Household Equivalent Income (Scaled)+	[0.068]	[0.068]	[0.070]
	-0.043	-0.044	- 0.050
Woman's Acceptability of Wife Beating Index+	[0.085]	[0.087] 0.020	[0.093] 0.024
Woman's Strength of Patriarchal/Traditional Beliefs+		[0.025] -0.370*** [0.124]	[0.024] -0.406*** [0.144]
Violence in Women's Family Background		[0.134]	[0.144] 0.166^{***} [0.042]
Violence in Spouse's Family Background			[0.043] 0.196^{***} [0.053]
Unknown Violence in Spouse's Background			[0.053] 0.031 [0.051]
Observations	1,401	1,401	1,401
Controls Included	Yes	Yes	Yes
Pseudo R^2	0.156	0.164	0.208

Table A.13: Sequential Addition of Sets of Variables: Separated and Divorced Women

Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Sexual Violence

	Set: Resources (1)	Sets: Resources & Beliefs (2)	Sets: Resources, Beliefs & Child. Violence (3)
Woman Had Money for Personal Expenses	-0.212***	-0.206***	-0.177***
Woman Had Ownership of Land or Crops	[0.042]	[0.042]	[0.042]
	-0.087	-0.083	-0.093
	[0.092]	[0.095]	[0.089]
Woman Had Ownership of Car or Vehicle	[0.002] 0.140 [0.193]	0.119 [0.193]	0.183 [0.198]
Woman Had Ownership of Savings	0.135	0.125	0.073
	[0.139]	[0.140]	[0.140]
Woman Had Ownership of House	-0.112**	-0.106*	-0.098
	[0.057]	[0.056]	[0.060]
Spouse Had Ownership of Land or Crops	-0.000	0.015	-0.025
	[0.109]	[0.109]	[0.102]
Spouse Had Ownership of Car or Vehicle	0.070	0.073	0.062
	[0.121]	[0.121]	[0.117]
Spouse Had Ownership of Savings	-0.033 [0.112] 0.070	-0.047 [0.107]	-0.059 [0.110]
Household Had No Ownership of Land or Crops	-0.070	-0.065	-0.055
	[0.075]	[0.075]	[0.075]
	-0.223**	-0.209**	-0.227**
Household Had No Ownership of Car or Vehicle	[0.092]	[0.092]	[0.104]
	0.016	0.022	0.014
Household Had No Ownership of Savings	[0.092]	[0.092]	[0.095]
	0.103	0.092	0.072
Household Had No Ownership of House	[0.080]	[0.084]	[0.092]
	-0.063	-0.068	-0.073
Household Equivalent Income (Scaled)+	[0.061]	[0.060]	[0.064]
	-0.070	-0.063	-0.057
	[0.082]	[0.084]	[0.084]
Woman's Acceptability of Wife Beating Index+	[0.082]	[0.034] -0.030 [0.021]	[0.034] -0.026 [0.020]
Woman's Strength of Patriarchal/Traditional Beliefs+		-0.214* [0.118]	-0.229* [0.117]
Violence in Women's Family Background			0.102*** [0.038]
Violence in Spouse's Family Background			0.208^{***} [0.041]
Unknown Violence in Spouse's Background			0.049 [0.041]
Observations	1,401	1,401	1,401
Controls Included	Yes	Yes	Yes
Pseudo R^2	0.159	0.167	0.211

Table A.14: Sequential Addition of Sets of Variables: Separated and Divorced Women

Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results Type of Violence: Economic Violence

	Set: Resources (1)	Sets: Resources & Beliefs (2)	Sets: Resources, Beliefs & Child. Violence (3)
Woman Had Money for Personal Expenses	-0.157***	-0.158***	-0.140***
Woman Had Ownership of Land or Crops	[0.039] 0.097 [0.116]	[0.039] 0.108 [0.118]	[0.039] 0.095 [0.110]
Woman Had Ownership of Car or Vehicle	0.193 [0.168]	0.176 [0.164]	[0.113] 0.224 [0.174]
Woman Had Ownership of Savings	0.028	0.004 [0.108]	-0.021 [0.107]
Woman Had Ownership of House	0.041 [0.070]	0.049 [0.071]	0.053 [0.072]
Spouse Had Ownership of Land or Crops	0.116 [0.132]	0.117 [0.130]	0.086 [0.125]
Spouse Had Ownership of Car or Vehicle	-0.135** [0.065]	-0.143** [0.063]	-0.153*** [0.059]
Spouse Had Ownership of Savings	$0.159 \\ [0.176]$	$0.150 \\ [0.176]$	$0.151 \\ [0.184]$
Spouse Had Ownership of House	-0.014 [0.095]	0.007 [0.098]	0.004 [0.093]
Household Had No Ownership of Land or Crops	0.008 [0.073]	0.018 [0.071]	0.009 [0.074]
Household Had No Ownership of Car or Vehicle	-0.073 [0.094]	-0.079 [0.095]	-0.105 [0.102]
Household Had No Ownership of House	-0.047 [0.096]	[0.099]	[0.107]
Household Equivalent Income (Scaled)+	[0.040] [0.061] -0.117	[0.043] [0.060] -0.110	[0.052 [0.060] -0.094
Woman's Accentability of Wife Beating Index+	[0.090]	[0.091] 0.017	[0.091] 0.024
Woman's Strength of Patriarchal/Traditional Beliefs+		[0.020] -0.290**	[0.019] -0.294**
Violence in Women's Family Background		[0.124]	[0.123] 0.079^{**}
Violence in Spouse's Family Background			[0.034] 0.124^{***}
Unknown Violence in Spouse's Background			[0.039] -0.040 [0.036]
Observations Controls Included Pseudo R^2	1,401 Yes 0.127	1,401 Yes 0.136	1,401 Yes 0.166

Inclusion of Widowed Women:

Table A.15:	Descriptive Statistics:	Mean	Values	by	Marital	Status,	Sample of	f 45-65	Year	Old	Women,
	Including Widowed W	omen									

	Married & Common Law Mean St. Dev. (1) (2)		Separated & Divorced Mean St. Dev. (3) (4)		Wic Wc Mean (5)	lowed omen St. Dev. (6)
	N O 10	F 00		.		
Woman's Age	53.12	5.69	53.35	5.48	57.73	5.65
Woman's Edu. Level: Basic Education	0.59	0.49	0.44	0.50	0.61	0.49
Woman's Edu. Level: Medium Education	0.17	0.38	0.21	0.41	0.09	0.28
Woman's Edu. Level: Technical or University Ed.	0.14	0.34	0.27	0.44	0.09	0.29
Woman's Activity: Employed	0.37	0.48	0.64	0.48	0.41	0.49
Woman's Activity: Unemployed	0.10	0.30	0.06	0.24	0.14	0.35
Woman's Activity: Not in Labour Force	0.53	0.50	0.30	0.46	0.45	0.50
Woman's Ethnic Identity: Mestiza	0.74	0.44	0.70	0.46	0.72	0.45
Woman's Ethnic Identity: Indigena	0.07	0.25	0.08	0.27	0.09	0.29
Woman's Ethnic Identity: Afrodescendent	0.06	0.23	0.08	0.27	0.05	0.23
Woman's Ethnic Identity: Other Ethnicity	0.14	0.34	0.14	0.34	0.13	0.34
Woman's Monthly Income (From All Sources)	138.68	294.50	237.78	285.50	174.01	295.69
Household Receives BDH Transfer	0.31	0.46	0.20	0.40	0.42	0.49
Household Income	737.37	706.22	651.16	646.14	438.76	451.67
Household Equivalent Income	355.45	350.35	326.78	305.17	229.28	211.53
Income Prop. to Household Income	18.51	23.92	48.26	40.10	47.28	39.91
Woman Earns 51% or More of Household Income	0.10	0.30	0.42	0.49	0.40	0.49
Woman Has Children	0.98	0.13	0.96	0.20	0.98	0.15
Woman's Number of Children	3.81	1.62	3.23	1.72	4.37	1.66
Age at First Birth	21.11	4.64	21.40	4.90	20.26	4.23
Woman Had First Child as Teenager	0.43	0.50	0.39	0.49	0.44	0.50
Age When First Legally Partnered	20.52	5.23	20.76	5.32	19.73	5.05
Woman Partnered as Teenager	0.50	0.50	0.47	0.50	0.58	0.49
Woman's Marriage/Relationship Length	32.60	7.57	16.27	10.84	24.99	11.46
Size of Family in Household	4.57	2.03	3.98	2.04	3.65	2.24
Household is Multi-Generational	0.29	0.45	0.00	0.49	0.40	0.49
Non-Family Members Live in Household	0.04	0.40	0.40	0.40	0.40	0.45
Male Head of Household	0.04	0.20	0.04	0.15	0.00	0.05
Household in Urban Area	0.55	0.21	0.15	0.35	0.67	0.25
Household III OTDall Alea	0.71	0.40	0.04	0.00	0.07	0.47
	N=3,038		N=523		N=357	

	Married or Common-Law (1)	Separated or or Divorced (2)	Widowed Women (3)
Woman Has/Had Money for Personal Expenses	-0.014	-0.328^{***}	-0.098
	[0.033]	[0.068]	[0.088]
All Assets Owned by Female	0.024	0.080	-0.379***
	[0.067]	[0.092]	[0.087]
All Assets Owned by Male	0.031	0.259^{**}	-0.350***
	[0.056]	[0.101]	[0.091]
All Assets Are Co-Owned	-0.021	0.224^{**}	-0.467^{***}
	[0.049]	[0.099]	[0.070]
Mixed Ownership of Assets	0.030	-0.054	-0.344^{***}
	[0.055]	[0.125]	[0.081]
Woman's Monthly Income (Scaled)+	-0.034	-0.073	-0.003
	[0.061]	[0.130]	[0.126]
Woman's Acceptability of Wife Beating Index+	0.037***	0.031	-0.020
	[0.012]	[0.034]	[0.032]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.079	-0.529**	0.379
0 1	[0.088]	[0.222]	[0.245]
Violence in Women's Family Background	0.203***	0.123	0.113
v o	[0.028]	[0.076]	[0.081]
Violence in Spouse's Family Background	0.199***	0.210**	0.135
1 0 0	[0.032]	[0.093]	[0.093]
Unknown Violence in Spouse's Background	0.100***	0.178**	-0.049
j i i i i i i i i i i i i i i i i i i i	[0.031]	[0.090]	[0.087]
	[0.00-]	[0.0000]	[0.00.]
Observations	3.038	523	357
Province Controls	Yes	Yes	Yes
Pseudo R^2	0.156	0.263	0.285

 Table A.16:
 Comparison: Sample of 45-65 Year Old Women - Type of Violence: Physical Violence

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results
M Cor	Iarried or mmon-Law (1)	Separated or or Divorced (2)	Widowed Women (3)
	0.000*	0.000	0.150*
Woman Has/Had Money for Personal Expenses	-0.068*	-0.279***	-0.170*
	[0.036]	[0.066]	[0.087]
All Assets Owned by Female	0.061	0.049	-0.343***
All Assets Osurad has Mala	[0.065]	[0.093]	[0.093]
All Assets Owned by Male	0.022	0.287	-0.234
All Arrete Are Co Ormed	[0.049]	[0.085]	0.200***
All Assets Are Co-Owned	-0.020	0.134	-0.389
Mirred Ormanship of Accesta	0.040	[0.107]	0.041*
Mixed Ownership of Assets	0.000	-0.005	-0.241 [0.197]
Women's Monthly Income (Scaled)	0.109*	[0.142]	0.029
woman's Montiny Income (Scaled)+	-0.108	-0.303	-0.056
Women's Accentability of Wife Posting Index	0.009]	[0.117]	0.020
woman's Acceptability of whe beating index+	[0.011]	0.034	0.050
Women's Strength of Detrienchel /Traditional Deliefe	0.001	0.030]	0.066
woman's Strength of Fatharchar/ fraditional beneis+	-0.008	-0.829	0.000
Violonce in Women's Femily Deckmound	[0.062]	[0.244]	[0.250] 0.141*
violence in women's ranny background	[0 029]	[0.072]	[0.080]
Violongo in Spougo's Family Background	[0.026]	[0.075]	[0.060] 0.179*
Violence in Spouse's Family Dackground	[0 022]	[0.002]	[0.000]
Unknown Vielence in Spouse's Background	[0.033]) 195***	[0.093]	0.066
Chknown violence in Spouse's Dackground	[0.021]	[0.086]	0.000
	[0.031]	[0.080]	[0.069]
Observations	3 038	523	350
Province Controls	Ves	Ves	Ves
Pseudo B^2	0 139	0 295	0.247

 Table A.17:
 Comparison: Sample of 45-65 Year Old Women - Type of Violence: Psychological Violence

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 45-65 years old, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has/Had No Money for Personal Expenses;' 'No Ownership of Assets;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Married or Common-Law (1)	Separated or or Divorced (2)	Widowed Women (3)
	a and shale	a successful.	
Woman Has/Had Money for Personal Expenses	-0.051**	-0.145**	-0.012
	[0.022]	[0.072]	[0.039]
All Assets Owned by Female	0.053	-0.017	-0.125***
	[0.039]	[0.084]	[0.041]
All Assets Owned by Male	0.058^{*}	0.371^{***}	-0.058
	[0.034]	[0.107]	[0.037]
All Assets Are Co-Owned	-0.001	0.170	-0.150^{***}
	[0.024]	[0.124]	[0.037]
Mixed Ownership of Assets	0.025	0.157	-0.076***
	[0.033]	[0.131]	[0.028]
Woman's Monthly Income (Scaled)+	-0.041	-0.159	0.110^{*}
	[0.030]	[0.116]	[0.061]
Woman's Acceptability of Wife Beating Index+	0.003	-0.027	-0.022
	[0.006]	[0.030]	[0.016]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.081**	-0.250	-0.212*
	[0.040]	[0.170]	[0.109]
Violence in Women's Family Background	0.045***	0.165***	0.056
v o	[0.014]	[0.056]	[0.037]
Violence in Spouse's Family Background	0.082***	0.249***	0.169***
	[0.019]	[0.072]	[0.055]
Unknown Violence in Spouse's Background	0.029*	0.024	0.037
1 0	[0.018]	[0.069]	[0.038]
		L J	LJ
Observations	3,038	523	339
Province Controls	Yes	Yes	Yes
Pseudo R^2	0.130	0.295	0.417

 Table A.18: Comparison: Sample of 45-65 Year Old Women - Type of Violence: Sexual Violence

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 45-65 years old, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has/Had No Money for Personal Expenses;' 'No Ownership of Assets;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

	Married or Common-Law (1)	Separated or or Divorced (2)	Widowed Women (3)
Woman Has/Had Money for Personal Expenses	-0.019	-0.266***	-0.033
	[0.014]	[0.071]	[0.043]
All Assets Owned by Female	0.018	0.050	-0.091***
	[0.024]	[0.075]	[0.026]
All Assets Owned by Male	0.003	0.211^{*}	-0.065***
	[0.018]	[0.117]	[0.021]
All Assets Are Co-Owned	-0.018	0.078	-0.152^{***}
	[0.015]	[0.104]	[0.033]
Mixed Ownership of Assets	-0.006	-0.128**	-0.078***
	[0.018]	[0.061]	[0.020]
Woman's Monthly Income (Scaled)+	-0.023	-0.037	-0.113*
	[0.027]	[0.107]	[0.066]
Woman's Acceptability of Wife Beating Index+	0.003	-0.003	0.015
	[0.004]	[0.028]	[0.013]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.039	-0.731***	-0.148
	[0.027]	[0.181]	[0.097]
Violence in Women's Family Background	0.049***	-0.019	0.002
	[0.011]	[0.061]	[0.036]
Violence in Spouse's Family Background	0.029**	0.132*	0.015
	[0.012]	[0.068]	[0.036]
Unknown Violence in Spouse's Background	-0.001	-0.067	-0.107***
1 0	[0.011]	[0.061]	[0.039]
		L J	L]
Observations	3,038	523	297
Province Controls	Yes	Yes	Yes
Pseudo R^2	0.130	0.278	0.434

 Table A.19:
 Comparison: Sample of 45-65 Year Old Women - Type of Violence: Economic Violence

 Marginal Effects from Probit Estimates of the Probability of Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 45-65 years old, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to the marginal effects from Probit estimates of the probability of experiencing intimate partner violence. The base categories that are omitted from the regression are: 'Has/Had No Money for Personal Expenses;' 'No Ownership of Assets;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

Table A.20: Change of Dependent Variable: Probability of Intimate Partner Violence Normalized by Length of Relationship: Married & Common-Law Women

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Has Money for Personal Expenses	-0.009***	-0.008**	-0.003***	-0.002*
Woman Has Ownership of Land or Crops	-0.010** [0.004]	-0.011** [0.005]	0.001	-0.002
Woman Has Ownership of Car or Vehicle	$\begin{bmatrix} 0.004 \end{bmatrix}$ 0.000 $\begin{bmatrix} 0.004 \end{bmatrix}$	0.006	$\begin{bmatrix} 0.004 \end{bmatrix}$ 0.001	$\begin{bmatrix} 0.003 \\ 0.003 \\ \begin{bmatrix} 0.002 \end{bmatrix}$
Woman Has Ownership of Savings	$\begin{bmatrix} 0.004 \end{bmatrix}$ 0.010 $\begin{bmatrix} 0.007 \end{bmatrix}$	0.008	$\begin{bmatrix} 0.002 \end{bmatrix}$ 0.003 $\begin{bmatrix} 0.002 \end{bmatrix}$	[0.003] 0.014 [0.008]
Woman Has Ownership of House	$\begin{bmatrix} 0.007 \\ 0.001 \\ \begin{bmatrix} 0.002 \end{bmatrix}$	0.002	0.003**	[0.003] 0.001 [0.002]
Spouse Has Ownership of Land or Crops	$\begin{bmatrix} 0.002 \end{bmatrix}$ 0.002 $\begin{bmatrix} 0.004 \end{bmatrix}$	-0.004	-0.001	$\begin{bmatrix} 0.002 \end{bmatrix}$ 0.000 $\begin{bmatrix} 0.002 \end{bmatrix}$
Spouse Has Ownership of Car or Vehicle	$\begin{bmatrix} 0.004 \end{bmatrix}$ 0.001	0.002	0.003**	0.002
Spouse Has Ownership of Savings	[0.005] 0.011* [0.006]	0.004	0.001	-0.001
Spouse Has Ownership of House	0.000	0.002	0.002	0.003
Household Has No Ownership of Land or Crops	-0.003	-0.003	-0.001	-0.000
Household Has No Ownership of Car or Vehicle	[0.003] 0.008***	[0.003] 0.008**	[0.001] 0.002***	[0.001] 0.004***
Household Has No Ownership of Savings	$\begin{bmatrix} 0.003 \\ 0.002 \\ \begin{bmatrix} 0.002 \end{bmatrix}$	[0.004] -0.002	$\begin{bmatrix} 0.001 \\ 0.002 \\ \begin{bmatrix} 0.001 \end{bmatrix}$	-0.001 [0.002]
Household Has No Ownership of House	0.006***	0.009***	0.000	0.002
Household Equivalent Income (Scaled)	$\begin{bmatrix} 0.002 \end{bmatrix}$ 0.004	0.003	-0.000	-0.002
Woman Earns 51% or More of Household Income	-0.005 -0.007**	-0.007** [0.002]	-0.002*** [0.001]	-0.002
Woman's Acceptability of Wife Beating Index	0.003**	0.002*	-0.000	0.002
Woman's Strength of Patriarchal/Traditional Beliefs	$\begin{bmatrix} 0.001 \\ 0.005 \\ \begin{bmatrix} 0.007 \end{bmatrix}$	-0.005	-0.004** [0.002]	-0.005* [0.002]
Violence in Women's Family Background	0.015***	0.017***	0.005***	0.005***
Violence in Spouse's Family Background	0.018***	0.019***	0.004***	0.005***
Unknown Violence in Spouse's Background	[0.003] [0.003]	[0.003] [0.003]	[0.001] -0.001 [0.001]	-0.001 [0.001]
Observations	10,801	10,801	10,801	10,801
R-squared Controls Included	0.113 Yes	0.107 Yes	0.037 Yes	0.044 Yes

Coefficients from OLS Estimates of the Exposure to Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to coefficients from an OLS regression on the exposure to intimate partner violence. The base categories that are omitted from the regression are: 'Has No Money for Personal Expenses;' 'Has Co-Ownership of Crops;' 'Has Co-Ownership of Vehicle;' 'Has No Ownership of Savings;' 'Has Co-Ownership of House;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

Table A.21: Change of Dependent Variable: Probability of Intimate Partner Violence Normalized by Length of Relationship: Separated & Divorced Women

Coefficients from (OLS Estimates	of the Exposure	to Intimate Partner	Violence - Main Results
---------------------	---------------	-----------------	---------------------	-------------------------

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Had Money for Personal Expenses	-0.008	-0.024	-0.019	-0.014
	[0.018]	[0.021]	[0.015]	[0.016]
Woman Had Ownership of Land or Crops	0.028	0.057	0.018	0.037
	[0.039]	[0.041]	[0.018]	[0.022]
Woman Had Ownership of Car or Vehicle	-0.004	0.040	-0.001	0.033
Women Hed Ownership of Servings	[0.095]	[0.063]	[0.020]	[0.028]
woman nad Ownersnip of Savings	-0.125	-0.128	0.051	[0.005]
Woman Had Ownership of House	0.024	0.028	0.001	$\begin{bmatrix} 0.027 \end{bmatrix}$ 0.017
woman mad ownership of nouse	[0.024]	[0.023]	[0.011]	[0.017]
Spouse Had Ownership of Land or Crops	-0.041	-0.053	0.013	-0.006
openant in a contract of the	[0.041]	[0.040]	[0.024]	[0.028]
Spouse Had Ownership of Car or Vehicle	0.000	0.045	-0.017	-0.036
	[0.094]	[0.054]	[0.020]	[0.026]
Spouse Had Ownership of Savings	-0.074	0.075	0.047	0.206*
	[0.106]	[0.138]	[0.030]	[0.122]
Spouse Had Ownership of House	0.053^{*}	0.048^{*}	-0.003	-0.011
	[0.030]	[0.029]	[0.014]	[0.021]
Household Had No Ownership of Land or Crops	0.008	-0.000	0.006	0.011
	[0.033]	[0.026]	[0.014]	[0.016]
Household Had No Ownership of Car or Vehicle	0.035	0.140**	0.010	0.035
Henry hell Hell N. O. analia (C. iana	[0.111]	[0.065]	[0.021]	[0.022]
Household Had No Ownership of Savings	-0.094	-0.133	0.010	-0.008
Household Had No Ownership of House	$\begin{bmatrix} 0.120 \end{bmatrix}$	[0.100] 0.028*	0.000	[0.022]
Household Had No Ownership of House	[0.027]	0.028	[0,009]	[0.011]
Household Equivalent Income (Scaled)+	0.059	0.039	-0.024	-0.029
Household Equivalent Income (Scaled)	[0.053]	[0.054]	[0.018]	[0.021]
Woman's Acceptability of Wife Beating Index+	0.010	0.009	0.003	0.008
······································	[0.008]	[0.008]	[0.006]	[0.006]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.056	-0.056	-0.093***	-0.064*
с ,	[0.046]	[0.056]	[0.035]	[0.038]
Violence in Women's Family Background	0.029	0.034	0.006	0.018
	[0.019]	[0.023]	[0.013]	[0.015]
Violence in Spouse's Family Background	0.025	0.007	0.019	0.000
	[0.017]	[0.027]	[0.012]	[0.018]
Unknown Violence in Spouse's Background	0.028*	0.020	0.021*	-0.001
	[0.016]	[0.018]	[0.011]	[0.016]
Observations	1,401	1,401	1,401	1,401
R-squared	0.254	0.266	0.180	0.173
Controls Included	Yes	Yes	Yes	Yes

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to coefficients from an OLS regression on the exposure to intimate partner violence. The base categories that are omitted from the regression are: 'Has No Money for Personal Expenses;' 'Has Co-Ownership of Crops;' 'Has Co-Ownership of Vehicle;' 'Has No Ownership of Savings;' 'Has Co-Ownership of House;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

Table A.22: Change of Dependent Variable: Probability of Intimate Partner Violence Normalized by Length of Relationship: Married & Common-Law Women Common-Law Women Common-Law Women

	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
			(-)	
Woman Has Money for Personal Expenses	-0.024***	-0.031***	-0.040***	-0.040***
v i	[0.008]	[0.009]	[0.010]	[0.015]
Woman Has Ownership of Land or Crops	-0.047***	-0.055***	-0.017	-0.033
	[0.015]	[0.017]	[0.024]	[0.036]
Woman Has Ownership of Car or Vehicle	0.003	0.033	0.050	0.105**
	[0.020]	[0.025]	[0.040]	[0.044]
Woman Has Ownership of Savings	0.027	0.027	0.059**	0.075^{*}
	[0.019]	[0.023]	[0.024]	[0.042]
Woman Has Ownership of House	0.024^{**}	0.034^{***}	0.044^{***}	0.044^{**}
	[0.010]	[0.010]	[0.015]	[0.022]
Spouse Has Ownership of Land or Crops	0.003	-0.022*	-0.019	0.014
	[0.013]	[0.013]	[0.019]	[0.030]
Spouse Has Ownership of Car or Vehicle	0.028^{**}	0.021	0.056^{**}	0.068*
	[0.014]	[0.015]	[0.023]	[0.036]
Spouse Has Ownership of Savings	0.028	0.012	0.022	-0.003
	[0.018]	[0.019]	[0.023]	[0.030]
Spouse Has Ownership of House	0.012	0.021**	0.016	0.028
	[0.009]	[0.009]	[0.015]	[0.022]
Household Has No Ownership of Land or Crops	-0.019**	-0.019**	-0.023**	-0.007
	[0.008]	[0.009]	[0.011]	[0.018]
Household Has No Ownership of Car or Vehicle	0.040^{***}	0.029**	0.044^{**}	0.057^{*}
	[0.012]	[0.014]	[0.017]	[0.030]
Household Has No Ownership of Savings	-0.001	-0.011	0.033^{*}	0.013
	[0.013]	[0.016]	[0.018]	[0.025]
Household Has No Ownership of House	0.014^{**}	0.028^{***}	0.003	0.001
	[0.007]	[0.008]	[0.011]	[0.018]
Household Equivalent Income (Scaled)	0.007	-0.000	-0.010	-0.019
	[0.012]	[0.011]	[0.015]	[0.027]
Woman Earns 51% or More of Household Income	-0.024**	-0.015	-0.016	0.010
	[0.009]	[0.009]	[0.013]	[0.018]
Woman's Acceptability of Wife Beating Index	0.011***	0.009***	0.003	0.004
	[0.003]	[0.003]	[0.003]	[0.006]
Woman's Strength of Patriarchal/Traditional Beliefs	-0.023	-0.045**	-0.074***	-0.110***
	[0.020]	[0.021]	[0.027]	[0.036]
Violence in Women's Family Background	0.076***	0.077***	0.063***	0.107***
	[0.008]	[0.009]	[0.011]	[0.017]
Violence in Spouse's Family Background	0.077***	0.086***	0.065***	0.087***
	[0.009]	[0.008]	[0.011]	[0.015]
Unknown Violence in Spouse's Background	0.029***	0.030***	0.003	-0.006
	[0.008]	[0.007]	[0.009]	[0.013]
	0.000**	0.10.1**	0.001***	0.070***
Constant	-0.083**	-0.104**	-0.321***	-0.373***
	[0.041]	[0.046]	[0.066]	[0.079]
Observations	10.001	10.901	10 001	10.001
Described D^2	10,601	10,801	10,801	10,601
Controls Included	0.525 Vec	0.247 Voc	0.205 Voc	0.162 Vec
	res	res	res	res

Coefficients from Tobit Estimates of the Exposure to Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to coefficients from an OLS regression on the exposure to intimate partner violence. The base categories that are omitted from the regression are: 'Has No Money for Personal Expenses;' 'Has Co-Ownership of Crops;' 'Has Co-Ownership of Vehicle;' 'Has No Ownership of Savings;' 'Has Co-Ownership of House;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

Table A.23: Change of Dependent Variable: Probability of Intimate Partner Violence Normalized by Length of Relationship: Separated & Divorced Women Comparison of
	Physical Violence (1)	Psychological Violence (2)	Sexual Violence (3)	Economic Violence (4)
Woman Had Monay for Porsonal Exponent	0.070**	0.078***	0 191***	0 111***
woman had money for Personal Expenses	-0.070**	-0.078	-0.121	-0.111
Warman Had Onmanship of Land on Chang	[0.028]	[0.026]	[0.055]	0.124
woman had Ownership of Land or Crops	-0.028	0.030	-0.024	0.124
Warran Had Ormanbin of Can an Valiala	[0.075]	[0.069]	[0.085]	[0.090]
woman Had Ownership of Car or Venicle	0.034	-0.034	0.075	0.109
Warran Had Ormania of Casimur	[0.154]	[0.134]	[0.121]	0.000
Woman Had Ownership of Savings	-0.182	-0.152	0.091	-0.020
	[0.169]	[0.154]	[0.106]	[0.107]
Woman Had Ownership of House	0.036	0.019	-0.056	0.064
	[0.034]	[0.036]	[0.052]	[0.064]
Spouse Had Ownership of Land or Crops	-0.164**	-0.151**	-0.013	0.048
	[0.072]	[0.063]	[0.081]	[0.093]
Spouse Had Ownership of Car or Vehicle	0.011	0.071	0.009	-0.203**
	[0.136]	[0.090]	[0.084]	[0.103]
Spouse Had Ownership of Savings	-0.109	0.110	0.060	0.351*
	[0.153]	[0.164]	[0.104]	[0.202]
Spouse Had Ownership of House	0.090*	0.069	-0.011	0.024
	[0.053]	[0.048]	[0.058]	[0.089]
Household Had No Ownership of Land or Crops	-0.021	-0.044	-0.096	0.043
	[0.062]	[0.048]	[0.062]	[0.064]
Household Had No Ownership of Car or Vehicle	0.014	0.169^{*}	0.016	-0.018
	[0.155]	[0.101]	[0.083]	[0.084]
Household Had No Ownership of Savings	-0.120	-0.170	0.062	-0.072
	[0.167]	[0.142]	[0.085]	[0.089]
Household Had No Ownership of House	0.015	0.015	-0.018	0.046
	[0.034]	[0.030]	[0.041]	[0.058]
Household Equivalent Income (Scaled)+	0.100	0.035	-0.080	-0.122
	[0.078]	[0.080]	[0.078]	[0.102]
Woman's Acceptability of Wife Beating Index+	0.018	0.014	-0.010	0.029
	[0.013]	[0.012]	[0.019]	[0.021]
Woman's Strength of Patriarchal/Traditional Beliefs+	-0.142*	-0.156^{*}	-0.280***	-0.332**
	[0.080]	[0.083]	[0.108]	[0.133]
Violence in Women's Family Background	0.078**	0.086**	0.071*	0.089**
	[0.031]	[0.034]	[0.042]	[0.043]
Violence in Spouse's Family Background	0.105***	0.062*	0.147***	0.087**
	[0.029]	[0.037]	[0.037]	[0.044]
Unknown Violence in Spouse's Background	0.062**	0.036	0.066*	-0.024
. 0	[0.031]	[0.029]	[0.036]	[0.043]
Constant	0.085	0.170	-0.337*	0.050
	[0.178]	[0.158]	[0.193]	[0.208]
	[0.110]	[0.100]	[0.100]	[0.200]
Observations	1 401	1 401	1 401	1 401
Pseudo \mathbb{R}^2	0.258	0.250	0.236	0.186
Controls Included	Yes	Yes	Yes	Yes

Coefficients from Tobit Estimates of the Exposure to Intimate Partner Violence - Main Results

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. + denotes current variables (at time of survey) as opposed to variables qualifying their past relationship. Sample consists of respondents in the ENVIGMU who are 15 years old or older, married, common-law, separated or divorced and widowed. Physical, Psychological, Sexual and Economic Violence are indicator variables of whether the woman experienced that type of violence. Estimates correspond to coefficients from an OLS regression on the exposure to intimate partner violence. The base categories that are omitted from the regression are: 'Had No Money for Personal Expenses;' 'Has Co-Ownership of Crops;' 'Has Co-Ownership of Vehicle;' 'Has No Ownership of Savings;' 'Has Co-Ownership of House;' 'No Violence in Own Family Background;' 'No Violence in Spouse's Family Background;' 'Woman's Ethnic Identity: White;' 'Woman's Highest Education: No Education,' 'Woman's Activity: Out of the Labour Force;' 'Household in Rural Area;' 'Province: Pichincha.'

Table A.24:	Correlations:	Beliefs	Justifying	Wife	Beating
-------------	---------------	---------	------------	------	---------

	Disobedient	Care Children	Unfaithful	Goes Out
Husband Justified in Hitting Wife if Disobedient	1.00			
Husband Justified in Hitting Wife if Improper Care of Children	0.59	1.00		
Husband Justified in Hitting Wife if Unfaithful	0.41	0.50	1.00	
Husband Justified in Hitting Wife if Goes Out Too Much	0.49	0.56	0.60	1.00
	.6 1			

Note: table depicts correlations for the set of beliefs justifying wife beating.

 ${\bf Table \ A.25: \ Correlations: \ Beliefs \ Patriarchal/Traditional \ Norms}$

	Obey	Friends	Expenses	Equal Right	Obligation Sex	
Good Wife Must Obey All Orders from Husband	1.00					
Wife Can Choose Friends Even Disliked by Husband	-0.07	1.00				
Men Should be Responsible for All Family Expenses	0.34	-0.01	1.00			
Women Have Equal Right to Work and Earn Money	-0.03	0.13	0.06	1.00		
Wife's Obligation to Have Sex Even if Unwanted	0.23	0.01	0.15	-0.03	1.00	
Note: table depicts correlations for the set beliefs regarding traditional/patriarchal norms.						

Table A.26:	Correlations:	Strength	of Beliefs	Patriarchal	/Traditional	Norms

	Obey	Friends	Expenses	Equal Right	Obligation Sex
Strength: Good Wife Must Obey All Orders from Husband	1.00				
Strength: (Negative) Wife Can Choose Friends if Disliked by Husband	0.06	1.00			
Strength: Man Should be Responsible for All Family Expenses	0.38	-0.02	1.00		
Strength: (Negative) Women Have Equal Right to Work and Earn Money	0.03	0.21	-0.09	1.00	
Strength: It's a Wife's Obligation to Have Sex Even if Unwanted	0.26	-0.02	0.16	0.04	1.00
Notes to ble denister completions for the not of considered anistic state strong	1	+:			

Note: table depicts correlations for the set of variables depicting the strength traditional/patriarchal norms.

Appendix B

Children's Well-Being



Figure B.1: Children's Hours Spent Working in Chores by Gender

Source: National Survey of Child Labour



Figure B.2: Child's Share of Total Household Hours of Chores by Total Household Hours of Chores Performed by All Members

Source: National Survey of Child Labour



Figure B.3: Child's Share of Children's Total Household Hours of Chores by Total Household Hours of Chores Performed by All Children

Source: National Survey of Child Labour

	All Children (1)	Male Children (2)	Female Children (3)
Child's Total Hours of Household Chores (Scaled)	-0.021*	-0.038**	-0.018
	[0.012]	[0.019]	[0.014]
Child's Total Hours of Household Chores (Squared, Scaled)	0.000	0.005 [0.006]	-0.001 [0.003]
Child's Total Hours in Labour Market Activities (Scaled)	-0.029***	-0.033***	-0.022**
	[0.006]	[0.007]	[0.009]
Child Attends School	0.136***	0.151***	0.120**
Age Group: 12 to 15	-0.000	0.018	-0.018
	[0.010]	[0.013]	[0.014]
Age Group: 15 to 17	-0.001	0.012	-0.015 [0.025]
Oldest Child	-0.005	-0.009	0.000
	[0.010]	[0.014]	[0.014]
Identifies as: Indigenous	0.004	-0.005	0.009
Identifies as: Afro-descendant	[0.016]	[0.020] -0.055	[0.019] -0.012
identifies as. mito-descendant	[0.029]	[0.038]	[0.038]
Identifies as: Other Ethnicity	-0.026	-0.017	-0.038
	[0.026]	[0.033]	[0.034]
Boy	-0.007	-	-
Dual-Earner Household	-0.011	0.010	-0.034**
	[0.012]	[0.015]	[0.015]
Mother's Educ.: Primary	0.008	0.034	-0.022
Mothor's Educ - Secondary	[0.021]	[0.025] 0.066**	[0.026]
Mother's Educ.: Secondary	[0.023]	[0.027]	[0.030]
Mother's Educ.: University	0.022	0.028	0.019
	[0.028]	[0.035]	[0.034]
Father's Educ.: Primary	0.017	0.002	0.027
Father's Educ.: Secondary	0.016	0.001	0.032 0.022
	[0.030]	[0.036]	[0.035]
Father's Educ.: University	0.060**	0.073**	0.040
Household Equivalent Income (Cooled)	[0.027]	[0.029]	[0.036]
Household Equivalent Income (Scaled)	-0.014 [0.027]	-0.038	0.014
Household Size	0.006	0.018**	-0.005
	[0.006]	[0.007]	[0.008]
House has Children Below Age 5	-0.022	-0.033*	-0.011
House has Other Boys Aged 5-17	[0.015] -0.022*	-0.040**	0.000
House has owner boys riged o fr	[0.013]	[0.018]	[0.020]
House has Other Girls Aged 5-17	-0.015	-0.039**	0.012
	[0.012]	[0.017]	[0.019]
Household Has a Member with A Disability	-0.027 [0.022]	-0.025 [0.028]	-0.026
Head of Household is Female	0.018	0.005	0.033
	[0.038]	[0.052]	[0.040]
Urban Setting	-0.015	-0.004	-0.025*
Parent or PMK Present During Child's Responses	[0.014] 0.033***	0.033**	[0.015] 0.030*
	[0.013]	[0.016]	[0.016]
Observations Province Controls	14,810 Vac	7,516 Vac	7,294 Vaa
Pseudo R^2	0.0499	0.0585	0.0576

 Table B.1: Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variable: Total Hours of Household Work & Total Hours of Household Work Squared (Self-Reported)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, *p<0.1 denote levels of significance. Sample consists of respondents in the ENTI ages 8-17, living with both parents. The dependent variable, *happiness*, is an indicator variables equal to one if child feels happy most of the time. Estimates correspond to the marginal effects for the probability of the child self-assessing as happy. The base categories are: 'Child Doesn't Attend School;' 'Age: 8-11;' 'Not Oldest Child;' 'Child's Ethnicity: Mestizo;' 'Non-Dual Earners;' 'Less than Primary Educ.;' 'No Additional Members'; 'Male Head of Household;' 'Household in Rural Area;' 'PMK not Present.' 'Province: Pichincha' omitted.

All Children Children (1) Male Children (2) Female (3) Child Works in Domestic Activities -0.045*** -0.045*** -0.033* Child Works in Market Activities -0.041*** -0.015*** -0.033* Child Atends School 0.013 [0.013] [0.024] Child Atends School 0.017*** 0.101*** -0.055*** Age Group: 12 to 15 -0.002 [0.013] [0.014] Age Group: 15 to 17 -0.007 0.010 -0.023 Oldest Child -0.011 -0.002 [0.013] [0.014] Identifies as: Indigenous 0.014 0.006 [0.013] [0.013] Identifies as: Cher Ethnicity -0.022 -0.013 [0.013] [0.024] Boy -0.016 -0.018 [0.033] [0.033] [0.034] Identifies as: Cher Ethnicity -0.022 -0.016 -0.024 -0.018 Identifies as: Cher Ethnicity -0.021 -0.021 -0.021 -0.021 Dual-Earner Household -0.023 [0.023] [0.023] [0.033] <th></th> <th></th> <th></th> <th></th>				
Child Works in Domestic Activities -0.040^{***} -0.045^{***} -0.033^* Child Works in Market Activities -0.01^{***} -0.01^{***} -0.05^{****} -0.01^{***} Child Attends School 0.170^{***} 0.011^{**} 0.005^{***} 0.021^{**} Age Group: 12 to 15 -0.002^{***} 0.002^{**} 0.022^{*} Age Group: 15 to 17 -0.007^{***} -0.007^{***} 0.001^{**} -0.002^{**} Oldest Child -0.001^{***} -0.001^{**} -0.002^{**} 0.014^{**} -0.002^{**} Identifies as: Indigenous 0.014^{**} -0.002^{**} 0.014^{**} -0.006^{**} -0.022^{**} Identifies as: Other Ethnicity 0.029^{**} 0.014^{**} -0.020^{**} -0.030^{**} Identifies as: Other Ethnicity 0.020^{**} 0.031^{**} -0.030^{**} -0.040^{**} Identifies as: Other Ethnicity 0.022^{**} 0.031^{**} -0.021^{**} 0.022^{**} Identifies as: Other Ethnicity 0.020^{**} 0.021^{**} 0.021^{**} 0.021^{**} 0.021^{**} 0.021^{**} 0.021^{**} 0.021^{**}		All Children (1)	Male Children (2)	Female Children (3)
Child Works in Market Activities $[0.012]$ $[0.015]$ $[0.015]$ $[0.015]$ Child Works in Market Activities -0.084^{+++} -0.101^{+++} -0.053^{++} Child Attends School 0.177^{+++} 0.101^{++} 0.0123 $[0.023]$ Child Attends School 0.177^{+++} 0.101^{++} 0.0123 $[0.023]$ Age Group: 12 to 15 0.002 0.017^{-} 0.002 0.017^{-} 0.00224 Oldest Child -0.007^{-} 0.001^{-} -0.022^{-} 0.0225 Oldest Child -0.001^{-} -0.001^{-} -0.001^{-} -0.002^{-} Identifies as: Indigenous 0.014^{-} 0.006^{-} 0.013^{-} 0.001^{-} Identifies as: Afro-descendant 0.026^{-} 0.003^{-} 0.003^{-} 0.003^{-} Identifies as: Other Ethnicity 0.026^{-} 0.010^{-} $ 0.0009^{-}$ Dual-Earner Household 0.003^{-} 0.032^{-} 0.032^{-} 0.032^{-} 0.032^{-} Mother's Educ.: Primary 0.032^{-} 0.00	Child Works in Domestic Activities	-0.040***	-0.045***	-0.033*
Child Works in Market Activities -0.084^{+++}_{++} -0.005^{+++}_{++} -0.005^{+++}_{++} Child Attends School 0.170^{+++}_{++} 0.161^{+++}_{++} 0.0231 0.0024 Age Group: 12 to 15 0.0017 0.0022 0.017 -0.002 Age Group: 15 to 17 0.017 0.0021 0.0231 0.0023 Oldest Child 0.0017 0.0021 0.0233 0.017 Identifies as: Indigenous 0.014 -0.006 0.017 0.0221 0.0231 Identifies as: Other Ethnicity -0.026 -0.016 - - - Identifies as: Other Ethnicity -0.026 -0.011 - - - Dual-Earner Household -0.008 0.013 -0.032 - - Mother's Educ:: Secondary 0.023 0.0232 0.0331 0.0332 Mother's Educ:: University 0.023 0.0232 0.0331 0.0322 Mother's Educ:: University 0.026 0.033 0.0322 0.0331 Father's Educ:: University 0.066		[0.012]	[0.015]	[0.018]
0.018 0.023 0.024 Age Group: 12 to 15 0.002 0.017*** 0.161*** Age Group: 12 to 15 0.002 0.013 0.021 Age Group: 15 to 17 0.007 0.002 0.013 0.022 Oldest Child 0.007 0.002 0.013 0.014 Age Group: 15 to 17 0.001 0.014 0.006 0.012 Oldest Child 0.001 0.014 0.006 0.017 Identifies as: Indigenous 0.014 0.006 0.018 0.018 Identifies as: Other Ethnicity 0.026 0.033 0.033 Boy 0.026 0.033 0.033 Dual-Earner Household -0.026 0.031 -0.026 Mother's Educ:: Primary 0.031 0.0221 [0.032] Mother's Educ:: University 0.025 0.038 0.032 Mother's Educ:: University 0.026 0.033 0.032 Mother's Educ:: University 0.026 0.033 0.032 Household Equivalent Income (Scaled)	Child Works in Market Activities	-0.084***	-0.101***	-0.065***
Child Attends School 0.170^{++} 0.170^{++} 0.170^{++} 0.161^{++} Age Group: 12 to 15 -0.002 0.017 -0.022 0.017 -0.022 Age Group: 15 to 17 -0.007 0.010 -0.023 0.017 -0.022 Oldest Child -0.010 -0.013 $[0.013]$ $[0.013]$ $[0.013]$ Identifies as: Indigenous 0.0144 0.0006 0.0118 $[0.013]$ $[0.013]$ Identifies as: Afro-descendant -0.026 -0.0016 -0.0040 $[0.026]$ $[0.033]$ $[0.033]$ Identifies as: Other Ethnicity -0.026 -0.016 -0.040 $[0.026]$ $[0.033]$ $[0.033]$ Boy -0.001 $ [0.026]$ $[0.033]$ $[0.033]$ Dual-Earner Household -0.008 0.013 -0.027 $[0.022]$ $[0.024]$ $[0.026]$ $[0.030]$ Mother's Educ:: University 0.032 $[0.027]$ $[0.032]$ $[0.032]$ $[0.032]$ $[0.032]$ $[0.033]$ </td <td></td> <td>[0.018]</td> <td>[0.023]</td> <td>[0.024]</td>		[0.018]	[0.023]	[0.024]
Age Group: 12 to 15 $[0.030]$ $[0.034]$ $[0.032]$ Age Group: 15 to 17 $[0.010]$ $[0.013]$ $[0.014]$ Oldest Child $[0.017]$ $[0.022]$ $[0.025]$ Oldest Child $[0.013]$ $[0.014]$ $[0.006]$ $[0.013]$ Identifies as: Indigenous $[0.014]$ $[0.006]$ $[0.013]$ $[0.014]$ Identifies as: Other Ethnicity $[0.026]$ $[0.033]$ $[0.033]$ Identifies as: Other Ethnicity $[0.026]$ $[0.033]$ $[0.034]$ Boy $[0.026]$ $[0.033]$ $[0.034]$ Boy $[0.026]$ $[0.033]$ $[0.034]$ Boy $[0.026]$ $[0.033]$ $[0.034]$ Dual-Earner Household $[0.026]$ $[0.033]$ $[0.034]$ Mother's Educ:: Primary $[0.021]$ $[0.024]$ $[0.024]$ Mother's Educ:: Secondary $[0.022]$ $[0.032]$ $[0.032]$ Mother's Educ:: University $[0.023]$ $[0.032]$ $[0.033]$ Father's Educ:: University $[0.026]$ $[0.033]$ $[0.033]$ Father's Educ:: University $[0.026]$ $[0.033]$ $[0.034]$ Household Equivalent Income (Scaled) -0.016 -0.041 -0.041 Household Size $0.006^{\circ\circ}$ $0.066^{\circ\circ}$ $0.076^{\circ\circ}$ Household Size 0.005 $0.017^{\circ\circ}$ -0.032 Household Size 0.005 $0.017^{\circ\circ}$ -0.021 Household Size 0.006 $0.017^{\circ\circ}$ -0.021 Household Size 0.006 $0.017^{\circ\circ}$ -0.022 <t< td=""><td>Child Attends School</td><td>0.170***</td><td>0.177^{***}</td><td>0.161***</td></t<>	Child Attends School	0.170***	0.177^{***}	0.161***
lage Group: 15 to 17 $[0.010]$ $[0.013]$ $[0.010]$ $[0.013]$ $[0.010]$ Age Group: 15 to 17 -0.007 0.010 -0.023 Oldest Child -0.010 -0.014 -0.002 Identifies as: Indigenous 0.014 0.006 0.017 Identifies as: Afro-descendant -0.029 -0.043 -0.008 Identifies as: Other Ethnicity -0.026 -0.016 -0.040 Identifies as: Other Ethnicity -0.026 -0.016 -0.040 Identifies as: Other Ethnicity -0.026 -0.016 -0.040 Identifies as: Cher Ethnicity -0.026 -0.016 -0.042 Identifies as: Cher Ethnicity -0.026 -0.016 -0.021 Identifies Ethnic: University 0.032 -0.032 -0.032 Identifies Educ:: Primary 0.018 -0.001 -0.032 Identifies Ethnic:: University 0.026 -0.031 -0.032 Identifies Educ:: University 0.026 -0.031 -0.032 Identifies Educ:: University 0.066^{**} 0.037 $-0.$	Age Group: 12 to 15	-0.002	$\begin{bmatrix} 0.045 \end{bmatrix}$ 0.017	[0.052] =0.022
Age Group: 15 to 17 -0.007 -0.010^2 -0.002^2 Oldest Child -0.010 -0.014 -0.006 Identifies as: Indigenous 0.010^2 0.022^2 0.022^2 Identifies as: Afro-descendant -0.029 -0.043 -0.008 Identifies as: Other Ethnicity -0.026^2 -0.016^2 -0.040^2 Identifies as: Other Ethnicity -0.026^2 -0.040^2 -0.040^2 Identifies as: Other Ethnicity -0.026^2 -0.016^2 -0.040^2 Identifies as: Other Ethnicity -0.026^2 -0.016^2 -0.040^2 Identifies as: Other Ethnicity -0.026^2 -0.010^2 -0.040^2 Identifies as: Other Ethnicity -0.026^2 -0.010^2 -0.021^2 Mother's Educ:: Primary 0.013^2 0.022^* -0.021^2 Mother's Educ:: University 0.022^2^2 0.038^2^2 $0.013^2^2^2^2^2^2^2^2^2^2^2^2^2^2^2^2^2^2^2$	Age 610up. 12 to 10	[0.010]	[0.017]	[0.014]
Oldest Child $[0.017] = [0.022] = [0.025] 0.010 - 0.014 - 0.006 [0.009] = [0.013] = [0.014] 1.0014 - 0.006 & 0.017 [0.015] = [0.018] = [0.018] [0.015] = [0.018] = [0.018] [0.028] = [0.034] - 0.008 = [0.033] = [0.034] - 0.008 = [0.033] = [0.034] = [0.038] [0.026] = [0.034] = [0.033] [0.034] = [0.033] [0.034] = [0.033] = [0.034] = [0.009] [0.009] [0.009] [0.009] [0.009] [0.009] [0.001] [0.009] [0.001] [0.001] [0.002] = [0.015] = [0.015] [0.015] = [0.015] [0.015] = [0.015] [0.015] = [0.015] [0.015] = [0.015] [0.015] = [0.015] [0.016] - 0.008 - 0.013 - 0.032* + - 0.015 [0.021] = [0.021] = [0.021] [0.022] = [0.033] [0.022] = [0.033] = [0.032] [0.032] = [0.033] [0.023] = [0.033] = [0.033] [0.023] = [0.033] = [0.033] [0.024*0.015 = 0.015 [0.032] = [0.033] = [0.033] [0.032] = [0.033] [0.032] = [0.033] = [0.033] [0.032] = [0.033] = [0.033] [0.032] = [0.033] = [0.033] [0.032] = [0.033] = [0.033] [0.033] = [0.033] = [0.033] [0.034] [0.027] = [0.033] = [0.033] = [0.034] [0.028] = [0.030] = [0.033] = [0.034] [0.027] = [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.032] = [0.033] = [0.034] [0.033] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [0.034] [0.036] = [$	Age Group: 15 to 17	-0.007	0.010	-0.023
Oldest Child -0.010 -0.014 -0.006 Identifies as: Indigenous 0.014 0.006 0.017 Identifies as: Afro-descendant -0.029 -0.013 -0.008 Identifies as: Other Ethnicity -0.026 -0.016 -0.004 Boy -0.026 -0.013 -0.033 [0.034] Boy -0.026 -0.016 -0.040 -0.028 Dual-Earner Household -0.008 0.013 -0.032** Mother's Educ:: Primary 0.013 -0.022* -0.021 Mother's Educ:: Secondary 0.025 [0.030] [0.033] Mother's Educ:: University 0.025 [0.032] [0.032] Father's Educ:: University 0.016 -0.006 0.028 Identifies as: Other Boys Aged 5-17 0.016 -0.006 0.028 Household Size -0.021 [0.031] [0.041] [0.041] Household Size -0.023 -0.034* -0.001 Household Size -0.024 -0.031 -0.032 Household Size		[0.017]	[0.022]	[0.025]
	Oldest Child	-0.010	-0.014	-0.006
Identifies as: Indigenous 0.004 0.003 0.0013 Identifies as: Afro-descendant -0.028 -0.033 -0.008 Identifies as: Other Ethnicity -0.026 -0.016 -0.040 Boy -0.001 - - Dual-Earner Household -0.003 -0.003 - Dual-Earner Household -0.001 - - Dual-Earner Household -0.002 -0.015 [0.026] Mother's Educ:: Primary 0.013 -0.026 -0.015 Mother's Educ:: Secondary 0.025 0.038 0.015 Mother's Educ:: University 0.027 [0.032] [0.031] Father's Educ:: Primary 0.018 -0.001 - Father's Educ:: University 0.012 [0.022] [0.032] Father's Educ:: Primary 0.018 -0.001 - 0.017 Household Equivalent Income (Scaled) -0.014 -0.014 -0.014 -0.014 Household Size -0.028 -0.028* -0.002 -0.024* -0.002 <	Identifies on Indiannaus	[0.009]	[0.013]	[0.014]
Identifies as: Afro-descendant $[0.023]$ $[0.033]$ $[0.034]$ Identifies as: Other Ethnicity $[0.026]$ $[0.033]$ $[0.034]$ Boy $[0.026]$ $[0.033]$ $[0.034]$ Boy $[0.026]$ $[0.033]$ $[0.034]$ Dual-Earner Household $[0.026]$ $[0.013]$ $[0.024]$ Dual-Earner Household $[0.012]$ $[0.015]$ $[0.015]$ Mother's Educ:: Primary $[0.022]$ $[0.023]$ $[0.026]$ Mother's Educ:: Secondary $[0.022]$ $[0.033]$ $[0.026]$ Mother's Educ:: Primary $[0.022]$ $[0.033]$ $[0.032]$ Father's Educ:: Primary $[0.027]$ $[0.032]$ $[0.034]$ Father's Educ:: Secondary $[0.027]$ $[0.032]$ $[0.031]$ Father's Educ:: Secondary $[0.026]$ $[0.031]$ $[0.034]$ Household Equivalent Income (Scaled) $[0.027]$ $[0.032]$ $[0.034]$ Household Size $[0.036]$ $[0.034]$ $[0.041]$ $[0.041]$ Household Size $[0.036]$ $[0.037]$ $[0.034]$ $[0.041]$ <t< td=""><td>identifies as: indigenous</td><td>0.014</td><td>0.000</td><td>0.017</td></t<>	identifies as: indigenous	0.014	0.000	0.017
Identifies as: Other Ethnicity $[0.028]$ $[0.034]$ $[0.038]$ Boy -0.026 -0.016 -0.040 Boy -0.001 $ [0.009]$ -0.001 $ -$ Dual-Earner Household -0.008 0.013 -0.032^{**} Mother's Educ.: Primary 0.013 0.042^* -0.021 $[0.021]$ $[0.022]$ $[0.035]$ $[0.036]$ Mother's Educ.: Secondary 0.032 0.074^{***} -0.015 $[0.027]$ $[0.023]$ $[0.026]$ $[0.038]$ Mother's Educ.: Vniversity $[0.025]$ 0.038 0.015 $[0.027]$ $[0.032]$ $[0.031]$ $[0.032]$ Father's Educ.: Secondary $[0.027]$ $[0.032]$ $[0.031]$ Father's Educ.: Secondary $[0.027]$ $[0.032]$ $[0.031]$ Father's Educ.: Secondary 0.016^{**} 0.06^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 -0.014 Household Size 0.006^{**} 0.07^{**} -0.002 House has Other Boys Aged 5-17 -0.022^{**} -0.033^{**} -0.002 House has Other Boys Aged 5-17 -0.022^{**} -0.034^{**} -0.013 House has Other Boys Aged 5-17 -0.016^{**} -0.024^{**} -0.032^{**} House has Other Boys Aged 5-17 -0.022^{**} -0.033^{**} -0.031 House has Other Boys Aged 5-17 -0.016^{**} -0.031^{**} -0.032^{**} House had of Household is Female 0.037^{**} <t< td=""><td>Identifies as: Afro-descendant</td><td>-0.029</td><td>-0.043</td><td>-0.008</td></t<>	Identifies as: Afro-descendant	-0.029	-0.043	-0.008
Identifies as: Other Ethnicity -0.026 -0.016 -0.040 Boy 0.003 $[0.033]$ $[0.034]$ Dual-Earner Household -0.008 0.013 -0.032^{**} Mother's Educ.: Primary $[0.012]$ $[0.013]$ $[0.024]$ Mother's Educ.: Secondary $[0.023]$ $[0.026]$ $[0.033]$ Mother's Educ.: University $[0.023]$ $[0.024]$ $[0.026]$ Mother's Educ.: University $[0.023]$ $[0.026]$ $[0.030]$ Father's Educ.: Primary 0.032 0.038 0.015 Father's Educ.: University $[0.027]$ $[0.032]$ $[0.036]$ Father's Educ.: University 0.016 -0.006 0.028 Father's Educ.: University $[0.026]$ $[0.030]$ $[0.034]$ Household Equivalent Income (Scaled) -0.014 -0.014 0.047 Household Size $[0.036]$ $[0.037]$ $[0.038]$ $[0.040]$ House has Children Below Age 5 -0.023 -0.034^{**} -0.002 House has Other Boys Aged 5-17 -0.022^{*} -0.039^{**} -0.013		[0.028]	[0.034]	[0.038]
Boy 0.033 [0.034] [0.034] Boy -0.001 - - Dual-Earner Household -0.008 0.013 -0.032** Mother's Educ.: Primary 0.013 0.042* -0.021 Mother's Educ.: Secondary 0.032 0.074*** -0.015 Mother's Educ.: University 0.022 0.033 0.032 Father's Educ.: University 0.021 [0.021] [0.033] Father's Educ.: University 0.018 -0.003 0.032 Father's Educ.: University 0.016 -0.006 0.028 Father's Educ.: University 0.016 -0.006 0.028 Father's Educ.: University 0.026 [0.030] [0.034] Household Equivalent Income (Scaled) -0.014 -0.041 0.015 Household Size -0.023 -0.034* -0.002 House has Children Below Age 5 -0.023 -0.034* -0.002 House has Other Boys Aged 5-17 -0.024 -0.039** -0.002 House has Other Girls Aged 5-17 -0.024 <td>Identifies as: Other Ethnicity</td> <td>-0.026</td> <td>-0.016</td> <td>-0.040</td>	Identifies as: Other Ethnicity	-0.026	-0.016	-0.040
Boy -0.001 - - IDual-Earner Household [0.009] - - Mother's Educ.: Primary 0.013 0.015 [0.015] Mother's Educ.: Secondary 0.032 0.074*** -0.015 Mother's Educ.: University 0.022 0.033 0.042* Mother's Educ.: University 0.023 [0.023] [0.033] Mother's Educ.: Primary 0.016 -0.006 0.028 Father's Educ.: Secondary 0.016 -0.006 0.028 Father's Educ.: University 0.016 -0.006 0.028 Ioo260 [0.033] [0.034] [0.034] Father's Educ.: University 0.066** 0.047 [0.026] Ioo261 [0.033] [0.034] [0.041] [0.014] Household Equivalent Income (Scaled) -0.023 -0.034** -0.013 Iouse has Children Below Age 5 -0.023 -0.039** -0.013 Iouse has Other Boys Aged 5-17 -0.016 -0.039** -0.011 House has Other Girls Aged 5-17	_	[0.026]	[0.033]	[0.034]
Dual-Earner Household -0.008 0.013 -0.032^{**} Mother's Educ.: Primary 0.013 0.042^* -0.021 Mother's Educ.: Secondary 0.032 0.074^{***} -0.015 Mother's Educ.: University 0.032 0.074^{***} -0.015 Mother's Educ.: University 0.027 $[0.032]$ $[0.033]$ Father's Educ.: Primary 0.018 -0.003 0.032 Father's Educ.: Secondary $[0.027]$ $[0.032]$ $[0.033]$ Father's Educ.: University 0.016 -0.006 0.028 Father's Educ.: University 0.016 -0.006 0.028 Father's Educ.: University 0.06^{**} 0.06^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 0.015 Household Size 0.006^{**} 0.007^{*} 0.0034^{**} -0.003 House has Children Below Age 5 -0.023 -0.034^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.028^{*} 0.017^{*} Household Has a Member With A Disability -0.029 -0.024 -0.032 Household Has a Member With A Disability -0.015 -0.006 -0.038^{**} Head of Household is Female 0.021 0.036^{**} 0.037^{**} 0.033^{**} Province Controls -0.034^{**} 0.037^{**} 0.033^{**} 0.033^{**} Province Controls -0.042^{**} 0.037^{**} 0.033^{*	Boy	-0.001	-	-
Duar-Lainer Household $[0.013]$ $[0.013]$ $[0.013]$ Mother's Educ.: Primary $[0.021]$ $[0.021]$ $[0.021]$ Mother's Educ.: Secondary $[0.023]$ $[0.026]$ $[0.030]$ Mother's Educ.: University $[0.027]$ $[0.023]$ $[0.026]$ $[0.030]$ Mother's Educ.: University $[0.027]$ $[0.032]$ $[0.035]$ Father's Educ.: Secondary $[0.027]$ $[0.032]$ $[0.031]$ Father's Educ.: Secondary $[0.027]$ $[0.032]$ $[0.031]$ Father's Educ.: University 0.016 -0.006 0.028 $[0.027]$ $[0.032]$ $[0.036]$ $[0.031]$ Father's Educ.: University 0.066^{**} 0.066^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 0.015 House has Children Below Age 5 -0.023 -0.034^{**} -0.006 House has Other Boys Aged 5-17 -0.022^{**} -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.012 House had Other Girls Aged 5-17 -0.029 -0.024	Dual Farmar Household	[0.009]	0.012	0.029**
Mother's Educ.: Primary 0.0131 0.042^{**} -0.021 Mother's Educ.: Secondary 0.032 0.0241 $[0.026]$ Mother's Educ.: University 0.025 0.038 0.015 Mother's Educ.: University 0.025 0.038 0.015 Father's Educ.: Primary 0.016 -0.006 0.0321 Father's Educ.: Secondary 0.016 -0.006 0.0232 Father's Educ.: University 0.066^{**} 0.016 -0.006 Father's Educ.: University 0.066^{**} 0.047 0.0321 $[0.031]$ Household Equivalent Income (Scaled) -0.014 -0.014 -0.014 -0.014 House has Children Below Age 5 -0.023 -0.034^{*} -0.003 House has Other Boys Aged 5-17 -0.022^{*} -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.014 -0.039^{**} -0.019 House has Other Girls Aged 5-17 -0.021^{*} -0.039^{**} -0.011 House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.012 Head of Household is Female	Dual-Earlier Household	-0.008	0.015	-0.052**
Mother's Educ.: Secondary $[0.021]$ $[0.024]$ $[0.026]$ Mother's Educ.: University 0.032 0.074^{***} -0.015 $[0.027]$ $[0.032]$ $[0.032]$ $[0.035]$ Father's Educ.: Primary 0.018 -0.003 0.032 $[0.027]$ $[0.032]$ $[0.031]$ $[0.032]$ $[0.032]$ Father's Educ.: Secondary 0.016 -0.006 0.028 $[0.029]$ $[0.032]$ $[0.036]$ $[0.034]$ Father's Educ.: University 0.066^{**} 0.066^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 0.015 Household Size 0.005 0.017^{**} -0.006 House has Children Below Age 5 -0.023 -0.034^{**} -0.013 House has Other Boys Aged 5-17 -0.022^{*} -0.039^{**} -0.022 House has Other Girls Aged 5-17 -0.022^{*} -0.039^{**} -0.021 House has Other Girls Aged 5-17 -0.022^{*} -0.039^{**} -0.012 Household Has a Member With A Disability -0.022^{*} -0.032^{**} -0.032^{**} Urban Setting -0.015 -0.006 -0.023^{**} -0.013^{**} Iteration Condon of the seponses $[0.037]$ $[0.018]$ $[0.017]$ Observations $[0.014]$ $[0.018]$ $[0.015]$ $[0.017]$ $[0.013]$ Province Controls Pa_2^{**} Pa_2^{**} Pa_2^{**} Pa_2^{**} Pa_2^{**} Descretoric Place of Place During Child's Responses 0.036^{***} <td>Mother's Educ.: Primary</td> <td>0.012</td> <td>0.042^{*}</td> <td>-0.021</td>	Mother's Educ.: Primary	0.012	0.042^{*}	-0.021
Mother's Educ.: Secondary $0.032'$ 0.074^{***} $-0.015'$ Mother's Educ.: University $0.025'$ $0.038'$ $0.015'$ Father's Educ.: Primary $0.027'$ $[0.032]$ $[0.035]$ Father's Educ.: Secondary $0.016''''$ $-0.006''''''''''''''''''''''''''''''''''$		[0.021]	[0.024]	[0.026]
	Mother's Educ.: Secondary	0.032	0.074***	-0.015
Mother's Educ.: University 0.025 0.038 0.015 Father's Educ.: Primary 0.027 $[0.027]$ $[0.032]$ $[0.033]$ Father's Educ.: Secondary 0.016 -0.006 0.028 Father's Educ.: University 0.066^{**} 0.066^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.041 0.015 Household Size 0.005 0.0077 $[0.034]$ $[0.040]$ House hold Size 0.005 0.017^{**} -0.006 House has Children Below Age 5 -0.023 -0.034^{**} -0.013 House has Other Boys Aged 5-17 -0.022^{**} -0.039^{***} -0.021^{**} House has Other Girls Aged 5-17 -0.014^{**} -0.022^{**} -0.022^{**} -0.022^{**} Head of Household is Female 0.021^{**} $[0.017]^{**}$ $[0.022]^{**}$ $[0.031]^{**}$ Urban Setting -0.015^{**} -0.006^{**} 0.031^{**} 0.032^{**} Urban Setting -0.016^{**} 0.031^{**} 0.032^{**} 0.032^{**} Urban Setting -0.016^{**} $0.037^$		[0.023]	[0.026]	[0.030]
Father's Educ.: Primary $[0.027]$ $[0.032]$ $[0.032]$ $[0.032]$ Father's Educ.: Secondary 0.018 -0.003 0.032 Father's Educ.: University 0.016 -0.006 0.028 $[0.029]$ $[0.036]$ $[0.034]$ Father's Educ.: University 0.066^{**} 0.067^{**} $[0.029]$ $[0.036]$ $[0.034]$ Household Equivalent Income (Scaled) -0.014 -0.014 -0.014 -0.014 -0.015 $[0.027]$ $[0.034]$ $[0.040]$ House has Children Below Age 5 -0.023 -0.034^{**} -0.023 -0.034^{**} -0.013 House has Other Boys Aged 5-17 -0.022^{**} -0.039^{**} -0.024 -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.012 $[0.017]$ $[0.012]$ $[0.017]$ $[0.020]$ House has Other Girls Aged 5-17 -0.022^{**} -0.039^{**} $[0.012]$ $[0.017]$ $[0.020]$ Household Has a Member With A Disability -0.029^{**} -0.032^{**} $[0.022]$ $[0.028]$ $[0.031]$ $[0.037]$ $[0.037]$ $[0.038]$ -0.015^{**} -0.032^{**} $[0.013]$ $[0.014]$ $[0.018]$ $[0.015]$ Parent or PMK Present During Child's Responses 0.036^{***} 0.037^{**} $[0.013]$ $[0.016]$ $[0.017]$ $[0.033^{**}$ $[0.014]$ $[0.018]$ $[0.017]$ $[0.017]$ Province Controls Yes YesYes	Mother's Educ.: University	0.025	0.038	0.015
Pather's Educ.: Frinkly 0.013 0.032 0.032 Father's Educ.: Secondary 0.016 -0.006 0.028 Father's Educ.: University 0.060^{**} 0.066^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.041 0.015 Household Size 0.005 0.007 $[0.034]$ Household Size 0.005 0.017^{**} -0.014 House has Children Below Age 5 -0.023 -0.034^{**} House has Other Boys Aged 5-17 -0.022^{**} -0.039^{***} House has Other Girls Aged 5-17 -0.022^{**} -0.039^{***} House has Other Girls Aged 5-17 -0.012^{**} -0.011^{**} House has Other Girls Aged 5-17 -0.022^{**} -0.039^{***} House has Other Girls Aged 5-17 -0.029^{**} -0.017^{**} House has Other Girls Aged 5-17 -0.016^{**} -0.012^{**} House has Other Girls Aged 5-17 -0.029^{**} -0.039^{***} House has Other Girls Aged 5-17 -0.029^{**} -0.032^{**} Head of Household is Female 0.021^{**} 0.031^{**} Head of Household is Female 0.021^{**} 0.032^{**} Head of Household is Female 0.021^{**} 0.037^{**} Head of Portrols 0.036^{***} 0.037^{**} <t< td=""><td>Father's Educ - Primary</td><td>[0.027]</td><td>[0.032]</td><td>[0.035]</td></t<>	Father's Educ - Primary	[0.027]	[0.032]	[0.035]
Father's Educ.: Secondary 0.016 -0.006 0.028 Father's Educ.: University 0.066** 0.066** 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 0.015 Household Size 0.005 0.017** -0.006 Household Size 0.005 0.017** -0.006 House has Children Below Age 5 -0.023 -0.034* -0.006 House has Other Boys Aged 5-17 -0.023 -0.034** -0.002 House has Other Boys Aged 5-17 -0.016 -0.039** -0.002 Household Has a Member With A Disability -0.021 -0.023 -0.032** -0.002 Head of Household is Female 0.021 0.006 0.039** 0.011 Head of Household is Female 0.021 0.006 -0.039 0.031 Urban Setting -0.015 -0.006 -0.023 -0.033 Ivan Setting -0.016 -0.039** -0.032 Ivan Setting -0.015 -0.006 -0.039 Ivan Setting -0.016 -0.023 [0.031] Parent or PMK Present During Child's Responses <t< td=""><td>rations Educ., r milary</td><td>[0.027]</td><td>[0.032]</td><td>[0.031]</td></t<>	rations Educ., r milary	[0.027]	[0.032]	[0.031]
Father's Educ.: University $\begin{bmatrix} 0.029 \\ 0.066^{**} & 0.066^{**} & 0.047 \\ 0.026 \\ 0.030 \\ 0.034 \end{bmatrix}$ Household Equivalent Income (Scaled) -0.014 -0.014 0.015 Household Size 0.005 0.017^{**} -0.006 House has Children Below Age 5 -0.023 -0.034^{**} -0.013 House has Other Boys Aged 5-17 $[0.014]$ $[0.019]$ $[0.019]$ House has Other Girls Aged 5-17 -0.022^{**} -0.039^{**} -0.012 Household Has a Member With A Disability -0.029^{**} -0.021^{**} -0.032^{**} Head of Household is Female $[0.021]$ $[0.032]$ $[0.032]$ Urban Setting -0.015^{**} -0.006^{***} 0.037^{**} Parent or PMK Present During Child's Responses $14,810$ $7,516$ $7,294$ Province ControlsYesYesYesYesPresude R^2 0.0556 0.0261^{**} 0.0551 0.0556 0.0621 0.0621^{**} 0.0556^{**} 0.0551 0.0552^{**}	Father's Educ.: Secondary	0.016	-0.006	0.028
Father's Educ.: University 0.060^{**} 0.066^{**} 0.047 Household Equivalent Income (Scaled) -0.014 -0.014 0.015 Household Size 0.005 0.017^{**} -0.006 House has Children Below Age 5 0.005 0.017^{**} -0.008 House has Other Boys Aged 5-17 -0.022^{*} -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.022^{*} -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.014 -0.013^{*} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.002 House has Other Girls Aged 5-17 0.012^{*} $[0.017]$ $[0.019]$ Household Has a Member With A Disability -0.029^{*} -0.032^{*} $[0.017]$ $[0.031]$ Head of Household is Female 0.021 0.006^{*} 0.032^{*} $[0.031]$ $[0.016]$ $[0.013]$ Urban Setting -0.015^{*} -0.006^{*} 0.033^{**} $[0.013]$ $[0.016]^{*}$ $[0.013]^{*}$ 0.033^{**} <		[0.029]	[0.036]	[0.034]
$ 0.026 $ $ 0.030 $ $ 0.034 $ Household Equivalent Income (Scaled) -0.014 -0.041 0.015 $ 0.027 $ $ 0.034 $ $ 0.040 $ Household Size 0.005 0.017^{**} -0.006 $ 0.006 $ $ 0.007 $ $ 0.008 $ House has Children Below Age 5 -0.023 -0.034^* -0.013 $ 0.014 $ $ 0.019 $ $ 0.019 $ $ 0.022 $ $ 0.022^*$ -0.039^{**} -0.002 House has Other Boys Aged 5-17 -0.016 -0.039^{**} -0.002 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.011 $ 0.017 $ $ 0.019 $ Household Has a Member With A Disability -0.022 -0.024 -0.032 $ 0.012 $ $ 0.017 $ $ 0.013 $ $ 0.017 $ $ 0.032 $ $ 0.022 $ $ 0.028 $ $ 0.33 $ $ 0.031 $ $ 0.033 $ Head of Household is Female 0.021 0.006 $0.039 $ $ 0.013 $ $ 0.015 $ $ 0.033 $ $ 0.015 $ $ 0.033 $ $ 0.55 $ $ 0.033 $ $ 0.016 $ $ 0.017 $ $ 0.033^{**} $ $ 0.013 $ $ 0.016 $ $ 0.017 $ $ 0.033^{**} $ $ 0.016 $ $ 0.017 $ $ 0.014 $ $ 0.018 $ $ 0.015 $ $ 0.033^{**} $ $ 0.033^{**} $ $ 0.033^{**} $ $ 0.016 $ $ 0.017 $ $ 0.016 $ $ 0.017 $ $ 0.016 $ $ 0.017 $ $ 0.012 $ $ 0.016 $ $ 0.015 $ $ 0.016 $ $ 0.017 $ $ 0.03 $ $ 0.016 $ <td< td=""><td>Father's Educ.: University</td><td>0.060**</td><td>0.066**</td><td>0.047</td></td<>	Father's Educ.: University	0.060**	0.066**	0.047
Household Equivalent Income (scaled) -0.014 -0.014 -0.014 -0.013 Household Size $[0.027]$ $[0.034]$ $[0.040]$ House has Children Below Age 5 -0.005 0.017^{**} -0.006 House has Other Boys Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.014 $[0.017]$ $[0.020]$ House has Other Girls Aged 5-17 -0.016 -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.011 Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses 0.036^{***} 0.037^{**} 0.033^{**} $[0.013]$ $[0.016]$ $[0.017]$ $[0.017]$ $[0.017]$ Observations $14,810$ $7,516$ $7,294$ Province ControlsYesYesYesYesPresude R^2 0.0556 0.0621 0.0555	Household Equivalent Income (Cooled)	[0.026]	[0.030]	[0.034]
Household Size $[0.037]$ $[0.037]$ $[0.006]$ House has Children Below Age 5 -0.023 -0.034^* -0.013 House has Other Boys Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.017 Household Has a Member With A Disability -0.022 -0.024 -0.032 Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses 0.036^{***} 0.037^{**} 0.037^{**} Observations $14,810$ $7,516$ $7,294$ Province Controls Yes Yes Yes Presende R^2 0.0556 0.0621 0.0555	Household Equivalent Income (Scaled)	-0.014	-0.041	0.015
House has Children Below Age 5 $\begin{bmatrix} 0.006 \\ -0.023 \\ -0.023 \\ -0.034^* \\ -0.013 \\ \begin{bmatrix} 0.014 \\ 0.019 \end{bmatrix} \\ \begin{bmatrix} 0.019 \\ 0.022 \\ -0.039^{**} \\ -0.002 \\ \begin{bmatrix} 0.013 \\ 0.017 \end{bmatrix} \\ \begin{bmatrix} 0.020 \\ 0.020 \end{bmatrix} \end{bmatrix}$ House has Other Boys Aged 5-17 $-0.022^* \\ -0.022^* \\ -0.039^{**} \\ 0.011 \\ \begin{bmatrix} 0.017 \\ 0.019 \end{bmatrix} \\ \begin{bmatrix} 0.017 \\ 0.022 \end{bmatrix} \\ \begin{bmatrix} 0.028 \\ 0.031 \end{bmatrix} \\ \begin{bmatrix} 0.031 \\ 0.022 \end{bmatrix} \\ \begin{bmatrix} 0.028 \\ 0.032 \end{bmatrix} \\ \begin{bmatrix} 0.031 \\ 0.033 \end{bmatrix} \\ \begin{bmatrix} 0.037 \\ 0.052 \end{bmatrix} \\ \begin{bmatrix} 0.038 \\ 0.038 \end{bmatrix} \\ Urban Setting \\ -0.015 \\ -0.006 \\ -0.023 \\ \begin{bmatrix} 0.014 \\ 0.018 \\ 0.015 \end{bmatrix} \\ \begin{bmatrix} 0.015 \\ 0.033^{**} \\ 0.037^{**} \\ 0.037^{**} \\ 0.037^{**} \\ 0.033^{**} \\ \begin{bmatrix} 0.013 \\ 0.016 \end{bmatrix} \\ \begin{bmatrix} 0.017 \\ 0.017 \end{bmatrix} \\ \begin{bmatrix} 0.013 \\ 0.016 \end{bmatrix} \\ \begin{bmatrix} 0.017 \\ 0.017 \end{bmatrix} \\ \begin{bmatrix} 0.018 \\ 0.017 \end{bmatrix} \\ \begin{bmatrix} 0.013 \\ 0.016 \end{bmatrix} \\ \begin{bmatrix} 0.017 \\ 0.058 \\ 0.056 \\ 0.0656 \end{bmatrix} \\ \begin{bmatrix} 0.0621 \\ 0.055 \\ 0.0556 \end{bmatrix} $	Household Size	0.005	0.017**	-0.006
House has Children Below Age 5 -0.023 -0.034^* -0.013 House has Other Boys Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 -0.013 $[0.013]$ $[0.017]$ $[0.020]$ House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.011 Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses 0.036^{***} 0.037^{**} 0.037^{**} Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes		[0.006]	[0.007]	[0.008]
	House has Children Below Age 5	-0.023	-0.034*	-0.013
House has Other Boys Aged 5-17 -0.022^* -0.039^{**} -0.002 House has Other Girls Aged 5-17 $[0.013]$ $[0.017]$ $[0.020]$ House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.011 House has Other Girls Aged 5-17 -0.016 -0.039^{**} 0.011 Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses 0.034^{**} 0.037^{**} 0.033^{**} Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes Presende R^2 0.0556 0.0651 0.0555		[0.014]	[0.019]	[0.019]
House has Other Girls Aged 5-17 $[0.013]$ $[0.017]$ $[0.020]$ House has Other Girls Aged 5-17 $[0.012]$ $[0.017]$ $[0.019]$ Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female $[0.022]$ $[0.022]$ $[0.033]$ Urban Setting -0.015 -0.006 -0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses $[0.013]$ $[0.016]$ $[0.017]$ Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes	House has Other Boys Aged 5-17	-0.022*	-0.039**	-0.002
House has Order Onts right 6-11 -0.0010 -0.0010 $0.017]$ $[0.017]$ $[0.019]$ Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses 0.037^{**} 0.037^{**} 0.033^{**} Observations $14,810$ $7,516$ $7,294$ Province Controls Yes Yes Yes Presude R^2 0.0566 0.0621 0.0555	House has Other Cirls Aged 5-17	-0.013	-0.030**	0.011
Household Has a Member With A Disability -0.029 -0.024 -0.032 Head of Household is Female $[0.022]$ $[0.023]$ $[0.031]$ Head of Household is Female 0.021 0.006 0.039 Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses $[0.013]$ $[0.016]$ $[0.017]$ Observations $[14,810$ $7,516$ $7,294$ Province Controls Yes Yes Yes Presude R^2 0.0566 0.0621 0.0555	House has other Gills Aged 6-17	[0.012]	[0.017]	[0.019]
$ \begin{array}{c cccc} [0.022] & [0.028] & [0.031] \\ \hline \\ \mbox{Head} of Household is Female & 0.021 & 0.006 & 0.039 \\ [0.037] & [0.052] & [0.038] \\ \hline \\ \mbox{Urban Setting} & -0.015 & -0.006 & -0.023 \\ [0.014] & [0.018] & [0.015] \\ \hline \\ \mbox{Parent or PMK Present During Child's Responses} & 0.037^{**} & 0.037^{**} & 0.033^{**} \\ [0.013] & [0.016] & [0.017] \\ \hline \\ \mbox{Observations} & 14,810 & 7,516 & 7,294 \\ \hline \\ \mbox{Province Controls} & Yes & Yes & Yes \\ \hline \\ \mbox{Pesende} & R^2 & 0.0556 & 0.0621 & 0.0555 \\ \hline \end{array} $	Household Has a Member With A Disability	-0.029	-0.024	-0.032
Head of Household is Female 0.021 0.006 0.039 Urban Setting $[0.037]$ $[0.052]$ $[0.038]$ Urban Setting -0.015 -0.006 -0.023 $[0.014]$ $[0.018]$ $[0.015]$ $[0.033^{**}$ Parent or PMK Present During Child's Responses 0.037^{**} 0.037^{**} 0.033^{**} Observations $[0.016]$ $[0.016]$ $[0.017]$ Observations $14,810$ $7,516$ $7,294$ Province Controls Yes Yes Yes Province R ² 0.0556 0.0621 0.0555		[0.022]	[0.028]	[0.031]
urban Setting $[0.037]$ $[0.052]$ $[0.038]$ Urban Setting -0.015 -0.006 -0.023 Parent or PMK Present During Child's Responses $[0.014]$ $[0.018]$ $[0.015]$ Observations $[0.013]$ $[0.016]$ $[0.017]$ Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes Pescude R^2 0.0556 0.0621 0.0555	Head of Household is Female	0.021	0.006	0.039
Orban Setting -0.015 -0.023 Parent or PMK Present During Child's Responses $\begin{bmatrix} 0.014 \end{bmatrix}$ $\begin{bmatrix} 0.018 \end{bmatrix}$ $\begin{bmatrix} 0.015 \end{bmatrix}$ Parent or PMK Present During Child's Responses $\begin{bmatrix} 0.014 \end{bmatrix}$ $\begin{bmatrix} 0.018 \end{bmatrix}$ $\begin{bmatrix} 0.015 \end{bmatrix}$ Observations $\begin{bmatrix} 0.013 \end{bmatrix}$ $\begin{bmatrix} 0.016 \end{bmatrix}$ $\begin{bmatrix} 0.017 \end{bmatrix}$ Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes Pescude B^2 0.0556 0.0621 0.0555	Unber Cotting	[0.037]	[0.052]	[0.038]
Parent or PMK Present During Child's Responses $\begin{bmatrix} [0.014] & [0.016] & [0.013] \\ 0.036^{**} & 0.037^{**} & 0.033^{**} \\ [0.013] & [0.016] & [0.017] \end{bmatrix}$ Observations 14,810 7,516 7,294 Province Controls Yes Yes Yes Pseudo B^2 0.0556 0.0621 0.0555	orban Setting	-0.015	-0.006	-0.023
Image: constraint of the point of	Parent or PMK Present During Child's Responses	0.036***	0.037**	0.033**
Observations14,8107,5167,294Province ControlsYesYesYesPeriode B^2 0.05060.06510.0555		[0.013]	[0.016]	[0.017]
Observations $14,810$ $7,516$ $7,294$ Province ControlsYesYesYesPenudo B^2 0.05060.06910.0555				
Province Controls Yes Yes Yes Yes $P_{\text{cend}} = \frac{P_{\text{cend}}}{P_{\text{cend}}} = \frac{P_{\text{cend}}}{P$	Observations	14,810	7,516	7,294
	Province Controls Pseudo R^2	Yes 0.0506	Yes 0.0621	Yes 0.0555

 Table B.2: Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variables: Child Performed Chores; Child Worked in Labour Market; Child Attended School (During Current Week, Self-Reported)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, *p<0.1 denote levels of significance. Sample consists of respondents in the ENTI ages 8-17, living with both parents. The dependent variable, *happiness*, is an indicator variables equal to one if child feels happy most of the time. Estimates correspond to the marginal effects for the probability of the child self-assessing as happy. The base categories are: 'Child Doesn't Attend School;' 'Age: 8-11;' 'Not Oldest Child;' 'Child's Ethnicity: Mestizo;' 'Non-Dual Earners;' 'Less than Primary Educ.;' 'No Additional Members'; 'Male Head of Household;' 'Household in Rural Area;' 'PMK not Present.' 'Province: Pichincha' omitted.

Table B.3: Marginal Effects from Probit Estimates for the Probability of Feeling like a Happy Child by Gender of Child; Independent Variables: (i). Total Hours of Household Work; (ii). Child Performed Chores, Child Worked in Labour Market, Child Attended School; (iii). Threshold of Total Hours of Household Work by Age of Child (Self-Reported)

	Chile	dren Aged	8-14	Children Aged 15-17		
	All Children	Male Children	Female Children	All Children	Male Children	Female Children
	(1)	(2)	(3)	(4)	(5)	(6)
i. Intensive Margin: Total Hours of Household Work						
Child's Total Hours of Household Chores (Scaled)	-0.025***	-0.027***	-0.022***	-0.013	-0.007	-0.019*
Child's Total Hours in Labour Market Activities (Scaled)	$[0.007] \\ -0.040^{***} \\ [0.009]$	$[0.010] \\ -0.046^{***} \\ [0.011]$	[0.008] -0.033*** [0.013]	[0.009] -0.022*** [0.008]	[0.017] -0.025** [0.010]	[0.011] -0.014 [0.012]
Child Attends School	0.138** [0.060]	0.085 [0.074]	0.193^{**} [0.088]	0.159^{***} [0.048]	0.198^{***} [0.070]	0.105^{*} [0.057]
ii. Extensive Margin: Binary Variables for Working Activities						
Child Works in Domestic Activities	-0.040***	-0.046***	-0.031*	-0.019	-0.018	-0.041
Child Works in Market Activities	[0.012] -0.099*** [0.022]	[0.015] -0.132*** [0.029]	[0.018] -0.066** [0.027]	[0.032] -0.054* [0.028]	[0.041] -0.041 [0.034]	[0.050] -0.063 [0.040]
Child Attends School	0.180^{***} [0.054]	0.117^{*} [0.062]	0.241^{***} [0.085]	0.187^{***} [0.046]	0.241^{***} [0.066]	0.137^{**} [0.057]
iii. Threshold of Total Hours of Household Work						
Child Does $(0,7]$ Hours of Household Chores Per Week	-0.035**	-0.047***	-0.019	-0.019	-0.029	-0.026
Child Does (7,14] Hours of Household Chores Per Week	[0.015] -0.068*** [0.021]	[0.018] -0.060** [0.028]	[0.022] -0.067** [0.031]	[0.038] -0.029 [0.041]	[0.047] -0.032 [0.055]	[0.067] -0.045 [0.069]
Child Does (14,21] Hours of Household Chores Per Week	-0.128***	-0.195***	-0.079*	-0.014	-0.067	-0.018
Child Does More Than 21 Hours of Household Chores Per Week	[0.035] -0.094*** [0.036]	[0.055] -0.062 [0.049]	[0.042] -0.093** [0.045]	[0.044] -0.077 [0.054]	[0.069] 0.003 [0.067]	[0.073] -0.122 [0.091]
Child's Total Hours in Labour Market Activities Per Week (Scaled)	-0.039***	-0.045***	-0.032**	-0.022***	-0.026**	-0.013
Child Attends School	[0.009] 0.138^{**} [0.058]	$\begin{bmatrix} 0.011 \\ 0.080 \\ [0.071] \end{bmatrix}$	[0.012] 0.194^{**} [0.087]	[0.008] 0.163^{***} [0.048]	$[0.010] \\ 0.202^{***} \\ [0.069]$	[0.013] 0.117^{**} [0.057]
Observations	11,578	5,849	5,729	3,232	1,667	1,565

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, *p<0.1 denote levels of significance. Sample consists of respondents in the ENTI ages 8-14 and 15-17, living with both parents. The dependent variable, *happiness*, is an indicator variables equal to one if child feels happy most of the time. Estimates correspond to the marginal effects for the probability of the child self-assessing as happy. The base categories are: 'Child Doesn't Attend School;' 'Not Oldest Child;' 'Child's Ethnicity: Mestizo;' 'Non-Dual Earners;' 'Less than Primary Educ.;' 'No Additional Members'; 'Male Head of Household;' 'Household in Rural Area;' 'PMK not Present.' 'Province: Pichincha' omitted.

Appendix C

Child Labour



Figure C.1: Conceptual Definition of Child Labour in Ecuador

		All Cł	nildren	
	Work=1 & School=1 (1)	Work=0 & School=0 (2)	Work=1 & School=0 (3)	Work=0 & School=1 (4)
a. Official Definition Child Labour	(n=35,085)			
Mother Started Working Before $15 = 1$	0.022***	0.000	0.005***	-0.027***
Father Started Working Before $15 = 1$	0.018***	-0.007*** [0.003]	0.002]	-0.011*
Household Receives BDH Transfer = 1	0.016^{***} [0.005]	[0.003] [0.003]	0.005^{***} [0.002]	-0.023*** [0.006]
b. Market Activities vs. Domestic	Activities			
i Market Activities $(n=12, 153)$				
Mother Started Working Before $15 = 1$	0.019^{*}	0.001	0.009	-0.029**
3	[0.012]	[0.004]	[0.006]	[0.013]
Father Started Working Before $15 = 1$	0.038***	-0.009***	-0.003	-0.026**
	[0.010]	[0.004]	[0.005]	[0.011]
Household Receives BDH Transfer $= 1$	0.023**	0.001	0.010^{**}	-0.035***
	[0.010]	[0.004]	[0.005]	[0.011]
ii. Domestic Activities $(n=34,739)$				
Mother Started Working Before $15 = 1$	0.008	0.001	0.004	-0.013
	[0.014]	[0.001]	[0.004]	[0.013]
Father Started Working Before $15 = 1$	0.014	-0.002*	-0.004	-0.008
	[0.010]	[0.001]	[0.003]	[0.010]
Household Receives BDH Transfer $= 1$	0.027**	0.000	0.006**	-0.033***
	[0.011]	[0.001]	[0.003]	[0.011]
c. Unpaid vs. Paid Work				
i Paid Work $(n=8.623)$				
Mother Started Working Before $15 = 1$	0.000	0.009	0.008	-0.017
	[0.007]	[0.007]	[0.005]	[0.012]
Father Started Working Before $15 = 1$	0.018***	-0.008	0.003	-0.012
	[0.006]	[0.005]	[0.004]	[0.009]
Household Receives BDH Transfer $= 1$	0.015**	0.002	0.010**	-0.028***
	[0.007]	[0.005]	[0.004]	[0.011]
ii. Unpaid Work $(n=10,928)$				
Mother Started Working Before $15 = 1$	0.013	0.002	0.003	-0.019
0	[0.010]	[0.005]	[0.003]	[0.012]
Father Started Working Before $15 = 1$	0.028***	-0.007*	-0.001	-0.020**
	[0.009]	[0.004]	[0.003]	[0.010]
Household Receives BDH Transfer = 1	0.023**	-0.000	0.002	-0.025**
	[0.010]	[0.004]	[0.003]	[0.011]
Income Sheels Control	\mathbf{V}_{a}	V~~	\mathbf{V}_{22}	V~~
Location Controls	1 es Ves	1 es Ves	1 es Ves	i es Ves

Table C.1.a.: Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, All Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

		All Children					
	Work=1 & School=1	Work=0 & School=0	Work=1 & School=0	Work=0 & School=1			
	(1)	(2)	(3)	(4)			
d. Light Work vs. Excessive Work							
<i>i.</i> Light Work $(n = 24, 683)$							
Mother Started Working Before $15 = 1$	0.011 [0.018]	-0.002 [0.002]	-0.002 [0.002]	-0.007 [0.018]			
Father Started Working Before $15 = 1$	0.006	-0.001	-0.001	-0.004			
Household Receives BDH Transfer = 1	[0.010] 0.037^{***} [0.014]	0.001 [0.002]	0.003	-0.040*** [0.014]			
ii. Heavy Work (n=9,735)							
Mother Started Working Before $15 = 1$	-0.011 [0.013]	0.018^{***} [0.007]	0.013^{**} [0.005]	-0.019 [0.016]			
Father Started Working Before $15 = 1$	0.033*** [0.011]	-0.008* [0.004]	0.000 [0.003]	-0.026** [0.012]			
Household Receives BDH Transfer = 1	0.036*** [0.012]	-0.003 [0.004]	0.004 [0.004]	-0.037*** [0.014]			
e. Safe vs. Unsafe Working Condit	ions						
i. Safe Working Conditions $(n=9.454)$							
Mother Started Working Before $15 = 1$	0.002 [0.011]	0.009 [0.007]	0.005 [0.003]	-0.017 [0.013]			
Father Started Working Before $15 = 1$	0.026*** [0.009]	-0.008* [0.005]	-0.000 [0.003]	-0.018			
Household Receives BDH Transfer = 1	0.027*** [0.010]	0.003	0.006** [0.003]	-0.035*** [0.012]			
ii. Unsafe Working Conditions $(n=10,24)$	(3)	[]	[]	[]			
Mother Started Working Before $15 = 1$	0.024*** [0.009]	0.001 [0.005]	0.009^{*} [0.005]	-0.034^{***} [0.012]			
Father Started Working Before $15 = 1$	0.026***	-0.008*	0.000	-0.018**			
Household Receives BDH Transfer = 1	0.011 [0.007]	0.001 [0.005]	0.005 [0.004]	[0.000] -0.016^{*} [0.010]			
Income Shock Control Location Controls	Yes Yes	Yes Yes	Yes Yes	Yes Yes			

 Table C.1.b.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, All Children (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables denoting whether the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

	Male Children Work=1 & School=1 Work=0 & School=0 Work=1 & School=0 Work=0 & School					
	Work=1 & School=1 (1)	Work=0 & School=0 (2)	Work=1 & School=0 (3)	Work=0 & School=1 (4)		
a. Official Definition Child Labour	(n=17,909)					
Mother Started Working Before $15 = 1$	0.020**	0.003	0.008**	-0.031***		
Father Started Working Before $15 = 1$	[0.008] 0.019*** [0.006]	[0.005] -0.009*** [0.002]	[0.003] -0.002 [0.002]	[0.010] -0.008 [0.008]		
Household Receives BDH Transfer = 1	[0.006] 0.019^{***} [0.007]	[0.003] 0.004 [0.003]	[0.003] 0.008^{***} [0.003]	[0.008] - 0.031^{***} [0.009]		
b. Market Activities vs. Domestic	Activities					
i. Market Activities $(n=7.002)$						
Mother Started Working Before $15 = 1$	0.011	0.004	0.011	-0.026		
Father Started Working Before $15 = 1$	$[0.016] \\ 0.035^{***}$	[0.006] -0.012**	[0.008] -0.006	[0.017] -0.017		
Household Receives BDH Transfer $= 1$	[0.012] 0.015	[0.005] 0.006	[0.007] 0.017^{**}	[0.013] -0.038***		
	[0.014]	[0.005]	[0.007]	[0.015]		
<i>ii. Domestic Activities</i> (17,625)	0.011	0.000	0.000*	0.000		
Mother Started Working Before $15 = 1$	0.011	0.002	0.009**	-0.023		
Father Started Working Before $15 = 1$	0.009	-0.002]	-0.008**	0.002		
Facher Started Working Belore 15	[0.014]	[0.002]	[0.004]	[0.014]		
Household Receives BDH Transfer $= 1$	0.014	0.002	0.009**	-0.026*		
	[0.015]	[0.002]	[0.004]	[0.015]		
c. Unpaid vs. Paid Work						
i. Paid Work (n=5,010)						
Mother Started Working Before $15 = 1$	-0.004	0.009	0.007	-0.013		
	[0.009]	[0.010]	[0.008]	[0.016]		
Father Started Working Before $15 = 1$	0.019**	-0.012*	-0.001	-0.005		
	[0.008]	[0.007]	[0.006]	[0.012]		
Household Receives BDH Transfer $= 1$	0.018*	0.006	0.016**	-0.039***		
ii Unnaid Work (n-6.126)	[0.011]	[0.007]	[0.007]	[0.015]		
Mother Started Working Before $15 - 1$	0.007	0.004	0.003	-0.014		
Mother Started Working before 15 – 1	[0.013]	[0 008]	[0.003]	[0.015]		
Father Started Working Before $15 = 1$	0.019*	-0.008	-0.002	-0.009		
Facher Started Working Belore 15	[0.010]	[0.006]	[0.003]	[0.012]		
Household Receives BDH Transfer $= 1$	0.016	0.005	0.004	-0.026*		
	[0.012]	[0.006]	[0.003]	[0.014]		
Income Shock Control	Vos	Vos	Vos	Vos		
Location Controls	Yes	Yes	Yes	Yes		

 Table C.2.a.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Male Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the 'canton' in which the household is in.

	Male Children					
	Work=1 & School=1 (1)	Work=0 & School=0 (2)	Work=1 & School=0 (3)	Work=0 & School=1 (4)		
d. Light Work vs. Heavy Work						
i. Light Work (n=12,985)						
Mother Started Working Before $15 = 1$	0.016	-0.004	-0.003	-0.009		
	[0.023]	[0.003]	[0.003]	[0.022]		
Father Started Working Before $15 = 1$	0.000	-0.002	-0.002	0.004		
Household Receives BDH Transfer — 1	[0.017]	[0.002]	[0.002] 0.005*	[0.010] -0.033*		
nouschold fictories borr fransier – f	[0.018]	[0.003]	[0.002]	[0.018]		
ii. Heavy Work (n=5,179)	[]	[]	[]	[]		
Mother Started Working Before $15 = 1$	-0.018	0.029**	0.014^{**}	-0.025		
	[0.013]	[0.011]	[0.007]	[0.020]		
Father Started Working Before $15 = 1$	0.030***	-0.013**	-0.001	-0.016		
	[0.011]	[0.007]	[0.005]	[0.014]		
Household Receives BDH Transfer $= 1$	0.009	0.002	0.004	-0.015		
	[0.013]	[0.007]	[0.005]	[0.016]		
e. Safe vs. Unsafe Working Condit	ions					
i. Safe Working Conditions $(n=5.348)$						
Mother Started Working Before $15 = 1$	-0.011	0.009	0.002	0.000		
_	[0.013]	[0.009]	[0.004]	[0.017]		
Father Started Working Before $15 = 1$	0.023**	-0.007	0.000	-0.016		
	[0.011]	[0.007]	[0.004]	[0.014]		
Household Receives BDH Transfer $= 1$	0.030**	0.006	0.008**	-0.044***		
	[0.013]	[0.007]	[0.004]	[0.016]		
ii. Unsafe Working Conditions $(n=5,89)$	0)	0.000	0.010*	0.040**		
Mother Started Working Before $15 = 1$	0.028**	0.002	0.013*	-0.042**		
Father Started Working Refere 15 - 1	[0.014]	[0.008]	[0.007]	[0.017]		
Tather Started Working Delore 15 – 1	[0 009]	[0 006]	[0.005]	-0.007		
Household Receives BDH Transfer $= 1$	-0.002	0.008	0.008	-0.014		
	[0.010]	[0.007]	[0.006]	[0.013]		
Income Shock Control	Yes	Yes	Yes	Yes		
Location Controls	Yes	Yes	Yes	Yes		

 Table C.2.b.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Male Children (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variable denoting whether the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

	Female Children Work=1 & School=1 Work=0 & School=0 Work=1 & School=0 Work=0 & Scho					
	Work=1 & School=1 (1)	Work=0 & School=0 (2)	Work=1 & School=0 (3)	Work=0 & School=1 (4)		
a. Official Definition Child Labour	(n=17,176)					
Mother Started Working Before $15 = 1$	0.017***	-0.002	0.003	-0.018**		
Father Started Working Before $15 = 1$	[0.005] 0.017*** [0.005]	-0.004 [0.003]	0.002	-0.015** [0.007]		
Household Receives BDH Transfer = 1	[0.005] [0.005]	-0.001 [0.003]	[0.001] [0.002]	-0.013* [0.007]		
b. Market Activities vs. Domestic	Activities					
i Market Activities $(n=5, 151)$						
Mother Started Working Before $15 = 1$	0.025**	-0.002	0.002	-0.024*		
	[0.011]	[0.003]	[0.006]	[0.013]		
Father Started Working Before $15 = 1$	0.046***	-0.005	0.001	-0.042***		
5	[0.012]	[0.004]	[0.006]	[0.013]		
Household Receives BDH Transfer $= 1$	0.027**	-0.004	-0.000	-0.024*		
	[0.011]	[0.003]	[0.005]	[0.013]		
ii. Domestic Activities (n=17,114)						
Mother Started Working Before $15 = 1$	0.002	-0.000	-0.001	-0.001		
5	[0.018]	[0.001]	[0.004]	[0.017]		
Father Started Working Before $15 = 1$	0.025**	-0.001	-0.000	-0.024**		
0	[0.012]	[0.001]	[0.004]	[0.012]		
Household Receives BDH Transfer $= 1$	0.037***	-0.001	0.002	-0.038***		
	[0.013]	[0.001]	[0.004]	[0.013]		
c. Unpaid vs. Paid Work						
i. Paid Work (n=3.613)						
Mother Started Working Before $15 = 1$	0.011^{*}	0.002	0.005	-0.019*		
0	[0.006]	[0.006]	[0.004]	[0.011]		
Father Started Working Before $15 = 1$	0.010**	0.004	0.006	-0.020**		
0	[0.005]	[0.006]	[0.004]	[0.010]		
Household Receives BDH Transfer $= 1$	0.008*	-0.004	0.000	-0.004		
	[0.005]	[0.005]	[0.003]	[0.009]		
ii. Unpaid Work (n=4.792)	[]	[]	[]	[]		
Mother Started Working Before $15 = 1$	0.019^{*}	-0.001	0.002	-0.020*		
	[0.011]	[0.004]	[0.004]	[0.012]		
Father Started Working Before $15 = 1$	0.042***	-0.005	-0.000	-0.037***		
	[0.012]	[0.004]	[0.003]	[0.013]		
Household Receives BDH Transfer $= 1$	0.027**	-0.005	-0.002	-0.020		
	[0.011]	[0.004]	[0.004]	[0.012]		
	[*:***]	[]	[0.004]	[]		
Income Shock Control	Yes	Yes	Yes	Yes		
Location Controls	Yes	Yes	Yes	Yes		

 Table C.3.a.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Female Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variables for the 'canton' in which the household is in.

	Female Children					
	Work=1 & School=1	Work=0 & School=0	Work=1 & School=0	Work=0 & School=1		
	(1)	(2)	(3)	(4)		
d. Light Work vs. Heavy Work						
<i>i. Light Work</i> (<i>n</i> =11,698)						
Mother Started Working Before $15 = 1$	0.003	-0.000	-0.000	-0.003		
Esther Started Washing Defens 15 1	[0.024]	[0.002]	[0.002]	[0.024]		
Father Started Working Before $15 = 1$	0.018	-0.000	0.001	-0.019		
Household Receives BDH Transfer - 1	0.046**	[0.002]	[0.002]	0.043**		
Household Receives DDH Hanslei – 1	[0.018]	[0.002]	[0.002]	[0 017]		
ii. Heavy Work $(n=4.556)$	[0.010]	[0.002]	[0.002]	[0.011]		
Mother Started Working Before $15 = 1$	-0.014	0.010*	0.009*	-0.005		
6	[0.021]	[0.006]	[0.005]	[0.022]		
Father Started Working Before $15 = 1$	0.053***	-0.003	0.004	-0.054***		
C	[0.016]	[0.004]	[0.004]	[0.017]		
Household Receives BDH Transfer $= 1$	0.061***	-0.008**	-0.001	-0.052***		
	[0.018]	[0.003]	[0.003]	[0.019]		
e. Safe vs. Unsafe Working Condit	ons					
i. Safe Working Conditions $(n=4,106)$						
Mother Started Working Before $15 = 1$	0.009	0.004	0.004	-0.017		
6	[0.455]	[0.000]	[0.455]	[0.603]		
Father Started Working Before $15 = 1$	0.039	-0.006	0.001	-0.033		
_	[0.354]	[0.000]	[0.354]	[1.004]		
Household Receives BDH Transfer $= 1$	0.017	-0.001	0.001	-0.017		
	[0.236]	[0.000]	[0.236]	[0.221]		
ii. Unsafe Working Conditions (n=4,353	3)					
Mother Started Working Before $15 = 1$	0.020***	-0.001	0.004	-0.023**		
	[0.007]	[0.005]	[0.004]	[0.010]		
Father Started Working Before $15 = 1$	0.024***	-0.000	0.005	-0.029***		
	[0.006]	[0.004]	[0.004]	[0.010]		
Household Receives BDH Transfer $= 1$	0.027***	-0.007*	-0.001	-0.019*		
	[0.008]	[0.004]	[0.004]	[0.010]		
Income Shock Control	Yes	Yes	Yes	Yes		
Location Controls	Yes	Yes	Yes	Yes		

 Table C.3.b.:
 Summary of Main Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance by Type of Work, Female Children (cont.)

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)), an indicator variable denoting whether the household experienced an 'income fall in the last 12 months' and indicator variables for the 'canton' in which the household is in.

Table C.4.a.:	Full Results:	Average Margina	l Effects from	Bivariate Pro	obit Regressions	for Work &	& School	Atten-
	dance Using	the Official Defini	tion of Child	Labour, Male	and Female Chi	ldren		

		Male C	hildren		Female Children			
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Work=0; School=0 (6)	Work=1; School=0 (7)	Work=0; School=1 (8)
Mother Started Working Before $15 = 1$	0.020**	0.003	0.008**	-0.031***	0.017***	-0.002	0.003	-0.018**
Father Started Working Before $15 = 1$	[0.008] 0.019^{***}	[0.005] -0.009***	[0.003] -0.002	[0.010] -0.008	[0.005] 0.017^{***}	[0.004] -0.004	$\begin{bmatrix} 0.002 \end{bmatrix} \\ 0.002 \end{bmatrix}$	[0.007] -0.015**
Mother's Educ.: Primary $= 1$	$\begin{bmatrix} 0.006 \end{bmatrix} \\ 0.004 \end{bmatrix}$	[0.003] -0.009*	[0.003] -0.006	[0.008] 0.011	[0.005] 0.000	[0.003] -0.011**	[0.001] -0.004*	$\begin{bmatrix} 0.007 \end{bmatrix} \\ 0.015 \end{bmatrix}$
Mother's Educ.: Secondary $= 1$	[0.008] -0.009	[0.005] -0.012**	[0.004] -0.011***	[0.011] 0.032^{**}	[0.007] -0.009	[0.006] - 0.024^{***}	[0.002] -0.010***	[0.010] 0.043^{***}
Mother's Educ.: University $= 1$	[0.011] -0.023	[0.005] -0.017***	[0.004] -0.016***	[0.013] 0.056^{***}	[0.009] - 0.031^{***}	[0.005] -0.023***	[0.002] -0.011***	[0.012] 0.065^{***}
Father's Educ.: Primary $= 1$	[0.015] -0.010	[0.006] -0.017**	[0.004] -0.016***	[0.016] 0.043^{***}	[0.008] 0.002	[0.003] -0.011**	[0.001] -0.004	[0.009] 0.014
Father's Educ.: Secondary $= 1$	[0.011] -0.013	[0.007] -0.027***	[0.005] - 0.022^{***}	[0.015] 0.062^{***}	$[0.010] \\ 0.007$	[0.006] -0.017***	[0.003] -0.006**	[0.015] 0.016
Father's Educ.: University $= 1$	[0.013] - 0.053^{***}	[0.006] -0.027***	[0.004] -0.024***	[0.016] 0.104^{***}	[0.012] -0.006	[0.005] - 0.024^{***}	[0.003] -0.010***	[0.016] 0.039^{***}
Mother not in Labour Force $= 1$	[0.010] - 0.056^{***}	[0.003] 0.013^{***}	[0.002] -0.006**	[0.011] 0.049^{***}	[0.013] -0.048***	[0.003] 0.008^{**}	[0.002] -0.008***	$\begin{bmatrix} 0.014 \end{bmatrix} \\ 0.047^{***}$
Father not in Labour Force $= 1$	[0.007] -0.014	[0.004] -0.005	[0.003] -0.007	[0.009] 0.026	[0.005] - 0.031^{***}	[0.004] -0.001	[0.002] -0.006***	[0.007] 0.038^{***}
Age	[0.015] 0.014^{***}	[0.007] 0.003^{***}	[0.005] 0.006^{***}	[0.018] -0.022***	[0.008] 0.007^{***}	[0.008] 0.006^{***}	[0.002] 0.004^{***}	[0.013] -0.017***
Oldest Child $= 1$	[0.001] -0.009	[0.000] 0.012^{***}	[0.000] 0.006^{**}	[0.001] -0.009	[0.001] -0.003	[0.001] 0.005	[0.000] 0.001	[0.001] -0.004
Identifies as: Indigenous $= 1$	[0.007] 0.044^{***}	[0.004] -0.003	$\begin{bmatrix} 0.003 \end{bmatrix} \\ 0.008 \end{bmatrix}$	[0.009] - 0.048^{***}	[0.005] 0.048^{***}	$\begin{bmatrix} 0.004 \end{bmatrix} \\ 0.002 \end{bmatrix}$	[0.002] 0.010^{***}	[0.007] -0.060***
Identifies as: A fro-descendant = 1	$\begin{bmatrix} 0.016 \end{bmatrix} \\ 0.007 \end{bmatrix}$	[0.005] 0.016^*	$\begin{bmatrix} 0.005 \end{bmatrix} \\ 0.015 \end{bmatrix}$	[0.017] -0.038	[0.011] -0.007	$\begin{bmatrix} 0.005 \end{bmatrix} \\ 0.012 \end{bmatrix}$	$\begin{bmatrix} 0.004 \end{bmatrix} \\ 0.003 \end{bmatrix}$	[0.015] -0.007
Identifies as: Other Ethnicity $= 1$	[0.017] 0.020	$[0.009] \\ 0.000$	$[0.009] \\ 0.005$	[0.026] -0.025	[0.011] -0.006	[0.009] -0.006	[0.004] -0.003*	[0.015] 0.015
Has a Disability $= 1$	[0.016] - 0.065^{***}	[0.005] 0.185^{***}	[0.005] 0.028^{***}	[0.019] -0.148***	[0.010] -0.038***	[0.005] 0.145^{***}	[0.002] 0.016^{***}	[0.011] -0.123***
Household Receives BDH Transfer $= 1$	[0.005] 0.019^{***}	$\begin{bmatrix} 0.025 \end{bmatrix} \\ 0.004 \end{bmatrix}$	[0.008] 0.008^{***}	[0.027] -0.031***	[0.005] 0.012^{**}	[0.029] -0.001	$\begin{bmatrix} 0.006 \end{bmatrix} \\ 0.002 \end{bmatrix}$	[0.031] -0.013*
Mother's Age	[0.007] 0.001^{**}	[0.003] -0.000	$\begin{bmatrix} 0.003 \end{bmatrix} \\ 0.000 \end{bmatrix}$	[0.009] -0.001*	[0.005] -0.000	$\begin{bmatrix} 0.003 \end{bmatrix} \\ 0.000 \end{bmatrix}$	$\begin{bmatrix} 0.001 \end{bmatrix} \\ 0.000 \end{bmatrix}$	[0.007] -0.000
Father's Age	[0.001] -0.000	$[0.000] \\ 0.000$	[0.000] -0.000	$\begin{bmatrix} 0.001 \end{bmatrix} \\ 0.000 \end{bmatrix}$	[0.000] 0.001^{**}	[0.000] -0.000	$[0.000] \\ 0.000$	[0.001] -0.001
House has Children Below Age $5 = 1$	[0.000] 0.002	[0.000] 0.006^*	[0.000] 0.006^{**}	[0.001] -0.014*	$[0.000] \\ 0.000$	$\begin{bmatrix} 0.000 \end{bmatrix} \\ 0.004 \end{bmatrix}$	[0.000] 0.001	[0.000] -0.005
House has Other Children Aged $5-7 = 1$	[0.006] 0.002	[0.003] -0.001	[0.003] -0.000	[0.008] -0.001	[0.005] 0.005	[0.003] -0.002	$\begin{bmatrix} 0.001 \end{bmatrix} \\ 0.000 \end{bmatrix}$	[0.006] -0.004
House has Other Child. Aged $8-14 = 1$	[0.006] 0.016^{***}	[0.003] -0.000	[0.002] 0.004^*	[0.007] -0.019***	[0.005] 0.001	[0.003] 0.003	$\begin{bmatrix} 0.001 \end{bmatrix} \\ 0.001 \end{bmatrix}$	[0.006] -0.006
House has Other Child. Aged $15-17 = 1$	[0.006] -0.003	[0.003] 0.011^{***}	[0.002] 0.008^{***}	[0.007] -0.016**	$\begin{bmatrix} 0.004 \end{bmatrix} \\ 0.000 \end{bmatrix}$	$[0.003] \\ 0.003$	$\begin{bmatrix} 0.001 \end{bmatrix} \\ 0.001 \end{bmatrix}$	[0.006] -0.004
	[0.006]	[0.004]	[0.003]	[0.008]	[0.004]	[0.003]	[0.001]	[0.006]
Observations Income Shock Control	17,909 Yes	17,909 Yes	17,909 Yes	17,909 Yes	17,176 Yes	17,176 Yes	17,176 Yes	17,176 Yes
Canton Controls	Yes							

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=0, School=0); working and not attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables for the 'canton' in which the household is in.

: Full Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance Using the Official Definition	of Child Labour, Male and Female Children (cont.)
C.4.b.: Full Results	of Child La
Tablé	

		Male C	hildren			Female (Children	
	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;
	School=1	School=0	School=0	School=1	School=1	School=0	School=0	School=1
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
House Has Daughter of Head of Household Aged $18+ = 1$	-0.022***	0.001	-0.005**	0.027^{***}	-0.014***	0.000	-0.003	0.016^{**}
	[0.007]	[0.004]	[0.002]	[0.009]	[0.005]	[0.004]	[0.002]	[0.008]
House Has Son of Head of Household Aged $18+=1$	-0.021^{***}	0.014^{***}	0.004	0.003	-0.010^{**}	0.003	-0.001	0.009
	[0.006]	[0.004]	[0.003]	[0.008]	[0.005]	[0.004]	[0.002]	[0.007]
House Has Other Female Relatives Aged $18+=1$	-0.018	-0.003	-0.007*	0.028^{**}	-0.007	-0.006	-0.004^{*}	0.017
	[0.012]	[0.005]	[0.004]	[0.014]	[0.00]	[0.005]	[0.002]	[0.011]
House Has Other Male Relatives Aged $18+=1$	0.009	-0.000	0.002	-0.010	0.008	0.020^{***}	0.010^{**}	-0.037**
	[0.016]	[0.006]	[0.005]	[0.019]	[0.011]	[0.007]	[0.004]	[0.017]
Child's Grandmother Lives in Household $= 1$	-0.007	-0.001	-0.003	0.011	-0.003	0.009	0.003	-0.009
	[0.011]	[0.006]	[0.004]	[0.015]	[0.010]	[0.008]	[0.003]	[0.014]
Child's Grandfather Lives in Household $= 1$	0.032^{*}	-0.010^{*}	-0.002	-0.021	-0.015	-0.005	-0.004	0.024^{*}
	[0.019]	[0.005]	[0.005]	[0.021]	[0.010]	[0.008]	[0.003]	[0.013]
Urban Setting = 1	-0.040^{***}	-0.005	-0.014^{***}	0.059^{***}	-0.022***	-0.011^{***}	-0.008***	0.041^{***}
	[0.008]	[0.004]	[0.003]	[0.010]	[0.006]	[0.004]	[0.002]	[0.009]
Observations	17,909	17,909	17,909	17,909	17, 176	17, 176	17, 176	17,176
Income Shock Control	Yes	Yes	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Canton Controls	Yes	Yes	Yes	Yes	Y_{es}	Y_{es}	\mathbf{Yes}	Yes

not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.01 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and Rural' household)) and indicator variables for the 'canton' in which the household is in.



Figure C.2: Average Marginal Effects at Different Ages for All Children Child Labour: Official Definition & Child's Family Receives the BDH Transfer



Figure C.3: Average Marginal Effects at Different Ages for All Children Child Labour: Work in Market Activities & Child's Family Receives the BDH Transfer



Figure C.4: Average Marginal Effects at Different Ages for All Children Child Labour: Work in Domestic Activities & Child's Family Receives the BDH Transfer



Figure C.5: Average Marginal Effects at Different Ages for All Children Child Labour: Paid Work & Child's Family Receives the BDH Transfer



Figure C.6: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work & Child's Family Receives the BDH Transfer



Figure C.7: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work & Child's Family Receives the BDH Transfer



Figure C.8: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work & Child's Family Receives the BDH Transfer



Figure C.9: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work & Child's Family Receives the BDH Transfer



Figure C.10: Average Marginal Effects at Different Ages for All Children Child Labour: Unpaid Work & Child's Family Receives the BDH Transfer

		Male C	hildren			Female	Children	
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Work=0; School=0 (6)	Work=1; School=0 (7)	Work=0; School=1 (8)
i. All Children from Dual-Parent Households								
Mother Started Working Before $15 = 1$	0.021** [0.009]	0.003	0.008** [0.004]	-0.032*** [0.011]	0.017^{***} [0.005]	-0.002	0.002 [0.002]	-0.018** [0.007]
Father Started Working Before $15 = 1$	0.019*** [0.006]	-0.009*** [0.003]	-0.002 [0.003]	-0.008	0.017*** [0.005]	-0.004 [0.003]	0.002	-0.015** [0.007]
Household Receives BDH Transfer = 1	0.016^{**} [0.007]	0.005 [0.003]	0.008*** [0.003]	-0.028*** [0.009]	0.012** [0.005]	-0.001 [0.003]	0.002 [0.001]	-0.013* [0.007]
		n = 1	17,909			n = 1	17,176	
ii. Children from Dual-Earner Households								
Mother Started Working Before $15 = 1$	0.022^{*} [0.011]	-0.000 [0.004]	0.006 [0.005]	-0.028** [0.013]	0.020^{***} [0.007]	-0.003	0.002 [0.002]	-0.019** [0.009]
Father Started Working Before $15 = 1$	0.025*** [0.009]	-0.010** [0.004]	-0.003 [0.004]	-0.012	0.016** [0.007]	-0.003 [0.004]	0.002	-0.015* [0.008]
Household Receives BDH Transfer = 1	0.017	0.007*	0.012***	-0.036**	0.025***	0.002	0.005**	-0.032***
	[0.012]	$\begin{bmatrix} 0.004 \end{bmatrix}$ n = 1	[0.004] 12,429	[0.014]	[0.008]	[0.005] n = 1	[0.002] 11,898	[0.010]
Canton Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

 Table C.5: Summary of Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School Attendance Using Working 1+ Hours/Week as the Definition of Child Work, Male & Female Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. 'Work' is defined as a binary variable equal to one if the child stated that they worked an hour or more in the previous week, zero otherwise. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables for the 'canton' in which the household is in.

		Male C	hildren			Female	Children	
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Work=0; School=0 (6)	Work=1; School=0 (7)	Work=0; School=1 (8)
i. All Children from Dual-Parent Households								
Mother Started Working Before $15 = 1$ Father Started Working Before $15 = 1$ Household Receives BDH Transfer = 1	$\begin{array}{c} 0.013^{*} \\ [0.007] \\ 0.024^{***} \\ [0.006] \\ 0.013^{**} \end{array}$	$\begin{array}{c} 0.005 \\ [0.006] \\ -0.012^{***} \\ [0.004] \\ 0.007^{*} \end{array}$	$\begin{array}{c} 0.005^{**} \\ [0.002] \\ 0.000 \\ [0.002] \\ 0.006^{***} \end{array}$	-0.024** [0.009] -0.013 [0.008] -0.026***	$\begin{array}{c} 0.021^{***} \\ [0.005] \\ 0.016^{***} \\ [0.005] \\ 0.008 \end{array}$	-0.003 [0.004] -0.003 [0.003] -0.000	0.002* [0.001] 0.001 [0.001] 0.001	-0.021*** [0.007] -0.014** [0.007] -0.009
	[0.007]	$\begin{bmatrix} 0.004 \end{bmatrix}$ n = 1	[0.002] 7,909	[0.008]	[0.005]	$\begin{bmatrix} 0.004 \end{bmatrix}$ n = 1	[0.001] 17,176	[0.007]
ii. Children from Dual-Earner Households								
Mother Started Working Before $15 = 1$	0.019^{*}	0.001	0.005	-0.024** [0.012]	0.025*** [0.007]	-0.004	0.002	-0.024*** [0.009]
Father Started Working Before $15 = 1$	0.029*** [0.008]	-0.013** [0.005]	-0.000 [0.003]	-0.016 [0.011]	0.016** [0.007]	-0.002 [0.004]	0.001 [0.001]	-0.015* [0.008]
Household Receives BDH Transfer = 1	0.008 [0.011]	0.012** [0.005]	0.008*** [0.003]	-0.028** [0.014]	0.018** [0.007]	0.003 [0.005]	0.003** [0.002]	-0.025*** [0.009]
		n = 1	2,429			n = 1	11,898	
Canton Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

 Table C.6: Summary of Results: Average Marginal Effects from Bivariate Probit Regressions for Work & School

 Attendance Using Working Before Meeting the Minimum Age Requirement as the Definition of Child

 Work, Male & Female Children

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. 'Work' is defined as a binary variable equal to one if the child started working before the age of 15, zero otherwise. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'canton' in which the household is in.

		Male C	hildren			Female	Children	
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Work=0; School=0 (6)	Work=1; School=0 (7)	Work=0; School=1 (8)
i. Child in Guayaquil Official Definition of Child Labour	0.048	0.016	0.011	0.925	0.011	0.010	0.002	0.977
Child Works in Market Activities Child Works in Domestic Activities	$0.099 \\ 0.646$	$0.032 \\ 0.013$	$\begin{array}{c} 0.021\\ 0.014\end{array}$	$0.848 \\ 0.327$	0.024 0.762	$\begin{array}{c} 0.018\\ 0.003\end{array}$	$0.008 \\ 0.008$	$0.949 \\ 0.226$
Child Does Paid Work Child Does Unpaid Work for Family	$\begin{array}{c} 0.008\\ 0.015\end{array}$	$\begin{array}{c} 0.041 \\ 0.036 \end{array}$	$0.009 \\ 0.001$	$0.942 \\ 0.947$	$0.000 \\ 0.015$	$0.019 \\ 0.015$	$0.000 \\ 0.002$	$0.981 \\ 0.969$
Light Child Work Heavy Child Work	$0.538 \\ 0.027$	$\begin{array}{c} 0.014 \\ 0.036 \end{array}$	$0.007 \\ 0.009$	$0.442 \\ 0.928$	$0.635 \\ 0.091$	$\begin{array}{c} 0.004 \\ 0.010 \end{array}$	$\begin{array}{c} 0.003 \\ 0.004 \end{array}$	$0.357 \\ 0.895$
Safe Conditions Child Work Unsafe Conditions Child Work	$0.027 \\ 0.020$	$\begin{array}{c} 0.032 \\ 0.051 \end{array}$	$0.007 \\ 0.010$	$0.935 \\ 0.919$	$0.008 \\ 0.000$	$0.010 \\ 0.022$	$0.003 \\ 0.001$	$0.979 \\ 0.977$
Started Work Age: 5-14 Started Work Age: 15-17	$\begin{array}{c} 0.011 \\ 0.000 \end{array}$	$\begin{array}{c} 0.044 \\ 0.040 \end{array}$	$0.001 \\ 0.000$	$0.944 \\ 0.960$	$0.006 \\ 0.000$	$\begin{array}{c} 0.011 \\ 0.011 \end{array}$	$0.001 \\ 0.000$	$0.983 \\ 0.989$
ii. Child in Cotopaxi Official Definition of Child Labour	0.035	0.003	0.002	0.960	0.024	0.004	0.001	0.971
Child Works in Market Activities Child Works in Domestic Activities	$0.362 \\ 0.895$	$0.003 \\ 0.001$	$\begin{array}{c} 0.015 \\ 0.004 \end{array}$	$0.619 \\ 0.099$	$\begin{array}{c} 0.152 \\ 0.919 \end{array}$	$0.000 \\ 0.001$	$0.003 \\ 0.004$	$\begin{array}{c} 0.844\\ 0.076\end{array}$
Child Does Paid Work Child Does Unpaid Work for Family	$0.059 \\ 0.199$	$\begin{array}{c} 0.014 \\ 0.002 \end{array}$	$0.019 \\ 0.001$	$0.908 \\ 0.798$	$\begin{array}{c} 0.001 \\ 0.081 \end{array}$	$\begin{array}{c} 0.004 \\ 0.000 \end{array}$	$0.004 \\ 0.000$	$0.991 \\ 0.919$
Light Child Work Heavy Child Work	$0.840 \\ 0.194$	$0.002 \\ 0.003$	$0.003 \\ 0.008$	$0.156 \\ 0.795$	$0.857 \\ 0.285$	$0.000 \\ 0.000$	$0.000 \\ 0.001$	$0.143 \\ 0.714$
Safe Conditions Child Work Unsafe Conditions Child Work	$0.152 \\ 0.161$	$\begin{array}{c} 0.005 \\ 0.004 \end{array}$	$0.007 \\ 0.008$	$0.835 \\ 0.827$	$0.030 \\ 0.058$	$0.001 \\ 0.000$	$0.001 \\ 0.002$	$0.968 \\ 0.939$
Started Work Age: 5-14 Started Work Age: 15-17	$\begin{array}{c} 0.091 \\ 0.000 \end{array}$	$0.000 \\ 0.021$	$0.000 \\ 0.000$	$0.909 \\ 0.979$	$0.037 \\ 0.000$	$0.000 \\ 0.000$	$0.000 \\ 0.005$	$0.963 \\ 0.995$

 Table C.7:
 Summary of Results: Work & School Attendance by Type of Work, Boys and Girls

 Probability of outcomes at mean values for boys and girls in cities:
 Guayaquil and Cotopaxi

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'canton' in which the household is in.

	-	Official Chi	ild Labour			Market	Activities			Domestic	Activities	
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Work=0; School=0 (6)	Work=1; School=0 (7)	Work=0; School=1 (8)	Work=1; School=1 (9)	Work=0; School=0 (10)	Work=1; School=0 (11)	Work=0; School=1 (12)
Household Income Excluding Child	dren's Earn	uings (Scal€	ed to \$1,00	00 USD)								
Mother Started Working Before $15 = 1$	0.022^{***}	-0.000	0.005^{***}	-0.026^{***}	0.020*	0.001	0.009	-0.030^{**}	0.008	0.001	0.004	-0.012
Father Started Working Before $15 = 1$	$0.006] 0.018^{***}$	[0.003]	$\begin{bmatrix} 0.002 \\ 0.001 \end{bmatrix}$	[0.007]	0.012 0.038^{***}	[0.004] -0.010***	[0.006]	$[0.013] -0.026^{**}$	$\begin{bmatrix} 0.014 \\ 0.016 \end{bmatrix}$	[0.001]	$\begin{bmatrix} 0.004 \\ -0.004 \end{bmatrix}$	[0.013]
Household Monthly Income (Excluding Child Earnings) (Scaled)	[0.004] -0.005 [0.003]	$\begin{bmatrix} 0.003\\ -0.003\\ [0.002] \end{bmatrix}$	$\begin{bmatrix} 0.001 \\ -0.003^{**} \\ [0.001] \end{bmatrix}$	$\begin{bmatrix} 0.006 \\ 0.011^{**} \\ [0.004] \end{bmatrix}$	[0.010] -0.014** [0.007]	$\begin{bmatrix} 0.004 \\ -0.002 \\ [0.003] \end{bmatrix}$	[0.005] -0.009** [0.005]	$\begin{bmatrix} 0.011 \\ 0.025^{***} \\ [0.008] \end{bmatrix}$	[0.010] -0.031*** [0.008]	$\begin{bmatrix} 0.001 \\ -0.000 \end{bmatrix}$	$\begin{bmatrix} 0.003 \\ -0.006^{**} \end{bmatrix}$	$\begin{bmatrix} 0.010 \\ 0.037^{***} \\ [0.008] \end{bmatrix}$
Average Household Monthly Expe	nditures (S	caled to \$1	,000 USD	<u> </u>								
Mother Started Working Before $15 = 1$	0.022^{***}	-0.000	0.005^{***}	-0.026*** [0.007]	0.021* [0.046]	0.001	0.009	-0.031^{**}	0.009	0.001	0.004	-0.014
Father Started Working Before $15 = 1$	0.018^{***}	[0.000] -0.007***	0.001 0.001	-0.011* -0.06]	0.038*** 0.038***	-0.010^{4}	-0.003 -0.003 [0.005]	-0.025** -0.025**	0.014	0.002*	-0.004 -0.004 [0.002]	[010]
Household Average Monthly Expenditures (\$1,000 USD)	$\begin{bmatrix} 0.004 \\ -0.020^{***} \end{bmatrix}$	[900:0] [900:0-	[100.03]	$\begin{bmatrix} 0.000\\ 0.025^{**}\\ [0.010] \end{bmatrix}$	$\begin{bmatrix} 0.010 \\ -0.049^{***} \\ [0.014] \end{bmatrix}$	$\begin{bmatrix} 0.004\\ 0.002\\ [0.007] \end{bmatrix}$	[0.013 -0.013 [0.011]	[110.0] 0.060*** [0.016]	[0.010] -0.075*** [0.016]	$\begin{bmatrix} 0.001\\ 0.002\\ 0.002 \end{bmatrix}$	[con.o] 200.0- [200.0]	0.080^{***} 0.080^{***}
Observations	35,085	35,085	35,085	35,085	12,153	12,153	12, 153	12,153	34,739	34,739	34,739	34,739
Income Shock Control Canton Controls	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes

Table C.8: Robustness Check: Income Variable: Household Income Excluding Children's Earnings (Scaled to \$1,000 USD) and Average Household Monthly

duct accentuity school (WOTK=1, JOHOO) -1), not working and not accentuity school (WOTK=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'byy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'byy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Urban' (base: 'Father not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'nonschold composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'nonschold composition' in which the household (base) indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'canton' in which the household is in.

¥	
of Livestoch	
Ownership	
of Land,	
Ownership o	
Vehicle, (
wnership of	ons
cy, O	ressic
Tenand	oit Reg
Home	te Prol
Assets:	Bivaria
ehold 1	from .
Hous	ffects
Check:	ginal E
Robustness (Average Mar
C.9:	
Table (

	-	Official Chi	ld Labour			Market A	ctivities			Domestic .	Activities	
	Work=1; School=1	Work=0; School=0	Work=1; School=0	Work=0; School=1	Work=1; School=1	Work=0; School=0	Work=1; School=0	Work=0; School=1	Work=1; School=1	Work=0; School=0	Work=1; School=0	Work=0; School=1
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
					•							
Assets: Home Tenancy, Ownership of Ve	shicle, Uwn	ership of L	and, Uwn	ership of Li	vestock							
Mother Started Working Before $15 = 1$	0.018^{***}	-0.000	0.004^{**}	-0.022***	0.016	0.001	0.007	-0.023^{*}	0.003	0.001	0.003	-0.007
	[0.005]	[0.003]	[0.002]	[0.007]	[0.012]	[0.004]	[0.006]	[0.012]	[0.014]	[0.001]	[0.004]	[0.013]
Father Started Working Before $15 = 1$	0.017^{***}	-0.007***	0.000	-0.010^{*}	0.037^{***}	-0.009***	-0.003	-0.024^{**}	0.016	-0.002*	-0.004	-0.010
	[0.004]	[0.003]	[0.002]	[0.006]	[0.00]	[0.003]	[0.005]	[0.011]	[0.010]	[0.001]	[0.003]	[0.010]
Household Owns Dwelling, Fully Paid $= 1$	0.001	-0.003	-0.002	0.004	0.005	-0.005	-0.007	0.007	-0.004	-0.001	-0.006*	0.012
	[0.005]	[0.003]	[0.002]	[0.007]	[0.010]	[0.004]	[0.006]	[0.011]	[0.012]	[0.001]	[0.003]	[0.011]
Household Owns Dwelling, Partially Paid $= 1$	-0.002	-0.001	-0.001	0.005	-0.031	0.005	-0.003	0.029	-0.044	0.001	-0.004	0.047^{*}
	[0.013]	[0.008]	[0.005]	[0.016]	[0.023]	[0.013]	[0.015]	[0.028]	[0.028]	[0.003]	[0.009]	[0.025]
Household Rents Dwelling $= 1$	0.013	-0.001	0.002	-0.013	0.002	-0.005	-0.009	0.012	-0.040^{**}	0.002	-0.001	0.040^{**}
	[0.010]	[0.004]	[0.003]	[0.013]	[0.016]	[0.005]	[0.008]	[0.020]	[0.018]	[0.002]	[0.005]	[0.018]
Household Has Arable Land $= 1$	0.046^{***}	-0.005**	0.007^{***}	-0.049^{***}	0.062^{***}	-0.007**	0.007	-0.061^{***}	0.045^{***}	-0.001^{*}	0.003	-0.047^{***}
	[0.005]	[0.002]	[0.002]	[0.006]	[0.011]	[0.003]	[0.006]	[0.012]	[0.011]	[0.001]	[0.003]	[0.010]
Household Owns Vehicle $= 1$	0.002	-0.010^{***}	-0.005***	0.013^{*}	0.003	-0.010^{***}	-0.017^{***}	0.024^{**}	-0.023*	-0.002*	-0.011^{***}	0.035^{***}
	[0.006]	[0.003]	[0.002]	[0.007]	[0.011]	[0.003]	[0.005]	[0.012]	[0.013]	[0.001]	[0.003]	[0.013]
Household Owns Livestock $= 1$	0.020^{***}	0.001	0.005^{***}	-0.026^{***}	0.031^{***}	-0.003	0.005	-0.032^{***}	0.007	0.001	0.004	-0.011
	[0.005]	[0.003]	[0.002]	[0.007]	[0.011]	[0.004]	[0.005]	[0.012]	[0.011]	[0.001]	[0.003]	[0.011]
Observations	35,085	35,085	35,085	35,085	12,153	12,153	12,153	12, 153	34,739	34,739	34,739	34,739
Income Shock Control	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}^{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}^{\mathbf{es}}$	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Canton Controls	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes	Yes
	(; ; ; ; ;	**	÷	- - -		•	-				-	

the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=1, School=0); not working and attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'canton' in which the household is in.
		Official Ch	ild Labou			Market 7	Activities			Domestic	Activities	
	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;	Work=1;	Work=0;
	School=1	School=0	School=0	School=1	School=1	School=0	School=0	School=1	School=1	School=0	School=0	School=1
	(+)	(2)	(0)	(1)	(0)	(0)	(1)	(0)	(0)	(01)	(++)	(71)
Mother Started Working Before $15 = 1$	0.008^{*}	-0.001	0.001	-0.008	0.008	0.001	0.005	-0.015	0.005	-0.000	0.001	-0.006
	[0.004]	[0.003]	[0.002]	[0.006]	[0.010]	[0.004]	[0.005]	[0.010]	[0.014]	[0.001]	[0.004]	[0.014]
Father Started Working Before $15 = 1$	0.011^{***}	-0.007**	-0.001	-0.003	0.031^{***}	-0.009**	-0.006	-0.016^{*}	0.017	-0.002^{**}	-0.005*	-0.010
	[0.003]	[0.003]	[0.001]	[0.005]	[0.008]	[0.003]	[0.005]	[0.00]	[0.010]	[0.001]	[0.003]	[0.010]
Household Receives BDH Transfer $= 1$	0.002	0.003	0.002	-0.007	0.009	0.003	0.007	-0.018^{**}	0.026^{**}	0.000	0.005	-0.031^{***}
	[0.003]	[0.003]	[0.001]	[0.005]	[0.008]	[0.003]	[0.005]	[0.00]	[0.011]	[0.001]	[0.003]	[0.011]
Child Worked Last Year $= 1$	0.494^{***}	-0.007**	0.095^{***}	-0.581^{***}	0.604^{***}	-0.027***	0.091^{***}	-0.669***	0.168^{***}	-0.007***	0.082^{***}	-0.243^{***}
	[0.017]	[0.003]	[0.007]	[0.017]	[0.026]	[0.002]	[0.010]	[0.024]	[0.011]	[0.001]	[0.00]	[0.008]
Observations	35,085	35,085	35,085	35,085	12,153	12,153	12, 153	12,153	34,739	34,739	34,739	34,739
Canton Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
							i					
Robust standard errors in parenthe	sis. *** p	$< 0.01, ** _{\rm H}$	o<0.05, *	p<0.1 den	ote levels o	f significa	nce. Samp	le consists	of children	in the El	VTI who a	re between
ane ages of 5-17 and nve with their F and attending school (Work≡1. Sch	ool=1): n	sumates co ot working	and not a	to une aver ttending so	age margu chool (Wor	tat enecus . k=0. Scho	ol=0): woi	tate Frobit cking and n	esumates ot attendin	ior the pro ng school (Work=1.	or working School=0):
not working and attending school (Work=0.	School=1)	The reg	essions inc	clude contre	ols for ind	ividual cha	aracteristic	s ('age' of	the child.	an indicat	or variable
		· /								() 0		

Table C.10:	Robustness Check: Control for Child Working Last Year
	Average Marginal Effects from Bivariate Probit Regressions

for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables denoting household composition, an indicator variable for 'Urban' (base: 'Rural' household)) and indicator variables for the 'canton' in which the household ÷ 242

is in.

Table C.11: Robustness Check: Inclusion of Growth Rates Results: Official Child Labour & School Attendance Average Marginal Effects from Bivariate Probit Regressions

		Child Labour: 1	Legal Definition	
	Work=1; School=1 (1)	Work=0; School=0 (2)	Work=1; School=0 (3)	Work=0; School=1 (4)
Mother Started Working Before $15 = 1$	0.031***	-0.005	0.005**	-0.031*** [0.008]
Father Started Working Before $15 = 1$	0.020***	-0.005* [0.003]	0.002	-0.016*** [0.006]
Household Receives BDH Transfer = 1	$\begin{array}{c} 0.0005 \\ 0.017^{***} \\ [0.005] \end{array}$	[0.003] 0.006** [0.003]	0.007*** [0.002]	-0.030*** [0.007]
Observations	32,281	32,281	32,281	32,281
Income Shock Control	Yes	Yes	Yes	Yes
Canton Controls	No	No	No	No
Province Growth Rates	Yes	Yes	Yes	Yes

Robust standard errors in parenthesis. *** p<0.01, ** p<0.05, * p<0.1 denote levels of significance. Sample consists of children in the ENTI who are between the ages of 5-17 and live with their parents. Estimates correspond to the average marginal effects from bivariate Probit estimates for the probabilities of working and attending school (Work=1, School=1); not working and not attending school (Work=0, School=0); working and not attending school (Work=0, School=0); working and not attending school (Work=0, School=1). The regressions include controls for individual characteristics ('age' of the child, an indicator variable for 'oldest child in household,' an indicator variable for 'boy,' indicator variables for ethnicity: 'Indigenous,' 'Afrodescendant,' 'Other: White/Montubio/Other,' 'Mestizo' (base), indicator variables for 'Mother not in the Labour Force,' 'Father not in the Labour Force'), household characteristics (indicator variables for the 'canton' in which the household is in.

Table C.12: Robustness Check: Ho Credit in Last 12 Mont	usehold E chs - Avera	xperienced age Margir	l a Negati 1al Effects	ve Shock i from Biva	a the Last riate Prob	12 Month it Regressi	is and Ho ions	usehold Tc	ook Out Lo	an or Bor	rowed Mo	ney or a
	Work=1; School=1 (1)	Official Ch Work=0; School=0 (2)	ild Labour Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Market A Work=0; School=0 (6)	Activities Work=1; School=0 (7)	Work=0; School=1 (8)	Work=1; School=1 (9)	Domestic Work=0; School=0 (10)	Activities Work=1; School=0 (11)	Work=0; School=1 (12)
Household Experienced a Negative Sho	ck in the]	Last 12 Mo	nths									
Mother Started Working Before $15 = 1$	0.022^{***}	-0.000	0.005^{***}	-0.026^{***}	0.020* [0.019]	0.001	0.009	-0.031^{**}	0.009	0.001	0.004	-0.014
Father Started Working Before $15 = 1$	0.017^{***}	[eon.0]	0.001	-0.011*	0.039^{***}	-0.009*** -0.009***	-0.003 -0.003	-0.027** -0.027**	0.014	-0.002*	-0.004 -0.004	[010.0]
Household Receives BDH Transfer $= 1$	$\begin{bmatrix} 0.004 \\ 0.011^{**} \end{bmatrix}$	$\begin{bmatrix} 0.003 \\ 0.003 \end{bmatrix}$	0.004^{***}	$[0.018^{***}$	[0.010] 0.022^{**}	0.004 0.002	0.010** 0.010**	[0.011] -0.034***	0.026** 0.026**	[100.0]	0.006^{**}	-0.032*** -0.032***
Household Had (-) Shock in Last $12M=1$	$\begin{bmatrix} 0.005 \\ 0.011^{**} \\ [0.005] \end{bmatrix}$	[0.003] -0.001 [0.003]	$\begin{bmatrix} 0.001\\ 0.002\\ [0.002] \end{bmatrix}$	[0.006] -0.012** [0.006]	$\begin{bmatrix} 0.010 \\ 0.018^{*} \\ [0.010] \end{bmatrix}$	$\begin{bmatrix} 0.004 \\ -0.005^{*} \end{bmatrix}$	[0.005] -0.003 [0.005]	[0.011] -0.010 [0.011]	$\begin{bmatrix} 0.011 \\ 0.012 \\ [0.012] \end{bmatrix}$	$\begin{bmatrix} 0.001 \\ -0.000 \end{bmatrix}$	[0.003] 0.001 [0.003]	$\begin{bmatrix} 0.011 \\ -0.013 \\ [0.011] \end{bmatrix}$
Household Took Out Loan or Borrowed	l Money o	r a Credit	in Last 12	Months								
Mother Started Working Before $15 = 1$	0.022^{***}	-0.000	0.005***	-0.027***	0.021^{*}	0.001	0.009	-0.031^{**}	0.009	0.001	0.004	-0.014
Father Started Working Before $15 = 1$	[0.006] 0.017***	[0.003] -0.007***	[0.002] 0.001	[0.007] -0.011*	[0.012] 0.040^{***}	[0.004] -0.010***	[0.006] -0.003	[0.013] - 0.027^{**}	[0.014] 0.016	[0.001] -0.002*	[0.004] -0.004	[0.013] -0.010
0	[0.004]	[0.003]	[0.001]	[0.006]	[0.010]	[0.004]	[0.005]	[0.011]	[0.010]	[0.001]	[0.003]	[0.010]
Household Receives BDH Transfer $= 1$	0.011^{**}	0.003 [0.003]	0.004^{***}	-0.019^{***}	0.023^{**}	0.001 [0.004]	0.010^{**}	-0.035*** [0 011]	0.027^{**}	0.000	0.006^{**}	-0.033*** [0.011]
Household Loan/Borrowed in Last $12M = 1$	0.004 [0.004]	0.000 [0.002]	0.001 [0.001]	-0.005 -0.006]	0.025^{***}	[0.003]	[0.004]	$[0.019^{**}]$	$\begin{bmatrix} 0.013\\ 0.013\end{bmatrix}$	-0.001 [0.001]	0.000 0.003]	[0.010]
Observations	35,085	35,085	35,085	35,085	12,153	12,153	12,153	12,153	34,739	34,739	34,739	34,739
Califoli Collutois	8	IES	IGS	IGS	162	168	8	168	IES	IES	8	162
Robust standard errors in parenthesis the ages of 5-17 and live with their par and attending school (Work=1, Schoo not working and attending school (W for 'oldest child in household,' an indic	. *** p<(ents. Esti i=1); not ork=0, Sc cator varia).01, ** p< imates corr working an hool=1). 7 able for 'bc	(0.05, * p < espond torespond toad not attoThe regresy,' indicat	<0.1 denote the average ending schc sions inclue or variable	e levels of s e marginal ol (Work= de controls s for ethnia	ignificance effects fro =0, School= for indivi- city: 'Indig	 Sample m bivariat =0); worki dual chara genous, ' 'A 	consists of te Probit e ing and not acteristics Afrodescen	children ir stimates foi attending ('age' of th dant,' 'Oth	a the ENT r the prob school (W e child, an er: White,	I who are abilities of ork=1, Sc indicator /Montubic	between working hool=0); variable /Other,'
'Mestizo' (base), indicator variables t denoting household composition, an i is in.	or 'Mothe indicator	er not in t variable fo	he Labou r 'Urban'	r Force,' 'F (base: 'Run	ather not cal' househ	in the Lal Iold)) and	bour Fore indicator	e'), housef variables f	old charac for the 'can	teristics (i iton' in wh	ndicator iich the he	/ariables ousehold

	Work=1; School=1 (1)	Official Ch Work=0; School=0 (2)	ild Labour Work=1; School=0 (3)	Work=0; School=1 (4)	Work=1; School=1 (5)	Market Mork=0; School=0 (6)	Activities Work=1; School=0 (7)	Work=0; School=1 (8)	Work=1; School=1 (9)	Domestic Work=0; School=0 (10)	Activities Work=1; School=0 (11)	Work=0; School=1 (12)
Changing Age Threshold for Paren	at Work Hi	story to A	ge 10:									
Mother Started Working Before 10 = 1 Father Started Working Before 10 = 1	$\begin{array}{c} 0.016^{**} \\ 0.007 \\ 0.025^{***} \end{array}$	$\begin{array}{c} 0.003 \\ [0.004] \\ -0.002 \end{array}$	$\begin{array}{c} 0.006^{**} \\ [0.002] \\ 0.004^{**} \end{array}$	-0.025^{***} [0.008] -0.027^{***}	0.028^{**} [0.014] 0.036^{***}	-0.002 [0.004] -0.001	$\begin{array}{c} 0.006 \\ [0.006] \end{array}$	-0.032^{**} [0.014] -0.045^{***}	0.028^{*} [0.017] 0.015	$\begin{array}{c} 0.001 \\ [0.001] \\ -0.001 \end{array}$	$\begin{array}{c} 0.010^{**}\\ [0.005]\\ 0.001 \end{array}$	-0.038^{**} [0.016] -0.015
Household Receives BDH Transfer = 1	$\begin{bmatrix} 0.005 \\ 0.016^{***} \\ \begin{bmatrix} 0.005 \end{bmatrix} \end{bmatrix}$	$\begin{bmatrix} 0.003 \\ 0.002 \\ 0.003 \end{bmatrix}$	[0.002] 0.005^{***} [0.002]	[0.007] -0.023*** [0.006]	[0.011] 0.025^{**} [0.010]	$\begin{bmatrix} 0.003 \\ 0.001 \\ [0.004] \end{bmatrix}$	$\begin{bmatrix} 0.005 \end{bmatrix}$ 0.010^{**} $\begin{bmatrix} 0.005 \end{bmatrix}$	[0.011] -0.037*** [0.011]	[0.012] 0.026^{**} [0.011]	[0.001] 0.000 [0.001]	[0.003] 0.006** [0.003]	[0.011] -0.033*** [0.011]
Changing Age Thresholds for Pare	ent Work H	listory to I	Jummy Va	riables:								
Mother Started Work Age: $5-11 = 1$	0.075^{**}	-0.003 [n nna]	0.015* [0.008]	-0.087^{**}	0.078 [0.058]	0.012 [0.017]	0.058** [0.097]	-0.147** [0.058]	0.038	0.000	0.010 [0.019]	-0.048 [0.033]
Mother Started Work Age: $12-17 = 1$	[0.023]	-0.004 [0.009]	0.008	-0.048* [0.025]	[0.041]	0.016	0.045^{**}	-0.102^{**}	0.017 [0.034]	-0.000	0.002	-0.019 -0.033]
Father Started Work Age: $5-11 = 1$	0.034	0.005	0.011	-0.050 [0.049]	0.066* 0.036]	-0.007 -0.007 [0.014]	0.009	-0.069 -0.058]	0.090^{**}	-0.003 [0.005]	0.007	-0.094** 0.04**
Father Started Work Age: $12-17 = 1$	0.008	0.006	0.005	-0.019 [0.039]	0.029	-0.007 -0.007	-0.002 -0.002 [0.027]	-0.020 -0.020 [0.053]	0.086* 0.051]	-0.003 [0.006]	0.006	-0.089* -0.089*
Household Receives BDH Transfer = 1	0.016^{***} [0.005]	$\begin{bmatrix} 0.012\\ 0.002\\ 0.003 \end{bmatrix}$	$[0.005^{***}]$	[0.006]	$[0.025^{\circ}]$ 0.025^{**} [0.010]	$\begin{bmatrix} 0.001\\ 0.004 \end{bmatrix}$	[0.005]	-0.036^{***}	0.026^{**} [0.011]	$\begin{bmatrix} 0.000\\ 0.000 \end{bmatrix}$	[0.003] 0.003]	[0.011]
Observations Income Shock Controls Canton Controls	$_{ m Yes}^{ m 35,085}$	$_{ m Yes}^{35,085}$	$_{ m Yes}^{ m 35,085}$	$_{ m Yes}^{ m 35,085}$	$_{ m Yes}^{ m 12,153}$	$_{\rm Yes}^{\rm 12,153}$	$_{\rm Yes}^{12,153}$	$_{\rm Yes}^{12,153}$	$_{ m Yes}^{ m 34,739}$	$_{ m Yes}^{ m 34,739}$	$_{ m Yes}^{ m 34,739}$	34,739 Yes Yes
Robust standard errors in parenthe the ages of $5-17$ and live with their r and attending school (Work=1, Sch not working and attending school (' for 'oldest child in household,' an im 'Mestizo' (base), indicator variable denoting household composition, at	sis. *** p- parents. E ₄ nool=1); nc Work=0, ' Work=0, ' dicator va si for 'Mot n indicato	<pre><0.01, **] <0.01, **] <pre>stimates c t working t working School=1) riable for ' her not ir t variable</pre></pre>	p<0.05, * orrespond and not a . The regr boy,' indic 1 the Labo for 'Urbar	p<0.1 dend to the aver ttending sc essions inc essions inc ator variah uur Force,'	ote levels o age margin thool (Wor lude contro bles for eth 'Father no &ural' hous	f significa nal effects k=0, Scho ols for ind nicity: 'In nicity: 'In ot in the "	nce. Samp from bivar ol=0); wo ividual); wo digenous, ^{ch} Labour Fo nd indicat	le consists late Probit rking and n aracteristic 'Afrodesce rce'), hous or variable	of childrer estimates tot attendi s ('age' of indant,' 'O ehold char s for the 'c	i in the EN for the pro- ng school (the child, ther: Whi acteristics canton' in	VTI who a obabilities (Work=1, a indicat te/Montu te/Montu te/Montu thereatc	re between of working School=0); or variable bio/Other,' r variables household

Table C.13: Robustness Check: Changing Age Threshold for Parent Work History to Age 10 and Changing Age Thresholds for Parent Work History to